In Search of Authenticity

The Religiosity of Christian Evangelicals

Francesca E. S. Montemaggi
Cardiff University
School of Planning and Geography

This thesis is submitted to Cardiff University in fulfilment of the requirements for the degree of DOCTORATE OF PHILOSOPHY, 30th SEPTEMBER 2013.



DECLARATION:

This work has not been submitted in substance for any	other degree or award at this or any
other university or place of learning, nor is being submitted	ed concurrently in candidature for any
degree or other award.	
Signed(candidate)	Date 30/09/13

STATEMENT 1

This thesis is submitted to Cardiff University in fulfilment of the requirements for the degree of Ph.D.

Signed	(candidate)	Date 30/09/13
--------	-------------	---------------

STATEMENT 2

This thesis is the result of my own independent work and investigation, except otherwise stated. Other sources are explicitly acknowledged with references. The views expressed are my own.

Signed	(candidate)	Date 30	/09	/13

STATEMENT 3

I hereby give consent for my thesis, if accepted, to be available for photocopying and for interlibrary loan, and for the title and summary to be made available to outside organisations.

Signed (candidate) Date 30/09/13



NOTICE OF SUBMISSION OF THESIS: POSTGRADUATE RESEARCH DEGREES

SECTION A: TO BE COMPLETED BY THE CANDIDATE AND SUBMITTED WITH THE THESIS

CANDIDATE'S LAST NAM	IE MONTEMAGGI
CANDIDATE'S FIRST NAME(S) FRANCESCA EVA SARA
CANDIDATE'S ID NUMBE	CR 0735458
SCHOO	DL PLANNING AND GEOGRAPHY
TITLE OF DEGRE	EE PhD
FULL TITLE OF THES	IN SEARCH OF AUTHENTICITY. THE RELIGIOSITY OF CHRISTIAN EVANGELICALS
IS THIS A RESUBMISSION	N? NO
THESIS SUBMITTED FO EXAMINATION I	
CANDIDATE SIGNATURE	DATE

Summary

The dominant theories in the sociological literature identify religion with propositional belief in the transcendent and belonging to and attendance at religious services. This understanding of religion replicates a past Protestant model that fails to account for contemporary religious forms, including Protestant forms. Drawing on Georg Simmel, one of the fathers of sociology, the thesis makes a theoretical contribution by reinterpreting and systematising Simmel's notions of religiosity as a sensitivity, and of belief as trust, to enhance the understanding of contemporary individual religious identity. The use of these notions and of Simmel's nuanced perspective on individuality enables the development of a theoretical framework for the understanding of authenticity. The study focuses on individual religiosity, which has been largely neglected by the literature. The most influential theories that have paid attention to the role of the individual – rational choice theory and spirituality studies – have centred on individual self authority and neglected to conceptualise the self. Accordingly, the thesis provides a relational account of the self, on which to ground authenticity as a process of identity formation. The narratives and practices, which emerged from the empirical research, have revealed the construction of Christianity through the prism of authenticity, as a distinctive and 'truer' way of living. This construction of authentic Christianity testifies to a shift from doctrinal adherence to a more pluralistic discourse by emphasising an overarching ethics of compassion over exclusive claims to truth. Thus, pluralism offers Christian evangelicals the platform to articulate the particularistic tradition of Christianity, whilst retaining a claim to - what they understand to be - universal truth.

Acknowledgements

I would like to express my deepest gratitude to Dr Sara Delamont (Cardiff University, School of Social Sciences) for rescuing me from the quagmire of ethnographic research, for providing a direction, and for making herself available to answer all my questions related to the fieldwork. Above all, Sara has introduced me to the study of social life and made me see how everyday gestures, words and actions make up the fabric of our social world. I applaud Dr Huw Thomas, my supervisor, for his tenacity in reading my work over the past four years, notwithstanding the disciplinary divide and the abstruse Simmelian philosophy. In particular, I would like to thank Huw and Dr Richard Gale, my second supervisor, for their moral support and friendly chats and complaints about the state of academia. I am also very grateful to Dr Matthew Francis (Lancaster University) and Dr Matthew Guest (Durham University) for reading chapter two and commenting on it; to Dr Finn Bowring (Cardiff University, School of Social Sciences) for reading a draft of chapter six and promoting Sartre delicately; and to Dr Andrew Edgar (Cardiff University, School of English, Communication and Philosophy) for reading a rather rough draft of chapters five, six and seven, and yet somehow making sense of them. I am grateful to my friend Marion Carling who proof-read chapters four to seven, all the way from Connecticut and to my friend Nick Speed for proof-reading the particularly tedious methodology chapter. Finally, I would like to thank the people of Bethlehem church, especially the members of the home group I attended, for allowing me into their lives and trusting me with what they hold dear and true, for being accepting and supportive, and for being friends to a stranger.

I dedicate this work to my parents and to my late teacher Sante Amaduzzi, who in their
own peculiar ways have instilled in me the curiosity and love for the works of the
human mind.
"I went into science a great deal myself at one time; but I saw it would not do. It leads to
everything; you can let nothing alone."
Mr Brooke, Middlemarch (Eliot 1872/2008, p. 16)

Table of Contents

	Summary	p.	1
	Acknowledgements	p.	ii
	Dedication	p.	iii
	Table of Contents	p.	iv
1.	In Search of Religion	p.	1
2.	Modern Religion and the Individual	p.	15
3.	Methodology	p.	73
4.	Georg Simmel, 'Sociologist of Modernity'	p.	114
5.	Authenticity	p.	153
6.	Authentic Self and Tradition	p.	175
7.	Religion, Legitimacy, and Ethics	p.	203
8.	From Doctrine to Authenticity	p.	226
9.	The Authenticity of Being a Christian	p.	259
10.	Authentic Ethics	p.	286
11.	Conclusions	p.	317
	Appendix A	p.	331
	Appendix B	p.	333
	Appendix C	p.	335
	Bibliography	p.	336

Chapter One

In Search of Religion

1. Finding Georg Simmel

The thesis makes a contribution by proposing a theoretical framework in the sociology of religion for an understanding of the construction of individual religious identity. It achieves this by developing the concept of authenticity. I formulate authenticity in the thesis as a process of identity construction that requires interpretation of one's religious tradition (cf. chapter six), and participation in defining its boundaries by attributing value to its components (cf. chapter seven). This results from an interpretation and development of key notions in Georg Simmel's sociology of religion and ethnographic fieldwork in a Christian evangelical church in the UK. Simmel was one of the founders of sociology; yet, his work in sociology of religion has received relatively little attention. He has exercised enduring influence on scholars, such as Paul Heelas (2007 and 2008) and Kieran Flanagan

(1996; 2007; 2008). Nevertheless, Simmel's insights have suffered from lack of systematisation and application.

Simmel's approach to sociology of religion has been contrasted with that of Durkheim (Cipriani 1998; Simon 2006) and Troeltsch (Starr 1996). Watier (1996) presented Simmel's notion of religiosity as "diffused" religion, present in numerous manifestations, while Varga (2007) underlined its affinity with "spirituality". Similarly, McCole (2005) and Erickson (2001) explored its metaphysical aspects. Laermans (2006) analysed Simmel's work on religion in relation to his formal sociology, and Vandenberghe (2010) offered a phenomenological reading of it. Simmel's study of religion is part of his complex and wideranging sociology, which is underpinned by a shifting philosophical framework and inconsistent terminology. Thus, his work is not amenable to simple application; rather it requires careful consideration, interpretation and development.

What makes Simmel's insights on religion compelling is the fact that religion is part of a broader thinking on modernity and social relations. This allows an approach for the understanding of religious life, not as separate from the rest of social life, but as a key component of it (cf. chapter four). However, for the purposes of this introductory chapter, it is perhaps necessary to reflect on a more practical question that a reader might, by now, have in mind: how does a thesis on Georg Simmel and sociology of religion sit at the School of Planning and Geography at Cardiff University? The answer lies in a personal journey, which needs to be mentioned to acquaint the reader with my own research perspective. Before venturing into my doctorate, I was a Cardiff city councillor and a policy and public

affairs professional. I wanted to research the involvement in the public sphere of religious communities with a view to developing policy. That never happened as I wandered into a Simmelian universe 'in search for authenticity'.

It all began in the summer of 1995, when I had to decide which university degree to choose. My father wanted me to study law. I was interested in philosophy and religion. 'Law' was his answer. I offered to study political science. 'Law', he replied. So I enrolled at the University of Bologna to read law and left the country two years later. In Britain, I developed a strong commitment to issues of justice and democracy and, in the most roundabout way, I ended up working in politics and public affairs. In 2004, I was elected to Cardiff County Council where I served for two terms. I learned what democracy was about, by fighting with council officers, developers, fellow councillors and constituents. Then, came the general elections of 2005. Conservative politicians, of whichever party, seemed to be using the 'religious card' to appeal to reactionary values, while liberals seemed to regard religion as something one does 'in private', like playing bridge and using pornography. I asked myself why conservative religion had the monopoly of morality. After all, it was my liberal Judaism that called me to politics.

Needless to say that my 'faith' was no settled matter. I agonised over my relationship with God and religious practice. I toiled between the rationalist theology of Maimonides, which goes very well with my Kantian proclivities, the enigmatic "oneness" of the Kabbalah, and my "Father Ted" prayers. I could not fit any denomination. I respected tradition, which I saw as a human quest for a wider horizon by a very select group of human beings, not

always inspiring, let alone inspired. One day, I came clean with God: "I'm more interested in religious people than you.". A bit harsh, but it is people who make religion live or die, with or without divine assent. It was all clear: I had to research "real people" to find out what was going on out there, and to make it up to God. I pressed wise people I knew whether I should do a Ph.D., including Dr Huw Thomas, of the then School of City and Regional Planning at Cardiff University, and Lord John Alderdice, Speaker of the Northern Ireland Assembly, who did their best to dissuade me from the enterprise.

In 2008, the world was hit by the most severe economic crisis since the stock-market crash of 1929. Against the advice of my future supervisor, Dr Huw Thomas, I decided to leave work and self-fund an M.Sc. in Social Sciences Research Methods to be in a stronger position for applying for doctoral funding the following year. I said goodbye to ever being able to get on the housing ladder and launched into the most boring course available at British universities. I planned my project as a "rigorously scientific" research of three case studies to find out how different faith-based organisations (FBOs) were structured, in what type of "social action" they engaged, how they chose it and carried it out. The fieldwork for the study would have required semi-structured interviews with the leadership and active members, and questionnaires, analysed within the epistemological framework of critical realism.

Something got in the way and I was plunged into a labyrinth of ethnographic meanings (cf. chapter three) and Simmelian relativity (cf. chapter four). Simmel reminded me of my teenage literary love, Luigi Pirandello. Pirandello's plays and novels are infused with

Lazarus' philosophy, a teacher of Simmel, and echo the same multi-layered and emergent conception of the individual and reality. A contemporary of Simmel, Pirandello left his land of Sicily in 1889 to live in Bonn for two years, where he translated Goethe, who inspired much of Simmel's later writings. In my teenage years, the reading of *The Stranger* (Simmel 1908/1950) in an anthology, aroused my interest. My re-discovery of Simmel came when I read his essays on religion, following a casual mention from an academic in relation to his essay on the metropolis. Once I found Simmel again, I could no longer delay my destiny.

My choice of Georg Simmel was not only for the objective value of his insights on religion, but for an 'elective affinity' that made me gravitate towards him despite his thorny and contradictory thought. However, the re-discovery of a philosophy that has always exercised a strong attraction on me happened after my initial encounter with the evangelical church I had chosen for my Ph.D. research. Distinctive themes echoing Simmel emerged from the brief fieldwork, which I carried out for the completion of the M.Sc. in Social Science Research Methods. My research participants compared faith to music, talked of authenticity and understood belief as trust in God. Simmel seemed the 'perfect match'. The ethnographic research, as outlined in the next section, and Simmel's complex sociology have thus transformed the project completely.

2. Social Capital and Finding Bethlehem

Before my fateful encounter with Simmel, my original proposal for the master's dissertation and for the Ph.D., focussed on the role of religious ideas in the involvement of FBOs in "social action" in the local community, within a public policy framework (Baker, 2006, 2008, 2009, and 2010; Chapman 2008, 2009, and 2011; Dinham 2008a, 2008b, 2009, 2011, and 2012; Furbey 1999, 2005, 2006, and 2009; Farnell 2001, 2003, and 2009; Graham 2004 and 2008; Livezey 2000a and 2000b; Lund 1995; Smith 1996, 2002, 2003, and 2004). Within this context, social capital seemed like 'a good idea'. Accordingly, social action "bonded" actors in FBOs through communal activities thus fostering community cohesion, but also "bridged" social actors across organisations, thus increasing trust (Putnam et al. 1993; Putnam 1995 and 2000). It took an essay in the second semester of the M.Sc. course to make me ditch social capital as a normative notion with ambitions to be the solution to most ills, and to condemn its 'sibling', spiritual capital, for hollow sentimentalism (Montemaggi 2011).

The fuzzy notion of social action was similarly problematic. Whilst I adopted a sufficiently broad definition, which included all social activities taking place in the church and open to the local community, such as the 'soup run' for homeless people, youth activities, and parent and toddler groups, the notion reinforced a vision of FBOs as service providers and of religion as (Anglican) 'tea and sympathy'. This approach runs the risk of FBOs being interpreted according to government expectations and standards (Dinham 2012), and religion becomes an old-fashioned stereotype of the Church of England, which misunderstands the activities and diversity of religious communities (Ammerman 2005). Religion has too often been essentialised, as a uniform and cohesive system of belief and

practices, thus disregarding the complexity of religious life in the past and the present (Ammerman 2010; McGuire 2008).

My search for religious organisations, as case studies, took an unexpected turn. I approached a Christian evangelical church, which I later named 'Bethlehem', on the basis of a friend's advice. Bethlehem had a good reputation of being active locally. The vision of the church was to provide a sense of place for the local community by creating a welcoming environment and building relationships. I met with the then pastor, Felix, to discuss the activities of the church. He later called me asking for my research proposal, which I thought was odd as I had not decided on the church yet. Felix was preparing to leave the church so he wanted to make sure that everything was in place, not realising what a departing gift a researcher might be for a church in transition. He chose me before I chose the church.

Once acquainted with Bethlehem Church through its documents, I became interested in taking it as a single in-depth study. Bethlehem seemed to have a clear vision of what kind of church and community it wanted to be. One of the church's leaflets, in particular, set out in a detailed and structured manner how theological ideas were linked to action. It captured the vision, the internal organisation, and the relationship with the outside and wrapped it all in the 'sacred canopy' of faith. My supervisor's initial resistance to the choice of an evangelical Christian church disappeared as soon as he saw the leaflet. It was captivating and intriguing, well laid out and coherent. It was perfect, except none of my research participants had ever come across it.

3. Individual Religious Identity

The choice of Bethlehem turned the research into an in-depth case study. It became a slow and intimate search for what a group of people called Christianity. The culture of Bethlehem was crucial in guiding the development of the theory. At the beginning, the empirical data showed the personal identification with religious narratives, from which participants gained meaning in their daily lives (cf. chapter seven). The study of Georg Simmel's sociology and philosophy led to the development of a relational conception of the individual and of social reality (cf. chapter six). The concept of authenticity thus arose from theoretical developments, but also from local narratives of authenticity. Thus, the research always maintained a dialogue between the theory and the data. The resulting concept of authenticity, as a process of identity construction, is laid out in chapter six.

3.1 Research Objective and Research Questions

The objective of the research was to examine the construction of authenticity among a group of committed Christian evangelicals. This required an account of Christian individuality within a sociological framework. The main accounts of religious individuality, as explored in the next chapter, are those of rational choice theory (RCT) and spirituality, both of which lack an adequate social dimension. Consequently, I sought to develop an alternative approach whilst retaining their key strengths, namely: an attention to choice

and individuality. This was achieved by examining critically the sociology and philosophy of Georg Simmel. Simmel's notion of religiosity (*Religiosität*) as an attitude lent itself to a more nuanced understanding of religion that did not rest on set practices and beliefs. Equally, his relational sociology pointed to a potentially more rounded conception of the individual. Nevertheless, Simmel's work is notoriously fragmented and contradictory thus leading to the first major research question of the thesis:

RQ1. Does Simmel propound a coherent framework for understanding religious behaviour?

A preliminary critical reading of Simmel's key works suggested that there was no clear and coherent framework laid out; but that there were concepts which could be interpreted and developed to suit a composite theoretical framework. This work of interpretation and development constitutes a major task accomplished in the thesis. The theoretical contribution is thus the development and systemisation of concepts derived from Simmel's sociology with a view to providing a conception of authenticity that accounts for its relational dimension and for a processual understanding of identity. The conceptions of relational individuality and authenticity serve the purpose of providing an account of contemporary Christian identity, as understood by a small group of Christian evangelicals in Britain. Therefore, the research asked the following questions:

RQ2. How is authenticity understood by participants in the case study?

RQ3. How do the developed theoretical notions help understand the identity of research participants?

The study aimed to gain a deeper understanding of religious identity through *emic* (local) and *etic* (theoretical) categories (cf. chapter three). Ethnographic methods were chosen to enable me to acquire data that would capture the subtle meaning-making of individuals in a small group. This included participant observation, which enabled an appreciation of the social and cultural context of the group, and interviews, which provided a window on the self-understanding of participants. The research has been restricted to a small group in one single church, in a specific time, place and culture. Consequently, the generalisability of the findings is limited. However, the main aim of the study was to develop conceptual notions that may aid the study of religious identity.

3.2 The Contemporary Relevance of Authenticity

On a rainy afternoon of a typical Cardiff spring, my friend Gethin told me of a couple looking at necklaces on display in the gift shop of a church. After a good look, they asked for 'the one with the little man.' I gasped and asked 'Are you serious?' Gethin was serious and unmistaken: the couple did not know who 'the little man' on the cross was supposed to represent. When I told Nicholas, one of the elders at Bethlehem church, he was not surprised. At Bethlehem, quite a few people feel Christians to be a minority despite living in a Christian country. Being a 'nominal' Christian is, for them, an oxymoron. Being a Christian means taking upon oneself to live the Christian life consciously every day. It is a

commitment that goes against the grain of a superficial consumer society (cf. chapter eight). In the eyes of my research participants, Christianity is a distinctive way of life in contemporary multicultural Britain.

The discourse of religious diversity, in the popular and academic imagination, has often neglected to see Christians as part of the multicultural quilt of European society. Contrary to this picture, the present research shows that Christian evangelicals construct their identity as "countercultural," by which they mean being in opposition to mainstream culture. Christians at Bethlehem have adopted the term 'countercultural' to stress that they feel a cultural minority in a multicultural country. The counterculture of the 1960s and 1970s posed a critique to the mainstream culture, seen as white, Western, patriarchal, and, indeed, 'heteropatriarchal' (Valdes 1996), with Christianity legitimising the social order by appealing to a divine and moral order¹. It may thus seem paradoxical that the term countercultural is used by white middle-class Christians; yet, Christian authenticity implies, for participants, countering mainstream culture, which does not reflect Christian values.

Thus, the *emic* category of authenticity is not the adoption of spiritual, mystical, or existentialist religion following the demise of organised religion, but the acceptance of the pluralistic framework of the 21st century, which provides the platform for the articulation of Christianity as a distinctive and 'truer' way of living. On the other hand, the *etic* category of authenticity is a tool to identify the dynamics of self-understanding, the interpretation of one's religious tradition and rationale for behaviour. Authenticity, hence, is a process of

_

¹ Christian movements responded to the 'counterculture' in various forms. Examples are the 'Jesus movement' (Miller 1997, pp. 33-34) and charismatic Christianity (MacLeod 2007, pp. 137-140).

identity construction through the interpretation of Christian tradition. It emerges from – what is felt as – the displacement, as a result of pluralistic culture, of Christianity as "sacred canopy" (Berger 1967), the overarching religious, cultural, and symbolical framework of society in Britain. Paradoxically, it is pluralism that offers Christian evangelicals the platform to claim a place for Christianity in a highly diverse society.

Following Beckford (2003), pluralism is here understood as the recognition of equal dignity for all individuals and groups as a principle of liberal democracy. In contrast, religious diversity refers to the empirical presence of many religions, but also to the internal diversity within each religion. Pluralism denotes the normative acceptance of the dignity and value of such diversity. As Beckford has argued, religious diversity is "the coexistence of several distinct religious collectivities in a country," while religious pluralism "is the belief that such *diversity of collectivities* is good, especially if their interrelations are harmonious" (2003, p. 79, emphasis in the original). Religious pluralism values individual autonomy and choice in religious identity and expression. Thus, it "is best considered as a term denoting a normative or ideological view holding that the diversity of religious outlooks and collectivities is, within limits, beneficial and that peaceful coexistence between religious collectivities is desirable" (2003, p. 81).

The present research thus illustrates how Christian evangelicals reformulate Christianity through the prism of authenticity, which encapsulates religious distinctiveness and a universalistic ethics. By developing a theoretical framework for the understanding of authenticity, as a process of identity construction, I aim to highlight the relational and

processual nature of the individual, the relational character of religiosity, and to begin to unravel the relationship between cultural particularism and ethical universalism. The authentic Christian, in the local narrative, is not merely the person who has a relationship with Christ, and follows and enacts Christian teachings, but the one who embodies the aspiration of humanity of living up to universal ethical standards.

In the literature review which follows, the thesis begins by outlining the most influential conceptions of religion. Here, I focus specifically on the notion of religion underpinning much scholarship in sociology of religion, which is steeped into the Protestant model. I present a close critique of rational choice theory and spirituality studies to question their conception of individuality and conclude by providing a window on the flexibility of evangelical Christianity in the past two decades. The chapter cannot be exhaustive of the literature in sociology of religion; rather it aims to provide a critique of the notion of religion underpinning the most significant literature, to question the concept of individuality in 'supply side' theory and spirituality studies, and to bridge the gap between highly theoretical texts and empirical research. Chapter three explains the methodology, the methods employed in the empirical work, and the epistemological approach underlying the research. The epistemological framework is dependent on Georg Simmel's sociology and philosophy, which is explored further in chapter four.

The theoretical contribution of the thesis is outlined in chapters five, six and seven. Chapter five situates the contribution alongside wider philosophical debates over authenticity and ethics. The proposed concept of authenticity does not presuppose ethical content. This is to

avoid endorsing a particular understanding of religion and ethics as superior or holding a privileged ethical standpoint over others. Chapter six sets out the conceptualisation of relational individuality and the concept of authenticity underpinning the research, while chapter seven proposes the notion of sacralisation to account for the attribution of value and legitimacy to religious tradition made by individual religious actors, which defines the boundaries of authenticity. The chapter also identifies the ethics of compassion that functions as an overarching discourse among participants at Bethlehem and indicates how ethics is part of one's cultural identity.

Chapters eight, nine and ten present an analysis of the data through the theoretical concepts espoused in the previous chapters. Chapter eight explores the culture of Bethlehem and identifies the shift in emphasis 'from doctrine to authenticity', while chapter nine explores what authenticity means in relation to individual identity. Chapter ten reflects on the ethical boundaries of being a Christian and thus of authenticity, and sheds light onto the ethics of compassion as a form of ethics influencing the making of community and the interpretation of Christianity. Finally, chapter eleven brings the thesis to a close by considering the relationship between the theory and the data, the wider significance of the research and offers a personal reflection on the process of empirical research and theoretical construction in the study of religion.

Chapter Two

Modern Religion and the Individual

1. Introduction

The objective of the present research is to develop a theoretical framework for a better understanding of individual religious identity, through the notion of authenticity and to apply it to the narratives and practices of contemporary Christians in an evangelical church. The aim of this chapter is to highlight the assumptions regarding the concepts of religion and modernity underpinning the most dominant literature in sociology of religion. These are broad concepts, of which the present critique can only treat the most significant aspects. I thus concentrate on Max Weber's notion of the 'disenchantment of modernity', which has been central to the most dominant theory of religion, secularisation, but also to its responses, such as 'supply side' theory and spirituality studies. Weber's disenchantment is at the core of sociological understandings of modernisation.

Secularisation theories, reviewed in section two, have rested on Weber's idea of disenchantment, as the process of intellectualisation of knowledge giving rise to instrumental reason (cf. sections two and three). Section four critiques the assumption central to the notion of disenchantment of a 'pre-modern world' as a uniform society of God-believing people subscribing to common values. This notion contributes to the construction of religion according to a Christian Protestant model of propositional belief and formal organisation, which has underpinned much of the dominant scholarship in the discipline. The conception of modernity and of religion is thus filtered through a lack of historicity, which ascribes to the Protestant Reformation the principal role of breaking free from the magic pre-modern world.

This is evident in Charles Taylor's work on religion (2002 and 2007). In his previous philosophical work, Taylor (1989) emphasises the 'inward turn' of the Protestant Reformation as the intellectual origin of modernity. Yet, he seemingly disregards previous instances of rationalisation, differentiation, and individualisation. He also adopts a normative approach judging negatively the loss of – what he calls – the pre-modern "cosmic-religious embedding" (Taylor 2010, p. 23) that leaves the modern "self" detached from communities of meaning and value (cf. chapter five). This normative account of individualisation can be found in the apprehension of classical sociological theory at – what seemed to be – the inevitable atomism and fragmentation of society, which is at the root of sociological thinking (cf. section four). This concept of rationalised and individualised modernity is also shared by 'supply side' theory of religion, which is based on rational choice theory, and byspirituality studies, where, however, individual autonomy is largely

positive and considered to be the prominent element of contemporary religious and spiritual forms.

However, whilst individual autonomy lies at the foundation of both 'supply side' theory and spirituality, the individual is conceptualised differently in each account. Thus, sections five and six analyse in depth 'supply-side' theory and spirituality studies, which are the two most influential frameworks for the understanding of individual religiosity. A close examination, rather than a simple review, serves the purpose of laying the ground to construct a fully relational account of individuality (cf. chapter six), which is in contrast with the individual's instrumental rationality underpinning 'supply-side' theory and the normative appreciation of individual autonomy by much of the spirituality scholarship and by theories of authenticity, including that of Taylor (cf. chapter five).

Spirituality studies have perhaps been at the forefront of pointing to the increasing relevance of the notion of authenticity. Authenticity, however, is not confined to New Age spiritual practitioners, but is present in mainstream religion, including evangelical Christianity. Thus, section seven delves into the great innovation, diversity and fluidity of evangelical Christianity. Evangelicals have responded to the social and cultural shifts of the past fifty years by adapting but also by challenging social change. The individualistic Protestantism of evangelicals, their suspicion of authority and relative lack of ritualism, especially by comparison with the Orthodox or Catholic Churches, allowed them a particular sensitivity to the anti-authoritarian individualism of the 1960s and after. This is

reflected in the increased lay involvement, flexible structures and the emphasis on acceptance Miller (1997) identified in – what he termed – 'new paradigm churches'.

Section seven concludes with a brief note on the literature on congregations and faith-based organisations (FBOs) to highlight current trends in interpreting 'faith' as belief in the supernatural and as a psychological motivator. This approach neglects the social embeddeness of the individual and how social action is constructed. The action of FBOs is also often assumed to be a positive contribution, which neglects the negative action in which religious groups are sometimes involved. This is a serious flaw in the understanding of the relationship between religion and ethics, and the different interpretations and, indeed, constructions of ethics (cf. chapter seven).

The literature surveyed is predominantly American and British. This is partly due to the dominance of Anglo-American literature in the field, but also to the geographical and cultural location of the case study employed in this research: a Christian evangelical church in the UK. The chapter, as mentioned, provides an overview of the most significant literature, together with an examination of rational choice theory, underpinning 'supply side theory', and of the normative hues in Heelas' spirituality studies to contest current notions of individuality and authenticity. A critique of influential theories on authenticity is provided in chapter five as an introduction to the theoretical framework presented in chapters six and seven. Scholarship relevant to Simmel's sociology is thus present in chapters four, five, six and seven. The relevant literature is employed as an aid to the

exploration of Simmelian notions of the individual and religiosity with a view to developing them for the interpretation of current religious behaviour.

2. Theories of Secularisation

Peter Berger's *The Sacred Canopy* (1967) has been one of the most influential texts laying out a theory of secularisation. Berger understood religion as an agent of legitimation: religion sustained the social construction of the knowledge of reality by providing an overarching narrative. It was Protestantism that, for Berger, undermined the legitimising role of religion by stressing individualistic and rational elements of religious life. This led to a differentiation between sacred and secular spheres accentuated by the differentiation of society brought about by industrialisation. The resulting secularisation eroded religion's power to legitimise social structures. In turn, the secularisation of society and culture would lead to the secularisation of consciousness. In other words, individuals would now no longer look at the world through religious interpretations (Berger 1967, p. 108), but as individuals free to choose the framework they found more consonant with their perspective.

According to Berger, secularisation "has resulted in a widespread collapse of the plausibility of traditional religious definitions of reality" (Berger 1967, p. 127). The diversity of worldviews, which Berger calls pluralism, has thus undermined the solidity of

traditional religious narratives. This is further compounded by the now voluntary nature of religious allegiance. The increased individualisation of society weakens the dominance and authority of religious tradition, which now needs to be "marketed" (Berger 1967, p. 138, emphasis in the original). In an increasingly diverse religious landscape, religious groups must organise to "woo a population of consumers, in competition with other groups having the same purpose" (Berger 1967, p. 138). The greater individualisation of society relegates religious "reality" to the private sphere thus turning it into a subjective concern. Consequently, the reality of religion "is apprehended as being rooted within the consciousness of the individual" (Berger 1967, p. 152), rather as an external meaning-system shared with others.

Berger's contribution was the stepping stone for the paradigmatic status of secularisation in sociology of religion. He later abandoned secularisation and acknowledged that the multiplicity of worldviews does not necessarily impede the coexistence of a religion's truth with another (Berger 1999). However, in his rejection of secularisation, Berger still holds on to the idea of modernisation. For him, "modernisation has had some secularising effects, more in some places than in others", but "it has also provoked powerful movements of counter-secularisation. Also secularisation on the societal level is not necessarily linked to secularisation on the level of individual consciousness." (Berger 1999, p. 3). Berger sees in the thriving of 'orthodox' religion, especially Pentecostalism, a sign of counter-secularisation which ensures survival of religion, although he also predicts that 'strict' religions will adapt eventually.

Berger's theory and other secularisation theories rest on a concept of modernity, which is largely characterised by the processes of individualisation, rationalisation, and differentiation. Individualisation identifies the rise in individual autonomy due to, at least in part, the differentiation of society in separate spheres, which results in the loss of premodern community (*Gemeinschaft*) in favour of a more contractual society (*Gesellschaft*). Modern differentiated society is seen as dominated by instrumental rationality as a result of rationalisation. This vision is heavily indebted to Max Weber's notion of the "disenchantment" (*Entzauberung*) of the world, better translated as demystification. For Weber "the process of intellectualization", which had been "undergoing for thousands of years", was the main cause of the retreat of religion into the realm of the irrational (Weber 1919/1970, p. 138).

The unabated advance in knowledge had rationalised the world and de-mystified it. The world could be explained by science and it was, hence, an object of discovery and no longer the physical creation of God, whose understanding uncovered God's will in the world. Weber thought that it was Protestant ideas that spurred rationalisation by stressing the rational and individualistic relationship with the divine. Consequently, religious explanations for the meaning of the world lost their authority. Weber's idea provided an interpretation of modernity by stressing an incommensurable distance with a fictitious 'pre-modern world' (cf. section 4). Tschannen (1991), in his systemisation of secularisation theories, stressed the importance of differentiation, rationalisation, and worldliness. He did not refer to individualisation specifically, although, in many cases, it is an outcome of the wider process of modernity.

Differentiation engenders a separation between the religious and secular sphere with churches losing control over social functions. This, for Luckmann (1967), leads religion to lose influence over other spheres of life and to become privatised. Rationalisation is the dominance of instrumental rationality, while worldliness is understood by Berger (1967) as the retreat of the sacred from everyday life; and a distance from the transcendent realm by Luckmann (1990). Other important features, which Tschannen attributes to the theories of Berger, Luckman and Wilson, are autonomisation, the process whereby society gains autonomy in its functioning from religion; the generalisation of religion, which "permeates the world 'under' secular appearances" (Tschannen 1991, p. 408); and the pluralisation of religious institutions becoming more worldly and competing with one another (Tschannen 1991, p. 409).

Most secularisation theorists see the process of rationalisation as a consequence of the Protestant Reformation. Accordingly, the pre-modern 'enchanted' traditional community (*Gemeinschaft*) gives way to the modern 'disenchanted' association (*Gesellschaft*) through the loss of monopoly over the institution of society. Following Weber, Steve Bruce (1999) has claimed that the Reformation demystified the world by enabling the individual to interpret Scriptures without the intermediary of the clergy thus weakening the power of the Church to have control over individuals' practices and beliefs. Bryan Wilson's interpretation of secularisation rested upon the process of differentiation that caused the institutions of society to grow apart and religious institutions to lose their control of and access to social activities (1966, p. 250). Similarly, David Martin understands secularisation as dependent on social differentiation, but has stressed the role of Christian language.

Accordingly, social differentiation "erodes the links between Christian language and emerging secular languages, for example, those of science and politics, and breaks down the comprehensive institutional coverage once provided by the Church. The theological mode ceases to provide the overarching frame." (Martin 2005, p. 187).

Secularisation has been restricted to a largely European phenomenon, which "is extrinsic rather than intrinsic to the modernising process per se" (Davie 2002, p. 161), and reframed as non-linear (Yamane 1997). In *Public religions in the modern world*, José Casanova (1994) considered differentiation inevitable, but not so privatisation and the disappearance of religion. He argued that religion underwent a process of "deprivatization", term which Casanova uses polemically against secularisation theories and liberal political theories that "prescribe the privatization of religion as a modern structural trend necessary to safeguard modern liberties and differentiated structures" (Casanova 1994, p. 220), but also in relation to the "move or relocation of religion" (Casanova 1994, p. 221).

Casanova adduces, as evidence of this process of "deprivatization", the religious dimension of the Iranian Revolution in 1979, the role played by of the Catholic Church in the processes of democratisation in Spain, Poland and Brazil, the development of liberation theology in Latin America, and the rise of Protestant fundamentalism in US politics. Beckford (2010b, p. 123) has critiqued this thesis for assuming that public forms of religion cannot coexist with continuing privatisation and for neglecting the role of governments and states in using religion for their own purposes. Over the years, Casanova has come to understand religion

through the process of globalisation, which makes possible the coexistence of different forms of religion (Casanova 2011).

In short, secularisation theories do not prophesise the 'end' of religion, but the decline of institutionalised forms of religion and of the power of religious authorities (Chaves 1994) in a more differentiated society, where the individual can choose how to express their religiosity. Modernity, in most accounts of secularisation, is thus the fundamental interpretive tool employed in the study of religion. Nevertheless, it still seems to be understood through the lens of Weber's disenchantment, which too often leads to the conception of a uniform undifferentiated and irrational pre-modern world. Gorski's (2000) historical reflection on secularisation provides a more nuanced understanding of the twin processes of differentiation and de-differentiation.

"In the West, it is argued, Medieval unity was undermined by the Reformation split, which in turn gave rise to modern pluralism. This schema is oversimplified, because the differentiation of the three major churches (Lutheran, Reformed, and Catholic) during the Reformation era went hand-in-hand with a de-differentiation among church, state, and society at the territorial level." (Gorski 2000, p. 143)

Building on historical scholarship, according to which medieval religiosity is one of 'immanence' whilst early-modern religiosity is one of 'transcendence', Gorski understands the Reformation as a process of "religious rationalization" seeking to replace a religiosity

that was "magical, ritual, and communal" with one that was "ethical, intellectual, and individual" (Gorski 2000, p. 148). Accordingly, 'magical religiosity', for Gorski, is a characteristic of believers who seek out divine favour or intervention through formulas or rituals. This is contrasted with 'ethical religiosity', in which believers call for divine favour or intervention "by means of proper conduct or individual supplication". 'Ritual religiosity' "centers around the performance and repetition of certain commemorative or propitiatory acts (usually by a priestly class)", whilst 'intellectual religiosity' affirms "religious truths or teachings". Finally, the rituals and practices of 'communalistic religiosity' are in pursuit of "welfare – material and spiritual – of the given social community (*Gemeinschaft*), understood here to comprise both the living and the dead", in contrast with the rituals and practices of 'individualistic religiosity', which are "oriented primarily toward the welfare of the individual believer and, more generally, of the community of the living" (Gorski 2000, pp. 148-149). These distinctions, however, suggest temporal transformations of religiosity rather than the co-presence of all in different sections of the population.

From this brief overview of the secularisation debate, transpires the overwhelmingly Anglo-American perspective of the literature betraying perhaps the influence of Protestant Christianity in the underlying notion of religion. Scholars have conceptualised religion in the image of Protestantism, as critiqued consistently by prominent anthropologists, such as Rodney Needham (1972) and Talal Asad (1993). Accordingly, religion is essentialised as a uniform and cohesive system of belief and practices, thus disregarding the complexity of religious life in the past and the present (Ammerman 2010; McGuire 2008). Religion, in the secularisation theories' paradigm, is reduced to exclusive truth claims, which may or may

not be falsified by science or undermined by other religions. Belief is restricted to the adherence to official doctrine, regardless of whether this reflects the experience of believers, not only in contemporary society, but also in the mythical 'pre-modern world'.

The underlying assumption of what constitutes religion seems to be propositional belief in the supernatural within a highly organised and hierarchical institution. It is not self-evident why religion can only be made of beliefs, practices and rituals belonging to an established tradition and legitimised by religious authorities thus disregarding 'informal' practices and beliefs and the role played by individual actors in legitimising both formal and informal beliefs and customs (cf. chapter seven). This well-ordered understanding of religion constructs a mythical past and an inadequate present. It conjures an imaginary 'Age of Faith', when religion dominated society and the world was understood according to religious ideas, which was superseded by the 'Age of Reason'. The process of rationalisation, so fundamental to secularisation, is interpreted with more than a tinge of Tylor's intellectualism and Fraser's evolutionary thought. Thus religion is constructed as the myths of the ancients explaining the world that are surpassed with the advance of modern science.

In a rather entertaining article, Stark (1999) runs through historical studies showing how little people knew about religion in the past and how little they cared, including the clergy. Churches were often used to stage markets and dances. Even during the Sunday service, behaviour was not the most respectful and pious. It included jostling for pews, spitting, farting, telling of jokes and the occasional letting off of guns (Thomas, cited in Stark 1999, p.

258). Further, historical research shows a secular (*laïc*) culture of intellectuals in the middle ages (Imbach 1996). The lack of historicity behind the construction of the premodern world is explored in section four. The next section clarifies Max Weber's notion of disenchantment that is reliant on a unified pre-modern world and reflects on its historical background.

3. The Disenchantment of Modernity

Theories of secularisation, as mentioned, are often wrapped in the narrative of disenchantment. A fundamental flaw of the narrative is taking Weber's normative and epistemological notion in a decontextualised and deterministic fashion. Weber's 'demystification' (*Entzauberung*), reflected primarily an epistemological concern (Swatos and Christiano 1999). He wrote at a time of unshakable faith in scientific and technological progress, which was dominated by instrumental reason. He thus feared that instrumental reason had become the only authority legitimising what was acceptable knowledge. Weber did not share the optimism of science and progress of philosophical Positivism; for him, rationalisation was not liberating but imprisoning. He lamented the disenchantment of reality, where myth is stripped away to leave room for rationality.

Weber thought that knowledge of the world was on a linear and progressive path of intellectualisation. He imputed intellectualisation and rationalisation to the indirect

influence of Protestantism and Puritanism, which shifted the focus from the knowledge of nature, to be found among philosophers, to the study of God's "works" in the "exact sciences" (Weber 1919/1970, p. 142). Intellectualisation and rationalisation identify the process whereby magical knowledge is overcome and reason is no longer unified ushering in differentiated spheres of knowledge. Thus, religious meaning lost its authority and science became a rival to religion. Intellectualisation and rationalisation were the paradoxical outcome of the "routinisation" of religious asceticism. Accordingly, the methodical practices and a structured organisation, which aimed to protect specific ideas, ended up overturning the very same ideas. The process is thus paradoxical. For Weber, as Symonds and Pudsey (2008, p. 237) argue, "fundamental values or ends are undermined, lost or reversed in the very pursuit of those values or ends".

The 'ethic' of religion remained alive but in opposition to the scientific understanding of a now objectified world (Weber 1919/1970, pp. 142-143). Religion came to be seen as the irrational human sentiment fighting against enlightened scientific rationality.

"The increasing intellectualization and rationalization do not, therefore, indicate an increased and general knowledge of the conditions under which one lives. It means something else, namely, the knowledge or belief that if one but wished one could learn it at any time. Hence, it means that principally there are no mysterious incalculable forces that come into play, but rather that one can, in principle, master all things by calculation. This means that the world is disenchanted. One need no longer have recourse to magical means in order to master or implore the

spirits, as did the savage, for whom such mysterious powers existed. Technical means and calculations perform the service. This above all is what intellectualization means" (Weber 1919/1970, p. 139).

This process, as Weber notes, has been central to the development of culture throughout history; yet Weber contrasts modernity with pre-modern times in terms of the changed conception of truth. He argues that whilst the truth of science, in the past, was guided by the search for a deeper transcendental truth, be it 'true being' in Plato, 'true art and nature' in Leonardo, or 'God' in Protestantism, in modern times, the truth of science serves only itself. Truth is lost in the self-justifying logic of modern science. Accordingly, science comes to be seen as "free from presuppositions" (Weber 1919/1970, p. 143), in the sense that it has no ulterior motives; it is "value-free because it disregards all values in order to concentrate on the world of pure things and to constitute it as a theoretically closed object-domain" (Vandenberghe 1999a, p. 61). Yet, the work of science is based on fundamental presuppositions that define, and thus confine, science.

"All scientific work presupposes that the rules of logic and method are valid ... Science further presupposes that what is yielded by scientific work is important in the sense that it is 'worth being known'. In this, obviously, are contained all our problems. For this presupposition cannot be proved by scientific means. It can only be interpreted with reference to its ultimate meaning, which we must reject or accept according to our ultimate position towards life" (Weber 1919/1970, p. 143).

The impersonality of the economy rationalises relationships economically. The rationalised economy and the rationalised bureaucratic state are characterised by depersonalisation which eschews morality (Symonds and Pudsey 2006). Instrumental rationality brings no freedom. Reason is itself "crushed by the forces of its own creation, the materialistic determinism of capitalism" (Mackinnon 2001, p. 330).

"The fate of our times is characterized by rationalization and intellectualization and, above all, by the 'disenchantment of the world'. Precisely the ultimate and most sublime values have retreated from public life either into the transcendental realm of mystic life or into the brotherliness of direct and personal human relations" (Weber 1919/1970, p. 155).

Weber's notion of disenchantment has coloured much of the scholarship in sociology of religion; yet, in secularisation theories, it has been interpreted as the reason for the demise of religion, or of religious authority. Weber, like many other classical sociologists, expressed concerns regarding the unabated rise of instrumental reason. This ought to be read in its historical context. Nineteenth century's faith in progress and the machine was accompanied by angst. Before Weber, Tönnies (1887/1963) had lamented the differentiation of human association in his dichotomy between the "real and organic life" of *Gemeinschaft*, and the "imaginary and mechanical structure" of *Gesellschaft* (1887/1963, p. 33). For Tönnies, *Gemeinschaft* and *Gesellschaft* coexist, but the latter is the fruit of urban culture. Thus, *Gesellschaft* is proper of the modern metropolis, subject to a constant process of urbanisation and instrumental exchange between people.

The natural will (*Wesenwille*) of *Gemeinschaft* is contrasted with the rational will (*Kürwille*) of *Gesellschaft* (1887/1963, p. 103). The rationality of contractual relations is instrumental. Thus, "every person strives for that which is to his own advantage and he affirms the actions of others only in so far as and as long as they can further his interest" (Tönnies 1887/1963, p. 77). In a similar vein, Georg Simmel talked of a blasé attitude (*Blasiertheit*) of the metropolis (Simmel 1908/1950), and of the calculatory mentality of the money economy (Simmel 1907/2004). The core characteristic of the mentality, or spiritual life, (*Geistesleben*) of the metropolis is fragmentation. Modernity thus encapsulates the sense of transitory and detached relationships of urbanised society. The modernity, ushered in by industrialised capitalism and 19th century urbanisation, carries the 'sensation' of a society in constant flux.

The boom of the 1850s (Hobsbawm 1987, pp. 44-45) was followed by spectacular growth between 1870 and 1890 (Hobsbawm 1987, p. 35), and yet intellectuals of the time had a sense of general economic and social malaise (Hobsbawm 1987, p. 36). Industrialised capitalism caused Europe and America to leap into the unknown of a new world. Between the mid-18th and mid-19th centuries, the northwest of Europe defeated epidemics and famine. As a result, whilst in 1810 life expectancy was below 40 years of age, it reached nearly 60 years by 1939 (Vallin 2006, p. 46). The 19th century saw the most profound economic and social changes in centuries from which fundamental differences between the 'West' and the rest of the world originated. Industrial capitalism lifted the 'West' from

poverty and early death. It gave life to a mass society, which became more literate, prosperous and democratic through economic development and political strife.

Modern capitalism inserted people within a system of constant transformation (Marx 1883/1976) that was not confined to the economy but touched every aspect of life. The primary *locus*, but also the 'product' of modernity, was the city. Under the thrust of industrialisation, cities sprang and grew exponentially. Impersonality and anonymity made the city the place where the individual was free but detached from – what were seen as – familiar bonds. It is unsurprising, therefore, that the great unfolding of the industrial revolution provoked almost a sense of nausea for its extreme heights and horrendous lows. Nevertheless, the melancholic nostalgia of the pioneers of sociology has too often been taken normatively and has led many contemporary sociologists to bemoan the loss of meaning and community associated with modern atomisation. Accordingly, the solitary modern individual is left drowning in the fragmented world of modernity, not to mention the abyss of post-modernity. Modernity has turned the 'West' into a society of individuals, who have been 'bowling alone' (Putnam 2000) for over a hundred years.

4. The Reformation and the Modern 'Self'

The previous section reflected on Weber's notion of disenchantment and its historical context. The narrative of disenchantment, as mentioned, is essential to the interpretation of

modernity as the threefold process of rationalisation, differentiation and individualisation. Following Weber, modernity is often associated with the Protestant Reformation, which initiated the process of rationalisation. In contrast, historians have begun to revise the notion of enchantment and to employ it as a tool revealing the construction of legitimate knowledge by an elite, rather than a loss of a pre-modern worldview (Saler 2006). The construction of a uniform pre-modern world undermines a proper understanding of modernity and, therefore, of contemporary religion. This is evident in Charles Taylor's work, which has been greatly influential in a wide range of disciplines. Taylor's assumptions are here contested to form the basis of my argument for an understanding of authenticity, as a process of identity construction, rather than an ethical aspiration, as formulated by Taylor.

Charles Taylor understands the disenchantment of modernity as the rejection of the premodern "cosmic-religious embedding" (2010, p. 32). In his monumental opus *A Secular Age* (2007), Taylor argues that modernity has brought a move from belief in God to belief in the self. Taylor proposes an understanding of modernity through shifts in culture, rather than 'acultural' accounts, by which he means materialistic explanations (Taylor 1995). Taylor's linear narration of the philosophical development of modernity neglects historical and sociological evidence. It is an effort in *eisegesis* rather than *exegesis*. Instead of providing a hermeneutical account of past philosophical thought, Taylor seems to read his idea of modernity into their texts. By excising any materialistic consideration, he decontextualises philosophical writings and constructs a narrative to suit his image of the pre-modern

world, modernity and contemporary society thus offering us no tool to understand social reality.

For the purposes of this chapter, I want to focus on three main concerns that are directly relevant to the conception of religion in contemporary times: Taylor's conception of the pre-modern world; the democratisation of ethics; and the inward turn, which Taylor imputes to the Reformation. In the first instance, Taylor claims that, in the West, there has been a movement "from a society in which it was virtually impossible not to believe in God, to one in which faith, even for the staunchest believer, is one human possibility among others" (Taylor 2007, p. 3). It follows that secularism, for Taylor, is the "rejection of cosmic-religious embedding" (2010, p. 32). Taylor's 'God' is clearly the Protestant Christian personal God and 'belief' is Protestant belief in the transcendent supernatural. Thus, in this version of secularisation, all forms of religion that preceded modernity are conflated with belief in the theistic God.

Taylor's assertion that Western society has moved from a culture of 'belief in God' to one where it is just an option, is simply flawed historically. Taylor's method of grounding his theory on the writings of philosophers and thus of an elite disregards the lived reality of people in the past. The everyday religion of people was not the diligent application of the ideal religion of theologians and clergymen. Indeed, as mentioned previously, even during the middle ages there was a 'secular' (*laïc*) culture among intellectuals. Most importantly, religion has changed forms, function and status throughout history. The construction of different paradigms for the understanding of humanity vis-à-vis nature and the

supernatural was mirrored necessarily by very different constructions of religion through the ages. In other words, the terms 'god' and 'religion' have a completely different meaning in Roman antiquity, medieval Europe, early and late modern Europe, and contemporary Europe.

Taking as an example ancient Rome, religion was pervasive of society but not as personal belief in God; rather it was a communal civic endeavour. Roman 'public religion' was constituted by norms and customs codified by practice. It was thus based on orthopraxis (Ando 2008; Scheid 2003). Roman rituals aimed at the wellbeing of the civic and political community, rather than a personal relationship with the deity. The "emperors' identification with gods", for instance, "did not constitute in itself a way or means of deification" (Camia and Kantiréa 2010, p 378). On the contrary the "imperial cult was born of the need of the subjects to define the place of their lord in a hierarchical system of values, and in this sense, it represented none other than the highest level in the scale of honours" (Camia and Kantiréa 2010, p. 381).

This example serves to challenge not only a uniform view of the 'pre-modern world', but also of rationalisation. Weber, himself, considered ancient Rome an example of rationalisation (Weber 1922/1965, p. 12), which is supported by more recent research (Keane 1984, p. 40; Kennedy 2004, p. 1046). For Weber, the legal reasoning of Roman jurists was an example of "formal rationality", which anticipated the modern legal system².

² Roman law was in use in most European countries as *Ius Commune* until the Napoleonic Code (1804) in France and the Civil Code (1900) in Germany (Stein 1999).

Therefore, rationalisation and differentiation, perhaps, ought to be understood as recurring processes or, at least, in relation to historical examples. Consequently, Taylor's second point of ethical democratisation, as a consequence of rationalisation, also needs to be disputed.

Taylor argues that the reformation brought ethical concerns to a much wider population than the clergy with the effect of democratising ethics. Taylor imputes this moralisation to Protestant rationalisation, which frees (instrumental) reason from a theological framework. Consequently, the modern "self" is "buffered" from traditional authorities (Taylor 2007, pp. 37-38), although it is not clear whether this "buffered self" is "buffered" from all authorities except its own self. Again, Taylor discounts pre-modern capitalist processes of differentiation and democratisation, due to his disregard for the historical process of urbanisation and technological changes, such as the printing press. For instance, the urban revolution of the Italian *comuni* of the XI and XII centuries and the invention of credit in Italian city-states laid the foundation for modern capitalism (Cipolla 1976/1997). Indeed, the growth of urban society is accompanied by the growth of schools and literacy (Cipolla 1969, p. 45), which were detached from religious authorities thus challenging their monopoly of power.

"Between the sixth and the eleventh centuries the Church had acquired a virtual monopoly on literacy and education. ... But there is no doubt that beginning with the eleventh century, especially in those areas where commercial life grew more intensively, the Church lost its monopoly as new arrangements and new

institutions slowly developed to cope with new situations and needs" (Cipolla 1969, p. 43)

The lack of a historical, economic and political perspective credits the Reformation with a revolutionary power able to transform society completely and initiate a new 'era'. The ascendency of the Protestant Reformation on past and present scholars of religion seems to have prevented a historical and multidimensional account, not only of religion, but also of modernity. Thus, the 'inward turn', operated single-handedly by the Reformation, has given us the modern "buffered self", seen by Taylor as detached from community and a wider moral horizon. The solipsistic view of the modern individual glosses over the different phases and aspects of individualisation. The Reformation stressed the more intimate conception of the self, inherited from Cicero's humanism (Davies 1997) and reflected in Augustine, as argued by Taylor (1989). However, this is not the only example of a 'reflexive' individuality. Simmel, for instance, considered the Italian Renaissance of the early fourteenth century as the origin of modern individuality for loosening the ties of the middle ages (Simmel 1971, p. 217).

The Italian Renaissance is often seen as the cultural movement when individuality is intensified and when notions of the 'self' become more accentuated. The ideas of "self-consciousness, self-expression, self-presentation and self-fashioning" (Burke 1997, p. 18) are but some elements of Renaissance subjectivity that will be expanded by industrialised modernity. Self-knowledge, the presentation of the self to others, sincerity, and even a fragmented self that are identified as so central to the 'subjective turn' (Taylor 1989) were

already present in Renaissance Italy. The self-reflexive sensibility, at that time, was thus not typically Puritan, but more a reflection of a re-discovery of ancient culture and, arguably, of the role of the city offering "alternative ways of life" and encouraging "a sense of individual choice" (Burke 1997, p. 22). The self, 'between Petrarch and Descartes', became "both more unified than before and more sharply distinguished from the outside world of family and community" (Burke 1997, p. 27). The "new discourse stressed self-reflection and self-control" and "lay the basis for modern subjective sensibility" (Smith 1997, p. 57)

The narrow interpretation of modernity according to the 'doctrine' of the rationalisation operated by the Protestant Reformation fails to appreciate individualisation throughout time, such as in antiquity (Stroumsa 2012) and early modernity (Burke 1997). Undoubtedly individuality received a significant boost under the thrust of industrialised modernity. However, I believe the role of the great economic and political transformations of the 18th and 19th centuries has been neglected in favour of a mythical image of the Reformation as the origin and agent of change. This simplistic account of modernity fails to explore the influence of distinctive features of Protestantism, such as the importance ascribed to sobriety, sincerity, and introspection, as developed in the cultural milieu of industrialised modernity, including in its reaction in the Romantic Movement.

Those traits, nourished by Romantic philosophy and poetry, have shaped a particular understanding of individuality and authenticity, which has remained a reference point

for those escaping economic and cultural upheavals. Lynch detects in the spiritual milieu the "post-Reformation turn to the self" and "the Romantic turn to nature" (Lynch 2007, p. 11), discussed in section six. In contrast, the more utilitarian side of modernity marks "supply-side" theory, based on rational choice theory and is advocated mostly by American scholars. The next section develops a close analysis of rational choice theory to shed light on the underpinning construction of individuality, which is countered by the relational account I propose in chapter six.

5. A Critique of Rational Choice Theory

"Supply-side" theory developed within the framework of rational choice theory (RCT), which outlines a neat and simple conception of the basic rationale and process of decision-making that human beings apply in their daily lives. The framework seemingly provides a way to understand behaviour, in particular organisational behaviour, by providing a model of individual decision-making. Notwithstanding fundamental critiques (Bruce 1993; Demerath 1995; Ellison 1995; Boudon 1998; Chaves 1995; Spickard 1998; Sharot 2002), rational choice theory seems to have conquered the hearts and minds of a sizeable and influential part of academia spreading like a virus across disciplines. Rational choice theory is predicated on actors choosing freely an optimal course of action to 'maximise' their gain. The essential element of 'maximising behaviour' imputes to actors the fundamental rationale of self-interest. The actor chooses between a set of alternatives. Lack of

information does not invalidate the model as "an actor will act on the basis of his or her *beliefs* about possible action opportunities and their effects" (Hedström and Swedberg 1996, p. 128, emphasis in the original).

The theory is, therefore, aimed at predicting trends resulting from the choices made by individuals. One of the most influential formulations of this theory is lannaccone's notion of 'religious capital' (Iannaccone 1990). Following Gary Becker's use of rational choice theory (1976) lannaccone proposes religious capital as the "familiarity with a religion's doctrines, rituals, traditions, and members", which, in turn, "enhances the satisfaction one receives from participation in that religion and so increases the likelihood and probable level of one's religious participation." (Iannaccone 1990, p. 299). 'Religious capital' is thus accumulation of familiarity, knowledge and relationships within the religious sphere. The core characteristic of religious capital is the benefit, or perceived benefit, the agent receives by participating in religious activity. The agent is thus willing to invest time and money to pursue religious activity and increase their religious capital. This affirmation is problematic for it excludes negative experiences of one's religious background and how these affect one's religious identity or rejection of religion.

lannaccone (1995a, p. 85) accepts that RCT takes a utilitarian approach and is, therefore, "unrealistic" for it disregards the complexity of human culture; yet he fails to acknowledge the dynamics of value formation, both at the individual and communal level. Religious capital has been considered a "paradigm shift" (Warner 1993, p. 1044; Finke and Stark 2003, p. 96), mostly due to the rising dominance of rational choice theory. Rational choice

theory has duly highlighted the importance of individual religiosity and choice. The person is not merely part of a religious group but an agent choosing freely. However, its fundamental assumptions of 'instrumental rationality' as the principal drive in decision-making cripple the theory. According to rational choice theorists, individuals make 'rational' choices by pursuing what is in their own 'self-interest' and differences in values and cultures only count as variables.

Thus, what an individual values is a mere preference which is to be pursued as a goal within a framework dominated by self-interest. Satz and Ferejohn (1994) explain that profit maximisation is "an externally derived theory of interests" (1994, p. 87), which supplements RCT. Kiser and Hechter (1998), drawing on Weber (1922/1965), recognise that there are non-instrumental orientations, such as "value-oriented action", "emotional action" and, also, habitual action. However, when faced with lack of evidence on the action orientation, the assumption is that "actors are instrumentally oriented. Instrumental action is least ambiguous and therefore most understandable to the analyst, who may not share an emic perspective with her subjects" (1998, p. 801). The rationale of instrumentality is assumed to be present in all human beings.

There are several problems with these conceptions of choice, instrumentality, and self-interest. In reference to choice, this is reduced to outcome-driven behaviour, which dismisses any other concern of the actor, any dynamic of the group, and any environmental and circumstantial influence. It is a mechanical conception of choice, which is reduced to the outcome of a calculation. Social relations seem to count only as a cultural reference

impinging on one's preferences, rather than a shaping of the individual. In reference to instrumentality, rational choice theorists claim that instrumental actions are based on the assumption of the shared "pursuit of wealth, power, and prestige" (1998, p. 802). However, "wealth, power, and prestige" are not goals, but values.

Consequently, rational choice theorists are not as 'instrumentalist' as they think they are; rather they are disingenuous about the underlying values informing the theory. The self-interest assumed by RCT is not informed by a theory of value. Therefore, it is difficult to understand how actors seek to maximise profit when profit does not equate with money. Further, rational choice theorists, by isolating individuals, prevent any reflection on what can be deemed 'efficient' or 'rational'. Even a consideration of costs and benefits requires an explanation as to what the agent considers a cost and a benefit and how these are balanced. Indeed, actors may choose very destructive and 'irrational' behaviour in order to pursue 'wealth, power and prestige'.

More significantly, the logic of economic self-interest, here implied, is generalised to all and in all circumstances, given the claims rational choice theorists make to explain the most diverse behaviour in the most disparate situations (Hechter and Kanazawa 1997). Money might make the world go round, but the logic of economic self-interest is but one of many and is not necessarily the dominant logic in human behaviour. It is thus arbitrary to assume that self-interest is the most fundamental motivation in all human beings. Arguably, an actor's pursuit of wealth, power and prestige could be determined by a deep-seated need for love and longing for human relationships, meaning and purpose in life. Whilst it is

difficult and often impossible to establish the underlying motivation for the pursuit of wealth, it is likely to impact on the actors' choices much more than an external and mechanical understanding of rationality.

This is not to dismiss self-interest as a powerful orientation in action; rather it is to appreciate that instrumental action is embedded within wider systems of meaning and that social relations shape and inform decision-making. For instance, the sense of belonging to a group engenders a rationale that affects one's interaction with members of the group and one's identity. Further, rational choice theory is heavily biased towards an assumption of self-interest. This is partly mitigated by the consideration of the context of the action. However, an adequate understanding of the societal, cultural and psychological constraints is prevented by a very narrow conception of rationality. One's identity is not irrational, neither are one's feelings, but these construct their own logic making the RCT approach hollow.

Finally, rational choice theory neglects altruistic motivations by giving primacy to self-interested behaviour. Many rational choice theorists seem reluctant to recognise that not all behaviour is motivated by self-interest and adduce the excuse that self-interest should not be confused with selfish interest (Hechter and Kanazawa 1997). However, such distinction fails to acknowledge altruism, which is by definition not self-interested, let alone selfish. Altruism, as noted by Monroe (1994), within the framework of rational choice theory is simply denied and transformed into "deferred self-interest", where the altruistic act has the aim of one's own personal gratification. Monroe stresses the importance of the

intentions over the consequences. If an action results in a benefit to another, regardless of the actor's motivation, it cannot be considered altruistic. Accordingly, altruism is "behaviour intended to benefit another, even when doing so may risk or entail some sacrifice to the welfare of the actor" (1994, p. 862).

5.1 The Actors' Meanings

A perhaps more fundamental problem with RCT is, as mentioned previously, the insulation of the actor from reality whilst claiming superior knowledge of the actor's motives. The theory removes actors from everyday reality imposing motivations on them that are alien to their way of thinking. It is, of course, reasonable that a social theory should shed light onto the mechanisms behind behaviour that might not be acknowledged fully by actors. However, it is difficult to see how this can be applied to allegedly self-interested action and, in particular, to motivation. By definition, motivation needs to move the subject to act in a certain way. Thus, one needs to be aware of one's motivations in order to make a 'choice'. Whilst there can be unconscious or sub-conscious influences on one's actions, these do not count as 'rational' if the subject is not aware of them nor can the subject make a free choice on how to act. After all, the theory is of rational 'choice', but if actors ignore the reasons why they act in a certain way, are they still pursuing profit? Are they exercising choice? To still consider 'rational choice' what might be habit or a choice made on the basis of reasons other than utility seems to stretch the definition of rational behaviour too far. In doing so, rational choice theorists impute the wrong motives to actors and exercise guess-work rather than analysing the data, including the narratives of the actors.

In summary, RCT's utilitarian framework reduces individual autonomy to a choice based on a 'costs and benefits analysis' thus failing to account for the social embeddeness of individuals, the normative nature of religious behaviour, and, paradoxically, the role of individuals in shaping religious norms by interpreting them and attributing value to them. In Iannaccone's formulation of 'religious capital', adherence to religious norms is seen as an investment on the part of the individual to derive a benefit from participation in the religious community (Iannaccone 1990, 1991, and 1998). Thus, rational choice theory fails to appreciate that the religious person does not adhere to norms out of a calculation of costs and benefits; rather compliance is often an expression of identity and commitment, which are inextricably linked to social relations.

5.2 Weberian Rational Choice Theory

Hedström and Swedberg (1996) present an alternative version of rational choice theory that is heavily reliant on Weber's sociology. Thus, rational choice theory is here a mechanism of "ideal-typical action", based on methodological individualism, to explain "aggregate social systems" (Hedström and Swedberg 1996, p. 129). Hedström and Swedberg refer to situational analysis to "explicate mechanisms that make individual action *social*, in the Weberian sense of the term. Actions are social when the choice of one actor influences or is influenced by the choices of other actors" (Hedström and Swedberg 1996, p. 134). The ideal type is an abstracted construction, which models action. The interests, beliefs and opportunities of an actor are elements of the mechanism. They are

factors of, what appears to be, the same weight in the 'final calculation' producing action. Such interests, beliefs and opportunities are decontextualised in the ideal-typical framework.

"The link between the context of action and beliefs, interests, and opportunities of actors, as well as the link between individual actions and collective outcomes, are not part of rational choice theory as such" (Hedström and Swedberg 1996, p. 129).

Hedström and Swedberg misappropriate Weberian sociology by expunging the pivotal reflection on subjective processes. They translate *Verstehen* as 'intentionality' (1996, p. 138) stripping it of its fundamental characteristic, which is that of, almost empathetic, understanding. *Verstehende* is the application of empathy in comprehending the meaning of an action from the actor's perspective. Consequently, actors' meanings, for Hedström and Swedberg, are factors to compute rather than to understand (*verstehen*). It is true that Weber privileged a rationalistic approach to *Verstehen* that reflected 'purposive-rational action' (*zweckrationales Handeln*). However, he recognised the epistemic value of psychological and emotional understanding. Thus, "the 'recapturing' of an experience [*die Nacherlebbarkeit eines Erlebnisses*] is important for accurate understanding [*Evidenz*], but not an absolute precondition for its interpretation" (Weber, cited in Harrington 2001, p. 314). Thus, as noted by Harrington, the "[P]sychological acts of sympathy and imagination could be used to grasp the 'artistic' or 'emotional' context of action" (Harrington 2001, p. 314). This aspect was pivotal to Weber's *Protestant Ethic and the Spirit of Capitalism*

(1905/2001). In the *Ethic*, Weber explored the individual subjective processes, which are not confined to purposive rationality.

This version of rational choice theory, whilst leaving out self-interest as the paramount logic, takes Weberian typologies in a non-problematic fashion. Typologies are abstract constructs, which have less to do with the empirical reality they try to explain than with the mind of the social scientist creating them. Yet, there is no acknowledgement of the bias and context of rational choice theorists and how these play on the framework. Indeed, it can be argued that Weber too showed a strong tendency to read phenomena through the lenses of rationalisation. As noted by Vandenberghe (1999a), the sociologist, for Weber, has no access to the mind of individuals; therefore, it is necessary to construct an ideal-type of possible meanings that can explain action. However, as the ideal-type stops being a construct for the understanding of motivations and becomes an explanatory device for rational action, meanings are lost and the utilitarian logic prevails.

"...it is necessary to construct an ideal-type of the hypothetical meanings or motivations that would explain the observed course of action. ...the problem arises when Weber subsequently tends to reduce this understanding of motives to the understanding of purposively rational action (*zweckrationales Handeln*), sliding thus from a hermeneutically sensitive methodological individualism to the purely utilitarian one which is nowadays advocated and promoted by the world-wide movement of the rational choice theorists" (Vandenberghe 1999a, p. 73).

Weber's *The Protestant Ethic and the Spirit of Capitalism* (1905/2001) and *The Sociology of Religion* (1922/1965), are far more focussed on the actors' subjective processes, including their emotions. The *Ethic* is a classical example of a text often cited and often misunderstood by rational choice theorists. RCT scholars seem to be under the impression that the Protestant believers Weber discusses in the *Ethic* are seeking salvation and thus engaging in productive work in order to secure it. This understanding turns on its head not only Weber's analysis, but the fundamental tenet of Protestant theology of *sola fide*. Protestantism was a reaction to the use of good works and indulgences to secure a place in heaven. The purchase of indulgences and the pursuit of 'good works' would have been much more 'rational' in obtaining salvation. Protestants believed that they were saved by faith alone, which could not be proven. For Weber, this caused 'salvation anxiety' which influenced the systematic activity in which they engaged.

Rationality, here, clearly does not equate with profit maximisation or even with choosing the course of action that would most likely guarantee a certain outcome, as supposed by rational choice theory; rather it identifies a methodical procedure. Further, it must be stressed that Weber considered such systematic behaviour the unintended consequence of the doctrine of *sola fide*, rather than the result of intentional, let alone instrumental, action. Salvation anxiety is the unintended consequence, which generates systematic behaviour. Borrowing from Goethe, Weber attributed to such systematic behaviour an 'elected affinity' (*Wahlverwandtschaft*) with capitalist activity, not a causal relationship. Elective affinity does not suggest causality, but an attraction which includes a choice (McKinnon 2010, p. 116). It can be argued that Weber's reflections in the *Ethic* and his sociology of world

religions do not fit neatly with his wider sociology, especially that of ideal types. There is undoubtedly room for misunderstanding and picking and choosing according to one's preference and objective. However, the brief analysis provided here has highlighted the fundamental distortion of Weber's thesis by rational choice theorists, which no amount of theorising could possibly legitimise.

Weber's approach to the study of religion, and society in general, is ground-breaking for recognising the importance of how individuals interpret and live culture. The diversity of religious behaviour warrants an investigation of the actors' meaning (*Sinn*) of their religious ideas and practices. The *Ethic*, despite its focus on economic behaviour, is a testament to the importance of the meaning-making processes the individual activates. As expressed by Talcott Parsons in his introduction to Weber's *The Sociology of Religion*:

"The central problem was whether men's conceptions of the cosmic universe, including those of Divinity and men's religious interests within such a conceptual framework could influence or shape their concrete actions and social relationships, particularly in the very mundane field of economic action" (Parsons 1965, p.xxi).

Weber highlights the inner dimension of the actor and how this is translated into 'rational' (methodical) behaviour. His famous method of *Verstehen*, the understanding of human subjectivity and its interpretative role, is central in order to identify what shapes behaviour. Human beings interact with their environment and attribute different meanings

to every aspect of life. Therefore, individual consciousness plays an important role in 'translating' ideas about the world into one's relationship to them and resulting action. Weber's attention to subjective processes has been mechanised in rational choice theory depriving the subject of meaning. Paradoxically, rational choice theory is the best example of what Weber lamented of the modern era: the succumbing of reason to mechanistic rationalisation.

6. The Authenticity of the Spiritual Revolution

The previous section argued that rational choice theory interprets religious action, as the rational pursuit of self-interest by freely choosing actors, only by discounting the actors' rationales for their choices. Inspired by Max Weber's sociology, rational choice theorists have paradoxically forced the instrumental reason Weber lamented of the modern age on religious decision-making. In contrast, the market logic has been seen as a loss of values bemoaned by scholars who have argued that religion has fallen prey to modern capitalism and has been reduced to a 'commodity' (York 2001; Miller 2004; Carrette and King 2004). Accordingly, the marketisation of religion has turned the 'goods' of religion into commodities and religion into either agent or victim of commodification. The "silent takeover of all aspects of life by the corporate world" (Carrette and King 2004, p.170) has led to "the colonisation of our collective cultural heritage by individualist and capitalist forms of spirituality" (Carrette and King 2004, p.171).

The twin processes of the marketisation and commodification of religion imply that consumer choice eschews considerations of value, and that, worryingly, left to our own devices, we neglect what is to be considered 'authentic' or 'valuable' in religious life. The concern regarding commodification of religious symbols and practices through the 'profane' process of marketisation reveals the normative assumption, according to which there are authentic and inauthentic forms for the expression or worship of the sacred. The theme of authenticity comes out strongly in Paul Heelas' (2008) account of New Age spirituality. Authenticity is, here, the exercise of one's autonomy but also the expression of one's deepest essence. Heelas, one of the most prominent proponents of 'spirituality', has been influenced by Georg Simmel's notion of mystical religiosity (Simmel 1997). Spiritual practitioners, for Heelas, seek to resist the consumer mentality and the consequent 'McDonaldisation' of society (Ritzer 1993). Therefore, in Heelas' account, spirituality tends to be liberating. Thus the analysis will concentrate on his work for it highlights a particular conception of modernity and spiritual sensitivity.

The term spirituality is often used to stress subjective experience and immanence. This is also reflected in the views of lay informants who see spirituality as more abstract and non-theistic, and religion as connected with organisation and belief (Schlehofer et al. 2008). Barker distinguishes between religiosity as believing in a transcendent, personal God, whilst spirituality refers to belief in the God within (Barker 2008, pp. 189-190). For Houtman and Aupers (2007), spirituality leaves behind religion's focus on transcendence to shift towards immanence and the self. Accordingly, the sacred loses its transcendent

character and "becomes more and more conceived of as immanent and residing in the deeper layers of the self." (Houtman and Aupers 2007, p. 315).

The term spirituality is often employed to denote those forms of spiritual expression that do not require, and indeed might reject, an external authority and a rigid organisation, whilst favouring a form of free expression of spiritual proclivities that are meaningful to the individual. Spirituality studies have highlighted this subjective dimension by demarcating the boundaries between spirituality, as focussed on the self and immanence, and traditional religion, as projected towards an external authority (Heelas 1996, 2006a, and 2008; Heelas et al. 2005; Houtman and Aupers 2007; Tacey 2004). Nevertheless, more recent research has shown the blurred boundaries between religion and spirituality (Ammerman 2013; Palmisano 2009; Pessi 2013) in their use of space, and in practices and beliefs (Bender 2010; Woodhead 2012).

Heelas' endeavour has been particularly influential in understanding spirituality as alternative to religion and, indeed, in conflict with it. Heelas opposes market-oriented theorists and secularisation theorists who interpret spirituality as self-absorption. His starting point is – what Taylor called – the 'massive subjective turn' of modernity, in which the personal experience of the individual comes to be valued more. In his research in New Age spirituality, Heelas finds that spirituality practitioners construct a 'liberatory' spirituality, which they contrast with – what is felt as – 'authoritarian' organised religion. Accordingly, spirituality shuns doctrine, beliefs and ethical injunctions in favour of

experiences of "the heart of life" (Heelas 2006a, p. 46), and practices that aim to facilitate inner search.

"Spirituality is taken to be *life itself* – the 'life force' or 'energy' that sustains life in this world, and what lies at the heart of *subjective life*. ... Whereas New Age spiritualities are experienced as emanating from the depths of life within the here-and-now, the spirituality of the Holy Spirit, the spirituality of obeying the will of God, or the spirituality of experiencing the God-head itself are understood as emanating from the transcendental realm to serve life in this world. Take away the theistic God of religious tradition, and there is little left of Christianity (or theistic traditions); take away the God of theism, and New Age spiritualities of life remain virtually intact" (Heelas 2006a, p. 46).

Practitioners claim that through spiritual practices an individual can realise authenticity, being true to oneself, which results in self-fulfilment. As Heelas puts it, spirituality "is experienced as dwelling within the here-and-now; as integral to life; as inseparable from, a natural aspect of, what it is to *be* alive" (Heelas 2006b, p. 224). Heelas language, however, is tinged with a normative stance suggesting that he views spirituality practices and beliefs as inherently valuable. Rejecting the accusation against spiritual practices of being 'atomistic' and 'selfish' (Heelas et al. 2005), Heelas claims that spirituality is relational and substantial. Thus, although practitioners say that "their participants should only 'accept' a teaching if it rings true to their own experience in the here-and-now", importance is "attached to 'going deeper', getting in touch with one's authentic or natural self, being true to oneself,

expressing love and care, living life to the full: all in the context of 'spirituality'. ... As for the ethical 'substance' of activities, ethics largely come with practice. ... What matters is experiencing the words *through* the practice of yoga, not least through the care and consideration of the practitioner in action." (Heelas 2006a, p. 232). Far from being a solid rebuttal to accusations of self-absorption, the paragraph above leaves one bewildered as to Heelas' interpretation of yoga as an ethical and relational practice.

Heelas also fails to acknowledge the prescriptive aspects of spirituality. The communal values, the rules of conduct and language, the "frames of reference, 'taboos', rules, 'principles', 'how to' instructions, structures ways of growing, and the 'inner' experiences themselves – all skilfully orchestrated by practitioners " (Heelas 2006b, p. 237), all these elements are interpreted by Heelas as aiding self-fulfilment. Among the rules of spiritual practice are the 'taboos' against argument, which are meant to contribute to "consensus and sharing rather than fragmentation or dispersion. … If support is shown for values which contradict the core values widespread in contemporary New Age spiritualities of life and which thereby serve to disrupt the individual, one-to-one or group quest for harmony and balance, 'sanctions' are likely to be called into play. A member of a spiritual group who persistently makes racist comments will be asked to leave. Some things are simply not said or done" (Heelas 2006b, p. 230).

Heelas considers this prohibition against arguing essential to the 'harmony' of group life and one's quest towards spiritual development thus betraying an uncritical and normative assessment of the research. However, such prohibition of argument may be experienced as

preventing discussion and the solution of disagreements. For Heelas, these practices ensure "participants experience neither the pitfalls of unfettered freedom nor the pitfalls of restrictive conformity" (Heelas 2006b, p. 238). The rules and taboos "are experienced as facilitating liberation from dis-ease [sic] or disharmony. Rules are followed for the sake of freedom. The voluntary exercise of obedience is experienced as enabling participants to express their spirituality through autonomous growth; to act as 'true' moral individualists" (Heelas 2006b, p. 233). Yet, there is no reflection on the relationship between obedience and individual autonomy.

Heelas characterises spirituality as revolving around self-authority and self-expression, whilst religion is articulated by religious authorities through structured forms. Consequently, spirituality captures the liberating quest for inner meaning and truth, whilst theistic religion is reduced to adherence to "rationalistic, codified, legalistic prescriptions" (Heelas 2008, p. 127). In his *Spiritualities of Life* (2008), Heelas distinguishes between "spiritualities of life", which give importance to "freedom, uniqueness and expressivity", and "life-as religion", which is characterised by conformity and outside authority (Heelas 2008, p. 55), with the latter option judged to be less valuable. Heelas conflates immanence with autonomy and transcendence with heteronomy. Therefore, only immanent spirituality can guarantee self-authority, whilst transcendent religion gives rise to external obligation.

"Grounded in the past, life-as religion provides detailed information about how one ought to live one's life. Beliefs, commandments, and ethical precepts are among the numerous vehicles which spell out the good life ... The emphasis is

on conformity, the 'match': being a good Christian by electing to live up to authoritative tradition" (Heelas 2008, p. 55)

Heelas acknowledges the spirituality present in Christianity although, being grounded on the 'transcendent', it is incompatible with 'inner-life spirituality'.

"Logically, it is impossible to reconcile an inner 'god' which facilitates *self*-actualization, the expression of the uniqueness or originality of the person, with the transcendent, theistic God of life-as religion-cum spiritualities, which emphasizes adherence 'to' and places limits on autonomous self-development and expression (notwithstanding the fact that one might have initially elected to take the conformist path)" (Heelas 2008, p. 57).

Heelas recognises that Christians participating in "networks, small groups, retreats and all those other activities operating on the 'edges' of theistic Christianity are more or less at one and the same time drawing on the two sources of authority and significance. The God within *and* the 'God without; ... 'perhaps 'selecting' the best of what the god within and the god without have to offer, to find some kind of experiential harmony; perhaps 'yoga in the aisles' coexisting in an uneasy relationship with 'worship in the pews'" (Heelas 2008, p. 58); or perhaps Heelas just got it wrong. Perhaps there is something more than individual autonomy, especially in relation to ethical living. Heelas eulogises the ethics that "speaks 'from the heart', not 'out of' duty" (Heelas 2008, p. 127). The relationship between universalistic ethics and individual autonomy is a fundamental question of 19th and 20th

century philosophy, which is, here, brushed aside by unrestrained enthusiasm for spirituality.

The echo of Simmel's reflections on mystical religion can be heard in Heelas; yet Simmel's conception of the individual as relational is reduced to the recognition of individuals' need for relationships (Heelas et al. 2005, p. 11). Finally, the frequent references to the 'authentic self' or 'real self', lack any reflection on the relationship between 'authenticity' with ethics, and on what interpretation of ethics is being put forward. The above examination of Heelas' understanding of spirituality allows us to highlight the conflation of forms, be they religious or spiritual, with ethics and freedom, which can be found in the spirituality literature, although mostly implicitly rather than explicitly as in Heelas. Heelas' normative stance might not be matched by other scholars, but sociologists of spirituality tend to overlook the organisational structures of spiritual practices by adhering to the 'paradigm of self-authority' (Wood 2010, p. 274).

A critic of 'spirituality' is Matthew Wood, who thunders against the "pre-eminence of self-authority", "taken to accord with the nature of social life in a postmodernized, privatized, individualized or globalized world" (2007, p. 36). Wood wants to retain the notion of power. Thus, drawing on Bourdieu and Foucault, he claims that the "[S]ubjective experiences of freedom and autonomy are *constituted through* (not merely *grounded or expressed in*) social authorities" (2007, p. 60, emphasis in the original). Wood rightly notes that New Age studies "dichotomize social power into two types of authority – self-authority and external authority – and believe that in the New Age, the former dominates the latter."

(2007, p. 70). For Wood, there are multiple 'relativised' authorities. His work shows "the relativization of multiple authorities, such that no single authority (or range of authorities) exerts a formative influence within the life of a group or individual, and that participation in them comes overwhelmingly from a certain class fraction" (2007, p. 71). He criticises spirituality studies for being focussed on the self and yet failing to ground their subjectivism on a "sociological account of the self" (Wood 2010, p. 272), a point which is addressed in the present thesis (cf. chapter six).

Tacey's interpretation of the 'spirituality revolution' reflects a normative theological perspective. He adopts the language of 'counterculture' against contemporary materialism, which is shared by research participants in the present case study. Thus, for him, it is "a people's revolution ... a counter-cultural revolution, a romantic rebellion against the rise of materialism, inhumanity, and economic rationalism. ... It is about personal autonomy and experimentation, with the use of direct experience of the world as a kind of laboratory of the spirit. ... The spirituality revolution is also about finding the sacred everywhere" (2004, p. 4). This aspect of immanent sacrality is the most interesting element of Tacey's thesis for a sociological reflection. It is reminiscent of Cipriani's 'diffused religion' (1989). Although immanence is too often understood as a post-modern novelty, rather than a 'pre-modern' inheritance (Gorski 2000), it is of significance in contemporary religious forms, be they evangelical environmentalism (Harden 2005; Smith and Johnson 2010) or religious politics (Pally 2011). If we concentrate on the aspect of 'diffused religion', the dichotomy of immanent and transcendent fades away.

Giordan (2012) identifies the spiritual as "the internal and experiential aspect of religion, whilst the institution is the rigid and external form" (2012, p. 67). Giordan, however, refrains from separating religiosity from spirituality preferring to employ the term spirituality as a tool for grasping a tonality of believing in the 21st century (Giordan 2004). Thus, spirituality can be found in the personal inner search and in the relationality of faith communities. For Giordan, the most important aspect of the 'paradigm' of spirituality is not the form it takes, but the shift from the institution to the individual. Spirituality is thus "a process of legitimisation of the sacred grounded on the free choice of the subject rather than the obedience to institutional belief" (2012, p. 70). In a similar vein, Hervieu-Léger remarks, in such societies where individual autonomy is a principle, the legitimisation "of belief is moving from religious authorities, guarantors of the truth of belief, to individuals themselves, who are responsible for the 'authenticity' of their own spiritual approach" (Hervieu-Léger 2006, p. 60). Thus, spiritual practitioners are not consumers, as RCT implied, but 'seekers'.

The explorations of spirituality of Wade Clark Roof and Robert Wuthnow identified 'spiritual seeking' as a cultural norm that is not confined to spiritual practices, but that also takes place inside religious congregations. In his book *A Generation of Seekers* (1993), Roof identifies in the 'baby-boomers' a generation with a culture significantly different from that of the pre-boomers. They distrust authorities and seek self-fulfilment, they value experience over adherence to intellectual statements of belief. In his later book, *Spiritual Marketplace* (1999), Roof claims that the culture of the boomers has solidified. The spiritual search takes different courses. Thus, Roof distinguishes between dogmatists,

born-again Christians, mainstream believers, metaphysical believers and seekers, and secularists. Roof's lucid analysis carries the message that 'spiritual seeking' is endemic to contemporary society rather than a truer or more authentic expression of life. Thus, "... personal autonomy has a double face, one that reflects the dislocations of institutional religious identities in the contemporary world, and a second that mirrors a deeply personal search for meaningful faith and spirituality" (Roof 2003, p. 146).

Therefore, the search for an authentic inner life, a reflexive spirituality aiming at self-fulfilment is not confined to spirituality practitioners, but is shared by forms of 'traditional' religion. Indeed, the 'spiritual' is found to be increasingly present within 'traditional' Christianity (Houtman et al. 2009; Droogers 2007). This emphasis on subjectivity allows individuals to shape how they understand and live their own 'spirituality'. Wuthnow (1998) distinguishes between 'dwelling spirituality' and 'seeking spirituality'. Accordingly, 'dwelling spirituality' is the spirituality lived in churches and centred around the family, typical of the 1950s; 'seeking spirituality' characterises the 1960s search for the spiritual dimension in the individuals' self-expression. Wuthnow seeks to go beyond the dichotomy of spiritual, as centred on the individual, and religious, as centred on institutions, and proposes a 'practice-oriented spirituality' integrating community and personal spirituality.

Wuthnow's dichotomy between dwelling and seeking spirituality is similar to Charles Taylor's distinction between – what he terms – the "paleo-Durkheimian dispensation" for which a relationship with the sacred was through one's belonging to a church, and the "neo-Durkheimian dispensation", which entitles people to join their denomination of choice,

which is connected "to a broader, more elusive 'church', and, more important, to a political entity with a providential role to play" (Taylor 2002, p. 93, emphasis in the original).

"The neo-Durkheimian mode involves an important step toward the individual and the right of choice. One joins a denomination because it seems right to one. And indeed, it now comes to seem that there is no way of being in the 'church' except through such a choice." ... "Coercion comes to seem not only wrong, but absurd and thus obscene" (Taylor 2002, p. 94).

Individual choice in deciding whether to be part of a religious, or spiritual, group is the turning point in contemporary liberal societies. This is accompanied by an expressivist turn, according to which religious life and practice "not only must be my choice, but must speak to me" (Taylor 2002, p. 94). The subjectivism of spirituality studies and of the post-1960s generations tend to give pre-eminence to "the 'sacralisation' of subjectivity" (Luckmann 1990, p. 135) and an individualistic self. In contrast, a distinctively Durkheimian approach comes from Gordon Lynch (2007), who stresses the social dimension of – what he calls – 'new spirituality'. Lynch is attuned to the role of religion in binding communities together.

Lynch's 'new spirituality' is expressed in progressive concerns and originates in the following: "the desire for an approach to religion and spirituality that is appropriate for modern, liberal societies, the rejection of patriarchal forms of religion and the search for religious forms that are authentic and liberating for women, the move to re-sacralize

science (particularly quantum physics and contemporary theories of cosmology), and the search for a nature-based spirituality that will motivate us to try to avert the impending ecological catastrophe." (2007, p. 10). Practitioners of 'new spirituality' construct progressive spirituality in opposition to – what they see as – "patriarchal" religion with "a 'top-down' notion of a God, separate from the cosmos, who seeks to order human life in an authoritative way." (2007, p. 11).

Like their New Age counterparts, progressive spiritual practitioners emphasise personal experience and spiritual search "over and against simple reliance on religious certainties" (2007, p. 24). Lynch stresses the role of "communal traditions and rituals as a framework for deepening and reflecting on personal experience" (2007, p. 58) and likens progressive spirituality to a social movement engendering a shared identity (2007, pp. 85-86). Lynch highlights the communal aspect and the political and social action of progressive spiritual practitioners, such as their involvement with environmental or civil rights' issues, reflecting – what Inglehart (1981 and 2008) called – the 'postmaterialism' of contemporary liberal societies.

The tendency to construct a dichotomy between religion and spirituality has impeded an analysis of the element of the communal identity of groups and the connection between spirituality and engagement in the public sphere. Indeed, spirituality has sometimes been accused of being self-indulgent and narcissistic (Chandler 2008). Research in the 'engaged spirituality' of spiritual practitioners is inconclusive (Oh and Sarkisian 2012). The dichotomy religion–spirituality has constructed both realms in essentialistic terms

neglecting the complexity of contemporary Western religious-spiritual forms. Equally, by ascribing an inherent authenticity and freedom of expression to spiritual practices, the spirituality literature has neglected to see these very themes in mainstream religion and the convergence between religion and spirituality. Thus, the next section explores the innovation in practices and sensibility among Christian evangelicals.

7. Evangelical Authenticity

In the past three decades, American evangelical churches have been at the forefront of reorganising how they 'do' church, often by distancing themselves from – what is seen as – institutional or organised religion. The most successful attempts were made by Bill Hybels in Chicago, Rick Warren in Los Angeles and Joel Osteen in Houston. The model of the *Purpose Driven Church* (Warren 1995) responded to the need for 'relevance' of churches for the boom generation. Spiritual seeking was, then, integrated in a religious setting by stressing the search for meaning in community. Christian Smith noted that "evangelicals are creating in their churches opportunities for spiritual nourishment, emotional support, friendships, self-improvement, practical help, and outreach to their surrounding communities" (Smith 1998). Smith (1998) described evangelicalism as "embattled and thriving". It was thriving because embattled. Evangelicals were responding to societal changes by forming a distinctive group identity made stronger by the contrast and conflict

with the outside world. They were renegotiating tradition by locating the spiritual seeking within a distinctive moral collective identity (Smith 1998, p. 219).

However, Smith conceived 'strength' as adherence to Christian religious beliefs; the importance given to faith in one's life; "confidence and assurance" in one's religious beliefs; participation in church activities; commitment "in both belief and action to accomplishing the mission of the church"; and strength of membership (Smith 1998, p. 21-22). Thus, the stress on doctrinal beliefs and the organisational aspects of church-doing tends to skew the picture of evangelicals in favour of dogmatism and separatism, which Smith later rejected affirming evangelicals' diversity (Smith 2000). Donald Miller captured the transformation of evangelical church-doing in *Reinventing American Protestantism* (1997). Miller charters the emergence of – what he calls – 'new paradigm' churches Calvary Chapel, Vineyard Christian Fellowship and Hope Chapel.

These churches have combined contemporary tastes with the 'traditional' Christian message. Accordingly, 'new paradigm' churches have no religious symbols in the building, the dress is casual, and attention is paid to creating a friendly atmosphere (Miller 1997, p. 13), there is a high level of involvement of the laity (Miller 1997, pp. 15-17) and looser structures. The culture of 'new paradigm' churches stresses acceptance (Miller 1997, p. 68) in an attempt to "meet people where they are at", as Miller was told by an informant (Miller 1997, p. 67). 'New paradigm' churches reject – what is felt as – the institutionalisation of religion, its formality, hierarchies and intellectualism, to locate religiosity in a personal

encounter with God. They are the inheritors of the Jesus Movement of the 1960s attracting 'baby-boomers' and 'busters', or 'post-boomers'.

Faith is emotional and experiential; it is transformative of the whole person. 'New paradigm' evangelicals are 'conservative' in doctrine and ethics but engaged in the local community. Miller's book is engaging, although light on theory. The most problematic aspect, however, was Miller's understanding of the changes actuated by 'new paradigm' churches as confined to forms. The changes, for Miller, do not challenge doctrine, but "the medium through which the message of Christianity articulated" (Miller 1997, p. 11). This stance risks 'authenticating' the crystallisation of a specific religious tradition in official dogma as 'true' to the original Christian endeavour (cf. chapter eight).

Sociologists cannot validate practices or beliefs as authentic to a faith tradition, nor can they invalidate them as inauthentic. Christian rituals, beliefs, laws and theologies have changed dramatically over the ages and have been shaped by political, economic and cultural events. Religion as a system of belief reflects the model of Christianity, which, according to influential scholars, began with Ignatius of Antioch in the first century C.E. (Boyarin 2006, p. 10), but which took hold in the third and fourth century. The progressive efforts at defining Christianity resulted in the categorisation of other cultures, such as *Hellenismos, Ioudaismos* and *Romanitas*, into belief-systems separate from the culture (Young, cited in Boyarin 2006, p. 11). Further, how well practices and beliefs adapt to the times guarantees the continuity of a tradition. Ultimately, a tradition rests on individual practitioners ascribing value and legitimacy to – what they perceive to be – their religious,

or spiritual, tradition; rather than on the official stamp of approval from religious authorities, let alone from sociologists.

Thus, authenticity is not an external category, but a group's construction of identity to legitimise changes in the light of tradition. Paradoxically, it is precisely this 'epistemic fallacy' that makes Miller blind to the dynamics of the constant modernisation of evangelicals through the claim, or narrative, of practising 'early Christianity', which is in stark contrast with Catholicism, for example, where the tradition of the church has normative value and is fundamental to Catholic identity and ethics. However, Miller, especially in his later book with Flory (2008), offers a window on the fluidity of evangelical identity and its adaptability to contemporary times. *Finding Faith* (Flory and Miller 2008) presents a typology of different ways in which how 'post-boomer' evangelicals engage with the world. This sheds light on the plurality of forms religion is taking today, from innovation to resistance, and the value placed upon social engagement.

Evangelicals have been undergoing a significant process of change in the past two decades. In the UK, Guest (2002, 2004, 2007a, and 2007b) has shown the diversity and spirituality of evangelicals. They have been particularly innovative in the "emerging" grassroots movement, which began in Britain, New Zealand, and Australia (Guest and Taylor 2006). In the 1980s, the "Nine O'Clock Service", a Christian alternative worship service aimed at young people, opened the door to experiments in alternative worship in Britain that emphasised experience and hermeneutical deconstruction. The emerging movement sought to marry experience with "contextual theology", which reflects critically on past

Christian tradition and the contemporary world. The emerging church is conscious of being in – what they interpret as – a 'post-Christendom' environment, where the church can no longer impose culture and is part of the wider pluralistic framework.

The emerging movement was a reaction to postmodern culture that forced a rethinking of 'truth'. Thus, the emphasis is on being open to 'unorthodox' opinions and the focus is on formulating authenticity (Guest 2007b; Guest and Taylor 2006). In contrast, in the US, the emerging evangelical movement has a much narrower spectrum of innovation (Labanow 2009, p. 5) and is "defined by a deeply felt disenchantment toward America's conservative Christian subculture" (Bielo 2011, p. 197). In the emerging movement we can find the motif of authenticity (Bielo 2011; Guest 2007b; Guest and Taylor 2006; Labanow 2009; Tomlinson 1995), which is becoming a feature of mainstream religion (Pessi and Jeldtoft 2012; Pessi 2013).

Authenticity, however, is not disjoint from social engagement. Thus, the spirituality of evangelicals is often an engaged spirituality that not only seeks to 'save souls', but also to 'serve society' (Unruh and Sider 2005). Unruh and Sider understand the social involvement of churches as 'spiritual capital'. The concept of "spiritual capital" has emerged in recent years in an attempt to distinguish the contribution of FBOs from secular organisations (Finke 2003; Baker 2009; Baker and Skinner 2006; Graham and Baker 2004). Its origins lie in Iannaccone's work on religious capital, which has, here, been emptied of its rationalistic and utilitarian logic. As a result, spiritual capital is taken to mean the positive contribution to society by religious organisations; yet it is also understood as the motivation for action

(Baker and Miles-Watson 2010). The contribution of FBOs is too often understood simply as the outcome of the 'faith' of individuals and 'faith' is conceived as the psychological motivation to action.

This normative notion fails to account for the success or failures of action. Volunteering is not always effective or carried out in a constructive manner (Lichterman 2006). Activities such as churches' soup runs have sometimes received criticism for perpetuating the street-based lifestyle of homeless people, instead of helping them out of homelessness (Lane and Power 2009). A more useful approach for the analysis of the contribution of FBOs to the public sphere would be that of 'social capacity' (Lichterman 2006); the analyses of the shared cultural forms and economic resources of social relations through the lens of Bourdieu's original conceptualisation of social capital (Bunn and Wood 2012); the aspect of reproduction and transmission of religion (Davies and Guest 2007; Guest 2007c, and 2008); and the framework of social movements (Beckford 2003). This would allow us to move away from the rose-tinted lenses of spiritual capital and be able to include the negative contributions to society of religious groups and individuals, which include violent action (Smith 1996); and investigations of instances of religion taking political action (Beckford 2010b).

The celebratory rhetoric of spiritual capital is totally absent in the work of Omri Elisha (2008a and 2008b). His nuanced study of evangelicals' social outreach reveals the tension between compassion and accountability. Elisha's evangelicals are torn between moral ideals of accountability on the part of those receiving charity and the 'moral ambition' of

compassion. There is no celebration in Elisha's analysis; on the contrary, he finds the social engagement to be enclosed in conservative attitudes. The focus on relationships and relationships through social engagement, rather than old-fashioned proselytising, is an important feature of evangelicalism today. Building relationships through engagement in society is articulated as adherence to their image of early Christianity, which is often contrasted with the inward-looking self-absorption of – what is seen as – mainstream religion, including mainstream evangelicalism.

Evangelicalism, in good old Protestant tradition, is anti-institutional and suspicious of authority. Thus, evangelicals form new movements and, once 'routinised', to borrow Weber's term, they then distance themselves from them to begin anew. It is rooted in the 'free church' tradition of the church as a "voluntary society" (Van Dyk 2007, p. 126), a society of individuals in personal relationship with God. It is "the 'tradition' of denying traditions", which characterises evangelicalism in its detachment from the church as a religious organisation and its thirst for being "people of God" (Van Dyk 2007, p. 133). Evangelicals are, therefore, 'radical'. They search the roots of Christianity overcoming the obstructions of historical reality.

This radicalism can drive them towards an 'inward asceticism' and rejection of society or call on them for engagement with the world to build the Kingdom of God. The evangelicals' search for authenticity, of being like Jesus, is a powerful narrative driving and legitimising change, but, first and foremost, it shapes individual religious identity. This is because redemption, for evangelicals, comes through individual transformation. Thus, evangelicals

are seeking 'authenticity', to be true to the example of Jesus, which is reinterpreted in every age and in everyone's life. Yet, it is precisely this search that forces constant change to go beyond the social constraints of a specific time. In seeking to transcend themselves, evangelicals reinvent evangelicalism.

8. Conclusions

The chapter has presented a critique of the conceptualisation of religion according to the Protestant model, centred around propositional belief, and of accounts of individual religiosity that severe the individual from the community. In section two, I have argued that secularisation theories have rested on the conception of modernity, constituted primarily by the processes of rationalisation, differentiation, and individualisation. These processes have been understood through the particular lens of Max Weber's notion of disenchantment. Consequently, they have been seen as unique to industrialised modernity and as an epochal break from a presumed pre-modern world of belief and cohesive communities. Section three has examined and contextualised Weber's notion of disenchantment, while section four, by critiquing the ahistorical image of the pre-modern world so fundamental to secularisation theories, has highlighted how this lack of historicity has led to the construction of religion according to a Christian Protestant model.

The cultural milieu of industrialised modernity was influenced by Protestant values, such as sobriety, sincerity, and introspection. Protestant inward focus, fostered by modern philosophies such as the Romantic Movement, contributed to the modern emphasis on individual autonomy. The image of the modern autonomous individual has coloured much of the sociological literature focusing on individual religiosity. However, the assumption of self-authority, in rational choice theory (section five) and the spirituality discourse (section six), detaches the individual from the group and thus ignores the dynamics that shape norms and behaviour (cf. chapter seven). Both frameworks too often neglect to analyse the social elements of spiritual groups: the belonging, structures, rules, and relationships creating normative behaviour and beliefs.

Rational choice theory's vision of the individual is thus one that is disembedded and disembodied. It is disembedded because the individual is taken as an isolated self, free from social constraints and relationships; and it is disembodied because the presumed 'instrumental rationality' discounts the role of the embodied reality of the person in reasoning. Paul Heelas' conception of spirituality is burdened by a normative and essentialistic take that elevates individual autonomy to the highest ethical principle and thus constructs spiritual practices as liberating for not being framed within a traditional religious environment. Nevertheless, spirituality studies have lately come to address the dichotomy between religion and spirituality that has dominated the field.

Roof's and Wuthnow's research in spirituality in congregations has shown that the 'spiritual seeking' of the boom and post-boom generations has transformed religion

radically. The literature on evangelicals has traced some of the most influential and innovative changes in 'doing church'. It suffers, however, from a lack of theorisation and of a closer analysis of what it means to be a Christian, how it is constructed at the individual and group level, and how it is articulated in practice. Finally, the literature on faith based organisations (FBOs) too often presumes the action of religious organisations to be for the benefit of wider society thus ignoring the conflictual nature of ethics (cf. chapters five and seven).

The selection of the literature, reviewed and critiqued in this chapter, aimed to highlight the Protestant bias in constructing religion as propositional belief, which, I argue, fails to understand contemporary Protestantism. The research on evangelical church-doing allows a glimpse of the diversity and innovation of contemporary forms of evangelicalism. However, the lack of a theoretical framework leaves this research unable to provide an account of identity construction at the individual and communal level. The notion of authenticity that is present in the narratives of spiritual practitioners and evangelical Christians is left mostly undefined. Thus, the thesis presents a critique of the most influential formulations of authenticity, those of Charles Taylor and of Alessandro Ferrara, in chapter five, and builds a conception of authenticity, as a process of identity formation, based on a fully relational understanding of the self, as developed in chapter six and applied to the analysis of the empirical case study in chapters eight, nine, and ten.

Chapter Three

Methodology

1. Introduction

Chapter one recounted how this research project emerged from the combination of different factors: personal questions over religious identity and ethics and their role in a pluralistic society; and the move from a policy-oriented approach to an in-depth study into the individual religious identity of a group of Christian evangelicals. Chapter two discussed how theories in sociology of religion have constructed religion in the image of Protestant Christianity. Much of sociological scholarship has taken belief in God or lack of it as the principal aspects of religion and as indicators of its transformation over time. Research on faith-based organisations (FBOs) presented a more positive picture of religion in the public sphere; however this is often marred by a normative approach (Cadge et al. 2011, Montemaggi 2011).

The present study seeks to challenge the established views of religion, and of individual and communal identity by shifting the focus to individual religious identity, understood as relational. It does so by reconceptualising the individual through a relational concept of authenticity. Authenticity is systematised, within a philosophical and sociological framework, with a view to aiding primarily the understanding of religious identity formation; however, its conceptualisation is sufficiently flexible for a wider application in different contexts. The approach, whilst discussing ethical concerns, does not propose a normative conception of authenticity (cf. chapter five). It is also careful to refrain from the conflation of ethics and religion (cf. chapter seven).

The theoretical conceptualisation of authenticity is a vehicle for the often undefined meanings of the local notion of authenticity. Nevertheless, the thesis does not endorse the local notion of authenticity, which is taken as an instance of identity formation. The use of ethnographic methods enabled an appreciation of the nuances of religious identity and culture. Ethnography has often been employed in the study of evangelical Christianity in sociology and anthropology (Guest 2007a; Versteeg 2001; Ammerman 2005; Elisha 2011; Bielo 2009 and 2011; Bender 2003; Smilde 2007; Hall 1997; Orsi 2003; McGuire 2008). It opens a window on a specific culture and on local social practices. Christian evangelicals, having been the subject of extensive research, are not an unfamiliar culture to explore. Thus, the ethnographic work served the purpose of challenging theoretical assumptions and of illustrating the developed theory.

The research aimed to generate theory for a better understanding of religious life by highlighting the meaning of religion for individuals in shaping their identity. The approach taken was to combine the development and systemisation of key ideas in Simmel's sociology with a process of guiding and refining theoretical concepts on the basis of qualitative data. The focus shifted from the 'outer level' of church-doing to the inner level of how individual actors interpret their tradition, the value they ascribe to it and how these processes of interpretation and value formation affect the actors' understanding of themselves and the world around them. This shift onto the subjective processes of actors and their relationality called for a theory that could account for both the individual and the relational dimensions.

Simmel's relational understanding of society also presented the opportunity for a multidimensional conception of subjective processes. Simmel allows more complex concepts of both the individual and society and of the relationship between the two (Fitzi 2012). His acute analysis of the transformations of modernity (cf. chapter four) enables an exploration of religious forms against a background of macro-level processes impacting on the daily lives of religious actors. Simmelian sociology, in turn, required a methodology attuned to the relational and subjective dimensions of religious life. Indeed, Simmel's conception of society as social relations and of actors' identity as socially embedded, influenced an important strand of sociology: symbolic interactionism.

Simmel's student Robert Park was one of the founders of symbolic interactionism at the University of Chicago where Herbert Blumer, a student of Park's, studied for his doctorate.

Symbolic interactionism gave prominence to the meaning-making of actors through their social interactions. The Simmelian and Blumerian symbolic interactionism guiding this study is outlined in section four of this chapter. The next section highlights the specific challenges to 'knowledge' raised by the research, namely essentialism, normativity, and reductionism. This is followed by the research design (section three), the epistemology (section four), and a reflection on the impact of the personal identity of the ethnographer in the research process (section five). Sections six and seven espouse the methods employed for data collection and analysis. The chapter concludes with a consideration of the ethical concerns of the research process.

2. Essentialism, Normativity, and Reductionism

Every study presents its own peculiar challenges to what constitutes 'knowledge'. This study is concerned with individual religious identity and the concept of authenticity arising from the field. This presented three distinctive challenges: firstly, the risk of essentialism, of affirming a set of attributes of religion and religious identity, and of validating as 'authentic' local experiences and narratives; secondly, the risk of normativity, of seeking to validate the ethics underpinning the practices and beliefs of research participants; and thirdly, the risk of reductionism, of reducing participants' values to mere narratives in the translation from a theologically-centred worldview to sociological knowledge. The epistemological approach taken, as mentioned above, is symbolic interactionism within a

Simmelian framework. Thus, the researcher interprets the reality under study and, in turn, the beliefs and actions of research participants are an interpretation of Christianity. The reality, of which we gain an understanding, is a construction resulting from the symbolic interactions of participants.

Consequently, with reference to the risk of essentialism, a constructionist-interactionist epistemology understands the narratives of research participants as an interpretation and construction of Christianity and of Christian identity. Thus, whilst the local, or *emic*, category of authenticity refers to 'truth' and is, therefore, essentialist, the theoretical, or *etic*, concept of authenticity is understood as a process of identity formation (cf. chapter six), which is employed to illuminate the local use of the term authenticity without endorsing its underlying values. Thus, references to authenticity are not meant to 'validate' local practices and beliefs, let alone the religious tradition of reference; neither does authenticity refer to an ethical construct. Authenticity is deemed to be a cultural process. It encapsulates what people believe to be 'truths'. It is a process that rests on the actors' interpretation of their tradition and – what they understand as – truth. Such interpretive understanding prevents a validation of the accounts of participants.

In reference to the risk of normativity, my efforts at expunging ethics from the concept of authenticity (cf. chapter five) seek to avoid a normative account of authenticity and of the values espoused by research participants. Although the data guided the theoretical progress, the theory sought to avoid a judgement on empirical reality. This was not done to dismiss ethical consideration, but to follow Max Weber's advice of keeping a 'clear head', a

'value free' stance (*Wertfreiheit*) during the research process. Weber was not appealing to ethical neutrality but to be free of bias and ensure "the impartiality of the scientist" (Hennis et al. 1994). He considered value judgments necessary in the heuristic phase in order to identify a problem and formulate a hypothesis, however the analysis of the problem in the research phase needs to be 'value-free' (Sztompka 2009, p. 41). Thus for Weber, "empirical science cannot tell anyone what he *should* do – but rather what he *can* do – and under certain circumstances – what he wishes to do" (Weber 1904/1949, p. 54, emphasis in the original).

The overarching reason for excluding ethics from the concepts of authenticity and religiosity lies in the concern of constructing inadvertently a normative dichotomy between 'good religion' and 'bad religion'. In the previous chapter, I argued that the distinction between spirituality and religion is too often normative making one form of spiritual/religious expression superior to the other. Just as religion has been judged authoritarian (Heelas 2008), spirituality has been dismissed as superficial and 'consumerist' (Chandler 2008; Miller 2004; York 2001; Carrette and King 2004). These normative accounts of religion and spirituality draw ethical judgements from their understanding of the forms and contents of religiosity/spirituality without assessing the associated ethical behaviours. Thus, I sought to avoid endorsing a particular form or content of religious expression as leading to ethical action. Equally, the thesis does not put forward an ethical ideal or framework on which basis to judge the beliefs and practices of Bethlehem church members or of any other religious actor. A normative account of religion would conflate religion with ethics, and thus neglect to appreciate the social and cultural

dimension of religion and its transformations throughout time. It would also deny the complexity and contested nature of ethics (cf. chapter five).

Finally, the translation of a theologically-centred world view to sociological knowledge poses the risk of reductionism. This is particularly the case when religion is understood through a functionalist approach. An example is the approach purported by Robert Segal (2005), who, whilst acknowledging the array of approaches to the study of religion, distinguishes between 'religionists' and 'social scientists'. According to Segal, 'social scientists' locate the origin of religion in a human need, whilst 'religionists' locate it in God. Origin is here to be understood not simply historically but as originating cause. "Social scientific theories deem the origin and function of religion nonreligious. … By contrast, theories from religious studies deem the origin and function of religion distinctively religious: the need that religion arises to fulfill is god." (Segal 2005, p. 51).

Segal's functionalist interpretation of religion rests on an intellectualist account of religion as a set of beliefs and practices related to the transcendent. It thus fails to acknowledge the deliberate effort of religious actors to understand their religious tradition in a way that responds to human needs. The point is not that religion is a projection of society and human needs, but a conscious response to them. The effort of religious actors to transform church-doing to be 'relevant', as shown in chapter eight, is a clear example of their self-awareness of the evolving nature of society and its impact on people's lives. Segal is rightly frustrated by phenomenological studies of religion that essentialise and authenticate religion (Segal 1999). Indeed, "too much sloppy research in sociology and, especially

religious studies, has been justified as 'phenomenological'" (McGuire 2002, p. 209). Nevertheless, ethnographic studies have provided a window on how religious actors understand religion and themselves, including their self-awareness in constructing religion.

Finally, I began the thesis by mentioning my personal 'quest' for progressive religion. I feel, therefore, obliged to provide a brief account of my own ethical beliefs. These are far from being systematised in a coherent theory. Despite Kant's overly rigid conception of ethics, I seem to have a Kantian vein that colours my character. It is a way of being much more than an intellectual standpoint. Having not been schooled in it while growing up, the categorical imperative seems to be part of my genetic code, which is terribly inconvenient whenever I would like to lie and cheat. This is compounded by a stoic attitude, which makes for an austere combination indeed. The Simmelian affinity, though, with its pluralistic universe lightens the burden. I am attracted to the pragmatist endeavour, but still uncomfortable at its vagueness. In short, on a good day, when the government refrains from doing too much damage, I am happy to doubt all truths and universalisms. On a bad day, when the latest anti-immigrant initiative is being flaunted in the media, I seek refuge in Kant: there are things that are just wrong, regardless of whether one agrees with them or not. I take comfort in the failure of the great, from Aristotle to Rawls, in coming up with an exhaustive theory of justice or a universal idea of the good. I think of ethics as a practice, best performed in company.

3. Research Design

The present study sought to develop theory in sociology of religion by challenging the Christian Protestant model of religion, which has dominated the literature (Cadge et al. 2011). Paradoxically, the scholarly Christian Protestant model no longer provides an adequate conceptualisation of religion and belief for Christianity as well as for other religions. The study also sought to bridge the distance between theoretical and empirical works. Too often the study of classical theory is confined to a philological analysis or is detached from concrete reality. Equally, empirical research only rarely yields systematic theory (Hammersley 1992; Snow et al. 2003). Grounded theory is the usual framework for ethnographies. It provides a systematic approach to the analysis of data. However, it is felt that grounded theorising is limited to the 'meso-level' and fails to acknowledge the fundamental ontological assumptions underpinning research.

The need to challenge, or begin to challenge, current paradigms in sociology of religion called for the development of a more robust theoretical framework that would address explicitly the ontological foundations of the research. Specifically, the Simmelian framework of this thesis proposes more nuanced concepts not only of religion and belief, but of individuality and social relations. The concepts of religiosity and belief are dependent on the Simmelian interpretation of the macro social processes of modernity and how these construct the individual and the social. Thus the theory developed in this study builds on Simmel's concepts of relationality, individuality and religiosity, which are dependent on Simmel's philosophical structure (cf. chapter four).

The development of the theory was supported by a study of a Christian evangelical church through the use of ethnographic methods. Ethnography, originally employed in Western anthropology as a descriptive account usually of a remote non-Western culture, came to play a crucial role in the study of urban life in the work of the Chicago School of Sociology, from the 1920s to the 1950s. Since then, ethnography has played an important part in studies that have privileged an interpretative approach to the knowledge of reality. Ethnographies are in-depth studies where the researcher participates in the life of a group, in its own setting, for a prolonged period of time. They provide a rich description of a culture, or of a group, by "understanding events *in context. Out* of context the nature of what occurred may be misunderstood" (Hammersley 1992, p. 23). However, for Hammersley, ethnographic description cannot be explanation. On the contrary, "ethnographic descriptions are theoretical only in the sense that they *apply* theories to the understanding of particular phenomena" (Hammersley 1992, pp. 24-25, emphasis in the original).

A core idea underlying ethnographic studies is that "people construct the social world, both through their interpretations of it and through the actions based on those interactions" (Hammersley 1992, p. 44). This need not be taken as a rejection of a reality independent of the observer; rather it stresses the role of social relations in the actors' interpretation of reality. Further, people's construction of their social world is mediated by the interpretation and, indeed, participation of the ethnographer. Ethnography is thus a 'method' of study that is 'interpretive', in that it captures the interpretations of research participants and of the researcher, and 'interactionist', for it assumes social reality to be the

result of social interaction. Thus, the focus, in the present study, on the actors' meaning-making processes was best met by ethnographic methods and by the development of a coherent Simmelian framework.

Scholars have often seen theory as secondary to the ethnographic endeavour, with the exception of 'grounded theory' (Glaser and Strauss 1967), and the 'extended case method' (Burowoy 1998). Hammersley noted that "there are few examples of explicit and sustained attempts at theorising" (Hammersley 1992, p. 39). Snow, Morrill and Anderson (2003) lamented a lack of awareness of the theoretical assumptions underlying ethnographic work (Snow et al. 2003, p. 183). The theoretical discovery is thus too often secondary to the naturalistic description (Snow et al. 2003, p. 184). Fruitful theoretical concepts have emerged from ethnographic work, for example Wood (2007) and Smilde (2007). However, the richness and fragmentary nature of Simmel's sociology called for a focus on theoretical systemisation, to which the ethnographic fieldwork was instrumental rather than the principal source of theory. Thus, the role of ethnography, in the present study, was to guide and illustrate theory, rather than to generate it.

The choice of the church in the study, here named Bethlehem, resulted from the aim to highlight individual religiosity and transformations in church-doing. Bethlehem was set up around 30 years ago in an area of rapid housing development that lacked a community 'hub', as Nicholas, one of the elders, explained in his interview. I was told about Bethlehem by a friend, who works for a religious organisation. I later met with the then pastor, Felix, to learn more about the church activities and religious identity. Bethlehem is a 'free

church', an independent evangelical church, of Baptist origin. The church attaches great importance to social engagement in the local community in the form of providing groups for local people around their needs, such as parent and toddler groups, youth groups, and an elderly group.

The engagement in the public sphere is seen as a way of going back to the example of early Christianity to recuperate the fundamentals of the Christian faith and practice. Evangelicals seek to 'be like Jesus', to follow his example and act in the world to bring about the transformation of society (Tidball 1994). Bethlehem's Protestant theology³ emphasises the role of the individual's faith, which is expressed in a relationship with Jesus. This focus on personal faith allows an examination of the individual dimension of faith. Thus, the case study is a window on how individual members participate within and outside the church, and how they understand and express their faith.

The emphasis on theoretical concepts has imposed limitations on the fieldwork: the focus was on a small number of participants (7-10), part of the congregation, and the data collection happened mostly at the weekly home study group and through interviews. The focus on individuals challenges the sometimes monolithic view of congregations, which fails to highlight "the internal incoherence of religious groups" (Cadge et al. 2011, p. 441). It also shows aspects of religiosity that have been less prominent in recent scholarship. The study seeks to shift the attention from a concern over the role of belief, or lack of it, in religion (Berger 1967; Davie 1994; Voas and Crockett 2005; Bruce 1993 and 2002), over

_

³ All theological ideas mentioned in the thesis reflect how they are understood at Bethlehem church.

church attendance (Iannaccone et al. 1997; Davie 2007), and over the positive social action of congregations (Baker 2006 and 2008), to the subjective processes of identity formation and the impact of social relations and practices on individual religious identity.

4. Epistemology

This section outlines briefly key epistemological concepts from Simmel's sociology and philosophy, which are explored in more depth in the next chapters. Reality is, for Simmel, in constant movement. Simmel follows Kant in distinguishing between *a priori* Forms⁴ and empirical Contents. For Kant, *a priori* Forms were universal and constant. By contrast, Simmelian Forms are always changing and depend on the content that they take. Forms give a shape to phenomena. We can think of Forms as abstractions that capture social patterns, such as social status, roles, and norms (Tenbruck 1994, p. 358). Forms are also "reciprocal orientations," which represent "a special layer or level of consciousness which is not part of the individual action systems proper" (Tenbruck 1994, p. 353). Individual action systems, as identified by Tenbruck, are Simmel's Contents. Thus Forms, or reciprocal orientations, are an abstract crystallisation of the Content.

_

⁴ The use of the capital 'F' for Forms, 'C' for Contents and 'L' for Life (*Leben*) is to distinguish the epistemological concepts of Forms, Contents and Life from all other meanings of these words.

"The reciprocal orientations are the conditions which account for stable patterns in interaction and for the regularities of overt behaviour (collective phenomena)" (Tenbruck 1994, p. 353).

Forms may be interpreted as social constructs that provide a mould for empirical occurrences, such as human drives and needs. In the later period of his life, Simmel understood Forms as making Life (*Leben*) able to be experienced and understood, albeit in a partial and temporary construct. In short, Forms are constructs that become 'obsolete' or crack under the force of Life, which we may call 'social change'. Life, for Simmel, is more than social, economic and cultural change; it is reality as a whole. However, for our purposes, we can identify Life with social change, which calls for new Forms. For example, the social and cultural changes of the 1960s have instigated innovation in contemporary church-doing (cf. chapter eight). Accordingly, the hierarchical and bureaucratic Form of church-doing, which emerged from the nineteenth century, no longer accommodated the social relationships, structures and culture of the new milieu.

The fundamental approach to reality of Simmel is one of 'relativity' (*Wechselwirkung*), or 'relationality', as explained in the next chapter. As such, it requires us to understand Forms as changing crystallisations of the mind. Such multiplicity of reality is 'understood' by a synthesis achieved through repeated sifting and ordering (Simmel 1900/2004, pp. 108-110). Following Kant, Simmel understands the synthetic view as ordering reality on the basis of the mind's constructs. The process is constructed, but also 'selective'. It is constructed because it reflects the 'epistemological eye' of the knower, the mind of the

social scientist; it is selective because perception and, therefore, analysis of reality depends on the 'distance' from the object of knowledge. It is the distance which makes the synthesis and, thus, knowledge possible (Deroche-Gurcel 1988, p. 356). This work of interpretation needs to be understood as an interaction in itself. Philosophical interpretation does not represent reality; rather it encapsulates the relationship of one's mind with reality. The knower looks into him/herself to understand the object.

"... we can always interpret 'external' events *symbolically* only according to 'inner' *analogies* and vice versa, and that both these analogy formations are performed in the *mode of the simultaneous*, that is to say, in the form of a *hermeneutic circle* or of an *interaction*" (Lichtblau 1991, p. 52, emphasis in the original).

We understand reality by synthesising data through the filters of our mental constructs, which give a resemblance of unity. In Simmel's words, "we inject into reality an ex-post-facto intellectual transformation of the immediately given reality" (Simmel 1950, p. 8, emphasis in the original). Thus, Forms identify patterned regularities such as norms, boundaries, and statuses. They are 'ideal types', which capture the tension between the constant synthesis of the mind and the multiplicity of Life. Forms are in relation with Contents through the synthetic process of the mind. The synthesis of the mind captures the everyday work of translation of chaotic reality into an ordered reality.

Simmel's complex ontology and epistemology has been influential on the development of symbolic interactionism in the United States, as mentioned previously. Blumer's pragmatist

version of symbolic interactionism is particularly close to Simmel's conception of social life (Low 2008), for it stresses the relational nature of the actors' meaning making processes. There are central features of symbolic interactionism that form the basis of my interpretation of Simmellian Forms. Blumer espouses three premises of his perspective, as follows:

"... the first premise is that human beings act toward things on the basis of the meanings that the things have for them ... The second premise is that the meaning of such things is derived from, or arises out of, the social interaction that one has with one's fellow. The third premise is that these meanings are handled in, and modified through, an interpretative process used by the person in dealing with the things he encounters" (Blumer 1969, p. 2)

Similarly, in Simmel, forms of sociation create frameworks of meaning, such as norms, statuses and roles. Indeed, Blumer affirms that "[H]uman group life is a process of formative transaction. Cultural norms, status positions and role relationships are only frameworks inside of which that process goes on." (Blumer 1969, p. 116). It should be clarified that, for the purposes of the present research, where the scope is micro rather than macro, it is the 'inside' process of meaning of norms, statuses and roles that is of relevance. This process of interaction is about attributing meaning. For Blumer, meaning arises from "the process of interaction between people. The meaning of a thing for a person grows out of the ways in which other persons act toward the person with regard to the thing." (Blumer 1969, p. 4). Thus, how the person interprets an object, taking into

consideration how others ascribe meaning to the same thing, guides action. The process of interpretation is the thread that runs through sociation; and sociation, or social interaction, happens through the interpretation of shared symbols. However, it is not simply an adherence to common customs, stereotypes and norms; rather it is a dynamic process which sustains but also challenges shared meanings.

"... interpretation becomes a matter of handling meanings. The actor selects, checks, suspends, regroups, and transforms the meanings in the light of the situation in which he is placed and the direction of his action. Accordingly, interpretation should not be regarded as a mere automatic application of established meanings but as a formative process in which meanings are used and revised as instruments for the guidance and formation of action. It is necessary to see that meanings play their part in action through a process of self-interaction" (Blumer 1969, p. 5)

Social norms are in flux. They are produced, reproduced, but also changed by human agency. Human agency, however, is not to be taken as the singular action of individuals as it is in rational choice theory. On the contrary, symbolic interactionism stresses the social dimension of action. Thus "joint action" is "a societal organization of conduct of different acts of diverse participants. ... The joint action has a distinctive character in its own right" (Blumer 1969, p. 17). This is reminiscent of Simmel's epistemology in *The Philosophy of Money* (1907/2004), where exchange "is the purest and most developed kind of

interaction, which shapes human life when it seeks to acquire substance and content" (Simmel 1907/2004, p. 82).

Such interaction in the process of exchange, the giving and receiving, gives rise to "a new third phenomenon" (Simmel 1907/2004, p. 90). Similarly, Blumer states that the inherent reciprocity of taking into account each other's meaning "not only relates the action of each to that of the other but intertwines the actions of both into what I would call, for lack of a better word, a transaction ... these two lines of action in their developing interrelationship constitute a singleness" (Blumer 1969, pp. 109-110). The meanings underlying joint actions "are themselves subject to pressure as well as to reinforcement ...they may be challenged as well as affirmed ..." (Blumer 1969, p. 18). Further, human action does not happen in a vacuum; it has a context, however, this is malleable to change. It is this process of meaning making which "creates and upholds the rules, not the rules that create and uphold group life." (Blumer 1969, p. 19).

There is, of course, a danger of reducing the individual to the group. Indeed, there is also the danger of falling prey to the circular argument according to which the individual is explained in terms of society and society in terms of individuals (Shalin 1986, p. 14). However, this needs to be understood as a "hermeneutical circle which demands as a matter of principle that the part be explained in terms of the whole and the whole in terms of its parts, i.e., that the individual be understood as a subject and object of the historical process and society as a continuously produced and a continuously producing factor in social interaction" (Shalin 1986, p. 14). We all engage in a hermeneutical exercise by

interpreting and defining the situation in front of us. This means that the researcher is also an interpreter of this process. Thus, my own religious, cultural and personal identity played a role in interpreting the local culture of participants, but also in constructing the field, as explored below.

5. Self-reflection

The swathes of articles and books on methodology and research methods digested during my training did not prepare me for the field because, as Landres rightly points out, the ethnographer *is* the field (Landres 2002, p. 105). For Landres, "... ethnographers actively manipulate their identities in order to get what they want in the field setting" (Landres 2002, p. 106). In my case, it was more a question of adapting to the 'local culture', rather than 'manipulating' my identity. Nevertheless, the embodied nature of ethnographic methods has required reflexivity about my own emotions, thoughts, aims, beliefs, preferences, and about the friendships formed in the field. Ethnography is "relational, emotional, and personal" (Atkinson et al. 2003, p. 56). Thus, the ethnographer is embedded in her own culture, social relations and identity outside the field, but also in the relationships she forms in the field.

"The ethnographer" as argued by Fetterman, "is a human instrument ... The information this tool gathers, however, can be subjective and misleading" (Fetterman 1998, p. 3).

Indeed, I recognised that "research is conditioned by value biases and factual preconceptions about the group being studied," and that "[I]nsider and outsider researchers will differ in the ways they go about building and testing their images of the group they study" (Styles 1979, pp. 148-150, emphasis in the original). Consequently, one's biases and preconceptions need to be scrutinised and reflected upon. However, this should not lead one to consider the insider's perspective as holding epistemic privilege (Burke 1984, p. 631; Wiebe 1999, p. 270; A. Geertz 2002, p. 230). The ethnographer aims to grasp the insider's culture rather than achieve "communion" (C. Geertz 1999, p. 63). Indeed, it is incumbent on the ethnographer to remain critical, whilst taking seriously the experiences and beliefs of research participants (Ingersoll 2002, p. 173).

Ethnography portrays a culture through the understanding of the 'native's point of view' (C. Geertz 1999) with a view to understanding better the internal social structures and dynamics. Thus, an important task is that of 'making the strange familiar' and 'the familiar strange' (Delamont et al. 2010) in order to see social patterns of language, customs and behaviour. Therefore, in my fieldwork, I moved 'close' to research participants to build rapport and gain insight into their view of religious life, but also reflected through my 'outsider identity' of being Jewish, as an analytical tool to fight familiarity. Thus, the self-reflection on my 'outside' identity served to analyse the interaction with research participants and discover social patterns of behaviour. It is in the interaction, rather than in an imaginary 'neutral observation', that patterns are identified and comprehended. The interaction sparks a constant negotiation with oneself and others over 'understanding'. As

mentioned previously, the 'self' of the ethnographer plays a part in the field and is formed by it.

The self in the field "is a product of the norms of the social setting and the ways in which the 'research subjects' interact with the selves the researcher brings to the field" (Reinharz 1997, p. 3, emphasis in the original). Accordingly, I built rapport by absorbing and performing the culture of the field. Like Peshkin, I wanted my non-Christian identity to be forgotten so that members at Bethlehem could be themselves (Peshkin 1984, p. 256). Self-reflection on my identity, in dynamic relation with that of the research participants, has been pivotal in carrying out the research. This concept of self-reflection goes beyond 'reflexivity' in the strict sense of disclosing and reflecting of one's positioning to maintain validity (Cresswell and Miller 2000, p. 127), or of being "sensitive to the interrelationship(s) between herself and the focus of the research" (Delamont 2009, p. 58); rather it identifies, more specifically, an analytical understanding of others, by looking at oneself, and analytical self-understanding, by seeing through the 'eyes' of others.

This approach is based on the epistemological dimension of 'understanding' (*Verstehen*) as originally put forward by Dilthey (1900/1972) and later by Simmel and Weber. *Verstehen*, as formulated by Dilthey, shares little with the concept of empathising with research participants in order to see the world through their eyes; rather it is a dynamic process whereby the researcher understands others through oneself whilst also understanding oneself through others (Harrington 2001, p. 320). We have no knowledge, argues Dilthey, of the inner reality of others; therefore we need to 'reconstruct' the person to make sense

of what we grasp. The sociologist thus synthesises the reality under observation through theoretical constructs, which capture social patterns. The account below provides an example of self-reflection in the field.

5.1 Understanding Others through Oneself

On a Sunday morning of August 2010, in the field of a camping site, I was having breakfast with the participants from the home group. In the middle of mundane chat, Nicholas looked at me and asked "So, what do you make of us?" Put on the spot, my mind scrambled for an answer which turned out to be a light-hearted, banal and clumsy "You are normal". It was met with laughter, which got me out of the spotlight. In that moment, I felt very 'visible'. I felt the researcher with white cloak, spectacles, clipboard and pen observing coolly an animal in its natural habitat. All this time, I've been trying to take the 'scientific spectacles' off, to leave behind the white cloak of my identity, and keep 'clipboard and pen' confined to my mind. After a whole year, I was, at least in Nicholas' eyes, a researcher studying them. All that 'building rapport' and 'fitting in' crumbled with just one question. As I walked back to my tent I mumbled 'Malinowski I ain't'. Yet, I had to come to terms with the fact that one never stops being a researcher no matter how close one gets and that there might always be participants who would think, and rightly so, that one is always on duty even during a camping weekend.

My answer did not raise any further questions, except from me. In one sentence, I ended my ethnographic aspirations. If I saw them as 'normal', what was I missing? If they were 'normal', what could I learn from them? The first question worried me the most and it was the starting point to answer the second one. How could I make the familiar *strange*? How could I see patterns of behaviour and meaning-making if the 'natives' were the dominant group of the culture of which I was part? The answer was staring at me in the mirror: I was the 'exotic exhibit' to observe and talk about, the Italian Jew among Christian evangelicals in Wales, rather than a professional ethnographer looking at 'natives'. Nicholas' question underlined my role of researcher and outsider. I thus considered more carefully how they saw me, how they behaved towards me to identify their own 'markers' of identity. During the camping trip I had been looked after by the two couples *as if* I were a student. Indeed, I was a student, an Italian Jewish student, and also a single female renting a room in a shared house, who found herself among couples, who worked and owned a home. The whole 'blending in' and establishing rapport went out of the window. Or better, I realised that I was as 'inside' as I could be, well 'immersed', but still with my head outside the water. It is in this liminal state that I could make sense of my fieldwork.

I had to observe myself to see what they saw in me and, in turn, how they saw themselves. By focusing on who I was in their eyes, I could see their identity a little more clearly. By reflecting on what made me an 'outsider' to the group, I could make visible the social boundary defining the 'in group'. Amidst British Christian married couples, often with children, my *self* was and is in a liminal state for being a foreigner, Jewish, single and with no fixed residence. I realised that I had always lacked a religious community where I could feel 'at home', where I could participate actively and express my religiosity. Indeed, lacking a community, in which to ground my identity of place, language and relationships, my

religiosity had very few certainties. Without a country, a religious community, a family and a home of my own, I was the Simmelian *stranger* (*Fremde*), the one who has a fluid identity that is constantly being challenged and re-shaped.

The stranger, as described by Simmel, is always "the potential wanderer: although he has not moved on, he has not quite overcome the freedom of coming and going." (Simmel 1908/1950, p. 402). The Simmelian stranger is not just an identity, but an epistemological category (Merton 1972). Thus, the identity of the stranger also sheds light onto the invisible lines of connections between the people making up the group, the culture and unwritten rules of the group. The stranger is not simply outside, but "an element of the group itself. His position as a full-fledged member involves both being outside it and confronting it" (Simmel 1908/1950, p. 402). To be a stranger, affirmed Simmel, "is a specific form of interaction" (Simmel 1908/1950, p. 402), which endows the stranger of 'objectivity', not in the sense of passivity or detachment, but a participation that is composed of "distance and nearness, indifference and involvement" (Simmel 1908/1950, p. 403). Thus, being a stranger is not to be taken as 'value-free' (wertfrei) objectivity, but as a complex relation with the research environment and research participants. Being 'outside', the stranger sheds light on the 'inside' and how that 'inside' is constructed. Being a single Italian student with no fixed residence and community, for instance, marked the construction of community that needs to respond to the aspirations and needs of middle class, relatively young couples with children, who are settled in a specific geographical location. These are generalisations, mental constructs, which reveal patterns of socialisation, which impact on the construction of religious identity and community.

6. Data Collection

I entered the field in June 2009 to carry out research in Bethlehem for three months, for my Master's dissertation. This became the pilot for the Ph.D. research. I gained access through Felix, the pastor of the church, whom I contacted after a suggestion from a friend. Following a meeting with the elders to explain the research and gain their approval, I was introduced to the church on a Sunday morning. Access through Felix allowed me to attend the 'home group' bible study in the home of church members. The first 'peek' of the culture of the church was through the leaflets displayed at the entrance of the main hall. Before the pilot study began, I analysed several leaflets describing the groups and structure of the church, and the 'statement of faith' (Appendix A). As mentioned in chapter one, most informants were not aware of the leaflets. I therefore considered it redundant to include their analysis in this thesis.

During the pilot study, in the summer 2009, I attended Sunday services, home group evenings, and most church-run activities, as a participant-observer. I carried out seven unstructured interviews, from which I draw for the doctoral research, with the following: the pastor (Felix), two church members (Dorothea and Will), two elders (Peter and James), and two church employees (Godwin and Lucy). Interviews provided rich data and gave an opportunity to informants to reflect on their personal experience. As the pilot study focussed on the specific activities in the community of the church, I also interviewed the Director of the Evangelical Alliance Wales and the Cardiff County Council officer coordinating homelessness action. These interviews were for triangulation purposes only,

to ensure an 'outsider' view (Hammersley and Atkinson 2007, pp. 183-185). Establishing rapport with people in the field was relatively easy. Participants were open and included me in non-church activities such as a barbecue and lunch. Some offered to talk about how they became Christians and what it meant for them, without being asked.

After completing the dissertation for the M.Sc., I continued the empirical work. This was partly due to my intention to maintain contact while working on the theoretical elements of the study, but also what I felt were the expectations of research participants. Thus, whilst I was reviewing the literature and gaining deeper familiarity with Simmel's work, I continued the fieldwork in a 'low key', by primarily attending the home group, where numbers of people present would oscillate between 10 and 15; church-wide fellowship evenings; and informal social events. This had the effect of limiting my access to other settings. "Gaining access to a research site" as Burgess notices, "is not a one-off event. It is instead a social process that occurs throughout a research project." (1991, p. 52). Consequently, when I attended parent and toddler groups nearly two years later, my presence seemed surprising. I also carried out another round of seven unstructured interviews at different moments, as outlined in the following section.

6.1 Interviews

I carried out a second set of interviews with church members (Celia, Selina, Arthur and Camden), one of the elders (Nicholas), the interim pastor (Walter), and an informal interview with Felix, the former pastor. Nicholas and the church members were all part of

the home group. All interviews, except Felix's informal one, were recorded and transcribed whilst some informal conversations were noted in the field notes, together with notes from participant observation. I chose to limit the interviews to members of the home group because I was more familiar with them, I considered them "knowledgeable" of the local culture (Rubin and Rubin 1995, p. 66), but also because the confines of the home group selected organically a particular group within the congregation. There are three couples in their late 40s and early 50s, and five couples in their late 20s and early 30s. People in the group have formed strong relationships of friendship. This meant that most of the informants represented a very cohesive and, to an extent, homogeneous group. Nevertheless, their opinions on important issues, such as salvation, varied widely (cf. chapter eight).

Interviews with church employees took place in a private room at the church; other interviews took place in the home of informants or at their workplace. One interview took place in a cafe. All the interviews were recorded with permission of the interviewees to ensure rigour and trustworthiness (Morse and Richards 2002; Maxwell 2004). I was aware that this would cause a degree of self-consciousness, however I tried to make the interviewees more comfortable by explaining that there was no set of questions and that they could digress. I also offered to switch the recording off at any point. The interviews focussed on personal religiosity, how interviewees live their faith and their relationship with the church community. They mostly began by asking informants for how long they had been part of the church, whether they were part of a church before that, or, more broadly, their spiritual/faith journey.

The interaction and dynamics of the interviews changed depending on the interviewees' personality and their role in the church. For example, the interviews with the elders were more focussed on the organisational and cultural aspects of the church, while the interviews with members were much more about their personal experiences and allowed more room for digression. The conversation flowed naturally in all cases and, in some, the interviewees felt able to talk at length recounting anecdotes from their lives. I was aware that my spoken and silent reactions tended to show understanding of what was discussed. Thus, my questions aimed at thinking of the same topic from different perspectives. However, I tried not to guide the interviewees and to keep the questions as open as possible.

Interviews were unstructured and trying to resemble a conversation as much as possible. The approach taken was that of the 'cultural interview' (Rubin and Rubin 1995), which aims at an understanding of the local culture. "Culture is about how people interpret the world around them by developing shared understandings" (Rubin and Rubin 1995, p. 20, emphasis in the original). Thus, I sought to avoid the imposition of theoretical categories as far as possible by letting interviewees talk at length about their experiences and how they viewed religious life. As Rubin and Rubin argue cultural interviews "probe for the special and shared meanings that members of a group develop, the kinds of activities that group members typically do, and the reasons why they do them." (Rubin and Rubin 1995, p. 28). My informants tended to be abstract, even when I pressed them to think of examples or when I used examples. That conceptual approach, albeit by no means shared by all, is also

part of the local culture that, by and large, privileges discussion of the bible and sermons reflecting on everyday life over more experiential forms of worship.

Cultural interviews require rapport between the researcher and the interviewee. Therefore, the choice of informants was confined to participants with whom I had developed a trusting relationship. The cultural interview also engenders empathy (Rubin and Rubin 1995, p. 12). I was conscious of the interview itself being a form of interaction that produces a self through narration rather than a more authentic mode of data collection (Atkinson and Silverman 1997). All "forms of talk – including narratives and interview accounts – are themselves examples of social action. People do things with words, and they do things with narratives. ... they construct their own lives and those of others; they justify and legitimate past, current, and future actions; they formulate explanations; they locate their own actions within socially shared frames of reference" (Atkinson et al. 2003, p. 117).

"The accounts produced by the people under study must neither be treated as 'valid in their own terms', and thus as beyond assessment and explanation, nor simply dismissed as epiphenomena or ideological distortions. They can be used both as a source of information about events, and as revealing the perspectives and discursive practices of those who produced them. ... all accounts must be examined as social phenomena occurring in, and shaped by, particular contexts" (Hammersley and Atkinson 2007, p. 120).

Interviews, as mentioned, are "social events in which the interviewer and for that matter the interviewee) is a participant observer" (Hammersley and Atkinson 2007, p. 120). The interview is thus another form of participation in the field, which varied depending on the setting and circumstances. Therefore, interview accounts ought to be integrated with participant observation to enable an analysis of social action: that is "the practices of everyday life, the performance of social selves, or the conduct of social encounters" (Atkinson et al. 2003, p. 116).

6.2 Participant Observation and 'Engaged Listening'

The fieldwork included observation of what people did and the local environment, such as the layout and furniture of the church, and the characteristics of the homes of participants, including their geographical location. Participant observation has been considered the primary tool of investigation in ethnography. However, it should be recognised that "ethnography is at least as much about conversation as it is about observation" (Forsey 2010, p. 563) and that the reality the researcher is trying to capture is not always easily observable. My preference for listening rather than observing is partly my own 'personal style' of experiencing the field (Forsey 2010). Thus, I adopted an 'engaged listening' (Forsey 2010, p. 567) to uncover how participants constructed their religious identity. "The aim of the ethnographer", as argued by Forsey, "is to listen deeply to an/or to observe as closely as possible the beliefs, the values, the material conditions and structural forces that underwrite the socially patterned behaviours of all human beings and the meanings people attach to these conditions and forces" (Forsey 2010, p. 567).

My participation sought to minimise the artificiality of responses or behaviour which note-taking can cause. The notes taken in the field were sketchy aiming at capturing 'data' (Delamont 1992, pp. 52-54). More detailed field notes were written after meetings and activities with accurate and 'thick' descriptions of the people, places, times, ways of speaking and my personal feelings and impressions. Field notes were written as 'factually' as possible, in a 'naturalistic' frame of mind (Emerson et al. 2001), to avoid imposing limited viewpoints or pre-determining the analysis. However, evocative terms were used to enable emotive memorisation of the events. The analysis of the data from participant observation involves, on the one hand, recollection relying on emotions; and, on the other, systematic reflection on the purposes of the study.

Participant observation and listening, as mentioned previously, concentrated, in large part, on the home group sessions and church meetings, and social events. At the home group, I participated more as my familiarity with the group grew. I asked to be put on the home group's 'treats rota', which allocated an evening a term to each to bring sweets and treats for the home group. In specific settings, I would bring my netbook. I chose to purchase a netbook due to its relatively small size to be less intrusive. The netbook accompanied me to the home study groups and fellowship evenings, but it was left at home during Sunday services and other events. There were obvious practical issues to do with the use of the netbook. It was convenient in recording data as many people spoke at the home study group and fellowship evenings. It was suitable, as I did not move around, but remained seated for most of the time. However, the netbook signalled a distance and an identity. It

was a reminder, sometimes more for me than for the research participants, of being 'at work'.

Van Maanen recommends immersion in the field by "[C]utting one's self off for a time and looking to build a life with one's new colleagues" (Van Maanen 1991, p.41). However, he refers to severing oneself from one's everyday physical, social and intellectual spaces, which is fundamentally different from severing oneself from one's religious identity. The distance created by that artificial boundary 'protected' my identity as a Jewish woman, especially when I found the talk on the Old Testament uncomfortable due to the particularistic and inaccurate interpretation of the text and Jewish culture. The intimacy of the home group made the netbook a 'shield' for my 'other' identity. The distance of the 'professional stranger' (Agar 1980) also allowed a glimpse of the social reality under observation. Thus, 'staying distant' was a way of 'getting close' (Gordon 1987).

The contrast between the settings where I would take notes and when I would refrain, such as during social occasions, also allowed different levels of engagement. The writing in the netbook at home group sought to report almost verbatim the discussion of participants, whilst the field notes recounted the impressions and events of the field. I was aware that writing itself is a filter through which we understand events and remember them. Indeed, the field, as argued by Atkinson, "is not an entity 'out there' that awaits the discovery and exploration of the intrepid explorer. The field is not merely reported in the texts of fieldwork: it is constituted by our writing and reading. … To begin with, the field is produced (not discovered) through social transactions engaged in by the ethnographer. …

Secondly, the 'field' is constructed by what the ethnographer *writes*. ... Thirdly, it is reconstructed and recontextualized through the reader's work of interpretation and contextualization" (Atkinson 1992, pp. 8-9).

7. Data Analysis

The database of the research is constituted by recorded interviews; notes of the discussions at the home group; and field notes from social and church activities and informal meetings. The focus, as afore mentioned, was on talk. Consequently, the analysis considered the broad narratives of research participants that constructed their view of 'being a Christian'. All interviews were transcribed fully and analysed through coding manually, whilst key interviews were analysed through NVivo software. The initial analysis identified emic categories, "indigenous cultural categories" (Hammersley and Atkinson 2007, p. 194), such as 'serving', 'spiritual gifts', and 'countercultural'. The perspective of research participants was then reconceptualised through the etic categories of the researcher (Fetterman 1998, p. 11). These etic categories sought to discern patterns and avoid normativity, as mentioned previously. The development of the Simmelian theoretical framework systematised further these categories. As I progressed with the theoretical framework, the coding through NVivo became redundant and unhelpful. The software could not facilitate analysis across emic and etic categories. Further, the software, whilst helpful in storing and retrieving data, carried the risk of fragmenting the data, which

become 'decontextualised' from their source to be then 'recontextualised' into analytical categories (Hammersley and Atkinson 2007, pp. 155-156).

It is important to recognise that "the process of analysis is intrinsic to all stages of ethnographic research" (Davies 1999, p. 193). Ethnographic research, therefore, needs to have "a characteristic 'funnel' structure, being progressively focused over its course" (Hammersley and Atkinson 2007, p. 160). Efforts were, thus, made to ensure an open mind for alternative interpretations and conceptualisations. I was conscious of the importance of scrutinising the data systematically so as to identify any "discrepant data" (Maxwell 1996, pp. 90-93) and assess whether the analysis needed to be reviewed. Thus, I arranged the interview with Selina because I had the impression she was quite distinctive from the rest of the group. Selina's sensibility is closer to Pentecostalism, which has a focus on experiential worshiping and a more conservative theology.

The texts of the interviews were analysed as narratives (Riessman 1993; Cortazzi 1993). Narratives are ways of structuring one's meaning, of giving sense and order to how one understands oneself. The researcher interpreting a narrative asks "why was the story told that way?" (Riessman 1993, p. 2, emphasis in the original). This includes how a narrative is organised and what emerges as the focal point. Informants in answering questions attempt to provide a coherent and persuasive account, not one that may appear illogical or meaningless. As "meaning-making structures, narratives must be preserved, not fractured by investigators" (Riessman 1993, p. 4). The texts were analysed using contrasts resulting

from the oppositions presented by informants between, for example, 'religion' and 'being a Christian'.

The analytical approach was abductive (Santaella 2005). Accordingly, a general pattern is drawn out from the observation of a case study. Abduction is, in Peirce terminology, a 'guess', a logical and, yet, intuitive reasoning which suggests a different understanding of the reality under observation. In the case study, the observation of Bethlehem called into questions concepts of social capital, belief and belonging, and prompted the development of a better understanding of the reality being studied. Abduction is an inference that calls for the construct of a new type. Abductive inference is "an attitude towards data and towards one's own knowledge: data are to be taken seriously, and the validity of previously developed knowledge is to be queried" (Reichertz 2004, p. 163). As such, abduction results in "mental constructs" (Reichertz 2004, p. 163, emphasis in the original) whose 'validity' lies in functionality. The constructs, at which one arrives through abduction, need to be judged according to their helpfulness in explaining a phenomenon or aspects of a phenomenon. In this thesis, the abductive inferencing challenged pre-constituted concepts of religion, belief, individuality and relationality.

7.1 Writing

Another component of the analytical process is writing. The need for coherence in communicating research findings inevitably raises questions of representation (Hammersley and Atkinson 2007, pp. 203-205). Scholars have experimented with a variety

of writing forms to overcome the 'crisis of representation', the questions over the authority of the researcher in interpreting the culture of participants. The present study being concerned with theoretical development sought to be faithful to local meanings whilst transposing them onto an abstract framework. The 'translation' from a theological discourse to a sociological perspective, as mentioned, might be contested as reductive. However, the claim made here is of providing a sociological perspective that helps the understanding of an aspect of the local culture precisely by severing it from its theological framework.

The writing form sought to balance the requirements of systematic theory with the portrayal of a culture. Ethnographic writing entails the use of evocative language to present a vivid portrait of the observed culture. Thus, chapter eight begins with an ethnographic account of the origins of Bethlehem in a narrative style to be more engaging. The account derives from the interviews of informants. The only factual data altered are the names and location to maintain confidentiality. The choice of a narrative style was in an attempt at conveying the atmosphere and culture of Bethlehem church. However, the subsequent extracts are primarily from interviews and serve the purpose to illustrate theoretical concepts, rather than provide an ethnographic description of the local culture. Some interview responses are reported in their full length to avoid a misappropriation of the data to serve the theory and to avoid the breaking up of more emotional accounts.

The names of the informants have been changed with the names from the characters of George Eliot's novels. This was partly due to George Eliot being one of my favourite

authors, but also an author showing a strong religious sensibility. Eliot translated Feuerbach's *The Essence of Christianity* (1841/1989). Like Simmel, she was influenced deeply by his philosophy. Eliot's novel Middlemarch (1872/2008) is infused with Feuerbach's concept of religion as consciousness and feeling. For Feuerbach, human beings saw their nature outside of themselves before being able to find it inside of themselves. Dorothea embodies that religious sense, which from evangelicalism turns into mysticism. The use of Eliot's characters is merely to conceal the identity of my informants and, perhaps, to evoke a sense of religious sentiment in approaching Bethlehem church. Thus, the names should not be taken as a reflection of any aspect of the participants' personality.

A further note needs to be spent on Casaubon, who is a mostly fictional character in the thesis and is employed to link parts of dialogue from interviews or conversations. Casaubon is also a device to avoid writing myself into the text. The presence of the researcher cannot be denied; yet explicit references to the thoughts and emotions of the researcher, whilst might be useful in a personal journal, do not serve the purpose of representing the reality of the field, rather they risk being a self-indulgent narcissistic exercise (Delamont 2009). Finally, writing often allows a process of discovery of what was not apparent during the observation. It helps connecting the fragments and seeing the patterns through a theoretical systematisation. The account of a small group of Christian evangelicals aims at capturing patterns of identity formation. The research was limited to a small group in one single church, at a specific time and place, and within a specific geopolitical culture. However, ethnographies do not aim to provide 'generalisable' findings, but a plausible account of a culture (Creswell and Miller 2000). The objective of the study

was to develop concepts that may aid the study of religious identity and possibly be extended to identity formation in other contexts.

8. Ethics

The ethical obligations of research include avoiding deception, asking for the permission to record, and being honest about the intended use of the research (Rubin and Rubin 1995, p. 94). Participants were informed of the aims of the research and of my personal background when I was introduced to the church during a Sunday service. This included my religious background and political role of elected councillor. During the entire project, there was no perceived or actual conflict of interest arising from my political role (Israel and Hersh 2006, p. 49). Consent was obtained for each interview by e-mail. Research participants at the home group were aware of my role as a researcher. However, a couple of new members joined later; thus it was difficult to know who was aware of the research at any given point. It should be recognised that in "complex and mobile settings, it may simply be impractical to seek consent from everyone involved. Unlike experimental researcher, ethnographers typically have limited control over who enters their field of observation" (Murphy and Dingwall 2001, p. 342).

During the interviews, informants were open in sharing personal reflections and anecdotes from their lives whilst being recorded. However, as the relationship with some key informants developed into friendship, I began to question what I could include as data without breaching confidence. I left out what I considered confidential, notwithstanding the fact that informants had volunteered very personal information about their lives during recorded interviews. I felt that leaving out personal information was necessary to live up to the trust that informants had in me, which made the research possible in the first place. After all, informants did not know how I would use and present the data.

Finally, researchers have a duty to guarantee that information about participants will be used in such a way as to make "impossible for other persons to identify the participants or for any institution to use them against the interest of the participant" (Flick 2006, p. 49). Research participants were asked for consent for observation and interviews through the elders and individually. It was made clear that every effort was made to maintain confidentiality and anonymity. It should also be recognised that research findings can be reinterpreted and misconstrued once out in the public domain (Lee-Treweek and Linkogle 2000, p. 18). Nevertheless, confidentiality and anonymity cannot be guaranteed under all circumstances.

Research participants do not always appreciate that their reserve is needed for anonymity to be maintained. This became apparent when I was contacted to present a talk on my research and the person making the invitation named the church of my case study. My most absolute denials were to no avail given that the information came directly from the former pastor. Most importantly, "participants and informants will remain identifiable to themselves" (Murphy and Dingwall 2001, p. 341). This posed an ethical question for the

present research once completed. The variance in interpretation of salvation had the potential to create divisions once the participants read the account. Therefore, I contacted Nicholas and agreed to discuss it with the new pastor to make him aware of the situation. I also contacted one of the research participants to discuss the matter more in detail.

9. Conclusion

The chapter presented the research design, methodology and research methods employed in this study. A fundamental question arising from the study of religion involves the risk of essentialising religious forms, of validating their ethical stance, but also of taking a reductionist approach to the study of religion. The theoretical framework aims to avoid these problems by stressing the concept of authenticity as a malleable and cultural vehicle for people's understanding of truth. I argued that an appreciation of the insider's culture is at the core of the ethnographic endeavour. However, it is important to remain critical of the accounts of participants and to focus on the social dynamics of religious life, rather than the 'numinous'. Empathy with participants or personal experience of the divine might aid in the representation of religious life through plausible descriptions (Davidman 2002; McCarthy Brown 2002; A. Geertz 2002); yet the focus of sociological research are human beings, not God. This does not exclude the study of religious experience (Neitz and Spickard 1990; Spickard 1991) or imply a reductionist interpretation of it.

The purpose of sociological knowledge is not to 'explain away' religion, but to interpret an aspect of social life. Thus, one's identity also plays an important role in seeing and interpreting social life. The methodology of the research included *Verstehen* through the epistemological tool of self-reflection. *Verstehen*, however, should not be misunderstood as empathic understanding of another's point of view. Indeed, empathic understanding can lead sociologists to give participants a voice without analysing it through sociological lenses. The 'original' *Verstehen* required the development of theoretical constructs, through which to understand others. Thus, self-reflection becomes an analytical and explanatory tool to discern how identity is played out in the field by research participants as well as by the ethnographer. It uncovers the unspoken rules of behaviour and the assumptions underlying practices and narratives.

The ethnographic researcher thus plays the roles of stranger and neighbour; 'hides and seeks' to grasp a very slippery reality; 'moves in' to appreciate the culture under observation, but also 'moves out' to reflect from another angle. This movement and reflection enables a better 'understanding', analytical inference and theoretical formulation. The present research does not endorse or provide a framework for the ethical assessment of local practices. This is not to reject the value of normative oriented projects employing action research, for instance. Embarking in sociological research cannot and should not lead one to profess neutrality. On the contrary, understanding of others and self-understanding can provide a useful reflection on ethics. Indeed, ethnographic research, with its strong focus on meaning-making and particularistic interpretations, can further our understanding of the complexity and sometimes conflictual nature of ethics.

Chapter Four

Georg Simmel

The 'Sociologist of Modernity'

1. Introduction

The second chapter examined the two main approaches to the study of individual religiosity, namely rational choice theory and spirituality, and found them wanting. Spirituality's focus on subjectivity tends to adhere to an overly individualistic conception of the person and a normative understanding of spiritual practices, whilst rational choice theory reduces subjective processes to a utilitarian calculation. In contrast, the relational sociology of Georg Simmel prevents a solipsistic understanding of the individual, while also avoiding an assimilation of the individual into the group, which would dissolve individuality. Simmel's understanding of the individual is deeply relational. His insights on religion provide the platform for a nuanced conception not only of religious sentiment but

of contemporary forms of individuality. Evangelicals, in the case study, show the traditional trait of Protestant individualism, but also affirm the interdependence of the individual and the need for community. The search for authenticity is an individual's journey that is rooted in relationships. This chapter presents the key features of Simmel's sociology, with particular attention to his epistemological notions, on which I build my theoretical framework in the following chapters.

Following a brief introduction on Georg Simmel as the "sociologist of modernity" (Frisby 1985), the chapter discusses the process of individualisation as 'the tragedy of culture' (Simmel 1918/1968), the conflict between 'objective' and 'subjective' culture, which engenders modern subjectivism. This is followed by an examination of his epistemology, articulated, first, through the Kantian distinction of Forms and Contents⁵, and underpinned by the principle of 'relativity' (*Wechselwirkung*); and later refined in his formulation of *Lebensphilosophie* (philosophy of life). The section on relational individuality stresses Simmel's attention to the relation and tension between individuality and sociality, which is resolved in *Lebensphilosophie*. *Lebensphilosophie* enables Simmel to insert the individual and the social within the broader philosophical concern of the relationship between the particular and the universal. The final section of the chapter analyses Simmel's writings on religion, how religion is affected by modernity and its role in overcoming modernity's 'tragedy of culture'.

⁵ As employed in chapter three, the use of the capital 'F' for Forms, 'C' for Contents and 'L' for Life (*Leben*) is to distinguish the epistemological concepts of Forms, Contents and Life from all other meanings of these words.

2. Georg Simmel, the 'Outsider'

Georg Simmel was born on the 1st of March 1858 in a Jewish middle class family of Berlin. He lived in the centre of the city, at the corner of Leipzigstraße and Friedrichstraße (Coser 1965, p. 1). The father, Eduard, had converted to Protestant Christianity. Simmel never really knew his father who died when he was still young. He was brought up by a domineering mother, Flora Bodstein, and a friend of the family, Julius Friedländer. In 1890, he married Gertrud Kinel, a close friend of Max Weber's wife, Marianne (Vandenberghe 2001, p. 8), and a philosopher in her own right. Their son, Hans, later died in the concentration camp of Dachau. Simmel had a long-standing relationship with poetess Gertrud Kantorowicz, from whom he had a daughter, Angela. His home hosted intellectual evenings with artists, philosophers and scholars, such as Stefan George, Rainer Maria Rilke, Heinrich Rickert, Max and Marianne Weber, Georg Lukács, and Ernst Bloch (Vandenberghe 2001, p. 10).

Simmel's famous lectures were attended not only by his students but by an enthralled public. Yet, an 'outsider' of the academic environment and politics, partly due to antisemitism (Coser 1965, p. 3; Frisby 2002, p. 18), Simmel secured a full professorship only very late in life at the University of Strasbourg. He was one "of the major minds German social science produced around the turn of the century," yet he remained "atypical, a perturbing and fascinating figure to his more organically rooted contemporaries" (Coser

1965, p. 1). He wrote more than two hundred articles and numerous substantive works. He trained in philosophy, writing his thesis on Kant, and considered himself a philosopher (Levine 2010, p. xi). He died of cancer of the liver on the 28th of September 1918, just before the end of the war. He influenced many philosophers and sociologists, including Martin Buber, Georg Lukács and Ernst Bloch, Karl Mannheim, and Martin Heidegger (Frisby 2002, p. 24). The American Robert Park, a student of Simmel's, brought his inheritance to the United States. Nonetheless, Simmel was neglected by American sociology, partly due to Talcott Parsons' dismissive views, notwithstanding his own theory drawing so much from Simmel (Levine 2010, p. xvii). Thus, Simmel, as he himself wrote, died without spiritual heirs, but with the consciousness that those who learnt from him took forward his thought in their own distinctive way.

"I know that I shall die without spiritual heirs (and this is good). The estate I leave is like cash distributed among many heirs, each of whom puts his share to use in some trade that is compatible with *his* nature but which can no longer be recognised as coming from that estate" (Simmel, cited in Frisby 2004, p. lvi).

3. The 'Sociologist of Modernity'

Georg Simmel was, for David Frisby (1985), 'the sociologist of modernity'. He captured the transitory and contingent nature of modernity, its subjectivism and fluidity. He wrote: "the

essence of modernity as such is psychologism, the experiencing and interpretation of the world in terms of the reactions of our inner life and indeed as an inner world, the dissolution of fixed contents in the fluid element of the soul, from which all that is substantive is filtered and whose forms are merely forms of motion" (Simmel, cited in Frisby 1985, p. 49). Simmel was particularly sensitive to the process of individualisation that, for him, was at the core of modernity. Simmel's sociology is characterised by his philosophical preoccupations. He understood modernity within a quasi-Kantian epistemology of Forms and Contents, and, later, of *Lebensphilosophie*.

Simmel's sociological insights are scattered across the many essays and books he wrote throughout his life. It is, therefore, unsurprising that he should have been described as 'unsystematic' (Scaff 1990 and 2005, Kemple 2007), epitomised as an 'impressionist' (Frisby 1981, Collins 1994), and, even, transformed into a postmodern (Weinstein and Weinstein 1993). Attempts have been made at bringing 'order' into the Simmelian oeuvre. Calabro' (2004), Backhaus (2003), Kaern (1990), and Segre (1990) focus on the phenomenological vein in Simmel, identifying natural reality as Simmel's *a priori*. Thus, there are for Simmel 'stable human qualities' (Segre 1990, p. 23), whilst, according to Vandenberghe, human drives, rather than being human qualities, are the product of socialisation (Vandenberghe 2001, p. 41). Levine (2012, p. 2) argues that, contrary to the traditional systematisation of Simmel's thought according to different periods: a Darwinian period, a Kantian period and a final Bergsonian stage of *Lebensphilosophie*, for Simmel integrates Kant and Goethe, who remain his principal interlocutors throughout his life.

Vandenberghe (2001), at first, identified in Simmel's recurrent dualism the underlying principle of his thought; he later rejected this position to emphasise the "aspiration towards wholeness" (Vandenberghe 2010, p. 27), which is present in Simmel's work. A similar position is taken by Pyyhtinen who stresses Simmel's conception of philosophy (Pyyhtinen 2008b, Lehtonen and Pyyhtinen 2008). Levine (2010 and 2012) glimpses an element of 'coherence' in Simmel's complex metaphysical spectrum. I concur with Vandenberghe's later position (2010) and with Levine (2012) in seeing a fundamental unity in Simmel's 'polyphonic' work. It is 'polyphonic' for his early writings are influenced by Spencer's Darwinian philosophy, while later ones resonate with Kant and Goethe. This should not be interpreted in terms of different phases, but of different emphases. As Levine (2012) argues, Kant and Goethe remain Simmel's 'interlocutors' throughout his life. Simmel was in constant dialogue with Kant, but also with Hegel, Schopenhauer, Nietzsche, Plato and Heraclitus. Simmel's 'metaphysical sociology' is not 'unsystematic' or incomplete; rather, like Michelangelo's *non-finito* (unfinished), it provides a glimpse into the 'totality of being', which cannot be crystallised into permanent forms. Simmel's non-finito captures the endless unfolding of Life itself.

4. The 'Tragedy of Culture', or Individualisation

In Simmel's sociology, the 'totality of being' is refracted in the myriad aspects of human relations. From these social relations emerge consistent processes and patterns that impact on individuals. For Simmel, in the modern age, we witness a heightened individualisation

due to the increase in "objective culture", culture that is independent of the individual, over "subjective culture". "Objective culture" refers to the plurality of cultural artefacts and products that are created in modern times. The individual is inundated by these products in an increasingly differentiated society. "Subjective culture" refers instead to the absorption of the cultural products by individuals, process through which the individual develops culturally and morally. Simmel calls this process the "tragedy of culture" (Simmel 1918/1968), which gives rise to a form of alienation. Modern culture overwhelms the person, who seeks refuge in a heightened subjectivism.

"The real cultural malaise of modern man is the result of this discrepancy between the objective substance of culture, both concrete and abstract, on the one hand, and, on the other hand, the subjective culture of individuals who feel this objective culture to be something alien, which does violence to them and with which they cannot keep pace" (Simmel 1909/1976, p. 251).

In modern industrialised society, the individual is "surrounded by an innumerable number of cultural elements which are neither meaningless to him nor, in the final analysis, meaningful. In their mass they depress him, since he is not capable of assimilating them all, nor can he simply reject them, since after all they do belong potentially within the sphere of his cultural development" (Simmel 1918/1968, p. 44). This cultural gap is a result of the division of labour of the modern economy, which creates "objectified cultural forms ... at a rate which exceeds the capacity of human subjects to absorb them" (Levine 1991, p. 107). Simmel acknowledges that human consciousness had to be preoccupied with means in

order for human beings to progress, to have the strength or interest to perform the immediate task without being crippled by the realisation of its ultimate insignificance (Simmel 1907/2004, p 231).

Nevertheless, the "increasing competition and increasing division of labour" render harder the attainment of "the purposes of life," which "require an ever-increasing infrastructure of means. A larger proportion of civilized man remains forever enslaved, in every sense of the word, in the interest in technics" (Simmel 1907/2004, p 232). The conflict of culture is the clash between Life and Form. Life needs Form to be actualised, but it is constrained by it. More specifically, individuals' vocations require actual social roles, which, however, may stifle the person's individuality. Consequently, the yearning for ultimate purpose and meaning of life is connected with the dominance of means over ends in modern culture (Pyyhtinen 2010, p. 52). Against this background, Simmel sees in expressionist art a rejection of Form itself, the search for artistic expression without mediation (Fitzi 2012, p. 189).

Simmel was ambivalent about modernity recognising the benefits to individuals, who became much freer in urban life. Modernity also represented "a great advance for humanity by virtue of encouraging an unprecedented degree of individuation" (Levine 1991, p. 105). Thus, the conflict of culture should not be taken simply as alienation, but as a dialectical mode of thinking (Levine 1991, p. 109). In order to appreciate Simmel's thought, we need to turn to the building blocks of his epistemology: the distinction between Forms and Contents, and the idea of relationality (*Wechselwirkung*).

5. Simmel's Relational Epistemology

Simmel, as mentioned in the previous chapter, employed the classical philosophical distinction between Forms and Contents. Forms are abstractions that can hold different empirical Contents, such as the drives of individuals, motives, and feelings. Simmel was inconsistent in his use of the term Form, which he interpreted as: a synthetic principle; as a principle of structuration of the social; and the "a posteriori crystallisation of energies or interactions in the cultural objects and social institutions," such as the sphere of values, science, art or religion (Vandenberghe 2002, pp. 54-55). Accordingly, in the first instance, Simmel's reference to Form needs to be understood as the subject interpreting reality. In the second instance, the Forms of 'sociation' (of being in social relations) identify empirical regularities into a pattern of sociation. Specific examples are status, role, and norm (Tenbruk 1994, p. 358). Finally, in the third instance, Forms can be understood as 'structures', or better, "schemata which constitute conditions for the intelligibility of the world as a whole or specific aspects of it" (Oakes 1980, p. 10). Forms, as a synthetic principle, are crucial in the apprehension of reality, and are fundamental to Simmel's epistemological principle. I clarify this notion in the section below.

5.1 Synthesis and Sociation

Simmel, like Kant, conceived knowledge of reality as partial and the result of a synthesis by the mind of the observer. He wrote: "human thought always and everywhere synthesizes the given into units that serve as subject matters of the sciences" (Simmel 1908/ 1950, p. 5). The result of the subject's synthesis is a Form that is an abstraction, "from a given complex of phenomena, of a number of heterogeneous objects of cognition" (Simmel 1908/ 1950, p. 7). However, contrary to Kant, for whom *a priori* Forms were universal and constant, Simmel considered Forms as dependent on the Content taken and always changing. Later, when Forms become part of Simmel's *Lebensphilosophie*, Forms are understood to be in opposition to Life (*Leben*). From the never ending flow of Life, Forms crystallise. However they do not fossilise Life; rather, the process of Life is such that Forms are adapted and surpassed by new Forms.

Forms are thus the result of 'synthetic judgement': the flux of reality is abstracted into Forms. The synthetic view is a repeated sifting and ordering (Simmel 1907/2004, pp. 108-110) of reality on the basis of the mind's constructs. Thus, the process is constructed and selective. It is constructed because it reflects the 'epistemological eye' of the knower, whose distance from the object of knowledge determines how the object is seen. The view is selective because the distance from the object makes the synthesis, and thus knowledge, possible (Deroche-Gurcel 1988, p. 356).

"Of course this form of interpreting meaning also connects with the *motives* of the persons who are acting, but it subjects their analysis to the a priori *demands* of thought, through which the transmitted events are first formed into a *historical* context" (Lichtblau 1991, p. 49, emphasis in the original).

The synthetic view is a translation performed by the knower through a reshaping of empirical occurrences and, even, subjective motives, which are Contents, according to the mind's Forms. Such Forms are, as afore mentioned, in constant movement and changing throughout time. Thus, the process of synthesis does not happen according to stable *a priori* Forms of the mind; rather it is relational. Simmel's relationality has two dimensions: a sociological and an epistemological dimension. From a sociological dimension, society is not a 'real entity' from which social relations and interactions derive, but the sum of these (Simmel 1890/1982, p. 19). Thus, a sociological object is 'unified' in as much as its parts are in reciprocal dynamic relation (Simmel 1890/1982, p. 17).

Accordingly, "one should properly speak, not of society, but of sociation (*Vergesellschaftung*). Society merely is the name for a number of individuals, connected by interaction. It is because of their interaction that they are a unit" (Simmel 1908/1950, p. 10). Within this framework, the actions of individuals cannot be explained according to the individual's ends, be they conscious or unconscious (Tenbruk 1994, p. 351); rather, actors are 'sociated', they are conscious of being part of social relations, and act with that consciousness. This relationality "implies a shift from substantial reality to *event*" (Pyyhtinen 2010, p. 68, emphasis in the original). Therefore, society is not a substance (*Substanz*), but an event (*Geschehen*) (Simmel 1908/1950, p. 11). *Geschehen* is "the underlying dynamic relations that give rise" to society; it is "society *in statu nascendi*" (Pyyhtinen 2010, p. 72, emphasis in the original).

Simmel writes that the consciousness of associating, or of being socialised, is a "knowing" (*Wissen*) rather than a "cognizing" (*Erkennen*). It is not the result of rational reflection, for "the mind does not immediately confront an object of which it gradually gains a theoretical picture, but that consciousness of the socialization is immediately its vehicle or inner significance" (Simmel 1910, pp. 377-378). This is so because the consciousness of being associated is dependent on the principle of reciprocity, or relationality. In this case, the relationality of individuals does not stop at being associated; rather individuals understand themselves and others through the typifications they hold of others, but also on the basis of the image they have of themselves in relation to others. Nevertheless, the individual is shaped but not exhausted by social relations. This is examined and developed in chapter six. At this point, it is important to outline the epistemological dimension of relationality, here called 'relativity'. Relationality in Simmel does not stop at being an exchange, but goes deeper and constitutes an epistemological principle.

5.2 The Principle of Relativity

In Simmel's sociology, individual actors are sociated; they are conscious of being part of social relations, and act with that consciousness. The consciousness of being associated is dependent on Simmel's broader principle of relationality (*Wechselwirkung*). Simmel develops Kant's notion of exchange (*Wechsel*) and turns *Wechselwirkung*⁶ (relationality) into the principle underpinning social relations. The principle of interaction

_

⁶ Wechselwirkung is composed of Wechsel (exchange) and Wirkung (effect). It was used by Kant in relation to Newton's 'movement of the planets' and implies a force of attraction and repulsion between the planets (Papilloud 2000, p. 104).

(*Wechselwirkung*), which underlines Simmel's epistemology, requires an all-round relationality. For Simmel everything is in interaction. Sociological knowledge is possible when the interaction presents regularities and patterns. However, this synthetic work of sociological knowledge and, equally, of philosophical interpretation, is also an interaction in itself. We understand reality by synthesising data through the filters of our mental constructs, which give a resemblance of unity.

In Simmel's words, "we inject into reality an ex-post-facto intellectual transformation of the immediately given reality" (Simmel 1908/ 1950, p. 8, emphasis in the original). This transformation is a frame of reference "detached from 'reality', which itself cannot be the immediate subject matter of science" (Simmel 1908/ 1950, p. 8). Relationality (Wechselwirkung) is not mere 'interaction', but epistemological 'relativity'. We cannot understand reality but in reference to its dynamic relations. This relationality is here called 'relativity' to stress Simmel's formulation in the light of Einstein's theory, to which he referred in a letter to his friend Heinrich Rickert.

"My kind of relativism constitutes a perfectly positive metaphysical worldview and it is as much scepticism as is relativism in physics as it is represented by Einstein and Laue. Obviously I have not made clear what I mean when I say that truth is relative. Relativity of truth does not mean to me that truth and untruth are co-relational; but rather, I mean that truth is a relation of contents. Neither of the contents is true in itself, just as no physical object is heavy by itself but

only in a reciprocal relation (*Wechselverhältnis*) with another" (Simmel, cited in Kaern 1990, pp. 78-79).

Simmel, with the principle of relativity, rejects the 'dogma' of "a single truth that needs no proof" (Simmel 1907/2004, p. 104), in favour of the 'interrelationship' of knowledge of the world, according to which "from every point we can attain by demonstration every other point;" such "reciprocity of proofs' is the basic form of knowledge" (Simmel 1907/2004, p. 104).

"Cognition is thus a free-floating process, whose elements determine their position reciprocally, in the same way as masses of matter do by means of weight. Truth is then a relative concept like weight" (Simmel 1907/2004, p. 104).

Truth is, therefore, dependent on our representations. Simmel recognises that, notwithstanding the fact that our actions are "carried out on the basis of representations that are not at all identical with objective being", they demonstrate a certain reliability, expediency and accuracy; "whereas other activities based on 'false' representations tend to injure us" (Simmel 1907/2004, p. 104-105). Thus, usefulness indicates whether our conceptions of reality are true or false. We call truth "those representations that, active within us as real forces or motions, incite us to useful behaviour" (Simmel 1907/2004, p. 105). Most importantly, for Simmel, objectivity rests on the mutual relationship between

representations. The "ideal of objective truth ... finds its supplement and therewith its legitimation through the other" (Simmel 1907/2004, p. 112).

Relativity is thus "the mode in which representations become 'truth', just as it is the mode in which objects of demand become values. Relativity does not mean – as in common usage – a diminution of truth, from which something more might have been expected; on the contrary, it is the positive fulfilment and validation of the concept of truth. Truth is valid, not in spite of its relativity but precisely on account of it" (Simmel 1907/2004, p. 116). Simmel does not undermine the validity of knowledge nor does he think that its validity is relative to a specific frame of reference, such as that of culture or language (Pyyhtinen 2008b, p. 72). On the contrary, Simmel's relativity is the underlying principle of truth, value and objectivity as "a new concept of solidity" (Simmel, cited in Pyyhtinen 2008b, p. 72), a positive doctrine of knowledge resting on reciprocal conditioning (Vandenberghe 2002, p. 41).

"Truth means the relationship between representations, which may be realized as an infinite construction, since, even if our knowledge is based upon truths that are no longer relative, we can never know whether we have reached the really final stage, or whether we are again on the road to a more general and profound conception" (Simmel 1907/2004, pp. 114-115).

In conclusion, Simmel's relativity, is a "regulative world principle", according to which "everything interacts in some way with everything else" (Pyyhtinen 2010, p. 43). The

validity of judgement, whether something is 'really real', is dependent on the relationality of Contents. It rests on a sense of wholeness that is "something of a pantheistic view concerning the whole universe, not only the whole of society;" yet Simmel does not view it as an absolute oneness, but "as a manifold, as a unity of individual, separate elements woven together by relations of reciprocity" (Pyyhtinen 2010, pp. 42-43). Simmel's relational understanding of reality underpins his conception of society and individuality, which is outlined below.

6. Relational Individuality

Simmel, as mentioned previously, understood modernity's tragedy of culture as contributing to the process of individualisation, which involved a heightened subjectivism. He understood the focus on the individual's subjective experience as the result of the dominance of objective culture over subjective culture. The tragedy of culture induced the person to look for authenticity, or "originality" (Simmel 1918/1968, p. 19). Thus, in modern life, there is an inescapable tension between the "restless life of the creative soul" and its own fixed cultural product, "which arrests and indeed rigidifies this liveliness" (Simmel 1918/1968, p. 31). Simmel, however, does not lament a loss of purity brought about by industrialisation; rather, for Simmel, culture is always a synthesis of one's subjective development with an objective value (Simmel 1918/1968, p. 37). He eschews

any romantic ideal of subjective authenticity over objective culture, and identifies in the search for "originality" the symptom of the conflict of modern culture.

"Originality reassures us that life is pure, that it has not diluted itself by absorbing extrinsic, objectified, rigid forms into its flows. This is perhaps a subliminal motive, not explicit but powerful, which underlies modern individualism" (Simmel 1918/1968, p. 19)

Simmel discerns a shift in the cultural understanding of individuality. He viewed the Italian Renaissance as the origin of modern individuality. The Renaissance signalled a loosening of the ties of the medieval community, "a time when individuals believed in no limits to the possibilities of presenting themselves to others with character, distinction and independence" (Simmel 1917/2007, p. 66). This core individuality, for Simmel, unfolds in the eighteenth and nineteenth centuries in two phases. The eighteenth century is characterised by what he called "quantitative individualism" whilst the nineteenth century exhibits "qualitative individualism" (Simmel 1971). Accordingly, "quantitative individualism" identifies the universal ideal of freedom and equality of the individual reflecting the Enlightenment's values of the eighteenth-century; whilst "qualitative individualism" identifies the uniqueness of the individual as espoused by the Romantic movement of the nineteenth-century.

In Simmel's words, "qualitative individualism" is "the individualism of uniqueness" [Einzigkeit], of the particular, whilst "quantitative individualism" is "the individualism of

singleness [Einzelheit]" (Simmel 1971, p. 224; Simmel 1908/1950, p. 81), which is universal. The understanding of individualism of the eighteenth century was firmly based on the ideas of freedom and equality. At its core was the Kantian conception of the individual as an end in itself based on a universal conception of humanity. By contrast, the nineteenth century saw the shift from a universal understanding of individualism to an individualism based on uniqueness. What mattered was that "one was a particular and irreplaceable individual" (Simmel 1971, p. 222). Distinction becomes important: nineteenth century's romanticism calls for the authentic individual, for being oneself to the full. This distinction is useful in understanding authenticity, as discussed in the next chapter. The twentieth century, for Simmel, needs to tackle the tension between particularism and universalism. In the attempt to reconcile the two, Simmel formulates a *Lebensphilosophie* (philosophy of life).

6.1 Lebensphilosophie

In his later work (1918/2010), Simmel re-conceptualised his neo-Kantian framework of Forms through a formulation of *Lebensphilosophie*. *Leben*, here, is modelled partly on Heraclitus' continuous flow. However, Simmel does not abandon Kant; rather he integrates Forms with *Leben*. Drawing on Schleiermacher, Simmel seeks a glimpse of eternity and of the universal in the contingent and particular. He does not conceive of a unity where the particular is subsumed to the universal, but of a dynamic relationship of the two. Simmel re-works his understanding of Forms as essential part of the Life (*Leben*) process. Forms

are in opposition to Life (*Leben*), but are also essential to its process. Thus, Life takes a Form in order to be experienced.

Simmel identifies two dimensions of *Lebensphilosophie*: 'more-Life' (*Mehr-Leben*), which is the fundamental dynamic movement of all life forms⁷, and 'more-than-Life' (*Mehr-als-Leben*), which transcends the Life process to crystallise into Forms (Simmel 1918/2010, pp. 13-17). Thus, *Mehr-Leben* refers to "the drive toward reproduction, common to all organic species" (Levine 2012, p. 37), whilst *Mehr-als-Leben* is the creation of Forms that become autonomous, which include cultural forms and social structures. These Forms, as noted by Levine, re-engage vital processes reshaping them and need thus to be understood as integral part of the Life process (Levine 2012, p. 37). This conceptualisation of Forms overcomes the tragedy of culture by allowing the individual to grasp the 'totality of being' in the constant overcoming of Forms.

Therefore, *Leben* is the fundamental principle of reality. It crystallises into Forms, which, in turn, constrain its constant flow and never reflect the total truth of Life. The relativity principle of *Wechselwirkung* tempers the process of synthesis and maintains the fluidity of Life by conceiving Life and Forms in dynamic relation. In *The View of Life* (1918/2010), Simmel "reversed his position dramatically. Instead of viewing the ongoing life process as threatened by the hypertrophy of objectified cultural forms, he found the ascendance of the idea of Life and the explosion of vital energies so relentless that cultural forms could no longer exert the kind of constraint that they had throughout history" (Levine 2010, p. xxii).

7

⁷ Life forms can be human or animal.

6.2 Self-transcendence

This dynamic relationship between Life and Forms is best understood as a perpetual process of self-transcendence. In Hauptprobleme der Philosophie (1910/1996), Simmel stated that the quest for the absolute, or transcendence, can be realised only through a partial and individual content (Simmel 1910/1996, p. 24). The absolute is never reached. Nevertheless, human beings, "as knowing beings", can transcend themselves by virtue of their capacity for interpretation and awareness of the limits of knowledge (Simmel 1918/1971, pp. 357-8). Human beings recognise the partiality of human knowledge. Knowing that "this one-sidedness" is a necessity of cognition places human beings above it (Simmel 1918/1971, p. 358). Thus, "we do not simply stand within these boundaries, but by virtue of our awareness of them have passed beyond them ... That we are cognizant of our knowing and our not-knowing ... this is the real infinity of vital movement on the level of intellect." (Simmel 1918/1971, p. 358). Human beings glimpse transcendence and thus transcend themselves in the consciousness of their 'not knowing'. Their 'not knowing' is a boundary that is inherent in all human beings. Thus, self-transcendence (Selbsttranszendenz) is a fundamental part of human nature (Jankélévitch, cited in De Simone 2002, p. 462).

Leben is 'this-side' of the boundary (*Grenze*), but it is also on the 'other-side'. The Form provides the boundary (Simmel 1910/2007, p. 73). The boundary is essential to the dynamic process of self-transcendence. Self-transcendence is Simmel's *a priori*, "the true absoluteness" (Simmel 1918/2010, p.10). This "self-transcending consciousness" poses the self between the relative and the absolute (Simmel 1918/1971, p. 364). Simmel does not

want the relationality of Life, its relativity and multiplicity, to be submerged into oneness (Simmel 1910/1996). The absolute totality, absolute *Leben*, is impossible to realise. Forms are necessary for Life to be experienced; yet Forms are temporary because they are within a relational world. Therefore, social reality, for Simmel, is underpinned by a constant becoming and overcoming. Fitzi explains that:

"Human beings are to be seen as 'beings of the limit', because their attitude to the world is determined by the fact that, in every dimension of experience, they find themselves constantly moving between two opposing limits ... the existence of limits is fundamental for their continued existence; the individual limits, however, are steadily overcome in a process that does not abolish its principle but each time establishes a new limit." (Fitzi 2012, p. 189)

In conclusion, modernity, for Simmel, is a time when the person is fragmented in an increasingly differentiated society. Unlike Heraclitus' constant flow, the process of Life (*Leben*) is the underpinning principle of reality, which is experienced through Forms. The fundamental 'energies' of Life take a shape, they take a Form: for example, the vital force of religious "yearning" (Simmel 1911/1997, p. 9) generates religious Forms. However, when a Form is no longer an adequate vehicle for Life's pulses, there is a crisis that can lead to changes to that Form or create new Forms (Levine 2012, p. 36). Forms cannot constrain Life, which enacts a never-ending process of self-transcendence. Within this framework, human beings are self-transcendent beings by virtue of being aware of "this one-sidedness", of being part of a wider reality. Humans are 'this side', but have a glimpse of

totality, the 'other side'. We thus turn to Simmel's conception of religion as a Form and how his insights can be made employed in the understanding of contemporary religious forms.

7. Religion and Modernity

Simmel referred to religion in many of his writings, but his main essay dealing directly with the matter was *Religion*, written in 1906 and redacted in 1912. Other relevant essays are *A Contribution to the Sociology of Religion* (1898/1955), *A Contribution to the Epistemology of Religion* (1902/1997), and *The Personality of God* (1911b/1997), thus spanning from 1898 to 1911. References to religion, however, are present in most of Simmel's works. Simmel, as his friend Max Weber described him, was 'musical', attuned to religious sensibility. Yet, he was 'methodologically agnostic' (Vandenberghe 2010, p. 7): God is not accessible. He stated that religion is "a product of human consciousness and nothing more" (Simmel 1902/1997, p. 121). However, he rejected the reductionism of imputing religion to a projection of human feelings arising from humans' natural state, such as interpreting the fear of God as a reflection of human awe in front of physical nature (Simmel 1912/1997, p. 145). His reflections were on human psychology and the effect the image of God humans hold has on them.

Simmel was acutely aware that the processes of differentiation, individualisation and rationalisation of modern science and technology posed an existential question for

religious forms of the time. Modern science defined the limits of rationality according to what was 'provable' within a framework of positivist science. Science defines as "believable reality only that which is scientifically probable" (Simmel 1909/1997, p. 4). It does not question particular dogmas, but, for Simmel, "the object of transcendent faith per se is characterized as illusory" (Simmel 1911/1997, p. 9). Modernity sweeps away the "object" of faith, as a set of claims regarding transcendence, forcing religion to turn inward. This is not to mean that religion merely recedes into the private sphere; rather that it assumes more mystical tones.

For Simmel, traditional forms of religion had become less binding in modern culture leading individuals to "satisfy their religious needs by means of mysticism" (Simmel 1918/1997, p. 20). A new Form of religion was required to contain the religious sentiment. Mysticism is the new Form that can express human yearning for transcendence. The fragmented modern individual can glimpse psychological unity through *unio mystica*. Therefore, religion is not just about morals but "reconciliation of the elements of man's nature" (Simmel 1904/1997, p. 41).

"The essence of mysticism is that we should perceive behind the given multiplicity of phenomena that unity of being which is never a given fact, and which therefore we can grasp directly only within ourselves as this unity. ... This unifying of the fragments and contradictions of our view of the world by attributing to them one common, all-embracing source may be the earlier achievement of religion, historically speaking, but it is perhaps only of

secondary importance. Of more fundamental significance, especially as far as modern man is concerned, is what religion makes of the contradictions of spiritual life. Just as this theistic or pantheistic mysticism reconciles the fragmentary nature of the world's elements by unifying them in God, so religious behavior brings peace to the opposing and incompatible forces at work within the soul, resolving the contradictions they create" (Simmel 1904/1997, p. 36).

8. The Form and Content of Religion

Religion needs to be understood within the epistemological framework of Forms and Contents. Accordingly, the Form of religion is the cultural product, what we may call, the 'social construct' of religiosity, which is a type of manifestation of the religious sentiment. Religion, as a Form, identifies the religious institutions, customs, rituals and laws that have arisen from social relations. However, religion is not the projection of social relations, nor is religiosity (*Religiosität*) the result of a social process, but the origin of it. Thus, religiosity is an innate disposition, a sensitivity, which is decoupled from a belief in the supernatural.

Cipriani draws a parallel between *Religiosität* and *Geselligkeit* (sociability). *Geselligkeit* is "the natural tendency people have to be together, to form a community, even if conflicts are present" (Cipriani 2000, p. 87). From *Geselligkeit* derives the cultural product of society.

Similarly, religion is the cultural product of the religious attitude. A phenomenological interpretation would view religiosity as an *a priori* essence. However, I argue that religiosity, as a mode of consciousness, is best understood as sensitivity to self-transcendence (cf. chapter six).

Simmel was inconsistent in his use of the terms religion and religiosity. Nevertheless, a fundamental point in his thoughts on religion, is his formulation of religiosity as an attitude, a "yearning" (Simmel 1911/1997, p. 9), which engenders a particular way of experiencing the world. This notion is akin to the mystical experience of religion, which was shared by Simmel's contemporary, William James (1902/1960), whom Simmel disliked profoundly. Nevertheless, Simmel does not reduce religiosity to mystical experience or sentiment, nor, indeed, to religious behaviours, rituals or beliefs. Thus, religiosity is not an attribute of someone who has faith; rather it is a disposition of the "soul", a way of seeing and experiencing life.

It is important to understand Simmel's references to 'soul' as 'psyche', for Simmel's study of religion is a study of the human psyche, rather than a theological reflection. Thus, religiosity, as a Form, acquires a broad and nuanced meaning. It is not a mere preference, but a way of being, which shapes how a person experiences and lives life. Simmel likens it to an artist's sensitivity to the aesthetic aspects of life. Religiosity does not necessarily manifest itself in religious behaviour, just as an artistic sensibility does not make one an artist; rather it is "the fundamental quality of being of the religious soul" (Simmel

1909/1997, p. 10), which can be deeply present in some individuals and only superficially in others.

"What makes a person religious is the particular way in which he reacts to life in all its aspects, how he perceives a certain kind of unity in all the theoretical and practical details of life ... Religiousness thus can be seen in this light: as a form according to which the human soul experiences life and comprehends its existence" (Simmel 1909/1997, p. 5).

Religiosity thus understood is not reduced to religious belief or practice. Being religious is not given by one's adherence to religious norms and practices, or belonging to a religious community, but by a more subtle and thus, difficult to define, personal characteristic that can be expressed in different ways. Consequently, walking in the woods, dancing or teaching can all be infused with religiosity. Religiosity is a 'mind-set' with blurred contours. Thus defined, religiosity suffers from a lack of clear boundaries, but enables a broader understanding of the religious sentiment, which is manifested in a multiplicity of ways. Just as the artist 'sensitivity' is not necessarily expressed artistically or identifies an artistic talent, religiosity does not necessarily lead to religious behaviour.

What makes Simmel's insights compelling for the understanding of the search for authentic religiosity of contemporary evangelicals is the conception of religiosity as a mode of consciousness, which allows us to appreciate it as fundamental to one's individual identity. A person is a 'religious person' and not someone 'with a religion'. However, we should not

disregard Simmel's framework of Form and Content. In the first instance, Simmel does not identify privileged forms of expression of the religious sentiment. On the contrary, any expression is a crystallisation, which will be surpassed or modified in the future. In the second instance, Simmelian Forms open the door to understanding the Form of religion in its synthetic dimension. Forms identify autonomous systems of understanding.

The synthetic principle of Forms gives rise to an autonomous "world" (Simmel 1918/2010, p. 55), like that of religion. As explained by Oakes, Simmel "describes forms as languages into which the world or aspects of it may be translated. These languages may be conceived as general schemata which constitute conditions for the intelligibility of the world as a whole or specific aspects of it" (Oakes 1980, p. 10). Accordingly, each Form "has its own definitive modes and its own characteristic language. Each form produces a representation of the world that is unique to the form itself" (Oakes 1980, p. 11). This understanding of Forms allows us to appreciate religion not just as an attitude, but also as an overarching perspective. In this conception, religion is a schema ordering one's life.

8.1 The Relationality of Religion

The key components of Simmel's religiosity are: its relationality; a propensity to believe; and a "totalising force" (*Totalisierungsvermögen*). These are important characteristics on which I build my theoretical framework in chapters six and seven. At present, it will suffice to outline how Simmel understood these notions. Simmel conceived the individual and all reality as relational. At times, he suggested that individual religiosity and religious

institutions are the crystallisation of social relationships and, in particular, of relationships of trust (Simmel 1902/1997, pp. 125-126; 1912/1997, pp. 157-158; 1898/1997, p. 118). Indeed, he identifies the sentiment of *pietas*, which originally was a sense of devotion to parental authority, and one's own country, as the sentiment underlying religion (Simmel 1912/1997, p. 161). The sentiment of *pietas* becomes religiosity by gaining autonomy from the social realm (Simmel 1912/1997, p. 158). It could thus be argued that this sense of social trust and devotion crystallises in religion.

The relationality of religiosity stresses the importance of social relationships. However, we should be careful not to give a too Durkheimian reading of Simmel's notion of religiosity. Although Simmel does not develop this point, he does not intend to reduce religiosity to social relationships or God to social unity (Laermans 2006, p. 486). According to Laermans, we can distinguish between two social types of religiosity: belief in the sense of social trust, and the experience of social unity, which expresses a feeling of dependency towards a higher principle. Simmel explains that the "individual feels bound to some general, higher principle from which he originates and to which he ultimately returns, to which he dedicates himself but from which he also expects elevation and redemption, from which he is distinct and to which he is yet identical" (Simmel 1912/1997, p. 156).

Simmel's comments on *pietas* are part of his distinction between religions in antiquity, which were whole cultures, and Christianity, which is a belief system. Accordingly, as mentioned in chapter two, in Greek and Roman cultures, "the religious obligation of sacrifice and prayer, indeed the cult as a whole, is not a personal matter but is imposed on

the individual as a member of a particular group. ... religious dedication, the outward appearance of something that is merely socially necessary, in fact constitutes an inherent feature of the fabric of social norms" (Simmel 1912/1997, p. 160). What makes *pietas* religious is the characteristic of "unselfish surrender and fervent desire, of humility and exaltation, of sensory concreteness and spiritual abstraction; and all this occurs not only in alternating moods, but in a persistent unity" (Simmel 1912/1997, p. 161).

Thus, religiosity is not merely a projection, or crystallisation, of social trust. Whilst Simmel himself did not elaborate the connection between religiosity and social trust, we can attempt to construct a more nuanced rendering of Simmel's relationality in relation to religiosity by examining his writings on groups. This is dealt with in the following section, which is essential to my notion of sacralisation and the role of tradition in the construction of authenticity, as put forward in chapter seven.

9. Belief, Social Trust, and Groups

I develop Simmel's conception of religiosity as relational by building on his notion of belief. Simmel contrasted belief, as *believing that*, with belief, as *believing in*. He explains that *believing in* God does not simply equate to holding something to be true; rather "it implies a certain spiritual relationship to Him, an emotional dedication, an orientation of life toward Him" (Simmel 1912/1997, p. 166). By analogy, our *belief in* others does not mean that we

believe in their existence; but that we assume a spiritual attitude in regard to them. In this sense, we can take Laermans' observation of religious belief in the sense of social trust (Laermans 2006). Simmel's notion of relationality is vague. At times, following Lactantius' etymological interpretation of religion from *religare*⁸, he seems to suggest that religion is relational for it provides unity by connecting the person to the world, which is sustained by religion's totalising force. At other times, religion is seen as the result of social relations, which is dependent on the projection of the sentiment of *pietas* onto the transcendent realm.

I retain the first (epistemological) interpretation (cf. section ten), but also develop the notion of *belief in* as the projection of human feelings onto a 'higher principle', which gives legitimacy to one's group and tradition (cf. chapter seven). I do so by linking *belief in* with the projection of human sentiments onto others, one's country and "the gods", put forward by Simmel in his work on the persistence of groups (Simmel 1898a, 1898b, and 1898c). Accordingly, I argue that the feeling of *pietas*, dependency on a 'higher principle', is projected onto the divine, the group and the tradition through which one experiences the relationship with the divine aiding the person's identification with the group. It is the consciousness of a relationship with something higher, such as the divine, which forms the group (Vandenberghe 2001, p. 42). Individualities are thus transcended in the spiritual bond human beings form in the social unity of the group.

⁸ In *Divinarum Institutionum* (IV, xxviii), Lactantius argues that religion derives from *religare* (connecting) against Cicero's interpretation of religion as *relegere* (to treat carefully), as written in *De Natura Deorum* (II, xxviii). http://www.documentacatholicaomnia.eu/02m/0240-0320, Lactantius, Divinarum Institutionum Liber IV, MLT.pdf

"Out of individuals existing side by side, that is, apart from each other, a social unity is formed. The inevitable separation which space places between men is nevertheless overcome by the spiritual bond between them, so that there arises an appearance of unified interexistence" (Simmel 1898a, p. 667).

The group assumes significance superior to that of the individual. It becomes 'immortal'. It is worth, however, consider other elements that impact on groups' continuity. Simmel refers to groups that, whilst changing in membership and forms, maintain continuity in time. He surveys disparate forms of association such as medieval municipal corporations, kingship, and the Catholic clergy. The group goes beyond the life of the individual. Continuity is maintained because change happens slowly and gradually. The new generation does not simply substitute the old generation causing an abrupt shift, but it is inserted into the life of the existing generation so that "the group seems as much like a unified self as an organic body in spite of the change of its atoms" (Simmel 1898a, p. 669). This 'unified self' is central to the Christian tenet of the church as body of Christ. Every Christian is, thus, an essential atom in the body of the church stretching infinitely beyond itself.

Simmel highlights constitutive elements of group cohesion, of which I shall only mention elasticity, conflict, honour, and the philosophy of the group. The tendency of a group to be more or less flexible in form is, for Simmel, dependent on the level of complexity and diversity of the group. Accordingly, a conservative policy will unite a diverse group to ensure social preservation. Yet, quantitative changes need to be accommodated by new

structures lest the structure becomes obsolete and collapse (Simmel 1898b, pp. 831-34). A group's "elasticity" (Simmel 1898c, p. 40) depends on the principles governing its structures. For example, the "aristocratic principle," which can apply to the clergy, the family and other social groupings, is characterised by a "severe conservatism" (Simmel 1898c, p. 37). Further, the type of economic production is also conducive to particular forms of groups. Thus, agricultural economics is conservative for it relies on long term arrangements and is epitomised by fatalism, due to the uncertainties of the weather; whilst the bourgeoisie has an attitude of "fluctuation" and will thus be more prone to adapt and metamorphose (Simmel 1898c, pp. 39-41).

In the final part of *The Persistence of Social Groups* (1898c), Simmel identifies antagonism as playing an important role in cementing group unity. This is not simply true in the case of external animosity against the group but of parallel animosity between two groups or sets of individuals. This is explored more fully in *The Sociology of Conflict* (1904). Simmel sees conflict as an active component of social interaction. Simmel's reflections on antagonism are relevant to the present study in reference to the 'rigidity' of Protestantism, which seeks to be true to 'early Christianity' and thus rejects established authority and reinvents religious forms (cf. chapter eight). Simmel contrasts the rigidity of Protestantism with the elasticity of the Catholic Church, which seeks to contain and subordinate antagonistic powers to the Church. By contrast, Protestantism, for Simmel, has relied on being a 'protest' against something. Thus, when deprived of an enemy outside, Protestantism has repeated the conflict "in its own camp and divided itself into a liberal and orthodox party" (Simmel 1904, p. 681).

Another important factor in binding groups is honour (which I develop further in chapter seven). Honour is a particular aspect of morality, which is not necessarily recognised as 'good' by the wider society. Honour, in fact, encapsulates the distinctive character of the group and forms the boundaries of conduct, but also of group membership. Thus, honour among thieves coalesces the class (*Stand*) of that particular social group. Honour is, for Simmel, "one of the most thorough means of maintaining the existence and specific significance of the group" (Simmel 1898a, p. 683). Honour appeals to specific ideas surrounding the essence of the group and the distinctive conduct, which form the boundaries of the group. A similar dynamic to that of honour is given by the philosophy of the group which, although deriving from the individual members, becomes the structure of the group.

Simmel's structures are not to be intended in a 'structuralist' sense; rather they "represent the idea or the power which holds the group together in this particular respect, and they, at the same time, consolidate the group coherence so that it passes from a mere functional to a substantial character" (Simmel 1898a, p. 685). These structures result from individuals projecting their "spiritual content" onto an abstract level of reality, such as country, friend and gods (Simmel 1898, p. 685). The 'gods' are endowed with attributes of the human psyche. This process gives strength and power to the 'gods' and turns them into a source of moral authority.

"Think, for instance, of the idea of the gods, whom men first endowed with all sorts of qualities, worthiness, and excellencies reflected from human souls. Then

the same men used these gods as a source of moral laws and of power to enforce them" (Simmel 1898a, p. 685)

In chapter seven I build on this concept to construct the process of 'sacralisation'. For the time being, it is important to recognise that a person's *belief in* others transforms how that person sees them. Accordingly, the person idealises the people who are the object of belief. She projects onto them hopes and ideas drawing on their specific characteristics. It is not a simple projection of the self onto others; rather the people around us arouse in us this spiritual attitude, which gives rise to a deeper bond. *Belief in* God thus shares the same process of idealisation.

10. Religiosity, Relationality, and Unity

The reflections on groups, presented in the previous section, show the subtlety of Simmel's thought. His conception of the projections of feelings of trust and social relations onto the transcendent sphere may be reminiscent of Durkheim (1912/2008) and, indeed, of Feuerbach (1841/1989). Nevertheless, I believe we should always bear in mind the epistemological value of relationality (*Wechselwirkung*), as the regulative principle underpinning his sociology. Accordingly, society is the sum of social relations and interactions (Simmel 1890/1982, p. 19). Everything is in interaction and therefore 'in relation' with everything else. I suggest we understand religiosity as a Form structuring the

experience a person has of reality: religiosity is relational in its ability to synthesise and make sense of reality. Religiosity provides connectedness in the sense that it connects all aspects of life and "resolves all contradictions" (Simmel 1904/1997, p. 36). It bestows, in Simmel's words, "unity".

"God conceived of as the unity of existence can be nothing other than the agent of this interlinking, this interaction between things" (Simmel 1912/1997, p. 201).

We thus come to the third element of religion, after relationality and belief: that of a 'totalising force' (*Totalisierungsvermögen*). I argue that religion provides unity to the person on the basis of its totalising force. In the individualised and fragmented modern world, religiosity's 'totalising force' goes beyond 'making sense' of life; it creates a paradigm for the religious mind. It provides a sense of unity to the person by ordering life's contradictions in a 'spiritual', or religious, dimension. Thus, one lives a 'religious life' in every moment, and one's religious identity, in the sense of self-understanding, becomes paramount. The multiplicity and contrasting aspects of life are unified within the religious consciousness. Religiosity not only unifies "the fragments and contradictions of our view of the world by attributing to them one common, allembracing source", but "brings peace to the opposing and incompatible forces at work within the soul, resolving the contradictions they create" (Simmel 1904/1997, p. 36). Thus, the totalising force of religiosity counterbalances the fragmentation of modern individual identity and provides a unified self-understanding by reconciling the

multiple aspects of the self that are played out in social relations. It is, at least in part, a solution to the tragedy of culture.

Simmel's quest for 'unity', to which religion gives expression, is not a mere search for meaning, but the response of religiosity to modernity's fragmentation of the self. The modern self is made of fragments searching for unattainable unity; religiosity is where the self is reconstituted. Simmel parallels this unity with the unity consciousness projects on the person (Simmel 1911b/1997). Accordingly, consciousness bestows wholeness to the person. Human beings are made of physical and psychological elements held together by consciousness, which gives an image of unity to the person. I argue that religiosity holds the person together, for it is a mode of consciousness. A similar process happens in art, reference to which might clarify further this notion.

Works of art, once separate from the artist and life, are objectified cultural forms contributing to modern "alienation", but they are also a means to overcome it (Dörr-Backes 1995, p. 127) for art provides a unifying perspective. Art, just like religion, is the locus for the integration of particularism and universalism. The modern paradigm, for Simmel, is to seek eternity and the universal in the contingent and particular. Simmel does not seek a unity where the particular is subsumed in the universal, but a dynamic relationship of the two, which, in art, he finds in the impressionism of Rodin (Dörr-Backes 1995, p. 126). Simmel sees in Rodin's art the self-transcendence of his own *Lebensphilosophie*, aiming to bridge the gap between being and becoming. In this modern Heracliteanism, as much as in Rodin's art, "all substantiality and solidity of the empirical perspective has turned into

movement. In restless transformation a quantum of energy flows through the material world, or, rather, is the world" (Simmel, cited in Pyyhtinen 2010, p. 58).

In summary, the previous development of Simmel's thought on religion highlighted three core concepts through which we can begin to construct a nuanced conception of religiosity. In the first instance, the use of Simmelian Forms allows an understanding of religiosity as a way of being, which goes beyond its specific manifestations, such as theological beliefs, customs, and rituals. This overarching sensitivity underpins all these dimensions and gives them meaning. Religiosity is a personal disposition, present in different degrees in different people and, indeed, absent in others, from which religion ensues in whatever form, thus not necessarily according to established religious canons. In the second instance, belief is intended as *belief in* and thus encapsulating a relational attitude of trust. This feeling is projected onto the divine but also one's group and tradition through which the divine is experienced thus binding and legitimising the group. In the third instance, I interpret religious relationality according to Simmel's epistemological principle of relativity (*Wechselwirkung*). Accordingly, religion provides unity to the person's mind. This is sustained by its totalising force (*Totalisierungsvermögen*).

11. Conclusions

Simmel's sociology of religion offers the opportunity to view religiosity not as an attribute of someone who has faith; rather it is a disposition of the person, a way of seeing and experiencing life. Religiosity, however, is not mere 'spirituality' – as we might call it today – but an overarching system, a Form. The advantage of using Forms lies in the capacity to encapsulate an overarching perspective rather than essentialising particularities such as religious belief or belonging to a religious community. Accordingly, religiosity is more than the 'sum of its parts', be they beliefs, rituals or emotions; rather religiosity is an all-encompassing mode of consciousness. In viewing religiosity in its dimension of Form, we free it from the constraints of constructs such as the propositional concept of belief, as an adherence to dogmas, or belief in the existence of supernatural entities. Influenced by Feuerbach (1841/1989), Simmel seeks to reach towards religion's 'essence', that is the inner psychological aspect of religion.

Simmel's 'musicality' in matters of religion assumes quasi-mystical tones, which could be advanced into explorations of the experience of the numinous. However, Simmel is interested in the modern human psyche. Thus, I pursue his reflections on religion in relation to his understanding of the person's psyche caught between individuality and relationality. Simmel's sociology does not subsume individuality to the group, nor does it concentrate on individuality at the expense of the group. It is a composite sociology that dares tackle the fundamental philosophical questions. Equally, his philosophy retains a

deep understanding of the social and captures the human quest for knowledge in a world that is being constructed and re-constructed continuously. Simmel bestows upon us an understanding of religion based on a relational individuality, which aspires to universality. Accordingly, in chapter six, I construct religiosity as a sensitivity to self-transcendence, on which I build my proposed concept of authenticity.

Chapter Five

Authenticity

1. Introduction

The previous chapter outlined Simmel's conception of religiosity as a sensitivity, which is distinctive of the heightened individualism of the modern age. Religiosity is an inner disposition that is lived out in one's everyday life. It is a call to "become who you are" (Simmel 1915/2007). However, as proposed in chapter six, the authenticity of religiosity is not limited to fulfilling one's particular individuality, but has a self-transcendent (universalistic) dimension. I argue that religiosity is a sensitivity to self-transcendence: the sense of being 'this side' (cf. chapter four) and the attempt at going beyond it. This is not manifested in necessarily ethical behaviour nor is it limited to a deep mystical experience of the numinous; rather it is the sense of a wider horizon of truth. The underlying argument of the thesis is that religiosity is being refashioned through the prism of

authenticity due to a heightened individualism of modernity and, at least partly, in response to pluralistic values.

This chapter focuses on the normative assumptions underlying the most influential formulations of authenticity today. It is a necessarily cursive account that aims to question the presumed pre-modern ethical uniformity and the conflation between ethics and authenticity. Thus, the second section explores Charles Taylor's concept of authenticity and MacIntyre's ethics. MacIntyre does not concern himself directly with authenticity, yet his ethical standpoint is very close to that of Taylor. His reflections, however, are useful in pointing to the inalienable influence of tradition on human beings. Accordingly, the intensification of subjectivity in the modern world is seen as unanchored from a universalistic ethics (Taylor 1991 and 2007; MacIntyre 1981), which leaves the individual alone in shaping one's identity and ethics. Their outlook on modernity is predicated on the notion of a pre-modern shared ethical framework, which is undermined by the 'inward turn'.

The next chapter outlines a relational conception of the individual, which counters the reflexive and, at times, solipsistic individualism of contemporary conceptions of the individual. Therefore, this chapter deals with two specific problems: the presumed loss of moral and religious consensus of the pre-modern era; and the ethical aspirations of contemporary philosophers to compensate for such a loss. I should clarify that I do not apply a strict separation between the terms 'moral' and 'ethical'. This is partly due to the

fact that their etymological meaning is the same⁹, but also due to the lack of consensus in differentiating the two in the literature. I tend to use 'ethical' and 'ethics' in reference to abstract principles while 'moral' and 'morality' in reference to practices. However, this is not grounded in a theoretical framework, but my own 'custom'. The following section discusses the assumptions underpinning Charles Taylor's and Alasdair MacIntyre's philosophy; while section three explores Alessandro Ferrara's notion of authenticity, which is partly inspired by Simmel, and Simmel's ethical reflection. The fourth section clarifies my political-philosophical standpoint and introduces the concept of authenticity I propose in chapter six.

2. The Modern *Fall* and Authentic Salvation

In his book *A Theory of Justice* (1972), John Rawls envisaged a value neutral state as guarantor of justice. Rawls' liberalism was far from the classical liberalism espoused by John Stuart Mill, imbued with classical moralities (Devigne 2006). Rawls' liberal conception of society was grounded in the notion of a bounded and self-sufficient individual, at the centre of a well-ordered and homogeneous society maintained by an "overlapping consensus" of principles chosen through the device of the "veil of ignorance," implying that all rational beings would choose the same fundamental principles. This was met with

_

⁹ 'Morality' derives from the Latin *mos* (custom, one's disposition/character), while 'ethics' derives from the Greek \mathbb{Z} ethos (custom) and \mathbb{Z} ethos (character). On a humorous note, 'ethics' as the philosophy of morality is the customary definition for Italian crosswords, which no doubt settles the matter once and for all.

opposition from – what became known as – communitarian philosophers, including Alaisdair MacIntyre and Charles Taylor, among others.

For his detractors, Rawls had detached individuals from their communities, where meanings are formed, and disconnected them from the processes through which moral imperatives, or virtues are developed. In *Political Liberalism* (1993), Rawls recognised that modern democratic societies are characterised by a plurality of beliefs ("comprehensive doctrines") that can often be irreconcilable. He developed further his idea of "overlapping consensus", the agreement, between citizens, on principles of justice that enable coexistence without prejudice to diversity. This is dependent on a "reasonable political conception of justice" (Rawls 1987, p. 10) that is shared by different comprehensive doctrines to guarantee a stable and secure polity.

In his seminal work, *After Virtue* (1981), MacIntyre argued that the conception of the 'good' is embedded in tradition, for "man without culture is a myth" (1981, p. 161). He revived Aristotle's account of virtues, whose practice develops the person morally, together with her understanding of the good. MacIntyre criticised the supposedly neutral and universalistic stance of liberal individualism. Kant's project of morality, based on rationality, had failed. Modern society, in MacIntyre's eyes, is affected by "emotivism", which "is the doctrine that all evaluative judgments and more specifically all moral judgments are *nothing but* expressions of preference, expressions of attitude or feeling" (1981, pp. 11-12, emphasis in the original). As such, agreement in moral judgement cannot be secured by any rational method (1981, pp. 12). Yet, it is not clear on what evidence

MacIntyre argues that there was ever a society where moral agreement was determined on the basis of rational argument, rather than might or persuasion.

MacIntyre criticised the universalistic claims' of liberalism for failing to acknowledge that liberalism is itself a tradition. Ironically, MacIntyre has refused to view liberalism as a complex and multifaceted tradition favouring, in its stead, a narrow conception of the 'liberal project' that claims neutrality. His communitarian conception of the aftermath of the Enlightenment seems to be the 'fall' of 'modern man' from the Garden of Eden of *Gemeinschaft*. Yet, whilst individuals are undoubtedly relational, the idealisation of communities by communitarian philosophy neglects to appreciate how communities have too often imposed homogeneity and hierarchies, and marginalised or excluded dissenters. MacIntyre's emphasis on tradition is valuable, however, it comes at the cost of individual freedom and the internal diversity of communities. He also seems to assume that all different communities will come to a similar understanding of the good, or one that is not antagonistic to the others, if they only practised the virtues from within their tradition.

In contrast, Taylor, in *The Sources of the Self* (1989), identified the subjective turn of modernity as the inheritance of the Protestant Reformation, and considered the turn inward not simply an affirmation of individual autonomy, but "a fragmentation of experience which calls our ordinary notions of identity into question" (Taylor 1989, p. 462). Thus, the 'condition of modernity' not only encapsulates the fragmentation of morality, but, paradoxically, of individuality. The subject turns inward and the 'I' becomes a fragmented 'self' in search of authenticity. The search for authenticity is, for Taylor, a form

of self-absorption originating in the three "malaises" of modernity: "the fading of moral horizons", "instrumental reason", and "loss of freedom" (Taylor 1991, p. 10). The extreme individualism of the modern age, not tempered by a "higher purpose" (Taylor 1991, p. 4), leads to an obsession with self-fulfilment.

Following Tocqueville, Taylor sees the democratic equality of modern times leading the individual to "self-absorption" (Taylor 1991, p. 4). Individuals are "enclosed in their own hearts" (Tocqueville, cited in Taylor 1991, p. 9) and thus unable to form communities. Premodern *Gemeinschaft* is lost for the atomised *Geselleschaft*. This leaves a moral vacuum and a search for impossible fulfilment. Echoing Trilling (1972), Taylor seeks to recuperate the ethical dimension. Thus, he argues that the modern individualistic idea of self-fulfilment is a valid idea that needs to be rescued from extreme subjectivism. Self-fulfilment is not vain egoism or relativistic self-referentiality. There is a moral ideal behind self-fulfilment which is "that of being true to oneself" (Taylor 1991, p. 15). This fulfilment is to be found in something "which has significance independent of us or our desires" (Taylor 1991, p. 82). It is in something 'bigger than ourselves', where we can find meaning.

Authenticity, for Taylor, is the foundation for a modern identity that results from one's moral stance. Thus, identity "is defined by the commitments and identifications which provide the frame or horizon within which I can try to determine from case to case what is good, or valuable, or what ought to be done, or what I endorse or oppose. In other words, it is the horizon within which I am capable of taking a stance" (Taylor 1989, p. 27). Taylor's moral horizon is based on "hypergoods", "goods which not only are incomparably more

important than others but provide the standpoint from which these must be weighed, judged, decided about" (Taylor 1989, p. 63).

This is explained clearly by Varga (2012), who subscribes to Taylor's notion of authenticity: authenticity "acknowledges that our wholehearted engagements have a grip on us as on persons embedded in horizons of significance that are partly constituted by qualitative distinctions of worth. While we constitute ourselves through our choices of commitments, these commitments are linked to ideas of the *good* that qualitative *distinctions of worth* are based on" (Varga 2012, p. 160, emphasis in the original). Here, wholeheartedness "means that betraying your commitment would also mean betraying yourself (centrality), and it involves being committed to both the actual project and to entertaining the commitment itself (continuity)" (Varga 2012, p. 160).

This notion of authenticity is based not on a solipsistic and self-enclosed self, but a "dialogical" self. As Taylor points out, "in the culture of authenticity, relationships are seen as the key loci of self-discovery and self-confirmation. Love relationships are … the crucibles of inwardly generated identity" (Taylor 1991, p. 49). For Varga, "authenticity is about articulating goods from a collective horizon by way of a commitment which requires responsiveness to these more or less locally constituted goods" (Varga 2012, p.161). However, it is not clear how our "interlocutors" (Taylor), or "a collective horizon" (Varga) may compel the individual in pursuing an ethical path. Taylor recognises that there are different views of what constitutes the 'good', which one needs to confront.

Taylor does not put forward a substantial conception of the good, nor does he subscribe to Kantian universalistic ethics. The self of Taylor has "moral sources", but no categorical imperatives. Thus, there are no criteria on the basis of which "hypergoods" should be debated and decided. At times, he seems to believe that there are universal ethical principles upon which agreement is possible, although he does not set out how such agreement is to be achieved. Thus, Taylor constructs a normative but procedural understanding of authenticity to be contrasted with the self-centred version of authenticity, fruit of the malaises of modernity. It is not clear, however, what makes this redemptive version of authenticity more than self-absorption apart from the 'dialogue' with other 'selves', who might be seeking the same authenticity in very different ways and in ways that might be antithetical to those of others.

Alasdair MacIntyre (1981) and Charles Taylor (1989, 1991 and 2007) unwittingly construct a mythical pre-modern world of cohesive communities, where people believed in God and thus shared an overarching moral universe. One is left confused as to their disregard for the plurality, complexity, and conflictual nature of the past. Undoubtedly, the many abuses of the Church and Christian monarchs were not simply the result of dogma, but were part of a wider geopolitical and economic structure. However, Christian 'morals' were imposed with violence by the ruler, be that a monarch or religious authorities, on their own subjects and neighbours. The fiercest materialist would still need to acknowledge the ideological-religious character of the massacre of the Cathars during the Albigensian crusade, in 1209; the expulsion of the Jews from England, in 1290; the persecutions of 'heretics' and non-Christians by the Spanish Inquisition and by various

European monarchs; the massacre of the day of St. Bartholomew, in 1572; and the burning at the stake of Giordano Bruno, in 1600, to list but a few examples.

A political incentive, if not motive, can be found in all these examples. However, this is partly due to the fact that religion was not about 'belief in God' or a 'cosmic embeddedness', but the conception of the order of nature, society and power an elite imposed on their competitors and the rest of the population. Far from being a cohesive and uniform world, the pre-modern era was torn apart by political-religious wars. Indeed, the principle *cuius regio*, *eius religio*¹⁰ had to be proclaimed after drawn-out wars, and, of course, did not include individual freedom of conscience. The 'cosmic embeddeness' of the Peace of Westphalia is very far from the normative conception sung by Taylor or MacIntyre. For Taylor and MacIntyre, the 'melting' of the pre-modern 'solids' (Marx and Engels 1848/2002, p. 223)¹¹ at the hand of heightened individualism has shattered communities and the overarching universalistic ethics.

_

¹⁰ The principle of the ruler determining the religion of the ruled was recognised in 1555 with the Peace of Augsburg, part of the Peace of Westphalia, concluded in 1648.

¹¹ The reference is used ironically to stress the normative tone of MacIntyre and Taylor and the many abuses of Marx's quote. Marx referred to the economic system of capitalism requiring constant transformation and its impact on society. "The bourgeoisie cannot exist without constantly revolutionising the instruments of production, and thereby the relations of production, and with them the whole relations of society. Conservation of the old modes of production in unaltered form, was, on the contrary, the first condition of existence for all earlier industrial classes. … All fixed, fast-frozen relations, with their train of ancient and venerable prejudices and opinions, are swept away, all new-formed ones become antiquated before they can ossify. All that is solid melts into air, all that is holy is profaned, and man is at last compelled to face with sober senses his real conditions of life, and his relations with his kind." (Marx and Engels 1848/2002, p. 223).

The condition of modernity is thus ethical fragmentation where we are no longer "governed by impersonal standards justified by a shared conception of the human good" (MacIntyre 2007, p. ix). Although both MacIntyre and Taylor accept pluralism, they presume that diverse societies, without a shared 'sacred canopy', are inherently conflictual and individuals no longer inhabit communities, which are sources of meaning. Alessandro Ferrara shares similar ethical preoccupations, although his formulation of authenticity is more nuanced and sophisticated, as outlined in the next section.

3. Authenticity and Simmelian Ethics

Taylor's notion of authenticity has been criticised by Alessandro Ferrara (1998) for being essentialistic. Ferrara, an admirer of Simmel, formulated his notion of authenticity by grounding it on Kant's reflective judgment, Aristotelian *phronesis* and Weber's notion of ethical responsibility (Weber 1919/1970). Kant's reflective judgment refers to those judgments that are not a generalised universal norm to be applied to given particulars, but judgments arising from given particulars. Specifically, the judgment on taste, from which Ferrara draws his notion of validity, is a subjective judgment that makes a universalistic claim of validity by expecting recognition from others. In other words, when we appreciate an object and call it beautiful, we expect others to share our judgment. Reflective judgment is, therefore, intersubjective. Similarly, Aristotle's *phronesis* identifies the ability to choose ethically independent of general laws. The conclusions made by *phronesis* cannot be

proven, only 'shown'. Thus, one's choices are shown to others as a model that needs the recognition of others.

Ferrara proceeds to formulate authenticity as an expression of the core of the actor's personality. He draws from Weber's sense of responsibility espoused in *Politics as Vocation* (1919/1991). The sense of responsibility, for Ferrara, "shares all the characteristics of authentic conduct", for it "combines considerations of expediency and value with a deep emotional resonance which stems from the link that this kind of action manifests with the *identity* of the actor" (Ferrara 1998, p. 6, emphasis in the original). Accordingly, authenticity implies faithfulness to one's character, or congruence with one's personal identity. Ferrara, here, takes from Georg Simmel's "individual law", according to which morality cannot be judged on the basis of single actions, but on the basis of the totality of individual life. The "individual law" is duty that is expressed in the entire life of a person: "law can stem from the life unity of the individual unfolding as obligation – or more precisely, the law must be the instantaneous arrangement of it" (Simmel 1918/2010, p. 124). In order to understand Ferrara's interpretation, I explain Simmel's individual law in the section below.

3.1 Simmel's Individual Law

In his attempt to reconcile the universal moral law with individual subjectivity, Simmel conceived of individuality as the door for universalistic value. He rejected the impersonal and objective Kantian universalism whilst seeking to avoid relativistic subjectivism.

Specifically, he rejected the general and abstract law of Kant's categorical imperative, according to which single acts are judged as ethical or unethical on the basis of their accordance with general moral principles imposed by human reason. In its stead, he believed that each act needs to be understood as part of a whole of one's life. Individuals live out, in their own values, universal norms (Vandenberghe 2000). For Simmel, ethical conduct, or better 'duty' (*Sollen*), is dependent on the individual person: the person expresses herself in the act. Ethical conduct requires the whole person; it is not exhausted by the action; rather it is an expression of the self.

"law can only stem from the life unity of the individual unfolding as obligation – or more precisely, the law must be the instantaneous arrangement of it" (Simmel 1918/2010, p. 124)

Morality, for Simmel, does not originate in Kant's universal reason, but it is apprehended "in the inner uniqueness or solitude in which it is experienced, then morality itself originates from the point where the person is alone with himself, and to which he finds his way back from the 'broad way of sin' – whose breadth signifies not merely its alluring ease, but also its accessibility for all" (Simmel 1918/2010, p. 115). Duty is not imposed by reason (Kant), but by one's subjectivity. However, duty is not subjective, but objective, whether it is recognised as such or not by the individual (Simmel 1896/2008, p. 37). This is so because one's subjectivity is connected with Life (*Leben*). Simmel identifies duty (*Sollen*) with Life itself (Simmel 1896/2008, p. 35). As explained by Levine:

"... life is a single process that encompasses the constituents acts, acts that cannot be judged as moral in isolation. The whole of life enters into each discrete action of the individual, such that there is a universality of the individual, not of the act or moral principle. We cannot, however, deny the objectivity of this individualized morality. Such denial causes humans to slip into pure amoral individualism. Rather we must recognize that the 'individual law', the universality of the individual, has a certain objectivity or 'materiality' to it. This conception, Simmel argues in a final added passage, means that the whole life is responsible for every act, and every act for the whole life" (Levine 2010, p. xx).

The 'ought', as Silver and his colleagues explain, is part of life. It follows that "the relevant ethical question is not how norms do or do not match reality, but how a norm is living and charged with emotional energy rather than dead or dull. This means that a genuine law, Simmel argues, must require me to give my whole self, including my being and my feelings; such laws must speak to each individual as the individual he is" (Silver et al. 2007, p. 272). Life becomes the realisation of the 'ought', in the sense of the expression of what we, as individuals, are meant to be. The 'lawfulness' of the 'ought' needs to reach deep into the heart of the individual, without it being relativised. In the 'individual law', Simmel attempts a reconciliation between particularism and universalism through the notion of 'eternal recurrence'.

Taking his cue from Nietzsche, Simmel adopts the idea of 'eternal recurrence' of existence as "the synthesis of the need for the infinite and the need for the finite" (Simmel 1907/1991, p. 176). Simmel does not subscribe to Nietzsche's idea fully, but adopts it in as far as it allows a synthesis between infinity and finitude. The normativity lies in the idea of recurrence instilling in the person the obligation of living every instant with the conscience we would have if living eternally. Simmel, explaining Nietzsche, stated: "We are responsible for our conduct in a new way, or at least we understand our responsibility differently, if we know that no moment of our life is ever over once and for all, but that we and humanity must experience it innumerable times just as we shape it now" (Simmel 1907/1991, p. 171). This is, however, based on the "persistence of the ego, which finds a new meaning and new consequences in the first instance in light of the reappearance of the same contents a second time" (Simmel 1907/1991, p. 173-174).

3.2 Ferrara and the Individual Law

Ferrara interprets the 'individual law' as "a principle of authenticity" (Ferrara 1998, p. 65). Accordingly, authenticity is not just the self-realisation of "one's needs and inclinations but also of the rest of the determinations of one's identity, including normative contents" (Ferrara 1998, p. 68). It is not a pursuit of personal wellbeing, but it includes the living out of duty. This means that even "the choice to renounce one's 'self-realization', if made and executed with reflective awareness is a choice *for* self-realization" (Ferrara 1998, p. 68).

Ferrara's sophisticated formulation of ethical authenticity is grounded in a teleological or, at least, a 'thick' conception of being human. He identifies qualities such as "coherence," "maturity," "depth," and "vitality" that suppose a high level of personal commitment and self-consciousness. This is in line with Simmel's position, which followed Nietzsche's idea of nobility. The "individual law" requires a high level of self-reflection and thus pertains to the "virtuosos" of ethics. As such, it is not a conception of ethics that can be generalised to all, but a path for the 'mystic' to follow. Ferrara's ethical authenticity is mostly concerned with personal development and assumes specific human characteristics as universally good. Thus, he seems to rely on each individual's reflexivity to reconcile divergent views. From this perspective, it follows that our actions ought to embody our ideal or, indeed, authentic individuality. The 'ethical life' is thus conceived by Simmel "as the perfection of the individual" (Lee and Silver 2012, p. 131). Lee and Silver explain that "[E]ach individual may have her own categorical 'ought' toward which her life aspires but what that ultimate 'ought' consists in can only be a product of vigorous self-reflection on the part of this individual in conversation with others" (Lee and Silver 2012, p. 134). However, intersubjectivity does not constitute objective morality. Indeed, it does not even ensure shared moral values. Lee and Silver seek to reconcile the 'objective value' of the individual ideal with the individual's relationality. They write: "Though ethics may manifest as rights and duties between individuals, the form of ethical life is not at all limited to its social facet. A's realization of his fullest potential through a social relationship may contribute to B's ethical fulfilment" (Lee and Silver 2012, p. 139).

An individual's self-realisation 'may' contribute to another's self-realisation or, indeed, it may not. It may, as it is often the case, be of obstacle to another or, even, harm another. In Lee and Silver, there is an unspoken assumption of the 'good life', which only needs to be understood and lived out by people. Ethical self-realisation, just as much as Simmel's 'individual law', betrays an implicit or, in the case of Simmel, explicit assumption of nobility. Authenticity is not an enterprise for the faint-hearted, but for a selected few. This approach not only deprives the individual law of universality, but of duty. The individual might feel the duty, but they might not. If ethics has no imperative, can it really motivate to act justly? My main concern is not, however, the question of universalistic or particularistic ethics, but the conflation of authenticity with ethics, to which I turn in the next section.

4. Agonistic Pluralism and Authenticity

The implicit or explicit 'ethics of authenticity' not only fails to provide an adequate conception of the conditions and norms to facilitate moral consensus; it also normativises a particular understanding of being human, as an intellectualist inward pursuit, which pays lip service to dialogue with other 'selves'. This might be suitable for philosophers, but it will not serve the needs and wants of the general population. My intent is to formulate a concept of authenticity that is not inherently ethical, but that can be informed by ethical concerns. One that is in relation to one's interpretation of tradition, but also to wider society and culture. Thus, before I begin to construct my conception of the subject and of

authenticity, in the next chapter, I wish to motivate further my disagreement with the ethical perspective of communitarian philosophers, such as Taylor and MacIntyre. This is a concise 'normative comment' to clarify my ethical standpoint.

The only 'normative comment' I allow myself to express is a disclaimer rather than the exposition of an argument. MacIntyre is right in reaffirming the importance of tradition. However, what constitutes tradition is a very selective account of events, norms, and practices and is exclusive of dissenting voices and experiences. My interest lies in individual agency and how individuals interpret what they understand to be tradition, how they redefine its boundaries, and how they innovate it. This is based on a broadly liberal stance, which respects particularism as long as it does not deny the fundamental rights of the individual. I consider the value of individual freedom of conscience over and above the value of 'communities'. I feel communitarian thinkers take for granted their ability to be individuals and free, which is still denied to many, especially to women, including in Western liberal democracies.

For Simmel, heightened individuality was a blessing of modernity rather than a curse. In *Sociology* (1908/1950), he argued that people are more closely involved in small groups, while larger groups require structures. The distant character of the larger group frees the individual from the control of other members. Thus, the larger entity enables diversity. I value the internal diversity of groups and societies, whose interests, needs and desires compete, clash, but also coalesce. I consider liberal societies agonistic, a position similar to that of Chantal Mouffe and Norberto Bobbio. Following Bobbio (1984/1991), Mouffe

(2005) accepts that liberal institutions are essential for pluralism and that total pluralism is not possible and that the exclusion of some points of view is necessary for groups to be able to co-exist peacefully. In an agonistic society, individuals as well as communities pursue their own interests, including the search for truth. This can certainly lead to antagonism and outright conflict. It is the 'art' of political compromise that seeks to accommodate divergent interests, visions of the good, and freedoms.

Compromise, as Simmel wrote, "especially that brought about through exchange, no matter how much we think it is an everyday technique we take for granted, is one of mankind's greatest inventions" (Simmel 1908/1955, p. 115). I believe that it is in the controversies, debates and exchanges that we come to have a view of the good, question it and refine it. The 'good' is not a static category, but the fruit of what we make of our existence. Even what we may call 'shared values', as human rights might be, are interpreted differently by different people, groups and cultures. Yet, this diversity should not be taken as a sign of impoverished communities, insurmountable conflict, or a lack of a universalistic framework. Taylor's, Ferrara's and MacIntyre's promotion of particularism, be that of individuals or of groups, detached from an overarching universalistic framework, relativises all conceptions of the good (Insole 2004). This very problem is at the heart of Simmel's distinction between 18th century quantitative (universalistic) individualism and 19th century qualitative (particularistic) individualism. In the post-Enlightenment era, qualitative individualism has become a pre-eminent value. Modernity has accentuated individual's consciousness. This has resulted in the internalisation of the universal Kantian

duty and its translation into a personal conscious choice to make life meaningful, and to grow morally and spiritually.

The Christian evangelicals, in the present case study are refashioning their identity around the notion of authenticity against the background of pluralistic values, multi-faith and multicultural diversity. Christian authenticity, observed in the fieldwork, is undoubtedly ethical (cf. chapter ten). Nevertheless, the conception of authenticity I formulate avoids ethics for two reasons: firstly, to eschew the same normativism and essentialism of the thinkers mentioned in this chapter; secondly, to account for a wider meaning of authenticity. In the first instance, it needs to be reiterated that authenticity, as 'being true to oneself', includes being true to one's character, one's wellbeing, one's ideals, which is not necessarily 'good' for others. One could conceive one's authenticity in opposition to others. One could even conceive of authenticity, as informed by the 'good', but a particularistic good that might be in conflict with other conceptions of the good.

In the second instance, a wider formulation allows us to appreciate that there may be things people value, which are not necessarily concerned with ethics, or solely with ethics. For example, although Christian evangelicals, in the study, practise an ethical authenticity, they do not deem ethics as exhaustive of authenticity. Indeed, evangelicals are conscious of the fact that ethical behaviour can be found outside of religion. Authenticity, more than ethics, identifies the attempt to be true to truth, which in the present case study is identified with the example of Jesus. The truth of Jesus is, for research participants,

universal in that it can or, indeed, should be recognised by all. It is transcendent, although experienced in the immanence of everyday life.

Accordingly, my conception of authenticity develops Simmel's notion of immanent transcendence: "transcendence reveals itself as the immanent condition of life" (Simmel 1918/2010, p. 17). I interpret it, perhaps reductively in order to apply it to my empirical work as a person's sense of truth, which takes the person beyond "this one-sidedness". The concept of authenticity, here proposed, refers to a process of identity construction that is informed by one's understanding of the truth, which may or may not include a conception of the good. Truth, whilst embedded in culture, enables the person to transcend her social and personal identity. Thus, actors construct their identity in relation to a meaning-system, which may include ethical norms, but that is not exhausted by them. It follows that there cannot be authenticity outside a tradition. The human condition is one of situatedness, as MacIntyre might argue. However, contrary to MacIntyre, I do not believe that modernity has left us a fragmented society where it is impossible to hold shared values. Pluralism, with its acceptance of particularistic viewpoints, implies mutual respect, which rests on a liberal universalistic conception of the polity. It entails an agonistic conception of culture and society that need not degenerate into antagonism.

5. Conclusion

The chapter reviewed the most prominent formulations of authenticity, namely those of Taylor and Ferrara, as well as MacIntyre's thought on tradition and ethics to highlight some of the recurrent assumptions on the individual, society and ethics in much philosophical reflection. These assumptions, as mentioned in chapter two, are recurrent in sociology as well as in philosophy. They are central to a particular understanding of modernity, which, however, lacks historicism and betrays normative concerns. Normative conceptions of authenticity are underpinned by an understanding of ethics that is seemingly nonconflictual and by a teleological view of human beings. This can lead to a hierarchy of goods and qualities with little room for interpretation.

A view of the 'good life' as part of one's identity neglects the fact that not all people might be concerned with developing and pursuing an ethical ideal. Further, the conflation of character with ethics may lead to conflating moral responsibility with moral quality. In other words, without external moral norms, one's conduct might be judged simply on the basis of one's consciousness rather than actions. A 'mature' understanding of moral questions does not necessarily lead to moral actions, as many philosophers of morality have shown in their own lives. Schopenhauer (cf. chapter seven) was acutely aware of the value of compassion; yet this did not stop him from mistreating his landlady, being scornful of people around him, and being disliked by his own mother.

The concept of authenticity, I propose in the next chapter, therefore eschews ethical assumptions to avoid essentialism, but also to provide a more faithful account of the notion of authenticity, observed in the field. Authenticity is thus a process of identity construction, which is articulated in relation to tradition and one's community, but also wider society. It is a practice of being true to – what is understood to be – truth. One's identity is thus not only defined by one's social relations and individual character, but also one's understanding of truth, which, contrary to Taylor's "commitments," is not necessarily ethical or universal. This conception of authenticity rests on a tripartite systematisation of the 'self' as 'social', 'individual' and 'authentic', which is explored in the next chapter.

Chapter Six

Authentic Self and Tradition

1. Introduction

Building on the previous chapter's critique of normative authenticity, I offer a formulation of authenticity as a process of identity construction. From the perspective of informants, (*emic*) authenticity is normative: it is a process of moral development at the communal and individual level. However, the thesis avoids endorsing the informants' narratives and experiences by distinguishing between *emic* and *etic* authenticity. Thus, the schema of *etic*

authenticity provides a framework for the sociological understanding of *emic* authenticity stressing the fact that the terms 'truth' and 'universal' need to be understood in relation to local narratives. The concept of authenticity, here proposed, is grounded in a relational understanding of the subject, the 'I', which counters the over individualistic assumptions of current understandings of individual religious identity and authenticity. The chapter begins by presenting a relational account of the self ('I and Thou') in three dimensions: a social self, an individual self, and a self-transcendent self. This systemisation develops Simmel's notions of self-transcendence, relationality, and individuality. The 'I and Thou' is an interpretation and development of significant notions in Simmel's relational sociology and philosophy, which privileges – what might be defined as – a Hegelian reading of Simmel's work.

This is followed by an account of the process of authenticity. The first part of the explanation stresses the distinction between authenticity and sincerity. Authenticity is an attempt at understanding 'truth', which is not limited to ethical concerns, and truth is always understood from within a tradition. This is illustrated through a reflection on dramatic art. Dramatic actors construct the identity of a character drawing on their own personalities, but also from acting styles, or traditions, and from shared patterns of behaviour. This reflection avoids a normative and essentialistic understanding of authenticity and an application in other fields, such as philosophy, politics and art, to name a few. This formulation of authenticity does not assume ethics or, indeed, belief and belonging. It is a framework for the understanding of an 'existential' identity.

The authentic identity is 'existential' in that it identifies the movement of self-transcendence of the person, the going beyond one's social context and individuality. Self-transcendence, however, need not be understood as a deep mystical encounter or philosophical reflections. 'Existential' identity results from one's social and personal dimensions and a wider horizon of truth. The wider horizon of truth may be felt as universal, however, it is always comprehended, experienced, and articulated through the particularistic narratives and symbols of one's tradition. In turn, what is considered sacred marks the boundaries of this process, as explored in the next chapter. This applies to religious tradition as much as political, artistic or philosophical traditions.

The 'I and Thou': a Systemisation of the Self

Drawing on Simmel's use of *Verstehen* (understanding) in social encounters, his conception of individuality, and the notion of self-transcendence, I put forward a relational systemisation of the self, which I term 'I and Thou'. The proposed 'I and Thou' schema provides an abstract model of construction of individuality in relational terms. The 'I' is conceived, at the same time, as social and individual, but also as 'self-transcending'. This systemisation is not Simmel's own classification, although Simmel refers to 'I and Thou' in positing the intrinsic relationality of human beings and of understanding. I thus employ the term to stress the relation between the individuality of the 'I' and the alterity of the 'Thou'. The focus being on the identity of the 'I', the 'Thou' represents the implied relationship that

constitutes the individual identity. I distinguish between three types of selves, to which I prefer referring as 'I' to underline their subjectivity, rather than reflexivity, and to maintain accordance with Simmel's terminology. These are: the 'social I', shaped by the social settings and social roles the person plays, where 'Thou' expresses social interaction; the 'individual I', which is specific to the person, like "a shading" (Simmel 1910, p. 382), thus distinguishable from 'Thou'; and the 'self-transcendent I', which captures the individual's attempt at transcending one's contingent and particularistic reality.

The tripartite systemisation of the 'self' is based on an understanding of the 'social I', the 'individual I' and the 'self-transcendent I' as types, for they capture the process of 'typification', or generalisation, of social interactions. Pyyhtinen (2008a) argues for an interpretation of Simmel's individuality as a 'type' and suggests three possible interpretations: "an already pre-existent model" independent of actualisation, according to which 'type' is similar to form; an individual type that needs to be created, such as an artist's peculiar 'mode of expression' which does not exhaust the person of the artist but identifies what is unique of the artist; and the individual's fulfilling of the moral 'ought' (*Sollen*) (Pyyhtinen 2008a, pp. 289-291). My choice of a tripartite schema serves the purpose of distinguishing between the individual 'type' as particularistic individuality and the universalistic horizon of self-transcendence. They are three separate but interlinked dimensions that seek to give expression to the social, individual and self-transcendent dimension of being human, as outlined below.

2.1 The 'Social I'

The individual, for Simmel, is shaped by social relations, although not exhausted by them. From social relations arise social generalisations that fix the image of the individual into a character. Accordingly, the 'I' is 'social' in the sense that emerges from and is constituted by social interactions. This should not be intended as a mere reduction of one's identity to social roles, but as a relational identity. Social interactions shape one's identity, but also create multiple typifications of the 'I' that are expressed at any given point. Human 'social' nature means that people 'act out' their persona, which is shaped by social relationships. Thus, the individual does not simply act in a social role; rather his 'personal character' is stylised in social interaction (Simmel 1908/1950, p. 48).

In the 'social dimension', the person experiences others, but also herself through typifications. When we interact we are mindful of the image we have of ourselves and relate to others according to this self-image. This changes according to the social situation in which we find ourselves and the people who participate in it. We thus form a 'picture' of ourselves and of others that captures the salient 'social information' and allows exchange. As Lichtblau explained it: "We always see the *other* only in the mirror of our own generalisations and typifications, and we gain our *self image* conversely only through a 'generalized other'" (Lichtblau 1991, p. 48, emphasis in the original). However, as Pyyhtinen suggests, Simmel's Thou should be distinguished from Mead's generalised other and other conceptions of the other as "the 'alien I' (such as the transcendental subject or the unconsciousness) with some external force (i.e. God), that is other to the I, as well as from the concept of 'otherness' employed by, say, feminist or postcolonial theory"

(Pyyhtinen 2010, p. 93). Thou is not alien to 'I', indeed, I and Thou participate in the same existence, albeit independently.

"... we experience the other person, the Thou, both as the most alien and impenetrable creature imaginable, and also as the most intimate and familiar. On the one hand, the ensouled Thou is our only peer or counterpart [Pair] in the universe. It is the only being with whom we can come to a mutual understanding and feel as 'one'. ... On the other hand, the Thou also has an incomparable autonomy and sovereignty. It resists any decomposition or analysis [Auflösung] into the subjective representation of the ego. It has that absoluteness of reality which the ego ascribes to itself" (Simmel 1918/1980, pp. 106-7).

'Thou' enables one's understanding of oneself as much as one's understanding of others depends on one's self-understanding. This self-reflective knowledge is grounded on the epistemological dimension of *Verstehen* (understanding), as originally put forward by Dilthey (1900/1972) and later employed by Simmel. Accordingly, we understand others by reflecting upon ourselves and we understand ourselves through our relations with others. *Verstehen* should not be misunderstood as empathic understanding of another's point of view, but a process of 'reconstruction' of our own as well as of others' reality (cf. chapter three). For Dilthey, I experience my own individuality "only through a comparison of myself with other people; at that point alone I become aware of what distinguishes me from others" (Dilthey 1900/1972, p. 231). We have no knowledge, argues Dilthey, of the

inner reality of others; therefore we need to 'reconstruct' the person to make sense of what we grasp.

Dilthey's notion of *Verstehen* is taken up by Simmel through the notion of the 'Thou' as understanding. 'Thou' as a category originates from experiencing "the other person or the Thou as a unified entity" (Simmel 1918/1980, p. 106). Simmel goes a step further and considers the "concept of the Thou or the other person and the concept of understanding" as expressing "the same ultimate, irreducible phenomenon of the human mind. The Thou, as we might put it, is its substance, and understanding is its function" (Simmel 1918/1980, p. 107). Understanding depends on alterity. Therefore, we understand others through ourselves, but also reconstruct the person through typifications, by creating images of others through which we relate and interact. 'Thou' as a category here stands to underline the consciousness of being associated, of being "by nature a political animal" (Simmel 1918/1980, p. 107), in the sense of being enmeshed in relationships.

The 'social self' is, therefore, the self that is conscious of associating and of being socialised (Simmel 1910, p. 377). This means that each member of a community looks at other members according to the constructs developed through the socialisation process of being part of the community. One looks at other members as 'a member of my group' (Simmel 1910, p. 380). It is the delimitation (*Begrenzung*) between individuals and between individuals and groups that ascribes clarity to collective structures (Simmel 1908/2007). Thus, the process of 'sociation', which invests others with their social place, their distance and closeness to us, "condenses" in specific patterns of social life (Simmel 1890/1982, p.

21). The more complex these patterns are, the more difficult it is to peel away all these layers and characterisations.

The process, however, is always ongoing and proceeds through subsequent and constant 'changes and reshapings' (Simmel 1910, p. 381), which prevent a complete characterisation of the individual. Thus, the individual does not become a static character, a mask imposed by social generalisations. On the contrary, Simmel's dynamic process of 'changes and reshapings' allows generalisations to be remodelled. It is this constant remodelling that prevents social interaction producing rigid characterisations of individuals. As mentioned, the fundamental principle of reality is, for Simmel, one of perpetual overcoming. Yet, it is not solely the dynamism of Life (*Leben*) that keeps individuals from being reduced to a 'mask', but personal individuality, without which social interaction would not be possible. Thus, the social 'I' does not exhaust the individual; there is an individuality that is personal.

2.2 The 'Individual I'

The individual dimension captures the distinctive character of the person, the personal side of the individual, that which is not "turned toward the group" (Simmel 1910, p. 381). The individual thus retains individuality. Social generalisations ascribe specific characteristics to the individual without ever capturing the actual self of the individual, but always painting a picture that forms a personality through which the individual engages in social relations.

"... the generalisation is always at the same time more or less than the individuality. That is, the individual is rated as in some particulars different from his actual self by the gloss imposed upon him when he is classified in a type, when he is compared with an imagined completeness of his own peculiarity, when he is credited with the characteristics of the social generality to which he belongs" (Simmel 1910, p. 381)

The social setting does not shape and absorb the individual completely, rather an "extrasocial being, his temperament and the deposit of his experiences, his interests and the worth of his personality, ... gives the individual still, in every instance, for everyone with whom he is in contact, a definite shading, and interpenetrates his social picture with extrasocial imponderabilities" (Simmel 1910, p. 382). It is through this individuality that the individual can come to be part of the group. Without one's own personal side, one could not interact with others. Individuality does not arise in isolation, but through the encounter with the "Thou', who recognises our individuality. "Thou' enables the self to overcome its own fragmentary nature. As Simmel put it: "We are all fragments, not only of the universal man, but also of ourselves" (Simmel 1910, p. 379). Thus, the relationship of the 'I and Thou' poses, at the same time, "conflict ... for the human consciousness, and the first consolidation as well" (Simmel 1984, p. 153).

The 'individual I', however, is not only the characterisation of the individual arising from social interaction and through which individuals interact with each other, but also the image we have of our own 'self' which is never concretised fully. Simmel uses the term *Individualität der Eigenheit* (individuality of one's own) thus emphasising "the irreducible singularity of the individual's life" (Pyyhtinen 2010, p. 154). In contrast with quantitative individualism (*Einzelheit*) and qualitative individualism (*Einzigkheit*), which refer to generalities, the individuality of one's own (*Eigenheit*) refers to the particular concrete individual. Individuality, for Simmel, is the overcoming of the contrast between mind and body. One's own specific "soul", or mind, belongs to one's specific body, and each unfolding into each other (Simmel 1910/2003, p. 13).

The specificity of one's own individuality grants individuals an 'image' of themselves. Thus, individuality refers to our 'personality', our 'character', in the same sense in which we recognise our actions as being 'in character' or 'out of character', expressing that unity underpinning our 'fragments'. It is one's consciousness or, in Simmel's words, 'personality' that bestows a sense of coherent unity to the person. Human beings are made of physical and psychological elements held together by consciousness, which gives an image of unity. Consciousness, this unified image of the 'I', or self, does not nullify its relationality. The 'I' is the outcome of constant inner as well as outer interactions. One's relationships with the outside world and the inner workings of the mind thus shape one's psyche.

The representations of memory are an example of how the mind gives a sense of a unified 'I'; one, however, that is perpetually altered by the dynamic relationship between past and

present. Thus, the past shapes us in the present as much as the present shapes the image we have of ourselves in the past. This perpetual process prevents the realisation of the 'personality', as Simmel puts it, "in the absolute sense" (Simmel 1911/1997, p. 51). One is never complete and, therefore, seeks unity perpetually. It is this longing for unity from which the search for authenticity emerges.

2.3 The 'Self-transcendent I'

The 'self-transcendent I' is based on Simmel's idea of self-transcendence, which encapsulates the recognition of human beings being 'this side', but also their longing for 'totality'. Vandenberghe notes that Simmel "continuously hints that there is something beyond reality, something that transcends mere life, encompassing all of its moments and giving it its unity" (Vandenberghe 2010, p. 7). Vandenberghe gives a metaphysical reading of Simmel's transcendence, which reflects Kant's move from the transcendent to the transcendental (universal). Kant's transcendental enacts a shift from the transcendent God to universal consciousness. For Vandenberghe, Simmel's transcendence is immanent because it is the unity to which one's subjective consciousness aspires.

The divine is then experienced from within, shifting the emphasis from religion to religiosity, piety and spirituality (Vandenberghe 2010, p. 7). Accordingly, Simmel's religion is grounded in human consciousness, and authenticity is attained through the cosmic Thou, which encompasses the self and the other, as well as the absolute Thou. It is through Thou that "the subject can thus realize its unique identity and attain authenticity" (Vandenberghe 2010, p. 21). Similarly, Podoksik suggests that Simmelian individuality needs to be reconciled with universality or, as he puts it, totality. For Podoksik,

individuality and totality are reconciled through a "radicalization" of qualitative individualism (Podoksik 2010, p. 139). I feel Podoksik's and Vandenberghe's Kantian interpretation of Simmel falls prey to essentialism. This may indeed be imputed to Simmel. In his book on Michelangelo (1910/2003), Simmel refers to the "metaphysical sentiment" of the universal, from which one's own destiny flows and from which one's life acquires meaning (1910/2003, p. 58).

Nevertheless, in both Vandenberghe's and Podoksik's reading, the totality of authenticity submerges the 'I' under an absolute 'Thou'. I feel it is a metaphysical understanding of Simmel's thought that should be resisted. Simmel does not seek a unity where the particular is subsumed in the universal, but a dynamic relationship of the two. I thus propose a more Hegelian approach to stress the simultaneous presence of opposites. Hegel recovers Heraclitus' conception of reality as becoming, unfolding in actuality, which is akin to Simmel's *Lebensphilosophie*. For Hegel, 'becoming' and knowledge happen through a dialectical process. Simmel's opposition between Forms and Life, especially in his late philosophical phase, shares Hegel's processual conception of actuality and knowledge. Life 'becomes' through Forms although it never reaches absolute *Leben*.

Against Romantics, Hegel argues that the absolute cannot be grasped through immediate experience (*Erlebnis*); rather it requires mediation, which he develops in his dialectics. Accordingly, knowledge of reality goes through a process of "sublation" (*Aufhebung*) of opposites. The synthesis, or sublation (*Aufhebung*), of the opposites of thesis and antithesis is a process of overcoming (*aufheben*), which retains the opposition. The verb *aufheben*

means cancelling but also preserving, so the sublation cancels and preserves thesis and antithesis. It is not a mere union of opposites; rather the contradiction is essential to the dialectical process. The Real, understood as the Idea (*Begriff*), or logos, has in itself its own unfolding. This happens first by becoming 'objective' through alienation, which is then overcome to return to itself, not in its original state, but in a mature and developed one.

Referring to Hegel, Simmel explains that opposites come together in a higher unity, in which the sense and value of each is preserved, and so is their opposition (Simmel 1910/1996, p. 57). The absolute is multiplicity, for the opposition is not nullified. The absolute is also never detached from empirical actuality. The finite is not separate from the infinite; rather the infinite 'becomes' actual through the finite. Simmel is very sympathetic to this way of thinking. For Simmel, Michelangelo attains perfection and redemption of life in life itself, in moulding the absolute in finite form (1910/2003, p. 64). The overcoming of the dualism of body and mind, accomplished by Michelangelo in his statues, is not a placid perfection, but retains the conflict of opposites. In Christianity, the reconciliation of the internal contradictions of being happens in the transcendent; in contrast, for Hegel, the reconciliation is not outside things but in themselves in the process of overcoming the opposite through its higher form (Simmel 1910/1996, p. 58).

'Being' thus reaches itself only in the infinite path of 'becoming' (Simmel 1910/1996, p. 59). The individual, for Hegel, needs to replicate the same path of unfolding to know the 'truth'. By philosophising, one overcomes the finitude of one's consciousness and elevates the 'empirical I' to the 'transcendental I', to Reason and Spirit (*Geist*). Hegel's view of

knowledge is, I believe, useful in order to appreciate Simmel's self-transcendence and immanent transcendence. I interpret this unfolding, or becoming, of the 'I' as coming to know oneself in relation to the world. Immanent transcendence is thus a process of going beyond oneself to know oneself in a wider perspective.

This brief reference to Hegel is merely to stress the role of becoming rather than to tackle Hegel's wider oeuvre. I suggest Simmel's "immanent transcendence" ought to be understood in a dynamic sense, based on the concept of Life going beyond itself. "Life is that which at all points wants to go beyond itself, reaching out beyond itself" (Simmel, cited in Pyyhtinen 2010, p. 146). For Simmel, the individual reaches out towards an ideal authenticity that brings together not only the fragments of their own selves but also "of the universal man" (Simmel 1910 p. 379). It is a higher unity that does not dissolve the inner contradictions. It is the development of one's individuality. I understand this as a process of identity construction by widening one's horizon of understanding of oneself and the world around. This may have an ethical component, but it is not essential to it.

Grounding the search of authenticity on self-transcendence means that the self is not made complete by a static unity. On the contrary, "transcendence reveals itself as the immanent condition of life" (Simmel 1918/2010, p. 17). Thus, authenticity is not merely the realisation of what is distinctive of the person, the individual uniqueness, rather it captures the person's self-transcendence. The 'self-transcendent I' expresses an individuality that transcends the individual. It is a search for reconciliation of the particular individual with truth, i.e. what is felt as going beyond the 'here and now'. Truth is understood within the

framework of one's tradition, which is not confined to religion, but can include philosophy, art, politics, as mentioned. It is also not necessarily ethical or applicable universally, although it may be understood thus by the agent. It captures what the actor understands as truth, which is refracted through the language, ideas and practices of tradition. It may be considered universal, but it must be grasped through culturally specific mediations and representations. Accordingly, it is a process of identity construction inscribed within one's frames of reference.

Simmel did not use the term authenticity; rather he theorised individuality stressing identity. Authentic, from the Greek *authentikós*, originally meant having authority over oneself. Modern authenticity tends to be associated with the Shakespearean verse "to thine own self be true" 12, uttered by Polonius in *Hamlet*, which has given the word a more pronounced sense of being in accordance with one's character (Guignon 2004). The accordance with one's character is often understood as ethical thus betraying an understanding of authenticity as being in accordance with one's 'better' self rather than oneself. Simmel has a normative approach and conceives of the development of one's individuality as an ethical task (*sittliche Aufgabe*) (Pyyhtinen 2010, p. 137). The term *Sittlichkeit*, employed by Hegel, is the noun deriving from the adjective *sittlich*, which means customary, deriving from *Sitte* (custom). Thus, the individual law is ethical and represents Simmel's attempt at combining individuality with a universalistic ethic. He sought to reconcile the individual with the universal in his concept of the individual law (cf. chapter five).

-

¹² Act 1, Scene 3, *Hamlet*.

In addition, notwithstanding the individual law being the ethical pursuit of the individual, Simmel has a relational understanding of ethics. Duty is not just abiding by ethical principles (as in Kant) but responsibility to a specific you. Morality (*Sittlichkeit*) is, for Simmel, of "an altruistic-social nature" (Simmel 1908/1950, p. 260). The definition of the moral is "the devotion of the 'I' to the 'thou' (in the singular or plural)" (Simmel 1908/1950, p. 260), from which flow philosophical doctrines of ethics. However, in recognition of contemporary pluralism and my understanding of ethics as agonistic, my intent is to avoid a conflation of authenticity with ethics. Thus, I propose a conception of authenticity as a process, where one's conception of truth, however understood by the actor, is interpreted through one's understanding of tradition, be it philosophical, artistic, or religious tradition, to name a few. Authenticity thus reflects the dynamism of Life, the continuous search for one's knowledge of oneself and the world around in a wider perspective. This may include personal moral development; it may also bring one to an antagonistic understanding of the world and violent action.

3. The Process of Authenticity

The pursuit of authenticity is grounded on a continually self-transcending process. I interpret the notion of self-transcendence to include the adoption of a wider horizon of truth, which may be understood as universal. It is thus a transcendence of one's social

conditioning ('social I') and of one's personal perspective ('individual I') to tend towards universality. "Tending' (*streben*) is characteristic of the Romantic Movement, but also of Hegel's and Simmel's conception of the individual. Human beings can transcend themselves through their awareness of being 'this side'. This need not be a deep mystical experience or philosophical insight, but a process of understanding of oneself and the world around. *Emic* authenticity, emerging from my informants' narratives, is trying to "be like Jesus". It is a doing articulated communally as seeking to create an inclusive church (cf. chapter eight), individually as seeking to be true to one's 'better' self (cf. chapter nine), and universally as incarnating compassion (cf. chapter ten). *Etic* authenticity is a process of constructing one's identity by seeking to grasp – what is understood as – truth intellectually, emotionally and in practice. The identity results from the person's attempt at being true to her understanding of truth. This is further explained in section four.

In this section, I want to reflect on the search for authenticity not being confined to morality. To illustrate this point, I draw from reflections on dramatic art. Dramatic acting allows me to draw a distinction between sincerity and authenticity (in contrast with Taylor and Trilling); to emphasise the processual and multidimensional nature of the concept, which does not solely depend on the subject, but is in dynamic relation with others and one's interpretation of tradition; and to espouse a concept of truth experienced through the particularism of individuality and culture. Accordingly, authenticity in acting refers to the process of conveying what is felt to be true about being human through the particularism of the actor, the story, the character, but also of the acting tradition, such as that of Stanislavski, Brecht or Artaud, to name a few. To an actor, a specific acting tradition might

feel truer than another. Acting traditions presuppose specific interpretations of what is truly human, which may or may not be supported by empirical evidence.

There is nothing moral or immoral about an acting style. The dramatic actor expresses authenticity through the language of the style he or she values. For Simmel, dramatic art is not merely a convincing and emotional representation of reality; rather representation is only the first step the actor takes to transform their internal reality and the reality given in the play into a work of art (Simmel 1911/1968, p. 95). I build on this notion to provide an example of authenticity through the art of dramatic acting. The actor's individuality is always present in dramatic representation. It is the source of the emotions used to animate the character, but also one's manner of speaking and moving. The actor's authenticity in dramatic art, however, does not lie solely in the projection of his or her personal feelings onto the character; rather it also rests on capturing something that may be deemed universal of humanity, which can be recognised by the audience. The conception of humanity is, however, always mediated by the actor's social and individual identity; by the actor's dramatic tradition, which provides interpretations of human characteristics and a repertoire of how to express them; and by the cultural and philosophical underpinnings of the play or script. For instance, the portrayal of mental illness may be informed by previous cinematic and theatrical representations, by the actor's observation and wider cultural influences.

Authenticity, in dramatic acting, is thus not the elimination of the actor's personal characteristics to replicate naturalistically a character in the story, but the portrayal of –

what are seen as – (universal) human traits through the actor's individual and social identity. The actor, but also the author and director, have an understanding of human traits, on which basis they build a character. Therefore, humanity is performed through the construction of a character by the actor and director. Authenticity is thus more than the originality, or distinctiveness of the actor. Actors are not simply faithful to their individuality nor are they solely representing the individuality of a character, but adding a further dimension of 'truth', to which the audience can relate. An actor's distinctiveness is part of the process of authenticity, for the self-transcendent aspect of authenticity requires it be expressed through the particularism of the actor's individuality and that of the character.

However, when actors are true only to their own, or to their character's distinctive characteristics, the acting crystallises in their distinctive manner of speaking and moving and results in mannerism. The actors, just as much as non-actors, are thus affected in their speaking and moving. Actors need to go beyond their own distinctiveness and that of the characters they are portraying to convey the human experience, in which the audience can participate. This is independent of the sympathy that a character or an actor might arouse or not. Dramatic art needs to induce empathy, an understanding of the reality being represented, but not necessarily sympathy¹³. Simmel wrote that acting is a crystallisation of a "spiritual reality" (Simmel 1911/1968, p. 95). I interpret this to mean that acting gives form to what is understood as true humanity by the actor, director, author, and by the audience. As mentioned, this may or may not be empirically founded. Thus, rather than

¹³ I am grateful to Dr Andrew Edgar for suggesting this distinction between empathy and sympathy in relation to dramatic art.

being essentialistic, this conception of authenticity allows sociological empirical discovery to inform its contents.

Authentic acting illustrates that authenticity is more than sincerity. Actors are not simply 'being themselves', they are not, strictly speaking, sincere; rather they are pretending. The 'pretence' of dramatic art, however, is a means through which – what is perceived as – the universally human is expressed through the particularism of a character. It is also expressed through the language and repertoire of meaning of acting styles and traditions. An acting tradition carries with itself an array of notions over human nature and how to represent it (realistically, naturalistically, ironically etc.). Actors' authenticity thus rests in the attempt of portraying what they or, for that matter, the author or director, see as universal of human beings. The universality that transcends the particular is always conveyed through the contingent attributes of a character, the personality and interpretative performance of the actor. Authenticity rests on that effort of conveying universalism through particularism.

The example of dramatic acting affords the possibility of conceiving authenticity as an attempt at being true to – what is seen as – transcendent of contingent reality and, yet, expressed through the particularism of the actor, the story and character. The 'universal', to which actors aspire, the 'truth' of human nature is not necessarily ethical, but a mere portrait of humanity, as understood within a specific cultural frame of reference. Thus, dramatic acting shows an articulation of authenticity that is neither normative nor essentialistic. The search for authenticity is thus informed by the social and individual

dimensions of the actor, the cultural specificity of the story, of the character and the actor's artistic tradition. Authenticity is a multidimensional process, for it is articulated by the actor being true to his/her individual identity, to the acting tradition of choice, and to what is understood as truth of the reality being represented.

Thus, authenticity is the attempt at reconciling universal truth (transcendental) with the particular: the actor seeks to portray – what he/she understands as – universal human nature through the particularism of a character and the actor's own personal characteristics. The social dimension of the actor, of the author and of the director, includes the shared notions regarding human nature held by a particular acting tradition. The search for authenticity encapsulates the tension between particularism and universalism. The universal, however, is always understood and performed within a particular framework of meaning. Authenticity is always inscribed within a tradition, be it a religion, an acting tradition, or other cultural framework.

4. Authenticity and Tradition

The systemisation of the 'I and Thou' provides a relational conception of individuality. The person is situated and embedded in social relations. The 'I' is always in relation with others, with one's individuality, but also with an inherited and/or chosen tradition. My understanding of tradition encompasses philosophical, political and artistic traditions,

among others. The present thesis examines individuals embedded in an inherited and chosen religious tradition. However, as the example of dramatic acting showed, the search for authenticity can occur in other fields. The political, philosophical or artistic worldviews, which illumine, guide and challenge politicians, philosophers and artists in their 'cultural productions', constitute traditions. Thus, a literary or musical style, a film genre or directorial cut, a style of acting, are not simply tools of a craft, but constitute a tradition, for they are enmeshed with meaning and values. Artistic styles, just as much as religious rituals, are not mere tools of communication; rather they are part of a meaning system. The search for grasping the truth is couched within tradition, but also ensures tradition is a body of knowledge and practices that are interpreted and lived out in one's daily life today.

4.1 Religious Authenticity in Three Dimensions

The religious person seeking authenticity is engaged in the process of interpreting and constructing religious tradition. I identify religious tradition with a body of religious symbols, rituals, laws, customs, beliefs and narratives associated with a religion regardless of their 'official' status. I argue that legitimacy derives from what is felt to be authentic by the group and individual religious actors (cf. chapter seven). Authenticity is thus an attempt at being true to – what actors understand as – truth. Truth has a social dimension, which is informed by one's social and cultural context; an individual dimension, informed by one's self-understanding; and a self-transcendent dimension, informed by one's sense of a universal horizon. The social dimension of authenticity reflects the values from the group's social context and identity, but also a critique of social *mores*. For instance, to

anticipate the discussion of the empirical chapters, the construction of Bethlehem church as inclusive and caring is seen as 'being authentic' in community, showing how church members construct the communal identity of the church.

The individual dimension indicates that truth needs to be in conformity with the self of the individual. It needs to 'make sense' to the person individually, affecting them emotionally and intellectually. It is, however, not simply experiential, it is not limited to what people find 'moving' or what resonates with their way of thinking; rather it results from the emotional and intellectual interpretation of tradition. Finally, the self-transcendent dimension of authenticity refers to the attempt to tap into – what is understood as – universal truth. In this case study, the truth of Jesus is considered by informants as universally valid. The self-transcendent dimension, in this case study, is born out of belief as *belief in*, trust in God, which leads to a relationship with God and others. Such a relationship prompts the person to live in accordance with the truth, as they understand it, in their everyday life.

The attempt to be authentic is sustained by the actor's interpretation of tradition to make sense in changed social circumstances, in one's life, and within a universal horizon. The process of authenticity, therefore, entails interpretation of tradition, to be relevant for today and consonant with what actors perceive to be timeless truth. Here, universality stresses the search for a wider horizon of truth that is not simply the expression of a cultural identity or social status, but that is valid for humanity as a whole. For example, my informants see Jesus as a revolutionary figure breaking free from traditional hierarchical

religion and inspiring them to do the same today. Therefore, 'enemies' of Jesus today are found in what is dubbed "traditional religion" (cf. chapter eight), which stifles the believer's inner search for truth. To be like Jesus requires being true to oneself, such as expressing one's religious sentiment in a way that 'rings true' to one, but also being true to 'something bigger' and act accordingly.

Truth may include ethical norms, which are the fruit of an interpretation of ethics from within a tradition; so it is always through a particularistic filter. Authenticity is articulated through practices, which may be valued, but are not necessarily ethical. Practices such as praying, fasting, singing, and celebrating festivals are not 'ethical' in and of themselves, that is, they are 'ethically neutral'. They might be of mental or physical benefit for the practitioner, but they are not inherently moral acts. This non-normative conception of authenticity leaves open the possibility of different ethical perspectives and understandings of the good that are negotiated and practised in everyday life.

Within this framework, I propose religiosity as a sensitivity to self-transcendence, which is not confined to religious forms. I base this construction on religiosity being a "life process" (Simmel 1912/1997, p. 209, emphasis in the original), which is required to transcend itself to acquire a Form in order to find expression and on – what Simmel calls – a "progressive development" (Simmel 1912/1997, p. 210), whereby religion acquires a "more purely religious form" (Simmel 1912/1997, p. 210). Thus, religiosity gives rise to a process, whereby religion becomes a Form. Further, in *Ethik und Probleme der modernem Kultur* (1913/2004), Simmel reflected on the framing of the religious question, following the

Enlightenment, as a question regarding the concrete truth of a supernatural reality. Accordingly, either there is a supernatural reality or faith is but a subjective fantasy (1913/2004, p. 44). Simmel opposed to this view a 'third' option: faith itself may be something metaphysical.

Accordingly, the transcendent lies in 'believing', in "the *process* of faith" (1913/2004, p. 44, emphasis in the original). This is, for Simmel, not merely subjective for the subject is objective. It is being religious that is transcendent, metaphysical and objective (1913/2004, p. 45). I stretch these reflections to interpret religiosity as a sensitivity to self-transcendence. Self-transcendent religiosity, the sensitivity to truth, may be inscribed within the religious frame and expressed through it, although it is not confined to it. In the present study, the self-transcendent religiosity of participants is articulated through the narrative of 'following Jesus', proper to the local tradition. Religiosity is thus not belief and/or belonging, nor is it moral conduct, but a sensitivity to self-transcendence, which can be more or less deep and which is expressed in everyday practices.

Understandings of religion as belief or relation to the transcendent fail to appreciate that 'immanent religiosity' has often been present within traditional religious forms. Arguably, most religious practices and rituals seek to make the transcendent immanent. The incarnation of Christ, in Christianity, is an example of the transcendent God moving into the immanent world. In Judaism, the welcoming of the *Shekinah*, the immanent presence of God, at the start of the Sabbath is another example of the transcendent participating in the immanent world. This suggests that immanence and transcendence are not in opposition,

but in dynamic relationship. Accordingly, I propose that the religious framework, rather than inserting the person in a transcendent perspective, 'makes sense' of the person's self-transcendence.

Authenticity is, hence, not just a search for replicating what is seen as normative of tradition, which in the present case is early Christianity, nor yet is it a modernising of tradition to be relevant to the individual today; rather it is a linking of individual reality to what is perceived to be truth through the prism of tradition. Authenticity is a process of constructing one's identity on the basis of one's conception of truth, which may be understood as universalistic, but is articulated through the particularity of one's tradition. The process requires personal participation and interpretation on the part of individual actors. Authenticity implies that one is no mere follower of religious customs, or artistic style, but an interpreter and a 'doer'. The person engages in the process of authenticity by employing the 'tools' in the repertoire for her own tradition. Thus, the person is in relation to tradition and tradition 'lives' through the person.

5. Conclusion

The concept of authenticity, proposed in this chapter, is founded on a relational understanding of the person, as systematised through the 'I and Thou'. Authenticity is, therefore, a process, rather than an ideal. Specifically, it is a process of identity construction

that is in dynamic relation with one's interpretation of tradition. This perspective stresses the agency of the religious actor in interpreting tradition to make sense of their self-transcendence. This should not be taken in a functionalist sense, identifying religion as a social construct to respond to the needs of humanity. The focus, from a sociological point of view, is decidedly on the human and the human construction of the sacred. Nevertheless, the theory does not dismiss the standpoint of the religious person, by which I mean the person of *Religiosität*.

A possible universal truth, here, is not denied; rather it is considered as comprehensible through the particularism of one's experience and culture. This conception underlines the subjective dimension of truth, which needs to ring true to oneself and be in accordance with one's personality. It is also social, for it is informed by social relations, and self-transcendent, pointing to what is seen as 'bigger', or 'beyond', and understood as truth. Truth, however, is always refracted in the plurality of traditions, be they philosophical, artistic or religious. Accordingly, religious tradition, as interpreted and practiced by religious actors, constitutes the framework for the understanding and articulation of truth.

Contra Simmel's own thought on individuality, I put forward a concept of authenticity, which may include ethical concerns, but is not exhausted by them. This is to respond to my participants' understanding of 'being a Christian' as being more than an ethical pursuit (cf. chapter nine and ten). It is also based on a conception of society as agonistic, albeit not necessarily antagonistic, where multiple ethical perspectives are practised in everyday life. It avoids abstract substantialist conception of being human in favour of empirical

discovery. The next chapter explores the boundary-making of authenticity by religious actors. It is argued that the sacred is constituted by what are understood by actors to be the confines of authenticity, which are not decided solely by religious authorities; rather they are interpreted and legitimised by individual actors.

Chapter Seven

Religion, Legitimacy, and Ethics

1. Introduction

In chapter four, I outlined how religiosity (*Religiosität*), for Georg Simmel, was a sensitivity, an attitude, a way of being, which shapes how a person experiences and lives life. This conception of religiosity distinguishes between the religious sentiment and religious expression. Thus, this sentiment need not be manifested in the religious behaviour of established traditions. I have interpreted religiosity as a mode of consciousness, which orders one's life and thus gives rise to an 'existential' identity. The person is a religious person, experiencing life religiously. Simmel described spiritual life as a continuous quest for God. However, it is in the quest itself that the goal is achieved (Simmel 1912/1997, p. 171-172). I thus feel that we can construct religiosity as a sensitivity to self-transcendence, as outlined in this thesis. This means that it is not limited to – what is generally recognised as – religious or spiritual behaviour, but may include other fields of human experience, such as philosophy, politics and art.

It was mentioned in chapter four, that the person expresses her religiosity, including experiencing the relationship with the divine, through her group and in the language of her tradition. Tradition, here, comprehends not only the historical construction of a religion, but all the narratives, ideas, and practices in which religious people participate. Tradition is "lived religion" (Hall 1997; Ammerman 1997; Orsi 2003; McGuire 2008), rather than solely official theological doctrines. Therefore, the emphasis is on the individuals' agency in constructing their own tradition and, above all, legitimising it. This work of attribution of value and legitimisation constructs the sacred, which is not an a priori category. It is, rather, constituted of itself in the sense that it is what marks an identity, it is "what we do/think as Christians", as participants might say. It is the fruit of religious people's 'sacralisation', whereby the person identifies with tradition and attributes value to the components of the tradition, such as specific ideas, symbols and practices, thus constituting the sacred. The attribution of value does not necessarily mean that the ideas, symbols and practices, which are objects of the sacralisation, are invested with 'magical' power, although it may in some cases.

Contrary to other interpretations, I do not identify the sacred with the numinous (Otto 1917/1924), nor do I consider it as what is considered 'good' within a given culture and society (Lynch 2012). It may include practices that are 'ethically neutral', as in the case of praying. Most importantly, the ethics, which may underpin instances of sacralisation, is not to be taken as applicable universally. The ethical standards professed by a community might indeed be in contrast with those of the wider community. Sacralisation marks the confines of an identity considered authentic, which is not limited to ethics. This can be

found in the divergent views over the place of environmental concerns in Christianity, evidenced in chapter ten, where all participants agree with environmental ethics, but not all considered them an authentic part of Christianity. Sacralisation thus limits the boundaries of authenticity, which depend on the interpretation of tradition by individuals as well as by religious authorities. Authenticity, in the present case study, is identified with 'following Jesus', which is fundamentally ethical, but also it is identified with a particular form of ethic: the 'ethic of compassion'. The second part of the chapter constructs the distinction between the 'ethic of purity' and the 'ethic of compassion' to aid the analysis of the ethical reasoning at Bethlehem. The next sections build on the systemisation and expansion I have provided so far of Simmel's sociology to propose the process of sacralisation.

1. The Term 'Sacralisation'

The term sacralisation has been employed in a variety of contexts. Historian Emilio Gentile (2000) develops this concept on the basis of his studies of totalitarian political movements. Sacralisation, for Gentile, refers to politics acquiring a "religious dimension" (Gentile 2000, p. 22). Thus, the sacralisation of politics takes place "when politics is conceived, lived and represented through myths, rituals and symbols that demand faith in the sacralised secular entity, dedication among the community of believers" (Gentile 2000, p. 21). Religion, for Gentile, is a system of beliefs and myths that define values and ethics. Gentile explains that

fascism "constructed its own system of beliefs, myths, and rituals, centred on the sacralisation of the state"; yet its totalitarian nature and politics meant that fascism became "an all-consuming existence" (Gentile 1990, p. 230).

Gentile's political religion gives meaning to the person. He writes that fascists were united not by "a doctrine but an attitude, an *experience of faith*" (Gentile 1990, p. 234, emphasis in the original). This latter aspect echoes unwittingly Simmel's definition of religiosity as an attitude pointing to the affinity between politics and religion in – what may be described in Simmelian language as – their totalising force. Gentile's work has primarily a "macro-level" focus. He thus provides a valuable account of the relationship between religion and politics, and the use of beliefs, myths and rituals in the sacralisation of politics, albeit leaving unexplored the individual dimension of religion and the role of individuals in constructing the sacred.

Sacralisation is also mentioned in relation to the phenomenon of 'fandom' (Jindra 2005 and 1994). Jindra argues that Star Trek 'fandom' can be considered a religious phenomenon due to the sacralisation of elements of culture, the formation of communities, a canon and a hierarchy. Sacralisation here entails "deep-seated American beliefs about the nature of humankind, the world and its future, and encourages the practices that parallel religious processes of codifying, forming a community and developing institutions to guide its practices" (Jindra 1994, p. 50). It thus seems to be based on beliefs and belonging, although there is no reflection on how fandom sacralisation affects one's identity construction and everyday practices. The wider phenomenon of celebrity is instead understood as a cult of

the individual (Rothenbuhler 2005) and a projection of a desire, rather than identification with the celebrity and identity construction.

In psychology of religion, sacralisation is concerned almost exclusively with the individual. Psychologists conceive of religion as an all-encompassing meaning system (Silberman 2005, Park 2005a and 2005b, Paloutzian 2005), which orients the religious person in how to understand life and organise values according to religious notions. For Pargament, religion "has to do with building, changing, and holding on to the things people care about in ways that are tied to the sacred" (Pargament 1997, p. 32). Accordingly, sacralisation explains why activities, such as teaching or giving food to the homeless become 'ministering', when performed as service to God, and are thus understood differently from the same activities performed as a secular task (Pargament 1997, p. 211).

Emmons and Crumpler (1999) distinguish between "sacralization" as "a process of imbuing external objects with sacred qualities", and "sanctification" as "an inner process of transformation" (Emmons and Crumpler 1999, p. 18). Sanctification corresponds to "a process by which personas are made pure or holy. Sanctification thus refers to moral purity or moral goodness (literally, 'saint-like')" (Emmons and Crumpler 1999, pp. 18-19). This notion of sanctification reflects the Protestant idea of sanctification by the Holy Spirit. In other words, the Holy Spirit begins a process of perfecting the believer once the believer has accepted God in their life. Consequently, it applies a very particularistic lens on individual experience; it assumes that there is such a state or quality of "moral purity,"

disregarding the complexity and contested nature of ethics; and it conflates the sensation of being cleansed with "moral purity".

The application of the psychological notion of sacralisation in different contexts shows the concrete effects of the attribution of sacred meaning to objects, events, and ideas. Thus, bargaining based on economic incentives during conflict resolution negotiations or which affects symbolical state policies, tends to backfire when it disregards sacred meaning (Fiske and Tetlock 1997; Dehghani et al. 2009 and 2010). In sociology of religion, Hans Mol (1976) proposed an understanding of religion as the sacralisation of identity. Accordingly, individuals and groups have a need for identity, understood as "a stable niche in a predictable environment" (1976, p. 55) to counter the process of differentiation of modernity. Sacralisation grounds and strengthens identity by projecting on it a feeling of awe.

Mol's functionalist approach presumes religion to be beneficial to individuals, groups and society by providing cohesion thus neglecting the diversity of religious expressions, especially disruptive or violent religion (Smith 1996). In contrast, the concept I propose sees sacralisation as a process of identification of religious actors with their religious tradition and their attribution of value to beliefs and practices by inscribing them into the religious framework. The process of sacralisation is critical to the construction and legitimisation of religious tradition on the part of individuals and the group. This identification and value attribution constitutes the sacred, thus providing boundaries to what is considered authentic.

3. The Process of Sacralisation

Religion, for Simmel, is a language (Oakes 1980, p. 10), a schema through which the religious person understands the world around her. Accordingly, the religious person understands herself and her everyday life religiously. I propose to understand religiosity as sensitivity to self-transcendence, which involves transcending one's social and individual background to live 'religiously'. It may be a deep mystical experience or, more simply, a specific mind-set over one's understanding of life. The mind-set of 'going beyond' the 'here and now' calls for action. The experience or sense of 'immanent transcendence' needs to be lived out in everyday reality. The search for authenticity is the search for expressing religiosity. This can include the adoption of a specific diet, such as Jewish *kashrut* and Muslim *halal*, or the performance of specific activities, like Christian ministering and Jewish *mitzvoth*.

There may be multiple reasons for adhering to a religious lifestyle. Focusing on the individual agency, we can glimpse at how individuals attribute value to a lifestyle, to ideas and symbols in the construction of their identity. The identification of a person with a way of being religious brings about the attribution of value and legitimisation to one's interpretation of tradition. I ground sacralisation in the following concepts: religiosity; religion's *belief in*; and religion's totalising force. I propose to understand religiosity (*Religiosität*), as a mode of consciousness, from which a religious paradigm ensues. Such paradigm has a totalising force (*Totalisierungvermögen*), in the sense of ordering one's world view. As mentioned in chapter four, I link Simmel's claim that human beings project

their own qualities and hopes onto the people near them, their country, and "the gods" with 'belief in' (trust).

The attitude of trust, which for Simmel is typical of religion, cements people's projections of feelings onto others, their tradition, and the divine. This results in a form of idealisation, elicited by the totalising force of religiosity. Just as people idealised others or their country, as proposed by Simmel, by projecting their hopes and ideas onto others and their country; here, religious actors project their religious sentiment, ideas and hopes onto their tradition sacralising it. The projection of religious actors' aspirations onto their tradition is dependent on the level of their identification with it. The intensity of the process of sacralisation reflects one's religiosity, which, as conceived by Simmel, is present in people in varying degrees. Therefore, it can be of deep significance or more superficial, depending on the degree of religiosity of the person. Consequently, religious actors vary in how they ascribe value to religious ideas and practices.

As mentioned, Simmel's religiosity is not confined to religious behaviour belonging to a recognised tradition, but has blurred boundaries. Simmel, however, does not explain whether the totalising force is a characteristic of religiosity (*Religiosität*) or religion (*Religion*). I interpret the totalising force as an aspect of religion, the world view to which religiosity has given expression. Accordingly, the ensuing world view is not limited to established religious traditions, but to philosophical, artistic, political traditions, to name a few. Most importantly, sacralisation rests with the individual, who legitimises religious practices and ideas of tradition. This approach to sacralisation emphasises the role of

individual religious actors in engaging with their tradition by interpreting and innovating its beliefs and practices. They participate in the making of their religious tradition by following or subverting the prescription of religious authorities.

Further, sacralisation is grounded on practice. Religious actors are thus practitioners who inscribe their actions within their tradition and attribute value to them. Even the Protestant notion of 'personal conviction' is 'canonised' through practice as much as belief. Membership in a church and serving are examples of practices through which believers sacralise their 'personal conviction'. Sacralisation is thus the identification with tradition, which inscribes oneself, one's beliefs and practices within the religious framework. This results in attributing value to the components of religion, which are seen as authentic, and their legitimisation. Sacralisation, which accounts for only the contribution of individual actors in legitimising tradition, does not happen in a vacuum, but within the web of social relationships the person inhabits.

The religious framework is moulded by social norms, customs and roles. Religion is not a separate sphere from society; rather religious actors reflect their social embeddedness in their interpretations of tradition. Therefore, the theological, personal, cultural and social background of the individual informs the process of sacralisation. Equally, this process happens within the religious group's hierarchies and structures. Some members of a group have official or unofficial authority and thus hold more influence. It is, therefore, important to explore those dynamics. However, sacralisation stresses the fact that the definition and legitimisation of religious tradition is not confined to religious authorities. Finally,

sacralisation is not necessarily ethical. For instance, the sacralisation of fasting is not in itself ethical, although it might aid reflection on ethical concerns, such as overconsumption in the West and wealth inequality.

Religion should not be conflated with ethics. Religious (and non-religious) traditions have ethical principles, but these are a component of tradition, which cannot be reduced to ethics. For instance, religious actors might value and follow a dietary or dress code without understanding it as ethical. A case in point can be found in Judaism, where diet norms, *kashrut*, have been justified by major Jewish philosophers and biblical interpreters on the basis of different reasoning. Accordingly, for Saadia Gaon the rationale behind *kashrut* was the rejection of animal worship (Isaacs 1996, p. 27), whilst for Moses Maimonides were health concerns and for Nahmanides *kashrut* was beneficial to the soul (Jacobs 1997, p. 127). Although a minority, prominent modern orthodoxy philosophers went as far as rejecting altogether the ethical justification for the whole of the Jewish legal system (*halachah*).

Yeshayahu Leibowitz claimed that at the heart of Judaism is not an ethical theory but the "recognition of a system of precepts as binding" (Leibowitz 1992, p. 3). Similarly, for Joseph Soloveitchik Judaism was a "total way of life" requiring the "surrender of human rationality and the sacrifice of one's human ethical sense" (Hartman 1985, p. 89). Leibowitz and Soloveitchik were reacting to the development of universalistic ethics in modern times and affirming the particularity of the Jewish tradition by stressing the theocentric nature of the legal system. In recent times, perhaps the most interesting development has been the rise

in 'eco-kashrut' (Waskow 1992; Kusnetz 2012; Goldberg 2013) demanding ethical criteria, such as food sustainability and workers' rights, to be included as part of *kashrut*, with organisations, such as Magen Tzedek¹⁴ and Hazon¹⁵ issuing certification of ethical standards. Thus, following *kashrut* may have different rationales within the same denomination or group.

Equally, practices such as praying, fasting, singing, and celebrating festivals are not inherently 'ethical'. They are, as an Israeli singer put it, the 'pilates of the soul' 16. They may harness one's concentration and mindfulness of religious ethical principles, but are not moral practices in themselves. How religious ideas, norms, and customs are interpreted and put into practice will vary depending on one's personal aptitude and social relationships. As such, it can lead to moral or immoral actions. Further, there may be different approaches and interpretations of ethics within the same religious tradition. Thus, when ideas and practices pertaining to ethics are sacralised, these should not be taken as inherently 'good' or applicable universally. Accordingly, religious actors may inscribe violent action within the religious framework bestowing on it legitimacy. The next section, therefore, explores an initial distinction between different forms of ethics. I suggest a distinction between the 'ethic of purity' and the 'ethic of compassion' to reflect my own empirical findings and of those of other studies. This is supported by the development of Simmel's thoughts on Schopenhauer and on love, as outlined below.

.

¹⁴ http://www.magentzedek.org/

¹⁵ http://www.hazon.org/resource/kosher-sustainable-meat/

¹⁶ Kobi Oz, Limmud Festival 2010

4. The Ethic of Compassion

In his book *Schopenhauer and Nietzsche* (1907/1991), Simmel contrasts Nietzsche's pity with Schopenhauer's compassion. For Schopenhauer, compassion means transcending individual division by sympathising with another. The "individual subject identifies itself with all of the others" (Simmel 1907/1991, p. 106). In contrast, Nietzsche interprets *Mitleid* as pity, which can involve manipulation by the pitied of the pitier (Cartwright 1984, pp. 85-86), but also helping others may increase the feeling of superiority the pitier feels towards the pitied (Cartwright 1984, p. 88). For Schopenhauer, "compassion for the suffering of sentient beings was the basis for actions having moral worth" (Cartwright 1984, p. 92). According to Simmel, Schopenhauer's view is based on the discovery of the subject that individuation is but an illusion concealing that someone else's suffering is ultimately one's own (Simmel 1907/1991, p. 110). Thus, compassion is based on the recognition of the unity of all being.

Morality, for Schopenhauer, is only the alleviation of suffering (Guyer 2012, p. 405). Compassion (*Mitleid*) is thus "a desire for another's well-being" (Cartwright 1988, p. 561). Cartwright explains that Schopenhauer identifies wellbeing (*Wohl*) as 'good' and misfortune (*Wehe*) as 'bad' (Cartwright 1988, p. 561). Compassion alleviates suffering thus bestowing moral worth to action. Compassion comes from "... the immediate *participation* (*Theilnahme*), independent of all ulterior considerations, primarily in the *suffering* of another, and thus in the prevention or elimination of it.... Only insofar as an action has

sprung from compassion does it have moral value; and every action resulting from any other motives has none." (Schopenhauer, cited in Cartwright 1988, p. 561, emphasis in the original). Compassion is "the fundamental incentive of all genuine, i.e., disinterested, justice and lovingkindness" (Schopenhauer 1839/1995, p. 187) for it is based on the identification of the subject with another.

"I nevertheless *feel it with him, feel it as my own*, and not *within me*, but *in another person*... But this presupposes that to a certain extent I have identified myself with the other man, and in consequence the barrier between the ego and the non–ego is for the moment abolished; only then do the other man's affairs, his need, distress, and suffering, directly become my own" (Schopenhauer 1839/1995, pp. 165-166, emphasis in the original)

In *Nietzsche and Kant* (1906/2008), Simmel explains that for Schopenhauer compassion constitutes the substance of all morality for it is the intimate and immediate form of solidarity among people, which is rejected by Nietzsche (1906/2008, p. 106). For Nietzsche, there is a natural distance between people (Simmel 1907/1991, p. 147), which should not be overcome. Simmel grounds his ethics in the individual law (cf. chapter five), which is not developed in conjunction with his thoughts on love and compassion. He is uneasy with the implication of Schopenhauer's grounding of compassion in the lack of individuation. Contrary to his student, Martin Buber (1923/1970), who will later interpret the 'I and Thou' as a unity and ground for ethics, Simmel resists this lack of differentiation.

This is because, for Simmel, the "absolute unity of essence dissolves not only an independent 'thou', but also the 'I'" (Simmel 1907/1991, p. 111).

In Simmel, the distinction between 'I and Thou' is affirmed rather than dissolved. It is, once again, in the relationality that we find morality, in the responsibility for a Thou. Accordingly, "the zenith of morality is reached when relations among human beings evince simultaneously full duality and full unity: a moral occurrence is just one in which duality is not permitted to disappear in the quest for unity" (Simmel 1907/1991, p. 116). This is reiterated in his fragment *On Love* (1984).

"The existential will of the I flows to the Thou with complete intimacy. It does not need a bridge, which separates just as it connects. Nevertheless, the dynamic at stake here is different from the metaphysical unity of all being as such, from which Schopenhauer, for example, derives charity and sacrifice. This is precisely the miracle of love. It does not nullify the being-for-itself of either the I or the Thou" (Simmel 1984, p. 155)

Simmel's reflections on 'universal philanthropy' and Christian love help us understand love in its 'ideal', in the sense of abstract, dimension. They also trace the move from love directed to a person to love as a state of being of the subject. In the first instance, universal philanthropy is "concerned with the person as an abstraction" (Simmel 1984, p. 182). Universal philanthropy has the abstract character of eighteenth century universalism. It rests on "the elimination of the individual differences of its objects" to the extent of

becoming "unconditional" (Simmel 1984, p. 185). It is a general emotion based on shared humanity and, therefore, not one directed to anyone in particular. Simmel contrasts this with Christian love, which, for him, "embraces the *total* person" (Simmel 1984, p. 185, emphasis in the original).

"... Although the sinner is also an object of universal philanthropy, this is really in spite of the fact that he is a sinner, and only because ultimately, he is a human being as well. Christian love, on the other hand, embraces the sinner – and precisely as a sinner – if not with a greater love than is bestowed upon the normal person, then at least without that 'in spite of'" (Simmel 1984, p. 185-186).

Simmel did not develop these reflections further¹⁷. However, Martin Buber and, later, Emmanuel Lévinas took Simmel's 'I and Thou' in an explicitly ethical direction. The total embrace, to which Simmel refers when describing Christian love, is echoed in Lévinas' moral conception of subjectivity, which rests on one's responsibility for the other. Buber's and Lévinas' relational ethics is in contrast with the abstract principle of Kant's categorical imperative. This tension has been played out in the debate between psychologists Lawrence Kohlberg (1973) and Carol Gilligan (1982). Kohlberg proposed six stages of moral development, where the highest stage identifies a conception of the good with universal justice. His theory has thus a strong affinity with the philosophy of Kant and Rawls, based on abstract principles, rights and duty.

_

¹⁷ Love is an incomplete collection of essays published posthumously.

Gilligan criticised Kohlberg's approach for adopting male experience and reasoning as universally normative (1982, p. 6). Women, she argued, see a moral dilemma as "a narrative of relationships that extends over time" (1982, p. 28), rather than a calculation that severs connections from relationships. Therefore, they would not score highly in Kohlberg's scale. In contrast with Kohlberg's 'ethic of justice', she proposed an 'ethic of care' to account for a worldview that privileges relationships. In support of Gilligan, Blum (1988) argued that the 'ethic of care' is just as universal as the 'ethic of justice'. More recently, the two ethics have been taken as complementary with Gilligan providing an expansion of Kohlberg's theory (Jorgensen 2006).

In sociology of religion, Wedam (1997), drawing on the Kohlberg-Gilligan debate, highlights how two religious groups used different ethics in discussing abortion. Both groups considered abortion immoral, yet one group applied an ethic of justice, implying "adjudication based on the principle of the right to life" (1997, p. 165); while the other group "applied an ethic of care in which the principle of nonviolence determined adjudication" (1997, p. 165). On a similar note, Edgell-Becker (1997), although not referring to Gilligan explicitly, draws the attention to the different "moral logics" present in congregational conflicts. She distinguishes between compassion and authority. Accordingly, "the logic of compassion or caring, a relational logic that keeps conflict, in most cases, from escalating into "winner takes all" contests. This logic emphasizes dialogue and compromise". In contrast, "the logic of religious authority, implementing the guidelines agreed upon in advance as originating from an authoritative text or person" (Edgell-Becker 1997, p. 140).

For Edgell, authorities justify choices on the basis of the authority of a text or person. However, this leaves out how individuals reinterpret the text (Bartokowski 1996) and how texts are used to justify practices, which may include those that have not been sanctioned by authorities or are in direct conflict with the will of authorities. Wedam suggests that different groups invoke different ethical principles, although it is not clear what makes one group choose one ethic over another. I construct the dichotomy 'ethic of purity' and 'ethic of compassion' on the basis of my empirical research without, however, endorsing the informants' normative judgements or favouring one form of ethic over the other. It is not my intention to develop a theory of ethics, but to identify the connections between ethical norms and social dynamics of identity and community making. I begin by outlining the ethical norms of compassion as people-centred. This is in contrast with norm-oriented ethical norms of purity.

5. Compassion and Purity

The discourse and practices at Bethlehem church are dominated by a relational ethical approach, which privileges compassion over judgement. This is partly motivated by the negative experience research participants have had of church as judgemental. I, therefore, build on Simmel's notion of "embracing of the *total* person" to construct a concept of compassion as the emotional participation in another's suffering, but also the recognition of the other's human dignity. Compassion is an encounter with another that enables the

actor to see the other as fully human. Thus, the other person is seen in her humanity rather than in the image we create through social categories, e.g. gay, woman, Asian, but also academic, parent, and pensioner. For instance, the religious actor, in encountering a homeless person, recognises the human dignity of the person, which lies beyond the social aspect of homelessness. Consequently, the homeless person is seen as 'God's child', in the language of research participants. Yet, as one of my informants, Camden, shows in chapter ten, this seems to apply not only to those who suffer, but even to one's spouse.

Compassion, therefore, gives rise to a state of mind of recognition of the divine in the human. There seems to be 'intentionality' to that state of mind. The actor wills the recognition of the other on the basis of the emotion of compassion. The aspect of intentionality highlights how compassion is a central value in the local culture that ought to be practised; rather than a feeling or a quality of specific persons. The 'ethic of compassion' underpins the actors' self-understanding and their rationale for action (cf. chapter ten). Similarly, the 'ethic of purity' delineates the boundaries between right and wrong and is a rationale present in the ethical considerations of religious actors. The terms 'compassion' and 'purity' are employed to convey a more multidimensional meaning that goes beyond justice and relationality.

I conceive compassion and purity as rationales affecting boundary-making. Taking inspiration from Mary Douglas' *Purity and Danger* (Douglas 1966/1989) and Simmel's extremely brief considerations of honour, I construct purity as a norm-oriented ethics, which coalesces the group and guarantees order. Douglas saw in the laws of Leviticus the

structures of a society in the interplay between order and disorder. Accordingly, "pollution ideas" and "danger ideas" exert social pressure establishing rules of "good citizenship" (Douglas 1966/1989, p. 3). So purity delimits correct behaviour. Building on Simmel's considerations regarding honour, the notion of purity can be expanded to stress much more the boundary-making of a community.

"... through the appeal to honor, society secures from its members the kind of conduct conducive to its own preservation, particularly within the spheres of conduct intermediate between the purview of the criminal code, on the one hand, and the field of purely personal morality, on the other" (Simmel 1898a, p. 680).

This is so because it regulates what is outside of the dominium of the law. "While civic law," explains Simmel, "employs physical force as its sanction, while personal morality has no other recourse than the approval or disapproval of conscience, the laws of honor are guarded by penalties which have neither the pure externality of the former nor the pure subjectivity of the latter" (Simmel 1898a, p. 681). Honour, therefore, is actualised through the group's condemnation and legitimisation setting the boundaries of behaviour and group identity. "By the demands upon its members contained in the group standard of honor the group preserves its unified character and its distinctness from the other groups within the same inclusive association" (Simmel 1898a, p. 681). Simmel does not conflate honour with universalistic ethics. On the contrary, it is a distinctive code. For instance, it is a matter of honour for the ascetic to let himself to be spit upon.

The 'ethic of purity' pertains to judgements over principles and actions, which are deemed ethical by the group and which are considered characteristic of the group. The 'ethic of compassion' is concerned with the acceptance of others, which makes boundaries more porous. Purity defines boundaries of 'right' belief and 'right' practice, while compassion privileges the experience of the person and is articulated through actions directed to specific people. I am not suggesting that one ethic is in any way superior to the other. Indeed, the two 'ethics' may be present at the same time. Therefore, there are two aspects to the two ethics: ethical norms and practices and boundary-making. With reference to aspect of ethical norms and practice, I illustrate the dynamics of the 'ethic of purity' and the 'ethic of compassion' in relation to an American video, which I accessed on a social medium website, regarding an anti-abortion campaign.

The interviewer approached anti-abortion campaigners who held signs showing – what they alleged to be – aborted foetuses. The campaigners sought to make abortion illegal in the United States. The interviewer asked them what they would like the penalty to be. The question was met with perplexity. The interviewer explained that an illegal action carries a penalty. Most of the interviewees were at a loss and did not want women, who underwent abortions, to be sentenced to prison. The intent of the video was to embarrass the campaigners. However, the most striking fact was the tension between the judgement over abortion ('ethic of purity') and the need for compassion ('ethic of compassion'), to which a couple of campaigners appealed explicitly. Thus, the distinction between 'ethic of purity' and 'ethic of compassion' helps seeing how actors translate into practice ethical norms.

I argued that purity is norm-oriented and concerns itself with 'right beliefs' and 'right action'. Being more detached from the concrete experience of the person, purity may be articulated more through strategic campaigning, which is what the anti-abortion campaigners were doing, than social action that has an immediate recipient. As I mentioned, the two ethics are present in the same group, so they are not mutually exclusive. The compassion approach of Bethlehem has led to a particular type of community and action that is people-oriented and service-oriented. Bethlehem does not engage in social justice campaigns locally or nationally; it privileges direct contact with people. The ethical discourse of the church plays a role, alongside the internal structures of leadership and membership of the church, in shaping the type of church community and the form action takes. Further research is needed to identify the rationale behind the choice of a type of social action over another.

With regards to the aspect of boundary-making, purity identifies the set rules of belief and behaviour that define the boundary of a group. For instance, belief in salvation of the soul through faith in Jesus excludes those who do not share that belief. In contrast, compassion stretches the boundaries of a group by seeking to accept the person regardless of the stage in their 'journey'. The practice of compassion may have an effect on rethinking boundary-making and the identity of the group (cf. chapter eight and ten). I thus construct the identity and community making of a group through the interplay of purity and compassion. It needs to be stressed that my interest in the formulation of the 'ethic of purity' and the 'ethic of compassion' does not lie in their normative potential, but in the possible sociological aspects of this distinction.

6. Conclusion

The chapter concerned itself with two main issues: the construction of the sacred and of the ethical rationale guiding ethical norms and practices and boundary-making. I have argued that individual religious actors contribute to the construction of the sacred by engaging with their tradition, which includes interpreting, innovating and subverting elements of tradition. Sacralisation, put forward in the first part of the chapter, is the process whereby individuals identify with their tradition, bestow value to its elements and legitimise them. Thus, the religious actor adopts religious tradition as an all-encompassing meaning system, but also contributes to shaping it by interpreting and legitimising it. Sacralisation is the basis for the construction of authenticity. What is sacralised, be it an idea, practice, object or place, identifies the boundaries of authenticity, which is inscribed within one's tradition. Being authentic is thus expressing one's religious sentiment through the sacralised narratives, ideas and practices of one's tradition. Religiosity, as the sensitivity to self-transcendence, is expressed within the particularism of the framework of tradition.

The latter part of the chapter responded directly to my empirical findings. The particular ethic, which is pre-eminent at Bethlehem, is one of compassion. The local narratives of my research participants construct compassion in contrast with judgement. Consequently, I posited a distinction between the 'ethic of purity' and the 'ethic of compassion' to stress the norm-oriented approach of the first and the people-oriented approach of the second. This

is not to be taken normatively. Neither ethic is superior to the other. The very broad terms 'purity' and 'compassion' were chosen to leave open the possibility of multiple meanings, rather than restrict the distinction to abstract normative categories. Accordingly, I suggested that both purity and compassion have an effect on the choice of specific practices, such as strategic campaigning versus providing a service to immediate recipients, and the dynamics of identity and boundary-making.

Chapter Eight

From Doctrine to Authenticity

1. Introduction

This chapter introduces the defining features of Bethlehem church: its origins, its founders, its narratives and practices. The local culture of Bethlehem is portrayed in the long account of section two, entitled 'Beginnings'. This is analysed in the subsequent sections. Following an explanation of the 'vision' of the church as 'modern, relevant and inclusive', the chapter reflects on the role of choice in being a Christian at Bethlehem and the construction of Christian authenticity, articulated as forming human relationships. Thus, it is argued that the narrative and practices of authenticity at Bethlehem point to a shift in emphasis from 'doctrine to authenticity', from propositional belief to conscious practice. Bethlehem sought to create a welcoming community where everybody would feel accepted. This results from the interpretation of 'following Jesus' as a call to form relationships in the church and in the local community. This chapter provides a broad view of the culture of Bethlehem and the local construction of authenticity. The reflection on doctrine and ethics reveals how the

'ethic of compassion' (explored further in chapter ten) relaxes the boundary-making of the church.

The narratives and practices of Bethlehem church show that the identification of authenticity with the practice of relationships has made doctrine, including the exclusive doctrine of salvation, secondary for many. This understanding of Christianity as a practice and the experience of relationships are discussed in sections five and six. Authenticity underpins the identity of Bethlehem church (cf. sections two and three) and is based on the conscious choice of believers, which engenders the commitment to be a Christian (cf. section four). The chapter begins with a glimpse of the local narratives around Christianity, as interpreted by my informants, and the role of the church in helping Christians practise authenticity. As explained in chapter three, the names and location have been anonymised by using the names of the characters in George Eliot's novels, whilst Casaubon is a fictional character employed to make the narration smoother. The account is presented in a narrative style to ease the reader into the world of Bethlehem and of its members. The excerpts in quotations are taken from interviews, while the home group excerpts are based on field notes and therefore are not verbatim. Finally, the use of italics is to underline significant sentences.

Bethlehem, as mentioned in chapter three, was set up around 30 years ago in an area of rapid housing development that lacked a community "hub", as Nicholas, one of the elders, explained. It is a "free church", an independent evangelical church, of Baptist origin. In the beginning, the church consisted of an informal group meeting in different venues to

worship God. As the group grew and became more settled in the local area, they decided to buy land and build the church. The people who set up Bethlehem have often had an experience of church as judgmental, divisive, and old-fashioned. The "founding fathers" of Bethlehem, Nicholas, elder, and Felix, pastor, wanted to communicate Christianity to the "unchurched", people who do not normally go to church, as told in the following section.

2. Beginnings

In the mid-1970s, Nicholas, a 15-year-old boy, found God. He told his family and they laughed and said 'Don't worry, it'll pass'. It didn't. Nicholas went on to found Bethlehem church. He wanted to communicate the truth he had discovered. So with his friend, Felix, he decided that it was time for something different, something that would not put people off, but would ring true. They were fed up with being preached hell, fire and brimstone and of singing unenthusiastically two-hundred year old hymns from dusty books on hard pews. They wanted to bring Christianity to the people. They began in the backroom of the Ship & Pilot, a pub just 300 yards from where the church now stands, on a small mound facing the sea.

The sixties had come and gone, but they had left a distinctive yearning for something authentic, something exciting, something true. The worship meetings moved from the pub to a school. The list of people willing to try something different and bored stiff of church

Sunday mornings grew steadily. It was time to build a home, a spiritual home, for the thirsty and the needy, for Christians searching for authentic Christianity and for the crowd outside seeking a soul in a world of strangers. Nicholas and Felix had a vision. They wanted a church that was not a club and a faith that was not just for the sake of believing. They wanted a place of comfort, where people could feel welcome. They wanted a building that would make that possible. It had to build bridges and include people. The building had to be open to the community to show a face of Christianity that didn't scare people off, that didn't judge or ask for money. But the 'founding fathers' also wanted something real that pierced through the hearts of people, that answered the big questions, the questions of pain and suffering, loneliness and emptiness. How do you do that? People don't want to go anywhere near a church. They think it's just self-righteous loonies looking down at you. They don't need salvation and they are quite happy with sin, thank you very much.

No use in talking about sin, sanctification and salvation if nobody understands what you mean. Nicholas looks towards the window and goes back a few decades, then says: "when we started to meet in homes, back in the eighties, that was just to meet together, study the bible together and pray together, in a small group. Bit by bit, we grew. There wasn't a strong focus on evangelism, on sharing faith other than in your personal life, until we had a public facility to hold meetings. ... As I was doing these things, we started to realise that we could invite people to these kind of events. I thought 'how are we gonna do this? How are we gonna present the information in a way that makes sense to people?' And it dawned on me 'why do we do church the way we do church?' ... So we started to look at that, we played with different types of styles. We did something called blue-prints for about a year or so,

once a month. We had a whole hour and half with videos, music, and a short message. ... The message was the same, but it was the way in which we were trying to communicate. Some non-Christians came along, I don't think anybody came to faith, ... but probably kept them on their journey, a little while... And that was in our mind when we opened this place and so we first started, probably the first 10 years or so, we said that our primary objective in running a church service on Sunday morning is for people who don't go to church". That was the vision: tell the story to those who don't know it.

Felix explains that the vision is for a modern church, where the services feel right, relevant to people's lives. "So they come along on Sundays, this isn't frightening; but the other thing is, is that it's serving society [...] a community that is committed to serve society and not be just a holy huddle inside a church building. [...] The word church has now become useless because we've ruined the word church, I mean it's a building now, it means stained glass, it's got an altar, funny talk there, that's not a church, the church is part of the Kingdom of God, it's people, it's relationships, it's a powerful spiritual thing, you know, but we threw that, so I use, the term kingdom ... helps break out of the word church, but community is a similar word to me, you know. Community is church in the community and moving out in the community ...".

Mr Casaubon didn't say a thing, he listened. He paid attention to every word, but wasn't sure he got the meaning of it. What had Jesus got to do with a café? It was a nice place, people were friendly and maybe God gave them a hand, but the whole thing about dying on a cross, salvation and sin didn't make much sense. He knew he liked the atmosphere. It was

an oasis of tranquillity and peace, only disrupted by the groups of people storming the place at regular intervals for classes of yoga, French and what have you. After Felix and Nicholas had left, Lucy noticed the puzzled look on Casaubon's face and said: "You see, the point is that ... those who started Bethlehem church came from all sorts of different church backgrounds wanting to strip all the stuff around church and get back to the real heart of their faith, which was trying to be more like Jesus". Casaubon nodded, but inside his head thoughts went round in circles and questions suddenly appeared from nowhere, questions he never pondered before, to do with God and religion. 'Wasn't Jesus meant to be God? How can you be like God? Isn't that heretical?' He finished his tea quickly before getting too entangled in theology. He left the café and looked back at the church. The night was falling and all you could see were the lights coming through the windows at the church centre.

As he lay on the sofa, Casaubon listened to Nicholas talking about the 'vision' for the church. "What is this vision thing?" asked Nicholas, "We need to be clear about this, about what we are about [...] I know people want to worship, there's nothing wrong with that, but small groups are perfect for worship. Let's face it, you actually can have too many prayer meetings, too many worship meetings, too many bible studies, because you become internally focussed and all you try to do is your thing in here, in your little holy huddle, whereas Christ never did that and I'm sure his church should never do that. [...] One thing that has become increasingly clear and challenging to me in this whole thing is this, if you're trying to boil down what Christianity is, you cannot do it simply [...] obviously there's too many things you boil it down to, what are they? [...] This one phrase keeps coming back to me, Jesus says "follow me". What does that mean? Where? How are you

gonna lead me? Where are we going together? And if you see how Jesus did his life, in the years of his public ministry, it wasn't just doing the way we do church. He didn't stand at the front of the synagogue and just preached. He did a bit of that as well, but that was probably minority at this time actually. But if we model Christianity in the way Jesus did his life, we wouldn't hold church services as often as we do. He was out there, in amongst these people's lives, sometimes he would speak spiritually to people, sometimes he wouldn't mention anything spiritual. Sometimes he healed people, sometimes he wouldn't, he would just speak spiritual things to them. So there's a whole spectrum of engagement with people, [...] like Willow Creek church in the States. I loved the way in which they understood the value of multimedia engagement with people. [...] We've run some comedy nights events, just to show it's not just about church and preaching Christianity. It's also about doing good and engaging and having fun actually. It's good to laugh."

Till drink to that' thought Casaubon who was now following the church meeting from a distance. It was meant to be an informal discussion on the vision, which was felt needed rekindling. There had been talk of expanding. It came to nothing though. Casaubon wasn't sure what happened. Some people wanted to work with the Council to provide services for the local community. They had to find money, a lot of money, and that didn't happen. It left a bitter taste in the mouth of Felix, who had tried so hard. If he could only have persuaded more people. They liked him a lot, but, you know people, they get tired of pursuing something so badly and, when it falls through, they just want to crawl back and lick their wounds. So they did and Felix left.

Casaubon wasn't clear about what happened. He knew that the proposal for the extension was put to the church and voted down. After Felix had gone, a new batch of elders was elected. They had enough to cut their teeth on with emerging competition to the café and nursery from a string of new shops nearby and an expanded school, the loss of the pastor and the stepping down of Nicholas, who were both the backbone of the church. Things started off fine. They hired Walter, as interim pastor, who had the same vision as Felix and Nicholas. He set out to learn about the local area. He wanted to gear the church to respond to the local needs and review Bethlehem's activities. Walter still lived in another town, further inland, and would travel nearly every day to Bethlehem to reorganise the church. He was clear that he would stay only for a couple of years or so and that the elders had to be in charge. Three years later, after a long soul-searching, Bethlehem found a new pastor.

3. Modern, Relevant, Inclusive

The picture that emerges, from the local understanding of religion and church-doing, paints a stark difference between the "traditional church" and Bethlehem church. The "traditional church" is a "holy huddle", as it is often described, an inward-looking club of like-minded people focussed on worship; an exclusive and unfriendly community, an old-fashioned institution where theological language dominates church services. In contrast, Nicholas and Felix, wanted to challenge the image of the church, as an ancient building resounding with traditional hymns played by an organ, and envisaged, in its stead, a church that was

"inclusive and welcoming". They experimented with innovative forms of church-doing by having services with a contemporary feel, groups and activities that met local needs, and, above all, to create a welcoming community where everybody would feel accepted. Thus, the vision of Bethlehem is to communicate Christianity to all, especially to those who have an image of Christianity as outdated, exclusive and dogmatic.

In the previous section, Nicholas explained how they began to think of new ways of doing church and asked: "Why do we do church the way we do church?" He realised that the "primary objective in running a church service on Sunday morning is for people who don't go to church". That was the vision: "tell the story to those who don't know it" (Nicholas, interview). In the words of Felix, the vision was for a church that was "modern", and "relevant" to people's lives. Thus, the dress is informal, the songs are contemporary, and the text of the passages read or of the hymns sung, during the service, is projected on a screen using power-point software. The sermons, mostly delivered by the elders and the pastor, are in everyday language and shun theological jargon. The idea behind it was for the church to function as a 'social enterprise' (Dinham 2011 and 2012), which required a different type of organisation, one with more flexible structures and supported by lay members.

Bethlehem is run by seven elders, a full time pastor and a group of trustees. Members are invited to provide feedback, ask questions and make suggestions at the fellowship meetings that are held two to three times a year. They are also encouraged to run activities and propose new ones. Members' involvement is essential not only to ensure that there are

services and groups for the local community, but as a hallmark of participatory culture. This model of church-doing seeks to move away from – what is seen as – the hierarchical and institutional church to that of a church set up and run by its own members. More participatory structures respond to the changed social environment of contemporary Western societies that are far less hierarchical than in the past.

The picture of Bethlehem that emerges is one that is consistent with the innovation in church-doing described by Miller (1997), showing the cross-pollination between British and American evangelical traditions (Hatcher 2010). Bethlehem, above all, sought to be a physical and emotional place not only for church members, but also for the local community. Accordingly, the attempt at refashioning "church-doing" for contemporary times led the founders to carry the message of "modern, relevant, and inclusive" onto the building itself. Bethlehem does not "look like" a church: there are no Christian symbols, such as crosses or paintings. It was designed and built as a community centre, often mistaken for a non-religious community centre, and is generally called "the Centre" by the local community and church members. The design was an exercise in translating the "Christian message" into practice. The aim of the building was to build bridges and create a friendly home open to all, seven days a week.

"When we sat down and I walked into the main hall and went 'are you sure this is a church, Winifred?' There's no priest, there's no altar, there's no gold or decoration, this just looks like a community hall and it was and it is and that's the idea. It's *reaching out to the community*" (Camden, interview).

Bethlehem also shares with 'new paradigm' churches, such as Saddleback Church, the same boundaries between a core group of committed members involved in 'ministries', and those who attend the church. Among the latter are those who attend religious services and those who just attend activities for the community, such as parent and toddler groups. A further distinction is between members, who have joined the church, and associates, who attend and participate but have not joined formally. At the time of writing, there are around 120 members and 80 associates. The act of joining the church requires accepting the 'statement of faith' (Appendix A), attending a short membership class, which explains the vision of Bethlehem and what is expected of members, and making a public statement during the Sunday service affirming one's willingness to join (Appendix C).

There are no set dues for joining the church, however, members are encouraged to give money and time to demonstrate 'stewardship', one's commitment to the church, but also ownership of it. Bethlehem, therefore, dedicates a Sunday service a year to stewardship, "where people are reminded to give themselves, their gifts, time and money", as Felix put it (Felix, interview). They are called to live out their Christianity in their personal and professional lives, but also as 'stewards' of the church. Individual believers have thus a stake in the church. These features reflect the central role of the individual in pursuing salvation and the tradition of Protestant voluntary society (Van Dyck 2007). The internal structures of Bethlehem also show the influence of the contemporary culture of participation. Thus, Bethlehem, in seeking to be "modern, relevant, and inclusive", has adopted the forms of contemporary organisations; however, to see its narratives and

practices as a form of 'branding' in today's consumer culture would miss their rationale and dismiss the importance of commitment, as explained below.

4. Choice and Commitment

The reflections in the section above stressed the role of participation as part of "relevant, modern and inclusive" church-doing. Miller (1997) argued that participation resulted from wider social processes of democratisation. It could be argued that churches are changing to attract and accommodate their audience. Indeed, many members of Bethlehem chose the church after having visited and attended others. Did they 'shop' for a church (Montemaggi 2013), as rational choice theorists would claim? Were they looking for a church that fitted with their views and lifestyle? Selina, who is a regular attendant of the 'women's prayer group', told me, during her interview, that she likes it because she feels free to share with other women what goes on in her life. Dorothea said that when her husband went to Bethlehem, he told her that he knew that they would "fit in this church" (interview). She added that she felt that it was "the right place" for her.

If we were to adopt a rational choice theory framework, it could be argued that they sought to 'maximise profit'. Accordingly, 'profit', in this case, is not pecuniary, but emotional. It could thus be argued that members of Bethlehem church are involved in the religious and social activities because they perceive that they will get something in 'return'. Thus, 'feeling

at home' and friendships are the 'goods' people gain by attending church. Dorothea, in her interview, contrasted Bethlehem church with other churches she had attended previously, and stated that she felt Bethlehem was the church for her, and that she felt accepted for who she was. This is in line with the emphasis on acceptance found in 'new paradigm' churches by Miller (1997, p. 21). Thus, Bethlehem, by being a 'spiritual home', as it is described by members, where everybody is welcome and where people care for one another, provides a comforting physical and emotional place.

Bethlehem 'fits'; it is 'the right place'; it 'meets people's needs'. It satisfies many of its members' needs by providing services, such as parent and toddler groups, and personal relationships. The guiding philosophy of the church is explicitly oriented towards meeting the physical and emotional needs of a person. Rational choice theorists might argue that Bethlehem members choose the church for its multiple benefits, including a caring community, friendships, and social groups. This understanding of choice, however, is an example of the failure of rational choice theory to comprehend the connection between choice and commitment. The following excerpts provide a glimpse of the reasons behind Celia's and Dorothea's choice of Bethlehem.

Celia and Arthur used to go to another church, but they had small children and wanted somewhere that had young families. Some friends recommended Bethlehem and so they gave it a try. It was "a breath of fresh air", reminisces Celia, "We were really, we were immediately embraced and welcomed [...] it did very soon feel like the right place for us to be." It was a big decision to move

house and change school for the children, but Bethlehem felt like home. They liked the place and the vision of "being Christ [...] to be like Jesus, become more like Jesus in the community, the idea of really serving and having opportunities to serve." Just seeing how everybody was involved in things made the place attractive. There was real commitment from the people of the church to serve the community. She was pretty clear that a church is more than a Sunday service. "It is no good to just come along and sit there on a chair [...] and then go home. That's not, that's not really what is all about [...] God has something for everybody to do, whatever that is, it might be to be at the front, it might be stacking the chairs. Everybody has a part to play" (Celia, interview).

"It's that sense of community, of home", says Dorothea. Dorothea and Will moved to the area a couple of years ago. They were welcomed from the time they got there and it was now their "spiritual home". Dorothea compares her experience of other churches. It's not that they were bad Christians, but there is a difference. "When I came to Bethlehem, I was accepted for exactly who I am." She doesn't like the "kind of religion [...] you go to church on a Sunday and the rest of the week it goes out of your head and it can be easy to fall into that trap and I didn't, I never wanted to fall into that trap". There's no point in hiding in Bible study meetings if you don't "live your faith in the community". In a previous church, she was once asked whether she would go to the bible study, she replied that she was going out on a pub crawl. She horrified her listeners, but that night she gave 'her testimony' twice, telling people why she had become

a Christian. "It was more useful than if I had been in the bible study." So she liked Bethlehem, she didn't have to pretend and behave. She didn't have to keep away from non-Christians. "At Bethlehem, they realised that a church can't run on an evangelist on his own, it's not all about one person, everyone should take a role and no one should be taking a back seat. It's not the minister that is the church, it's the people who comprise it" (Dorothea, interview).

The "spiritual home" is not merely a place of emotional and physical comfort, but a place of growth in one's relationship with God. The believer chooses God, yet this choice engenders commitment. By choosing God, the believer establishes a 'personal relationship' with God, which, in turn, calls on the believer to 'answer'. The initial choice thus becomes a commitment to 'live differently' and be in accordance with 'God's will' by following – what is often described by informants as – the "topsy-turvy" logic of faith. Thus, whilst commitment might have originated in an autonomous choice, the 'personal relationship with God' transforms that choice into dependence on God. In other words, believers choose to commit, but in doing so, they create a bond with the overarching authority of God. Prayer is asking God to manifest God's will. This requires trust in God and, therefore, acceptance of God's will, even when it seems to be against one's own perceived interests or what is considered 'rational' in the general culture one inhabits.

"When we ask for revelation from God, we need to be prepared to deal with what comes from it" (Selina, women's prayer group, 3rd July 2009).

"... make yourself open to the answer you don't want" (Will, home group evening, 25th November 2009).

We begin to see how religion reflects Simmel's understanding of religiosity as a mode of consciousness, where belief is intended as *belief in*, or trust. Believers have often remarked how difficult it is to understand what God 'says' and that they 'get it wrong'. The religious frame of mind calls on the person to maintain the 'awareness' of God in seeking to understand the will of God. Religious life is the fruit of a personal relationship with God, which requires a deep commitment and a distinctive way of living. Such a way of living is seen as 'countercultural' for it is centred on God and God's will, rather than society's ways. This includes being in opposition to the 'traditional church', seen as irrelevant to people's lives, and of the 'world-friendly church', which fails to challenge the cultural norms outside its doors.

Celia and Arthur, Dorothea and Will chose Bethlehem because of the church's inclusivity. They did not choose a church that 'fitted' them, but one that was open to all and encouraged them to be open to all. They felt the church was relevant to people's lives by being inclusive and having a positive impact on the local community. Informants identify strongly with the 'vision' of Bethlehem as a 'community-focussed' church seeking to meet the physical, emotional and spiritual needs of people locally. This vision is embodied in their ministries, where members form relationships with people in the local community. However, Bethlehem church is not merely an exercise in 'church-doing', in organising and

running a church to attract the unchurched, and provide services to the community; rather members at Bethlehem seek to "follow Jesus" by building relationships in the community.

The rejection of institutional religion is accompanied by an emphasis on authenticity: being true to the example of Jesus, which calls for a continuous effort to live up to that ideal through the practice of relationships. The notion of authenticity provides a distinctive identity and grounding in – what is deemed to be – the essence of Christianity. Yet, Christianity, as a practice of relationships, erodes the primacy of traditional doctrinal beliefs, such as the exclusive doctrine of salvation, and the strictness of specific ethical stances, such as regarding homosexuality. This understanding of authenticity and the ensuing practice of relationships trump the adherence to doctrinal statements.

5. From Doctrine to Authenticity

Godwin, manager of the centre, states in his introduction to the Alpha course: "We don't do religion, religion causes problems. We are about relationships". At Bethlehem, Christianity is understood as relationships and, hence, as concrete experience of God through others. It is not focussed on the experience of the otherworldly numinous; rather Christianity is seen primarily as divine love lived through human relationships, in the "here and now". The Christian faith is contrasted with "religion" that is made up of theology, the "heavy stuff", as a member described it.

"...the starting point is 'God is good'. If you start from a, like most people who don't know God and don't know faith, that God is either a God with a big stick or he's this tyrant in the old testament or he is not here, he is an absentee landlord, that kind of approach, then I think you are missing out on religion. If I call that, I don't like the concept of religion. I don't think religion is good [...] religion is adhering to a bunch of principles and a way of living what you think it will make you good with God, whereas faith, Christian faith and the Christian experience of life, is about not you, but about Jesus, about God coming down in the form of man and meeting you rather than you having to do something to attain to his standards" (Camden, interview).

In this dialectical opposition between the "real church" of Bethlehem and the "traditional church", between religion and faith, or the Christian life, we begin to glimpse how Christianity is being interpreted and articulated, and why doctrine becomes secondary. The distinction made by my informants between "religion" and "faith", or "being a Christian" testifies to a discomfort and, often, rejection of Christianity understood as adherence to doctrine. Authenticity, here, is formulated as a 'doing', a process, in which religious actors engage to live their faith. It is the continuous attempt at being true to what is understood to be "the truth of Christianity". The search for authenticity is the search for a Christian life that goes beyond institutionalised religion. It is a constant and never-ending quest for truth. This is primarily articulated in the conception of Christianity as relationships, which is deemed "countercultural". By the term "countercultural", informants identify the

commitment to a lifestyle that counters the materialism and self-interest of contemporary society. Thus, authenticity is a dialectical process of discarding the "falsity" of social life as much as the institutional aspects of Christianity.

Home group evening, 7th October 2009

Arthur: "We tend to think that we own what we have ... the culture around us tells us so. Materialism is a fundamental part of our society, we are part of society so it affects us ... The challenge of the bible is how we deal with our possessions and money".

Accordingly, being a Christian means challenging the dominant culture. Bethlehem seeks to be the refuge that goes against selfish and empty consumerism; a place of relationships in an individualistic competitive, and lonely world. The interpretation of Christianity as "relationships" signals dissatisfaction with contemporary society, seen as dominated by individual autonomy, to which Christians counter a conception of human beings as "interdependent". Thus, the possibility of forming relationships with people in the local community resonates with the idea of church that many members hold. When Celia, now an active member of Bethlehem, was looking for a new church with young families, she liked the vision of "being Christ [...] to be like Jesus, become more like Jesus in the community, the idea of really serving and having opportunities to serve" (Celia, interview). It is in community, in relationships, that Christians can be authentic and follow Jesus.

Home group evening, 22nd June 2011

Arthur: "instead of staying on your own, you should be forming relationships with neighbours, rather than crashing out on the sofa and eating crisps. Be out there engaging with someone. It's important to have community interaction".

Nicholas: "when we did a survey in the local community, 20 years ago, one of the biggest issue was loneliness. One of the reasons why people come into the centre is that they want to feel part of something".

The identity of Bethlehem church is based on the idea of a church "in community". As Felix explained: "The church itself is a community, but also it is 'in' community, in the society in which it lives and serving that community, blessing that community". This identity is underpinned by a theology centred on human relationships, which mirrors the relationship between the believer and Jesus. Christianity is thus not something to be believed, but something to be experienced and, above all, to be lived out. Christians are "followers", not "believers". The relationship with the divine is translated into human relationships. Relationships are understood as the realisation of "following Jesus": loving others by welcoming and accepting them for who they are. These are not exclusive relationships among church members, but open to the wider community.

Thus, being "relevant, modern, and inclusive" needs to be understood within the framework of authenticity. It is not an end in itself; rather it is a way to form and live relationships with others. Bethlehem is "relevant" because it is "real"; it speaks to people

without the mediation and artificiality of hymns and pews. The aspect of "relevance" is also part of the construction of authenticity. Bethlehem, therefore, is not simply adapting to culture in the pursuit of "relevance"; rather, by being "countercultural", it maintains the tension with culture. This notion of authenticity rests on Christianity transcending culture, which includes institutional Christianity, and thus being "true" for the whole of humanity and for all times. Thus, being a Christian is "being like Jesus": incarnating a timeless truth that often runs counter to today's customs, but also to "religion". The "countercultural" stance has the potential for undermining doctrine at its very core, as shown in the next section.

5.1 The Practice of Relationships and Salvation

The vision of acceptance and inclusivity is translated in allowing non-members to participate in the activities of the church. The relatively high number of associates in proportion to members means that there are associates who participate actively in the life of the church. This includes a woman from a non-Christian background volunteering in one of the groups. The significance of this openness should not be underestimated. Christian evangelicals generally subscribe to the doctrine of *sola fide*, according to which salvation comes only from belief in Jesus and not "good works". Therefore, a "good" person is not saved unless they believe in Jesus. At Bethlehem, this doctrine is secondary to the message and practice of relationships. Some of my informants struggle to reconcile a loving God with an exclusive doctrine of salvation, while others reject the doctrine outright and are

quite certain that salvation is not exclusive to Christian believers. At my direct questions over salvation, many informants felt that they could not "make that call".

Nevertheless, there are members who believe that people from other religions are not saved, although they have access to the truth of God. At a Bible study group, Caleb, the group leader, talked of "false teaching". Among the false teachers were American televangelists who advertise healing and ask for money. Caleb affirmed that "the claim of Jesus (of being the way to God) is sacrosanct", however this did not seem to be exclusive. He explained that: "The truth is like a circle and if we represent different religions through symbols, such as the star of David, the Crescent, they cover part of it but not all. Only Jesus Christ covers it all. However, most Christians' understanding of that truth is also partial so there's a need to be humble and learn from others, including those from different faiths". (Field notes, 4th July 2009). Thus, non-Christians can know God and have a relationship with God, although it is not clear whether this is sufficient for the salvation of the soul.

Talking to Felix (informal conversation, 28th June 2012), the former pastor, enquiring whether non-Christians could enter heaven, he replied "It depends on what you mean by heaven". He did not believe that non-Christians were damned, but that there was a difference, although he did not elaborate on it. He said: "I believe in Jesus' 'I am the way, nobody comes to the Father (God as the father), except through me". Nevertheless, he sees religion as people's "attempt to get hold of God, Christianity is part of it, but there's also revelation". The real difference, for Felix, is that "(Christian) faith is 'sanctification', it needs to transform, renewing your mind and make you think the way God thinks". Thus, faith is

having a relationship with God, evidenced by self-transformation, expressed in love for others. Again, Felix seemed to suggest that non-Christian souls were saved, although they did not enjoy the same relationship with God and the transformative aspect of sanctification. During a home group evening (24th June 2009), Camden, faced with the question of salvation, said: "there must be something else". In his interview, he said the following:

"I see God at work in Christians and non-Christians alike and one of my great friends in my work at the moment, who isn't a Christian. [...] He says a lot of things which I regard as true and someone who doesn't accept that he is religious in any way or Christian in any way saying things that have a truth means that in this world there is truth and we are all part of that truth. God will work in individuals to teach you truth and to show you more truth, but it's not something which is just for Christians cos it's not meant to. [...] So it's always the process of God working in us. The more you hang on to him, the more you trust him, the more you spend time with him, the more he's gonna change you, hopefully the better. That's the road we are all on and that's the road we should be on and even people who at heart are unconcerned, are not Christian, haven't dedicated their life to Christ are still on that path, they just don't know it yet".

A possible reading of the text above is Camden's acceptance of a pluralistic notion of truth. Camden seems to suggest that truth is shared human values, albeit originating in God, and feeling compassion taps into that truth. However, I would like to suggest a more subtle

appreciation of truth. Camden, whilst recognising truth outside of Christianity, affirms Christianity's privileged, but not exclusive, way to the truth. This reading is supported by a subsequent conversation with Camden. Camden and Felix point to Christianity being the way to be authentic as human beings and as Christians, which is expressed in having a relationship with God and with others. This shows a shift in narrative from eternal salvation to authenticity in a church that is a relatively mainstream evangelical church. Authenticity means being a Christian is a practice sustained by belief in Jesus.

Thus, Christianity is "following", rather than (doctrinal) "believing"; "being in relationships", rather than "being judgmental," and "serving God", rather than staying in the "holy huddle". Belief in Jesus does not necessarily entail belief in the doctrine of exclusive salvation. Belief in Jesus is a call to being like Jesus in relationship with others. The narrative of Christianity, as something to be experienced and lived in relationships and carrying a "countercultural" message, constructs how authentic religion, or Christian life, is understood and practised at Bethlehem. Authenticity is the practice of being true to the example of Jesus, who is seen as the embodiment of compassion. The practice of relationships is thus underpinned by compassion and, in turn, relationships reinforce the primacy of compassion over doctrine, as in the case of salvation, and over judgment, as in the case of homosexuality.

5.2 The Practice of Relationships and Homosexuality

During the three years of fieldwork at Bethlehem, I noticed a conspicuous absence of judgment of homosexuals. I have never come across homosexuality being mentioned in derogatory tones. The first encounter with the subject of homosexuality came at a study group (24th June 2009) led by Caleb, an elderly member of the church, who referred to a legal case where a Christian registrar had been fired for not wanting to perform civil partnership ceremonies. The reference underlined the sense of imposition of external values on Christians. However, there was clear and unequivocal condemnation of criminalisation or persecution of homosexuals. It took nearly a year for the matter of homosexuality to resurface in the form of homoerotic light-hearted innuendos among participants, during a social evening. Camden offered Humphrey a lift and joked about his wife being out of town. All the people present laughed and kept the joke going. I was surprised at the ease of the conversation.

Indeed, talking informally to Camden and Winifred, they were clear that they had no objections to homosexuals being allowed to get married. This, of course, is not to be taken as representative of the views of the majority at Bethlehem. Yet, the most illuminating example of the overarching value of compassion over judgment came from Walter. Walter was the interim pastor at Bethlehem after the founding pastor, Felix, left the church four years ago. After a few years, he was involved in the recruitment process for the permanent post of pastor. Walter participated in the final interview, which involved one candidate. He told me that he asked the candidate how he would deal with homosexuals, single people,

and people going through divorce. He told me he liked the answer as it was pretty much how he would have answered so I asked how he would have answered.

"We need to embrace people as fellow human beings regardless of the circumstances. In many churches, they reject the person, which means not receiving one another, not even within the church. You can't reject the person because of the circumstances they are in. That's what happens very often in faith communities. I have lovely times with people, who sometimes would come and cry on my shoulder. It's being a human being, for goodness' sake, you know? And showing myself vulnerable as well. I think that our evangelism is to get to the core of people, I have no right to tell you what to do, no right at all, but if I care about you, you might trust me enough to ask my opinion on things or to ask me how could this work for me? That's where I feel the church must be. Traditionally, the dogma of churches gets in the way 'I believe this so you must be wrong'. As a result, they can't see the person of another faith as a person. We are made in the image of God, this means we are equal. We need to identify with one another's humanity. This is core to Jewish tradition: how do you deal with the stranger?" (Walter, interview)

6. Discussion

The narratives and practices of committed members of Bethlehem, recounted above, show a long-standing search for a church that is "real". Disenchanted with institutionalised

religion, but, above all, with the exclusivity and judgemental culture of the "traditional church", this group of evangelicals sought to create a community that responded to people's spiritual and social needs in today's world. From this search emerges the notion of authenticity, as a 'doing', a process of discovery of how to be a Christian. Authenticity is articulated through the dialectical opposition between "religion" and "faith", "believing" and "following", "traditional church" and "real church", but also between "culture" and "counterculture".

Authenticity is lived in the practice of relationships, underpinned by the value of compassion. Human relationships thus mirror the relationship (authentic) Christians have with Jesus and are the concrete expression of that relationship. Through human relationships, Christians gain knowledge and experience of Jesus and become like him. Thus, being true to Christianity is not exhausted by belief or experience of the divine, but it is primarily about practice. Religious life is not about experiencing the numinous, but being like Jesus in one's daily encounters. Therefore, the attempt at being "relevant, modern, and inclusive", as a church, is not a form of refashioning, or of modernisation, of "church-doing", but a way to practise authentic Christianity.

In a pluralistic society, the practice of relationships and the value of compassion undermine traditional doctrines and ethics, such as the exclusivity of salvation and the condemnation of homosexuality. Although my informants, including Camden, are not necessarily doing away with exclusive salvation, they are relaxing those boundaries and rethinking what Christian "truth" is. The recognition and "embrace", to use Walter's words, of humanity

gives primacy to the "truth" of human relationships over doctrinal "truths". As mentioned in chapter two, this shift to authenticity and critical rethinking and refashioning of Christian identity is already present in the 'emerging movement', which is often conversant with 'hermeneutic deconstruction'. Bethlehem is not an 'emerging church' as such and, indeed, began as a Baptist church. I posit that the emphasis on practice and authenticity rather than the salvation of the soul and doctrine is gaining a wider appeal. This includes divergence from – what used to be – official doctrine. Research, in the United States (Putnam et al. 2010) and in the 'bible belt' of Norway (Repstad 2003, 2008 and 2009), shows that Christian evangelicals are also shifting in their understanding of doctrinal beliefs.

Putnam and his colleagues found that "83% of evangelical Protestant say that those not of their faith could go to heaven" (Putnam et al. 2010, p. 535). When asked whether this included non-Christians the figure went down to 54% (Putnam et al. 2010, p. 536), which is still a significant proportion. Evangelicals also recognised "basic truths in other religions" with just over 20% stating that only one religion is the true one (Putnam et al. 2010, p. 546). Putnam's reason for the high levels of tolerance in American society, including among religious people, is due to the high level of diversity and fluidity of religiosity. For instance, Putnam reports that about a third of all Americans "choose their religion rather than simply inheriting it" and that about half of married Americans have spouses from a different religious tradition (Putnam et al. 2010, p. 148).

In Europe, Repstad also documented a shift away from a more sectarian outlook and towards an understanding of Christianity, which stresses experience and community (Repstad 2009, p. 127). According to Repstad, "[D]ifferences are still expressed, but less dramatically than before. There is more talk now about practicing Christians and active Christians, less about saved people, not to mention about non-saved" (Repstad 2009, p. 128). Indeed, the subject of hell has become "an unpleasant dogmatic issue in conservative circles" (Repstad 2008, p. 22). Repstad also supports the connection between the change towards a more 'liberal' Christianity and the encounter of other worldviews: "Christian leaders ... when asked to reflect, they often say that changes in their views are caused by challenging encounters with people" (Repstad 2008, p. 20). This lends credence to the broader cultural accommodation of evangelicals (Hunter 1985 and 1987; McConkey 2001; Shibley 1996).

The main reasons, I advance, for the stress on authenticity rather than the exclusive notion of salvation are three: pluralism, contemporary economic and social life, and individualism. Thus, in the first instance, the encounter with others has made many question their doctrine of salvation and judgment of homosexuality. Pluralism is, here, not the presence of different religious and non-religious traditions, but the normative framework in which such diversity is seen as "a positive value" (Beckford 2003, p. 81). This leads to an articulation of Christianity as a distinctive identity, a part of multicultural Britain, but also to the acceptance of others on an equal footing, which may clash with doctrinal beliefs. This is compounded by the lesser influence of religious authorities in post-1960s Western culture (Brown 2001 and 2010; McLeod 2007).

The 1960s cultural and societal shifts have led to the legitimisation of individual autonomy thus opening the door for a wider reinterpretation of doctrine in the light of individual experience. This does not necessarily mean that individual autonomy dominates over external authority; rather that we witness a rise of multiple centres of authority (Wood 2007). Wood interpreted the shift as "the relativization of multiple authorities, such that no single authority (or range of authorities) exerts a formative influence within the life of a group or individual, and that participation in them comes overwhelmingly from a certain class fraction" (2007, p. 71). The increased cultural diversity of Western societies has allowed religious actors to be in contact with a variety of traditions and interpretations, and to rely for information on a wide range of sources. In post-1960s Western societies, the coexistence of multiple, often overlapping, identities has given rise to a pluralistic framework where diversity is not only accepted, but valued. Thus, pluralism, understood here as the acceptance and value of diversity (Beckford 2003, p. 81), provides a paradigmatic shift in cultural framework for the re-definition of Christianity.

In the second instance, I argue that the change in material circumstances of the West. As mentioned in chapter one, the nineteenth and early twentieth centuries oversaw remarkable economic growth and improved life with the most recent and spectacular phase taking place in the 1960s. This has had an impact on the focus of religion. From the late 1950s to the early 1970s, the West saw a massive change in living standards, industrial output, and technological development. Although "the Golden Age was a worldwide phenomenon" (Hobsbawm 1994, p. 259), it was the West that experienced the most radical changes. In the 1960s, west European unemployment averaged at 1.5% (Hobsbawm 1994,

p. 259). "World output of manufacturers quadrupled between the early 1950s and the early 1970s and ... world trade in manufactured products grew tenfold" (Hobsbawm 1994, p. 261). "In the 1930s, even in the rich USA about a third of household expenditure still went on food, but by the early 1980s only 13 per cent" (Hobsbawm 1994, p. 269).

The economic boom of the sixties had a profound cultural and social impact. The 'counterculture' together with the nascent consumer society emphasised individual distinctiveness and self-expression transforming all aspects of society deeply. I argue that these monumental changes have allowed a perspective focussed more on the "here and now" and less on the "thereafter". Salvation was about eternity, but eternity has become less of a concern. The authenticity of my informants seeks to give meaning to life today, not only in prospect of eternal judgment. Consequently, truth is not exclusive to Christianity; rather the distinctiveness of Christianity rests on the process of authenticity, which is expressed in (compassionate) relationships.

In the third instance, the traditional individualistic perspective of Protestant Christianity is, here, translated in the search for authenticity. Salvation is based on the faith of the individual and is evidenced by individual transformation. Against a pluralistic cultural context, which calls for distinctive identities in the sea of diversity, Protestant emphasis on inner search and personal faith is translated in the search for authenticity. Thus, evangelicals are seeking "authenticity", to be true to the example of Jesus that goes beyond the historical forms and teachings of the church. Yet, it is precisely this search that forces

constant change to go beyond the social constraints of a specific time. In seeking to transcend themselves, evangelicals reinvent evangelicalism.

7. Conclusions

The original vision of Bethlehem church was to provide a sense of community and to articulate Christianity in a way that could be meaningful to people's lives and sense of self today. Bethlehem fashioned its identity in opposition to the "traditional church" to embody a vision of Christianity that seeks to be more accessible to all, but also truer to the message of Jesus. The vision of Bethlehem, to be a welcoming and inclusive church for all, is grounded on relationships. It is a vision of authenticity for the person developed through human relationships, which reflect the relationship between the believer and God. Thus, Bethlehem does not simply strive to be "modern" and "relevant" to adapt to the times; it does not simply provide comfort to church and community members; rather it aspires to be the place where people can seek to be authentic, as human beings and Christians, by forming human relationships.

The chapter has shown how the narrative and practice of relationships is having an effect on the interpretation of doctrine. The encounter with others is eroding sectarian interpretations of doctrine. The study does not seek to provide a representative sample of Christian evangelicals in the UK; rather it highlights how many active and committed

evangelicals struggle with the doctrine of salvation and, often, reject it, on the basis of an overarching ethics of compassion (cf. chapter ten). Christianity is thus deemed by informants not a "religion", but "a life", a practice that requires commitment and involvement. It is "countercultural", in opposition to mainstream culture, seen as individualistic and materialistic, and in opposition to the inward perspective of the "traditional church". Bethlehem's culture is grounded on a conception of Christianity as "relationships".

The practice of relationships is centred on compassion for others, which reflects how Jesus is seen at Bethlehem. This results in a shift in emphasis from doctrine-based religion, as adherence to a set of theological convictions, to authenticity. This is not to suggest a complete foregoing of doctrine, but an important change in how Christianity is understood and practised in everyday life. The search for authenticity is the fruit of macro-level social and cultural changes, but also of the day to day practice of relationships in a pluralistic society. Thus, the shift in emphasis from doctrine to authenticity is part of a wider trend of refashioning religious and spiritual expressions in diverse and liberal Western societies. This has enormous implications not only for the innovations in church-doing and theology, but also for the engagement of evangelicals with the public sphere.

Chapter Nine

The Authenticity of Being a Christian

1. Introduction

The previous chapter showed the 'move' from doctrine to authenticity. Bethlehem, like 'new paradigm' churches, rejects – what is felt as – the institutionalisation of religion, its formality, hierarchies and intellectualism, and locates religiosity in a personal encounter with God, which is lived out in human relationships. Being a Christian, however, is not just an experience of the divine, but a commitment to practise a way of life. Informants oppose the view of Christianity as holding a set of beliefs or belonging to a church, and argue that being a Christian is about being true to the example of Jesus. Christianity is understood as a continuous effort to be true to the values embodied by the figure of Jesus. It is an attitude expressed in being compassionate towards others. In a pluralistic society, which recognises the value of different worldviews and traditions, religious identity is more than belief

and/or belonging (Davie 1994, 2002, and 2007; Day 2011), but a mode of consciousness, which calls for commitment to practise.

This is best understood through the framework of authenticity, outlined in chapter six. Authenticity is grounded on a relational and processual understanding of the individual in three dimensions: a 'social I', an 'individual I', and a self-transcendent I'. Authenticity is thus a process of constructing one's identity in relation to one's social context and relationships, to one's personal individuality, and to one's understanding of truth. Specifically, authenticity rests on the actors' awareness of truth, inscribed in their tradition's religious narratives, and their commitment to practise through the interpretation of their tradition. As mentioned previously, I identify religious tradition with the religious symbols, rituals, norms, customs, beliefs and narratives associated with a religion regardless of their 'official' status. Legitimacy derives from what is felt to be authentic by the group and by individual religious actors (cf. chapter seven).

Mirroring the multidimensionality of the self ('I and Thou'), as outlined in chapter six, authenticity is articulated as grasping the truth, which has a social, and individual, and a self-transcendent dimension. Accordingly, the search for authenticity, at the 'social' level, involves acquiring a Christian perspective by interpreting Christian tradition and committing to follow Jesus in one's daily life. This often requires rejecting mainstream cultural values that clash with it, such as consumerism. Being a Christian is thus a practice, rather than a belief system. Christians seeking authenticity are engaged in the process of interpreting and constructing religious tradition. Authenticity is social because one's

interpretation of tradition is informed by one's social and cultural context, including one's socio-economic status, education, and demographic characteristics, such as age, gender, and place of residence. The social dimension also refers to the religious group of the religious actor, in this case the home group and the wider church community.

Authenticity is 'individual' because it needs to strike a chord with the individual. The individual dimension indicates that truth needs to 'make sense' to the person individually, affecting them emotionally and intellectually and is always in reference to one's tradition. The person thus constructs her identity as a Christian by inserting herself within the framework of her understanding of Christianity. Self-transcendent authenticity refers to the attempt to tap into universal truth, understood through one's tradition. Building on Simmel's concept of self-transcendence, one's realisation that the world is but "one side", allows one to transcend it by virtue of this very recognition. The transcending is momentary; however the sensation of the other side, so to speak, of awe, of truth, sets forth a continuous process of self-transcendence, which is expressed through practice.

Authenticity is thus not an essence, but a process of constructing one's identity at the level of one's consciousness and one's behaviour. It rests on identifying what informants feel is true Christianity, which is often in contrast with wider society's values, thus marking the distinctiveness of Christianity. Authenticity is articulated through one's commitment to practising Christianity intentionally: with consciousness that one's practices are in service to God. It is not just a search for replicating what is seen as the model of early Christianity, nor is it an attempt to modernise tradition; rather, the search for authenticity constructs

one's identity in relation to one's tradition and one's sense of truth, which may be understood as universalistic, but is articulated through the particularity of one's tradition (cf. chapters five and six).

The process requires personal participation and interpretation on the part of individual actors. Accordingly, the interpretation of the figure of Jesus and of early Christianity is not a historically or philologically accurate exegesis, but an ideal that guides the practice of being a Christian. The image of Jesus is translated into narrative and practices for living life today. Christian tradition is subject to active reflection to make sense of one's everyday life. Thus, being an authentic Christian is, first and foremost, a practice.

2. The 'Social' Dimension

The analysis of Bethlehem's discourse and practices, in the previous chapter, points to the key aspects of how authenticity is viewed in the local culture. Being "relevant" and "countercultural" and, above all, forming relationships are important *emic* categories of authenticity. In this section, I explore further the construction of Christianity within the group. The first extract is from a home group discussion during the week dedicated to 'stewardship', in which members' commitment in terms of money and participation was discussed. The second extract refers to a discussion based on questions in a DVD series aiding the study of the bible.

Home Group discussion, 7th October 2009

Arthur: "We are stewards; we are not owners of anything. The culture we live in

is one that says to us 'that is mine'."

Nicholas: "It's countercultural"

Arthur: "The Bible says that God gives it all. I want you to manage your money

wisely to serve others and for kingdom's purposes. God says 'I'm keeping tabs,

I'm watching what you do'. The effect of culture on us is huge. How we guard

money is a reflection of the way we are. God is looking to challenge our

perspective."

Nicholas: "God wants to lead us, learn the right way."

Arthur: "how much of God's money am I keeping for myself? It's upside down.

[...] We are either countercultural or not at all. You're either in it or you're out of

it. If we do things begrudgingly [...] it's not the attitude God wants. It challenges

to our core.

Home Group discussion, 24th November 2010

Selina: "why is an attitude of obedience crucial to fulfil this role (of leader)? [...]

he's asking us to be different."

Arthur: "if we choose to be obedient is like walking in the right direction if we

choose not to be obedient, we fall a little bit and go backward."

263

Selina: "people have this expectation of you (as Christians). People, family, they say to you 'you're a Christian, you can't say that ...'. It's constantly thrown at you. [...] I'm a work in progress!"

The first extract highlights the sense of not owning what one has in life. It is an attitude of humility based on the belief that one's possessions, abilities and fortune are granted by God and, therefore, need to be used to serve God. This is contrasted with outside culture, where people feel what they have is fruit of their labour, that they own it and have control over it. Thus, Christians are "countercultural" for they recognise God's sovereignty over and above everything else. Consequently, one needs to follow "the right way", as put by Nicholas, and make use of what one is granted in this life according to the ethical ideals of the Kingdom of God, rather than what is valued by culture. There was mention of the need for finding a balance between keeping the necessary for everyday life and "giving back to God", although the discussion tended to be abstract, with few concrete examples.

Most of the discussion reiterated the notion of viewing everything one has as coming from God. Thus the focus was on one's "attitude", rather than the specific use of one's resources for ethical ends. Being a Christian requires a distinctive perspective. It is a mode of consciousness, from which action follows. This is evident in the second extract. Selina asks a question from the booklet used to guide the study of the bible, which refers to chapter 19 of the book of Exodus. She adds her interpretation by highlighting obedience as the aspect that distinguishes Christians from mainstream culture. Obedience requires being "different", recognising God's sovereignty, in

contrast with Western society where self-authority is celebrated. As mentioned, choosing to be a Christian engenders commitment.

Obedience is understood by Arthur as the way to stay on the right path. By walking in the right direction, he means living an ethical life, which is corroborated by Selina who points out the expectations other people have of moral behaviour on the part of those who define themselves as Christians. She exclaims that she is "a work in progress", which is more than a joking justification for human weaknesses; it points to Christian life as a continuous effort. Thus, being a Christian is a distinctive way of perceiving the world and practising life, according to Christian values, as understood by participants, but also according to the group's construction of Christian identity as "countercultural".

3. The 'Individual' Dimension

Individual consciousness is fundamental to the process of choosing God and committing to practise Christianity. Being a Christian is articulated as being distinctive, which is beyond holding the right beliefs or acting ethically. It means becoming aware of the truth of Jesus and responding to it. Religiosity is thus clearly a mode of consciousness. Below are Nicholas' and Lucy's accounts of their conversions, which show them becoming aware of the truth of Christianity, coming to believe and the impact it had on their lives. Both

Nicholas and Lucy were suspicious of 'religion' and thought religious people were 'loonies'. Nicholas' conversion spurred him into founding Bethlehem with the intent of communicating Christianity as a way of life relevant to people today. The finding of truth, for Nicholas, was accompanied by a rejection of 'false religion', the religion of 'fire and brimstone'. Conversion is, here, coming to recognise what is true. Thus, we can see religiosity as a mode of consciousness, where belief is *belief in*, trust in God, which engenders a relationship with God. The relationship with God, for participants, grounds their search for authenticity.

Nicholas: 'Belief in'

When he was a teenager, Nicholas began to frequent the youth club of a church with his friends. The church had a café and they thought they could "have a good laugh at those Christians". He didn't grow up in a Christian family and thought that Christians were "a bit like flat earth society people". So, at first, he was surprised by the "rational thinking" of that group of Christians.

"... there was actual rational thinking. People there who understood my questions and surprised me by having answers, so from a number of conversations [...] I guess it was a sort of challenge, you know, you say you don't believe in God, but maybe he is there, why don't you try him? [...] *trust him. It's about trusting*. Pray and just see what happens. If he is there may be he'll do something. And *I prayed, I started to trust that may be he is there*, and I just felt different. [...] and the more I looked into what they were saying and claiming,

and what the bible was saying, the more it made sense to me. [...] I made that step of faith and say 'God, if you're there, I want all this' [...] and something happened. I felt, literally felt quite high [...] Today, 40 years on, how do I describe that? I don't know. At that time, it was a something that happened there, the spirit of God, is one way of looking at it, emotions is another one. Something happened, and probably a mixture of the two, that made me feel [...] I just felt different. [...] I mentioned to my family that I'd become a Christian. They just laughed and said 'don't worry, it'll pass'" (Nicholas, interview).

Nicholas refers to 'trusting that God is there'. Simmel explains that religious belief can be expressed in theoretical form, but that it is not "the content of a mental image"; rather it is "an emotional fusion with Him experienced as a *real event*" (Simmel 1902/1997, p. 130, emphasis in the original). This notion of religiosity and *belief in* God reveals belief as a change in one's mode of consciousness. It emphasises the emotional dimension of belief, as pointed out by Nicholas. Finally, it enables us to understand why Nicholas prayed although he had no belief. Simmel explained that: "[T]hanks to this interpretation – according to which religious belief is a form of existence, is a subjective process itself, and is not, as in theoretical knowledge, its content – all kinds of facts are explained that would otherwise be paradoxical. For example, there is prayer to obtain faith – completely senseless behaviour from the standpoint of common rationality because one evidently can address the prayer only to one in whom one already believes. ... But that one acts in this way proves that one prays to obtain something else, an actual inner reality, a transformation of the way we are

that finds in the holding of something to be true only a point of conscious support or an external reflection" (Simmel 1902/1997, pp. 130-131).

Nicholas felt that something had happened, he "felt quite high". It was a deeply emotional experience that gave a new direction to his life. It was not, however, 'once and for all'. Nicholas, at the time of the interview, had been going through a period of questioning his faith radically. He had doubted God's existence and realised how much "the life of faith is a journey". He struggled with faith, but found "living the Christian life [...] a million times better than anything else". Christianity is, here, a way of life, which requires self-transformation as part of one's relationship with God.

"So, even if I couldn't intellectually understand it, something in there is hugely attractive. [...] There's something about this I don't wanna lose and so I step back in faith and I hold on and, suddenly, I discover he (God) is still there and I can experience him" (Nicholas, interview).

During the interview, Nicholas asked me "why does God demand faith?" Would it not be better to have a permanent physical presence of God in the world at all time?" The narratives of my informants suggest that it is precisely the step of faith, Simmelian *belief in*, which leads to the path of authenticity, the attempt to be true to the ideal of following Jesus. Trust in God gives rise to a relationship with God, which is partly defined by theological ideas. Being a Christian, in the theological discourse of Bethlehem, is defined by having a relationship with Jesus and, therefore, having a personal experience of him. It is the

personal relationship with Jesus, which calls on Christians to be like him by forming relationships with others. The consciousness of being in a relationship with Jesus and of other relationships being part of serving God is pivotal in the performance of authenticity.

Being a Christian denotes a mode of consciousness. From an *emic* perspective, it means becoming aware of the truth of Jesus and imitating Jesus in one's daily life. That consciousness is an essential part of authenticity. Consciousness of one's relationship with Jesus drives the sense of self-transformation, or "maturing", of participants. As mentioned previously, the conscious choice of following Jesus engenders commitment to practise the authentic Christian life. Authenticity, from an *etic* perspective, identifies this process of gaining a cohesive identity by seeking to grasp truth in one's social and historical environment (social dimension), but also in a way that is meaningful to the person (individual dimension). Just as the 'social dimension' of (*emic*) authenticity required critical reflection on institutional religion and materialistic society, the 'individual dimension' of *emic* authenticity centres on gaining a Christian consciousness by breaking away from mind-sets that are seen as self-centred and shallow.

At the individual level, the discovery of the truth (of Christianity) is not simply about emotional and intellectual adherence to a religious tradition, but a transformed consciousness and attitude. It is a "spiritual journey", on which all Christians are travelling. One's personal transformation may simply be a deepening of one's understanding and practice of Christianity, or a more dramatic shift from a previous lifestyle to a "Christian lifestyle". For instance, Nancy, a member of the church, who made use of recreational

drugs, gave them up as a result of becoming a Christian. This was not an easy or linear process. Nevertheless, she adopted a different perspective that enabled her to change her behaviour. Lucy's account of her conversion is particularly dramatic. However, the stress in the narrative is not on an experience of the numinous, but on self-realisation.

Lucy: Becoming Oneself

Lucy comes from – what she described as – a 'non-Christian' family, who "only went to church for weddings and funerals and christenings". She was not interested in religion. She began attending Bethlehem church for its parent and toddler group. After two years, a few people mentioned to her the 'Alpha course', an introductory course on Christianity. She told me that she was not looking for faith, but went because she is "not very good at saying no". After a few weeks, she noticed a change in her. Although she "fought against it" and wondered whether she was being "brainwashed", she began to believe. She recounted always being very volatile, shouting and swearing, especially in the car and then she found calm. Lucy's transformation is a stark example of authenticity as a process that restructures and intensifies one's identity. She became calmer by letting go of her sense of responsibility towards her children. The surrender to God instilled in her self-control.

"... as a family, we only went to church for weddings, funerals and christenings [...] so I had no interest at all. [...] I took the children to the nursery and mother and toddlers, still no interest. And then I think after the birth of my third child, I had made a lot of friends here. So, after two years of coming here, [...] three people [...] in the space of five minutes, said: 'why

don't you come to the Alpha course? You only have to come to the first night, come and see if you're interested' [...] I knew lots of my friends would be there, so I went along to the supper and really enjoyed that. I probably went because I'm not very good at saying no, if I'm honest. I think within about two to three weeks, I knew that something really major was happening in me, which I didn't understand at all and I guess, at the beginning, I really fought against it. I thought: 'no, I didn't come for this. This isn't, I'm not, I'm not a Christian, this isn't how I think' and I thought: 'am I being brainwashed? Am I actually being just coerced into believing in this?' [...] In a period of about three weeks, I really moved from not really caring whether God existed to actually thinking: 'yes, he's real and he's there and he's trying to talk to me'. It's kind of strange because it was a very quick journey. I was not aware myself that I had changed, but my husband said: 'you're so different'. Cos I've always been a very volatile, up and down, up and down, up and down, and shout and swear, always really shout. In the car, really terrible, you know, beeping the horn and being aggressive. And it just left me. It just went. I stopped swearing. I just calmed down. I felt happier. [...] My family noticed, my friends noticed and they were saying: 'what's happened to you? You're really, you're really content'. Even my daughter, who was six at that time, said: 'mum, you're much nicer than you used to be.' And that was really shocking. [...] I can't even explain it. Something in my, it was like something deep in my heart that changed and I felt that all the anger, all the lack of patience, all the frustration had just been replaced by a sense of calm that everything was ok and I wasn't a bad mother and I didn't have to try hard, because I was ok as I am. [...] At that time, I really didn't understand it, but I was quite happy [...] because I had changed for the better. [...] As time went on, [...] I borrowed a bible and I remember sitting for about three days and every time I opened it what I read really hit me, it was for me, it was about me. [...] I got to the point when I thought, d'you know, this is a real problem, because I'm not even challenging this, I'm just accepting this like a child, I'm not, I'm not even thinking about this as an adult. The next verse I read was from Luke where he says that 'we're expected to believe as children, not to question, but to believe as children.' To me that confirmed that it came from God, that this wasn't from within me, this was from outside of me. And that gave me a great peace". (Lucy, interview)

The above extract is an account of a rather dramatic conversion, which is not typical. People at Bethlehem have sometimes remarked that Lucy's change was very significant. It was not, however, a short-lived conversion experience; rather she has now been a Christian for over ten years. The perhaps extreme features of Lucy's account allow us to distinguish between 'individual authenticity' and 'self-transcendent authenticity'. 'Individual authenticity' refers to the attempt at being true to oneself (one's better self), in contrast with being true to universal truth by projecting oneself beyond one's immediate individuality and social context. The extract below shows Lucy's experience of Christianity, as a way to be true to her individual self, rather than a mystical experience. Lucy worked as a nurse for many years and has always had a desire to help others. Christianity provided her with validation.

"I think that my personality is a huge thing because, like I said, even as a child I liked to do things for other people. I've always had: 'she's very caring, she wants to help people'. And I did and maybe that was part of the change in me when I became a Christian that it was all kind of validated. It actually, you can live your life [...] with that personality, with that desire to help others without being weird. You know, it kind of gave it an expression, I suppose, in a way and [...] maybe that's why it happened so quickly. I mean I thought about it so much that maybe because that was me and my personality and then it fitted, it all came together, it felt right, it felt [...] I've always wanted to kind of be able to help others and now I understand maybe why; maybe that was always in me for a reason. [...] the way life and the world is constructed is not easy to really express that without people thinking: 'mhm, strange' or that [...] you're kind of doing for ulterior motives. And I think for me it was kind of 'yeah, this is fine, this is what the Christian life is all about' [...] you're not just out for yourself, but you're almost going to the other side and give yourself [...] in small ways for someone else, without expecting anything back" (Lucy, interview).

Lucy identifies the ethic of compassion as a key aspect of Christianity and what makes Christianity distinctive. Accordingly, within the Christian environment what may seem "weird" in the world outside becomes the norm. Christianity allows the expression of love and compassion, which, for Lucy, is the living out not only of faith, but of her own "personality". Thus, Christianity is liberating for Lucy and, in her opinion, for humanity as a

whole ('self-transcendent authenticity'), because human beings "want to be cared for and want to be nurtured and protected". Lucy does not see Christianity as the only framework that legitimises compassion, although compassion is characteristic of Christian authenticity. By stressing its opposition to a competitive environment outside, where caring acts can be mistaken for the pursuit of self-interest, Lucy demarcates authenticity from the 'falsehood' of leading a self-interested life. Authentic Christian life frees the person to go beyond the logic of competition and wealth accumulation. By being a Christian, Lucy can thus be "true to herself".

4. The 'Self-Transcendent' Dimension

The self-transcendent dimension hinges on the believer having a sense of awe for 'something bigger', the recognition of human finitude, of being 'this side' while having a sense of going beyond it (cf. chapter four). The move towards the 'other side', never achieved, is likened by participants to 'relinquishing control' (Lucy) and 'death and rebirth' (Arthur). Self-transcendence, within the religious framework, is grounded on the religious actor's trust in God and gives expression to the religious sentiment, described as *pietas* by Simmel. The sentiment of *pietas* was, for Simmel, a composite attitude containing "a peculiar mixture of unselfish surrender and fervent desire, of humility and exaltation, of sensory concreteness and spiritual abstraction" (Simmel 1912/1997, p. 161). Simmel's sentiment of *pietas* captures the believer's sense of dependence on God. The relinquishing

control of my informants is an expression of that sentiment. The self-transcendence in the examples provided, entails the surrender of the self.

Lucy: Self-surrender

"I've always wanted children. I loved being with my children, but there was always a part of me thinking: 'is this it? Is it gonna be like this for 10 years?' It's just give, give, give. And no matter how wonderful your children are, it's giving and giving and giving. It's never enough and that was perhaps the root to a lot of my frustration. And that was the biggest change in me. I thought: 'even being a mother to these children is an act of service. It's something I should do gladly [...] they are not actually my children. These are individual people that happened to have been born to me and my husband, but they are not my possessions, they are not my burden. My whole perception changed, even on what being a mother was. I was just the one who had been tasked, or trusted, to help them grow up" (Lucy, interview).

Lucy's "recognition" that her children are not her "possessions" and that she is "tasked" to bring them up inserts her and her task of child-rearing within the religious perspective. Lucy realises that she is not a "bad mother", she is "ok", for she goes beyond understanding her life as confined to the "here and now". Her role of mother is part of God's design. Her life is transformed by going beyond the finite perspective of this world or, in Simmelian terms, 'this side'. Self-transcendence, however, is not other-worldly; rather it 'elevates'

concrete reality, in this case child-rearing. The movement of self-transcendence, which inserts Lucy in a dimension that is beyond concrete reality, not only frees Lucy from feeling inadequate and burdened by her daily tasks, but also gives value to those tasks. Lucy's changed perspective, however, is not simply a form of personal development or self-fulfilment. She gives up control and recognises that she does not 'own' her life and, indeed, her children. Lucy's self-transcendence happens through self-surrender. This may be reminiscent of mystical theologies, such as that of Meister Eckhart, where the self is 'annulled' in God. However, there is not a mystical sublimation of the self, the self does not cease to exist in God, nor is there an abdication of one's will; rather individuality is enhanced by relinquishing the sense of control over one's life.

Therefore, self-transcendent authenticity is not just about being true to oneself, but about going beyond the self and being true to truth, however truth is understood. Religion inserts the person within the wider historical perspective of the religious tradition and also within the transcendent perspective of truth. Lucy comes to recognise Christianity as true and this changes her perception deeply. Being a Christian calls for a new identity or, at least, one that is aware of the spiritual dimension in everyday life. This is evident in the next section, where Arthur describes the self-transcendent dimension of authenticity as a metaphorical death and rebirth. He stresses that self-transcendence is part of one's relationship with God. He also explains that the fulfilment of one's potential is not to be seen as self-fulfilment, "like some self-help philosophy"; rather it is linked to something "eternal". It is the realisation of one's potential that is in accordance with what God wants people to be.

"... he sees, he sees my life, its warts, its failures, not failures, its frailties. But I also think God takes a lot of delight in seeing the potential in us, in everyone of us, I believe, occasionally being released. That, to me, is incredibly authentic and incredibly privileged position. I think all that I've got is God given. I make him, cos it's relational, I know that I make him sad for want of a better expression, like we do with our human relationships. [...] But there's nothing more fulfilling in my view than seeing people, you know, you love and care for or grow in a relationship with, you see them flowering in their potential [...] God sees us all like that, like little children, you know, when we start to release, sorry, fulfil potential that he's put within us, which I think it's unlimited as well, he smiles and he enjoys that; he enjoys seeing us becoming more and more the type of person that he intends us to be. [...] It's not like some self-help philosophy or whatever. To me, it's more genuine than that, it's more linked to something which is supernatural, eternal and yet it's real now, in the here and now. [...] I'll never forget the sermon by this old minister. He said unless you wanna enter the kingdom of heaven you need to be like the seed and the seed needs to die to re-grow. And there's a wonderful picture there of death and then out of the death comes new life. I thought this, this is just me. I don't think I'm bad, but if I want to really know God, to have this relationship with him, rather than just knowledge, I need to die to myself and I'll be reborn. And I think God's spirit then touched me for the first time. And it's been a process ever since. Some days

I really struggle, sometimes it's not so much of a struggle. I think I grow more and more aware all the time".

The becoming aware of the truth of God, which is beyond one's immediate experience, moves the believer to self-transcendence. Thus, the relationship is felt as causing a metaphorical death and rebirth. In this case, it is a process of moral development that Arthur defines as "present continuous" to emphasise its on-going nature. Authenticity is thus not an essence or a state to be attained, but a process that moulds one's existential identity. The person needs to go beyond the self continuously. Arthur's (*emic*) framework of authenticity is the theological notion of sanctification. Accordingly, the believer goes through a process of transformation when they come to recognise the truth of Christianity and to have a relationship with Jesus, which is lived out in human relationships. This theological transformation of the self of the believer rests on the idea that Jesus has saved believers with his sacrifice thus bridging the gap between human beings and God.

According to the local understanding of salvation, human beings are "broken" – as informants say – and not deserving of heaven on their own merit. They can never be "good enough" to be saved. This is based on a particular reading of the Old Testament, according to which humans are burdened by the original sin. Salvation by adhering to God's demands is unattainable; it can only come from Jesus' forgiveness of sins through the substitutionary atonement¹⁸. The believer is saved as a result of Jesus' death being a substitution for the punishment of believers. Belief in Jesus, however, not only brings eternal salvation but

.

¹⁸ There are different views of the 'substitutionary atonement' within Christianity. Like all other theological ideas mentioned here, the notion reported above reflects the local culture.

transformation of the person through the process of sanctification. Sanctification is the process of transformation of the person enacted by the Holy Spirit.

Sanctification is a process whereby the Holy Spirit grants believers the "fruits of the spirit", as Nicholas explained in his interview, such as love, joy, peace, patience, kindness, goodness, humility and self-control. These characteristics "grow" in the person who "submits" to God. Faced with inherent human brokenness, the Christian believer submits to Jesus to receive forgiveness and salvation. Surrender is a crucial aspect of self-transformation. Complete sanctification comes with eternal life. In contrast with theological sanctification, *etic* self-transcendence is the sense of going beyond one's finitude. Thus, self-transcendence is not limited to Christians or, indeed, to religious people. Self-transcendent authenticity is a process of identity formation, whereby actors have a feeling of awe and acknowledge 'something bigger'. This moves the person to go beyond the self to attain a connection with what is understood to be truth. Authenticity represents the aspiration of transcendence of the mundane; the glimpsing at a wider horizon of truth. It is not other-worldly; rather it makes transcendence "immanent" (cf. chapter four).

5. Discussion

The application of the concept of authenticity, as a three-dimensional process of identity formation, to ethnographic data and interviews reveals different aspects of individuals'

religious identity. Authenticity does not stop at the individual's inner search and emotional experiences; it is a more complex process that includes one's social as well as individual identity. The role of religious tradition is fundamental to the process. However, tradition should not be understood as adherence to norms and practices; rather it results from an active interpretation and legitimisation on the part of believers. Tradition is the framework through which actors glimpse the truth. It comprises customs, rituals, beliefs, laws and ethical principles that enable the religious person to have a relationship with the divine. Religious actors engage with tradition emotionally and intellectually at the individual and at the group level.

The commitment to authenticity is an attempt at translating truth by interpreting tradition. This work of interpretation reflects the social context of religious actors, their individual personality, and the individual's striving for a universal horizon. Accordingly, the interpretation of truth by religious actors through tradition reflects inevitably their social context, but it is also a deliberate attempt to make sense of tradition in the light of contemporary life. Tradition's norms, narratives and practices need to be "relevant" to people's lives. In turn, the process of interpreting and applying tradition in every generation guarantees its continuity in time, by making tradition relevant in every generation. In other words, the forms change to preserve what is seen as the substance of religion, which, in this case study, is the truth of Jesus.

The first section showed how contemporary social and ethical concerns regarding the competitive and materialistic aspects of Western society are used to formulate Christian

authenticity. The group identifies the authentic ethics of Christianity, in cooperative human relations, by discarding the false ethics of materialism of mainstream society. The references to obedience point to the sovereignty of the ethics of Jesus over and above society's ethics. Jesus is seen by the group as love and compassion, which come to identify Christian authenticity. Consequently, the ethos of Bethlehem church is to be a caring and welcoming church (Appendix B), open to all, and accepting of people. The church seeks to practise love and compassion by providing a sense of community for church members as well as for local residents. The compassionate Christianity of Bethlehem is focussed on being caring and compassionate, which is contrasted with – what is felt to be – the selfish individualism of Western society.

Christianity is "countercultural" because it reflects divine ethics, rather than human ethics. It is a call to be like Jesus in the world. This compassionate Christianity is also the basis on which Lucy constructs her authentic self. Here, tradition is interpreted in the light of the social context of the church, but also of one's individuality. For Lucy, her personal attitude of compassion towards others finds expression in being a Christian. Thus, at the individual level, the search for authenticity involves being true to truth that speaks to oneself and validates one's personality. The individual dimension of authenticity provides unity to the self and self-understanding within the religious framework of tradition. It is subjective, for the truth of Christianity needs to be experienced by the person emotionally and intellectually and ring true to the individual. It is also objective in as far as it is an interpretation of a shared tradition.

An interesting aspect of the process of authenticity is the relinquishing of control over one's life that has the paradoxical effect of intensifying one's individuality. The surrender to God is not a feature only of self-transcendence, as mystical accounts may convey. On the contrary, it is present at the social level, where participants reflect on the recognition of the sovereignty of God and that everything is a gift from God. Surrender is present at the individual level, when Lucy embraces the Christian framework that allows her to be herself. At the self-transcendent level, Arthur shows how authenticity requires "death and rebirth" to truly appreciate one's relationship with God; while, for Lucy, self-transcendence enables her to inscribe her task of bringing up her children in a different perspective.

The self-transcendent dimension of authenticity lets us glimpse at the human sensation of awe, the recognition of 'something bigger' and true. Self-transcendence thus means gaining a 'higher', or 'deeper', degree of consciousness of truth – however understood by the actor. The move of self-transcendence gives the believer a wider perspective. In the present case, self-transcendence refers to the believer's relationship with Jesus, which calls for being like him. This is impossible to achieve in its fullness; it thus becomes a continuous attempt to re-awaken one's consciousness. Authenticity is a fleeting moment, for human beings cannot be constantly aware of truth and living according to truth. They cannot be self-transcendent in every moment. Authenticity is a constant effort on the part of the believer to connect with and understand truth. Believers connect and relate with the divine by being in relationship with it. It is a deeply personal endeavour which needs to "ring true" to the person, as put by Selina. It also needs to stir the person to go beyond her usual mind-set

and way of acting. The sense of self-transcendence allows the believer to experience the divine in the here and now.

Thus, the experience of self-transcendence is a very immanent one. Yet, authenticity is not confined to emotional and mystical experiences of the numinous; rather the acknowledgement of truth beyond immediate experience entails a doing. Authenticity is always sought and performed, rather than attained. It is a process of identity formation, which requires commitment in everyday life. The self is not sublimated in the overarching truth, or God. It is not annulled, but reinforced, as a "follower of Jesus", who commits to serve God. Authenticity is not an exercise in self-authority or individual choice; it requires surrender of the self, which is an acknowledgement of one's finitude, to then commit to God. It is a constant effort to be "servant of God". The person thus forms her authentic identity as a Christian, through a process of interpretation of tradition, socially and individually, and through the emotional and intellectual experience of being in relationship with Jesus, grounded in awe: the awareness of 'something bigger' and true.

5. Conclusions

The individual makes sense of truth (Jesus) through a process of identity formation, which involves interpretation of tradition. Tradition is 'social' and socially constructed, thus developing over time. However, the overarching narrative of any tradition is to serve as the

way to truth, which, in the present case study, is Jesus. The interpretation of truth by religious actors needs to be consonant with the social context, and thus relevant to today's life. It also needs to connect with the emotions and intellect of the individual. It also needs to point to a universal horizon of truth that can enable the person to transcend herself and her immediate social reality.

Self-transcendence is based on the feeling of awe for 'something bigger', the sensation of humans being 'this side' only, and the ability to think of something 'beyond'. Lucy and Arthur show that self-transcendence is experienced as a surrender of the self. However, the relinquishing of control over one's life is to 'regain it', thus the surrender effects an intensification of individuality. This may have the effect of self-fulfilment, but that is not its rationale. Self-fulfilment is but an outcome of acting according to one's 'new' consciousness. Self-transcendent authenticity is a change in consciousness, which is articulated in a different way of thinking and acting. Accordingly, Lucy sees child-rearing as a task, part of God's design. The surrender of the self has the effect of intensifying one's individuality, but it also connects the individual to a universal horizon of truth, which in this case is understood through the figure of Jesus.

My informants understand the truth of Christianity as, at the same time, universal and particular. The tension between universality and particularism, already seen in the previous chapter, is at the core of the conception of Christian authenticity. As mentioned, Jesus embodies the overarching ethics of compassion. My informants recognise that compassion is present in people of other religions and in people who are not religious; yet

Christianity offers a privileged path to truth, due to the sanctification process of the Holy Spirit that transforms the person. It contains a universally valid truth, which can be known by non-Christians, but which requires a specifically Christian framework for its fuller appreciation. As mentioned in the previous chapter, there is diversity among participants in formulating and dealing with this question. I thus suggest that Christian authenticity is best understood as a privileged framework for the knowledge and experience of truth.

Chapter Ten

Authentic Ethics

1. Introduction

In the previous chapter, I examined authenticity in its social, individual and self-transcendent dimensions. It has been argued that authenticity is not a state or a quality, but a process of identity construction. Such identity is relational, although its relationality is not limited to social relations, but also to one's individuality and one's understanding of truth (through one's interpretation of tradition). In this thesis, I proposed to understand religiosity (*Religiosität*) as sensitivity to self-transcendence. The self-transcendent dimension refers to the awareness of 'this side', to which I connect the sense of awe (*pietas*), the sense of 'something bigger' of the religious sentiment (cf. chapter four). Self-transcendence is not 'neutral' or confined to one's inner feelings; it is inscribed within one's religious culture, which, in this case study, means "following Jesus". This chapter focuses on the ethical dimension of authenticity, as evidenced by Arthur's reflections.

Arthur's extract, in chapter nine, reveals the process of authenticity as a process of moral development. 'Following Jesus' means making a conscious decision of having a relationship with Jesus, which is lived out in practice. Self-transcendence calls for conscious action in accordance with the ideal of Jesus. The ensuing action is called serving, which is explored in this chapter. Serving is constructed as authentic action by individual religious actors and the group. The empirical data show how actors engage with religious tradition in forming their own identity. Actors do not accept religious norms, narratives and practices supinely. Indeed, the process of sacralisation (cf. chapter seven) helps us appreciate how tradition rests on individual actors granting religious significance to norms, narratives and practices and thus legitimising them.

Authenticity thus rests on sacralisation, the process whereby individuals engage with tradition attributing value to it and legitimising it. Accordingly, for something to be authentic, it needs to be recognised as a legitimate part of tradition, but also enabling authenticity in the present time. Sacralisation does not necessarily imply that sacralised ideas and practices are ethical; rather they are a valuable and legitimate part of tradition. However, as the data show, (*emic*) authenticity is a process of moral development. Authenticity, in this case study, is grounded in ethics, and, in particular, in the ethic of compassion. In contrast with the ethic of purity, which is norm-oriented, the ethic of compassion is person-oriented. The final section thus explores the distinctiveness of Christian authenticity through the idea of compassion.

2. The Ethical Concerns of Authenticity

In the course of three years of observation, I rarely encountered a discussion on ethics. Even when probed, during interviews, participants did not spell out the ethical rationale for action nor the underlying ethical principles. We may argue that this suggests that ethics is 'assumed', in the sense that it is not felt necessary to articulate the ethical rationale for one's behaviour. This might seem redundant if we conflate religion with ethics, but, as argued in chapter seven, not all religious norms and behaviours are ethical. Rituals, norms, symbols are not necessarily of ethical value nor do they have an ethical rationale. A ritual may have an experiential value and purpose. It may be of uttermost importance in invoking the divine, such as prayer, without having ethical value. Religious experiences may focus the person's mind to follow ethical precepts, but are not ethical in and of themselves. They can be considered ethically neutral.

Emic authenticity has ethical connotations whilst *etic* authenticity is not determined by ethical content, although it may include it, and regards ethics as not necessarily shared by the wider community. Accordingly, in the previous chapter, Arthur's description of authenticity captures a process of moral development. The experience of the divine does not exhaust authenticity; being authentic thus requires commitment to a moral lifestyle. The most explicit references to ethics are in the examples below. The first is from the interview with Felix, when he recounts his effort of reviving the soup run for the homeless and Godwin's explanation of what Christianity is for him.

"They just couldn't run it anymore, you know, we were literally gonna say 'well, the people aren't here, we won't do it anymore', but it was, this was a slightly different case where, as a pastor, I thought: 'as a church, we should be involved with caring for poor people and we're not doing much in that area and here is one of the few bits we're doing, so let's try and keep that alive'. So, I went down myself. I shared it with the church, suddenly there were 14 people who came forward and said: 'look, we really wanna help'. So it's a fantastic, thriving ministry now, but it, it almost died, I mean ..." (Felix, interview).

"... for me Christianity is about reaching out to people, it's about *feeding the hungry*, it's about *clothing the poor*, it's about whatever it means to make life better for people, whatever it means. Ultimately, it would be great then if those people buy into Christianity, but if they don't, I believe, that we've given a little bit of Jesus Christ to them. We made their lives better. That's how I live my Christian life". (Godwin, interview).

When mentioned explicitly, action was rarely within the frame of a moral dilemma over which course of action to take. This is not to say that research participants did not face moral dilemmas or did not have doubts on how to act, but that doubts were often framed as a personal sense of inadequacy or failure when confronted by a difficult situation. They reflected a personal struggle. Arthur, referring, in his interview, to the gap between what he says and what he does, called it "part of the wrestle for me about Christianity". It is not,

however, merely a question of sincerity; rather it captures the performative nature of authenticity. Authenticity is a way of thinking, but also a way of doing. It is a 'doing', which implies intentionality: the consciousness of one's actions being a reflection of one's relationship with God. Thus, authentic action is intentional, rather than casual. Being an authentic Christian requires the awareness of the spiritual significance of one's actions. Accordingly, serving is not "good works" to merit heaven, but the practice of being a follower of Jesus.

Home Group discussion, 24th June 2009

Celia: "as a Christian I have absolutely no doubt about my salvation. But it doesn't mean there is no challenge daily in how we act. Am I out there using those gifts he has given me? I want him to find me using those gifts, being a mother, what he would want me to do, becoming the person he wants me to be".

Camden: "make sure your actions show Jesus".

Serving is often described as "putting others' needs before one's own". Although what constitutes serving is not clear cut, it is inherently moral. Above all, serving is the attempt to practise authenticity. As such, it is marked by a specific frame of mind, according to which the action is to serve God. Below are a few explanations and definitions of serving from participants' interviews.

"What is serving? I think *it's preferring someone else's needs to your own* and not necessarily thinking of yourself as better than people, [...] always thinking about others as better than yourself. I don't think that we've got that sort of 'summed up', I don't think Christians do it by any sense, but I think that's the end deal [...] what you can do to help them (others) in whatever situation. And I see it as doing it for God when I do it for someone else". (Dorothea, interview).

"What's the difference between serving and caring? I think [...] I'm the centre manager here and under a normal business model [...] I would have people under me who would do the rolling their sleeves up [...], but that's not the model that we hold up to here. [...] If I walk into the kitchen and the kitchen staff are busy, I roll my sleeves up and worked the dishwasher. [...] I think it's about putting yourself in somebody else's shoes, it's about what it takes for me to make sure the team operates well, that we work well together. It can be throwing bodies at tasks, but it can also mean, I've gotta do some stuff that I probably I don't wanna do: take the rubbish out, do the dishwasher, you know. Serving is about making everybody else's burden lighter, making their work easier. [...] that's what Christianity is all about". (Godwin, interview).

"The whole serving thing to me is [...] a kind of an *attitude* really: *what you do,* you do to please God. And sometimes the things you do to please God are not necessarily the things you would have wanted to do or chosen to do, but you know that it's the right thing and [...] there's a kind of a real joy in that. [...] I really do

not wanna do this or you had to do it at the expense of something that you would rather do, but there's a kind of a sense deep within that 'yeah, this is the right thing to do, this is what God wanted me to do'. So I think, yes, serving is definitely, it's like an act of devotion, without sounding too holy, that is, it's something I want to please God. God's done so much in my life that I want to use my life to, to do good things in his name. I don't want people to say 'oh Lucy is such a good person'. I want to say 'God's amazing, look what he's enabled her to do, look how he uses her'. Whereas sometimes giving is [...] we do for our own gratification. Perhaps we need that in a certain way, to be built up, but I think the service is really about where I stand before God and where he's kind of pushing me and nudging me to do". (Lucy, interview).

(referring to the parent and toddler 'ministry') "How is it a ministry? [...] I love the little ones, I do, don't get me wrong, I love them very much and, but it's not my sole reason in doing it, [...] but my heart is for the families in the community, [...] it's such a time in life when you can feel very isolated, [...] my heart is that they might come to know God, they might come to know of the saviour that loves them and, [...] they might experience God's love for them, as well as meeting them on a human level [...] providing somewhere where they can come and feel comfortable and at ease and get to know some other people". (Celia, interview).

The above extracts all explain serving as putting someone else's needs before one's own.

The action may include something one does not want to do. It is an effort which becomes

effortless because, as Lucy says, it is done to serve God. God enables the person to be selfless. Serving is doing something for others, but avoiding doing something for one's own gratification. Thus, the motivation is fundamental to the act. Authenticity, as mentioned, presupposes intentionality: the path of authenticity requires the believer to choose and commit to follow Jesus. Thus, for it to be serving, action needs to be carried out with religious intention, with an awareness of its spiritual significance. It is not merely directed to the recipient, but it serves God. Spiritually, the action contributes to building the Kingdom of God. Serving, in turn, raises one's awareness of the spiritual significance of all of one's actions.

The participants' reflections on serving also point to the particular form serving takes: that of loving others. Therefore, it is not a mere act of devotion to God, as prayer might be; rather it requires a recipient. It is within a relational framework. The way serving is constructed in Bethlehem is characterised by a strong focus on caring for people. Serving is "the essence of Christianity", as expressed by Selina in her interview, and, most importantly, it is when Christianity is "genuine" (Selina) or, we may add, authentic. Serving is "doing things to make people better" (Lucy, interview), but it is in service to God. Thus, one's sentiment of devotion towards God is externalised and projected in one's relationships with others. The authentic Christian aims to see others as God sees them and for others to "see something of Jesus in us", as put by Celia; so that "they might experience God's love for them". Relationships are not ends in themselves, but a way in which Christians try to follow the example of Jesus by loving others.

3. The Sacralisation of Authenticity

Serving is the practice of authenticity in Christian life. Authenticity, as mentioned, rests on sacralisation, the recognition of a practice or a belief as authentic part of tradition. Thus, sacralisation imposes boundaries on authenticity. For a practice or belief to be considered authentic, they need to be seen as a legitimate part of the tradition, but also to be relevant to the believer's life. Thus, individual religious actors play a role in shaping and constructing their tradition and granting legitimacy to specific practices and beliefs. The example below illustrates the fuzzy boundaries of authenticity. The exchange is taken from a group discussion on an episode from a DVD series on how to be a Christian, called *Living Distinctively*, which focussed on environmental issues. There was general agreement that care for the environment was important as were issues of fair trade, inequality in the world, and resource management. However, there was disagreement as to its level of priority and the theological framework in which to understand environmental concerns. The DVD featured a woman living in a house with some land for growing vegetables and rearing pigs.

Home group discussion, 15th June 2011

Camden asks: "What do you honestly think about Ruth?" (the environmentalist in the DVD)

Elinor: "She wasn't that radical, apart from the pigs"

Arthur: "this is one of the subjects where we have stereotypical images. [...]

They get a bad press"

Harriet: "the media often portray environmental campaigns as done aggressively"

Camden: "it must be hard for the kids growing up thinking that the pigs reared in the allotment will get killed"

Winifred: "I don't think so, I grew up like that and I have no problem eating them"

Tertius: "the stress is on buying local, but this has a huge impact on *the third* world which relies on us buying non-locally"

Camden: (reading from the DVD booklet) "How does attitude to environment reflect theology?"

Rosamond: "I've never thought about it. It's stressful enough when you're going shopping to look at where it comes from"

Nicholas: "I'm not convinced by her view of theology"

Will: "it would have been helpful if they had quoted scripture"

(Winifred reads some verses from the Bible)

Camden: "it's about *stewardship*. We can't be wasteful, we can't be careless. *It's* ultimately about being ethical to our fellow-man"

Arthur: "for our society it is as cheap as you can" (consumer goods)

Harriet: "... and people can't afford it as it is now"

Nicholas: "food prices are going to go up with the economy of India and China growing. We might not be able to care about the environment if feeding the family becomes difficult. I think the bigger issue is resource management [...] energy..."

Harriet: "it's got to be an individual's decision. As a society we cannot possibly do it, because there's such a difference between those who can afford it and those who can't"

(The discussion turns to 'fair-trade')

Mary: "I don't want to be cynical, but it's fair-trade to a point. They (farmers) are not getting the right amount. ..."

Arthur: "there is huge inequality in the world. I wonder what he (God) thinks about that"

Celia: "what she's (Ruth) saying is that that *it comes from our heart*, to reflect how we understand God"

Harriet: "I agree with you. *Different things touch you. I prefer giving to the poor* than spending hours checking where things come from"

Camden: "what are the more important commitments?"

(Winifred reads from the booklet a statement on the environment being part of being a Christian)

Camden: "we focus too much on food, what about travelling to work?"

(Camden proposes an exercise of spending 10 minutes thinking of what people can do to make their life choices more sustainable with regards to food, travel, and energy consumption)

Camden: "If Jesus were around today, would be go around the supermarket looking at where things come from? I'm not convinced that the answer is yes"

"Yes, he would", replied, his wife, Winifred.

Tertius is not convinced: "This is all very well but 'we're making a mess of everything else. What about people who have nothing in Africa?"

Winifred clearly saw environmental issues as a legitimate part of being a Christian. The environment, for Winifred, should be part of the sacred framework. However, during the discussion, this seemed not to be shared by the majority of the group. Although the group agreed that respect for the environment was important and everybody should adopt a more sustainable lifestyle, care for the environment was not sacralised. I suggest that there are three main reasons for this. In the first instance, there is uncertainty, among group members, as to what should be deemed a sustainable lifestyle. The group lacked clear guidelines and criteria upon which to judge the best policy. For example, the focus on buying local produce is questioned in the light of the importance of international trade for the sustenance of the 'third world'. There were also comments on the price for sustainable consumption, such as organic products, which are not affordable for all.

Arthur highlighted how "for our society it is as cheap as you can", thus environmental concerns clash with consumerism. However, in the eyes of participants, the price of an

environmentally-friendly lifestyle seems to confine issues of sustainability to the few rather than the many. In the second instance, environmental concerns are not seen as part of the theological discourse, as shown by Nicholas' doubts regarding the theological grounding of environmentalism. That evening there were 13 people present, so the turn-out was good and the discussion was representative of the group in the case study. However, subsequent conversations revealed a wider support for Winifred's position. The lack of time to discuss the issue prevented the group from seeing environmental concerns within Christian tradition. The environment is becoming increasingly part of evangelical concerns (Pally 2011; Smith and Johnson 2010; Harden 2005); nevertheless the group did not have access to environmental theologies and time to reflect on them.

In the third instance, serving is primarily conceived as giving to other human beings. The 'poor' are seen much more as a priority, especially if in the 'third world', as the comments by Harriet and Tertius show. Indeed, the charity action in which Bethlehem church is involved relates mostly to missions in Africa. The long-standing concern for the poor and the tradition of missionary work inscribe action aimed at relieving of poverty within the sacred framework. In particular, Africa occupies a prominent place in Christian evangelical imagery, which is explored in the next section.

4. Africa and Authenticity

Africa was often mentioned during the discussion, reported in the previous section, over the importance of environmental concerns for Christians. It has a special place in the hearts of my informants that is unsurpassed by anything else. Several of them have spent prolonged periods of time (from one month to a year) in missions in Africa engaging in very practical and, often, heavy labour to aid African villages. Although it is acknowledged that there is need for charity in many other places, including the UK, Africa is invested with religious significance. The extracts below offer a glimpse on various aspects of authenticity. They all come from Arthur's interview, where he recounts the first time he went to Africa with the organisation Samaritans' First. He was met by a young woman working for the organisation, who explained the challenge they faced and stressed the frustration that might arise from a 'Western' attitude of 'fixing things' when faced with the inability to solve problems.

"She said: 'probably, most of what you're used to doing it in terms of your society is fixing things.' [...] That's what you're paid to do, that's what I'm paid to do in work. [...] She said: 'You're gonna see things you've never seen ever before. Some of those things are quite disturbing, actually they are all pretty disturbing in the great scheme of things.' She said: 'I suspect most of you are going to revert back to yourself, your normal type, *you're gonna fix this.*' She said: 'you will not be able to fix this, what you see, cos the scale of the challenge is so vast that it

might just knock you completely. It can shake your faith, it could, could psychologically really cause a bit of discomfort.' Then she opened the bible and she told the story about Jesus and the 5000 people, they didn't have any food. [...] You've got five loaves of bread, two fishes, loads of people. The automatic reaction is 'ok, we can't do any of that. It's a waste of time, but they brought them to Jesus and, how it happened I don't know I don't even really wanna go down the road of trying to explain it, but the principle here is *he took something* that seemed a total waste of time and multiplied it to such an extent that there were baskets and baskets, leftovers. And the principle was sometimes all we can do is bring to, bring to God our limited offering that's pathetic and stupid, seemingly hopeless as that might be and he has, because he's so different, he's God, on a different plane to our thinking, to our mentality, he was able to do all sorts of things. That was unbelievable for me. That was major shaping of my mind-set really, my heart, attitude and also my understanding of the greatness of God. Also probably my own, man generally, willingness to trust him, because we could do an awful lot more by trusting and by giving what we do have [...] But my contribution, what God's called me to do, I think he's called me to make a contribution in whatever way that is, bring my little bag of seemingly inconsequential offerings that I lay before the king of kings and the lord of all creation. And I don't actually know what he's gonna do with it." (Arthur, interview)

There are two important points I wish to highlight on this passage. Firstly, the culture of being in control, typical of the West, as exemplified by the idea of 'fixing it', is considered meaningless before utter deprivation. To be able to work face to face with extreme poverty, one is required to relinquish control and accept that one's action is but a drop in the ocean. Secondly, Arthur presents it as a turning point in his knowledge of God. He felt in awe of God, as he came to terms with his own finitude. It is a moment of personal spiritual growth, when Arthur understands that all he can do is to bring his "inconsequential offerings" without knowing what will come of them. Thus, faith is, here, trust in God; rather than belief that God exists or that God intervenes to help the needy. God may 'multiply' what one brings, but belief in God is an attitude born out of awe. This sentiment calls for transcendence of one's self-authority to act in accordance with the ideal of Jesus. The transcendence of the self inscribes the person within the religious framework giving rise to what is felt to be an authentic identity. In this case study, the local narratives of authenticity construct a moral identity. As Arthur puts it, authenticity happens when "men and women being made in God's image [...] reflect something of God's characteristics" (interview).

"... To be moulded and shaped for good, I know these things Francesca are very subjective, but I think the changes in my character from what would be, if I wasn't a Christian, to what is work in progress, now are just massive. I'm so glad that even if it must be frustrating to God at times cos I don't, I'm not changing as quickly as ... It's not change, I'm not being refined as, maybe, as speedily as he would like. But I just know they're good, they're good and I feel fulfilled. [...] I just know there's a load of impurity left within me that God is constantly blasting

with a big [...] kind of hot flame in order to [...] He's looking to create in all of us something that's precious. *It's not free from pain, it's not free from stress* and whatever, but it's precious nonetheless. I know someone would argue that's a contradiction, but to me that's part of life. This progress going through."

Authenticity is, here, presented as living out one's relationship with God by fulfilling one's spiritual and moral potential. The relationship with the divine sets in motion the process of moral development, which is a work in progress of being "refined" rather than changed. The use of the word 'refine' refers to a song from a verse in the bible (Malachi 3, 2¹⁹). From a theological point of view, the 'refinement' is the effect of 'sanctification' born out of the relationship with God. The context of Africa brings to light the inner process of authenticity, which is not merely acting in accordance to ethical principles, but becoming aware of the spiritual dimension of one's life. Africa is the 'Other', the opposite of comfortable life in the UK. It is bare and raw. Stripped of Western materialism, it is *the* place of faith, where "things" and "society" do not obstruct the relationship between believers and God. It provides a source of intense religious experience. Serving in Africa is not merely an act of charity, but a way to 'find oneself'.

Thus, Africa has a transformative power. However, the inner transformation of the believer is not for one's personal sake, but to acknowledge the sovereignty of God and to be in a relationship with God. Africa offers the opportunity to 'surrender' to God. The overwhelming poverty of African villagers, but also their faith has a shattering impact on

.

¹⁹ "But who may abide the day of his coming? And who shall stand when he appeareth? For he is like a refiner's fire, and like fullers' soap".

the consciences of those who have been there. It is a powerful experience of humility that transforms the person. The human struggle for survival is a reminder of the inconsequentiality of humanity. Human beings have no real control over their own destiny. Yet, once again, it is in acknowledging one's powerlessness that one's individual autonomy is intensified. It is intensified by one's awareness of the truth of God.

It has been argued that authenticity is a process of identity construction, which rests on the interpretation of tradition by religious actors. It requires awareness of what the actors understand as truth, however it is not exhausted by experience of the numinous. On the contrary, authenticity is consciousness of one's religious identity and ensuing action. Authenticity calls for action that reflects one's awareness of truth. *Emic* authenticity is thus a process of moral development. Informants consider the ethical attitude underpinning authenticity distinctive of Christianity and, yet, of universal value. Although the ethics present at Bethlehem church are nuanced, *emic* authenticity is primarily articulated as an attitude of compassion. Compassion is the principle underpinning serving. Being a Christian, as explored in chapter eight, is lived out in human relationships where the person 'embraces' the other (cf. chapter seven), who experiences God's love for them. The next section thus examines more closely the 'ethic of compassion'.

5. The 'Ethic of Compassion'

This section explores the type of ethics that guides authenticity. It is, of course, difficult to assess ethical decision-making in the private lives of individuals. It would have been too intrusive to follow participants, at work or at home, to gather data on their moral behaviour. It would have also created an artificial situation, where participants would have felt self-conscious. However, instances of caring, especially among participants were clearly prominent. For instance, Celia told me of a time when she was ill and members of the home group brought food for her and her family for a week to the amazement of their neighbours. Similarly, during the fieldwork I recorded that the home group organised a food rota to help out Elinor after the birth of her child. After a fire that destroyed the interior of Lucy's house, people from the church hosted Lucy and her family and continued to provide food and support for weeks, as Lucy recounts in the first extract. Caring is underpinned by – what I term – an ethic of compassion that inspires recognition of the dignity of the person. This is evident in Celia's remarks commenting on her ministry in the parent and toddler group, in the second extract.

"We ate somewhere different every night. People fed us, people took my washing, people bought books for the children. It was just incredible. There was a sort of sense that, you know, you're family. Coming around you when you most needed them. And then what was really nice was when we moved to the rented house, there was a little flurry of activity and people brought bits and pieces and

things to make it home. And then people kind of left us alone to settle down, which I really appreciated. [...] the genuineness of our family here, of how people truly care and show that they care, cos it's one thing, isn't it? to say: "Gosh, I'm so sorry to hear what's happened", but actually to turn up on your doorstep, to phone you up and say: "come and eat with us". To me, it's really putting it into action. And we've received so much of that, it's been incredible. So, yeah, that's been [...] spiritually a very significant event in our lives. I think as a family as well [...] for the children as well. It's been a real lesson for them in, you know, in caring and how we should care for one another" (Lucy, interview).

"my heart is that they might come to know God, they might come to know of the saviour that loves them and [...] through coming to the church, that they might experience God's love for them as well as meeting them on a human level, providing somewhere where they can come and feel comfortable and at ease and get to know some other people, some other in the same position as them. [...] being Christians among them, by welcoming them, by loving them, by listening, by getting to know them, by hearing the needs that they might have and maybe do something about that, maybe, really making them feel welcome and hopefully having them to see something of Jesus in us, [...] that's our prayer" (Celia, interview).

In Celia's extract, we glimpse at the core of authenticity as relational, ethical and self-transcendent. Celia, at the parent and toddler group, meets the parents at a "human level", helping and supporting them in their role of parents, but it is also through that relationship that Celia and the rest of the team live out their faith. Their love for the parents seeks to mirror divine love. The relationship is the vehicle for experiencing the love of God, where the receivers of love are both the parents and Celia. Indeed, the parents may feel the love of God in the love they receive from Celia and the team, whilst Celia transcends herself and experiences the love of God that, in turn, transforms her attitude towards the parents. This is evidenced by her comments after the interview, reported below. The relationship with lesus has, for Celia, transformed how she experiences the parents she has met at the group.

Celia mentioned that one day she was walking down the street when she saw one of the mums from the parent and toddler group. She felt compassion for that mother and remembered a verse where Jesus feels compassion for people. She felt the same compassion. She cared and she was made to feel that way by God. It isn't just her. That is not what those activities at the church are about. That's not what serving is about. It is not individuals, but God calling you to do it. God calls you to serve and God enables you to feel that care and compassion for people. (Field notes)

This is echoed in Camden's comments.

"It's a matter of being so filled with something that you have to do it. So filled with compassion for homeless people, for instance, that you have to spend time with them and you have to try and help them. It would hurt you not to. That's the kind of, I think, experience that God wants for you [...] God loves each and anyone of us equally. We are all children to him. He loves everyone of us in a deeper way than I love [name of daughter]. I'm never gonna understand that. I don't want to understand that, I just wanna be in awe of that. I wanna say: 'that is amazing'. And just simply to cry and appreciate it and see people through God's eyes. That's something Liberia did for me. I saw Winifred [Camden's wife] through God's eyes when I came back and it brought me to tears. It was just nuts. I have a deep love for Winifred, but I had a new respect for her as my wife, cos she wasn't just my wife or my partner or the biggest love in my life. I didn't have [name of daughter] back then. She wasn't just the biggest love in my life; she was so much more 'cos she was God's child" (Camden, interview).

The feeling of compassion and the ability to love are the result of the process of authenticity. From the perspective of my informants, compassion is the result of 'sanctification' (cf. chapter nine). The believer submits to God and thus draws from God the ability to love or act righteously. God fills the believer with compassion and enables the believer to feel compassion. Camden's experience of authenticity, in Africa, deepens his love for his wife. Camden's love for his wife is thus an instance of self-transcendence. After his experience in Africa, Camden loves differently; he loves with compassion. Compassion

is not solely empathy with the suffering of another, but empathy with shared humanity, as expressed by Camden in the following extract.

"When you see that in the unsavoury in society no one wants to speak to, when you see them as a child of God, it's heartbreaking for you as an individual. [...] When you meet someone for the first time, in a few seconds you call these assumptions about them without knowing anything about them. The way they walk, the way they talk, whether they have eye contact with you, what you know about them in their job, anything like that. God doesn't see like that. It's really, really compelling when you try to see someone through God's eyes, if you do manage to do that. [...] it's just nuts" (Camden, interview).

The attitude of compassion reflects the attempt of Christians to be like Jesus and, accordingly, see others like Jesus sees them. The ethic of compassion underpins the practices of informants and the running of the church. As mentioned in chapter eight, the vision of Bethlehem is to be accepting of all. The ethic of compassion impacts on how church activities are conducted and how evangelism is understood. Compassion means seeing "the whole person", their physical and emotional needs; rather than recipients of evangelism, as expressed by Godwin and Selina.

"God has created us to have a physical body, emotions, psychology, the spiritual and I don't see that God actually separates any of that. I think he's interested in the *whole person*. [...] you wanna give the message, but the reality is, what their

needs are in that moment in time are much more crucial [...] You've got somebody and try and share what you believe in, but they are on their knees with hunger, it's not gonna do any, it's not gonna do any good. The whole thing is evangelism, but we endeavour to get to the message of it at some point. Jesus, when he walked the earth, he went out and he, he healed people, he didn't always necessarily saved them for want of a better phrase. *Sometimes just healed people because that's what they needed and then he carried on. So, that's the model: give people what they need*, be willing to share what you believe, but literally it is about giving them what they need".

"... Somebody asked what was evangelism and the definition I give is that it's just being Jesus to people. Ehm, and that's the reason why we do it, is to show love to people, isn't it? And it's to be genuine, because Christianity is not being inward-looking and just thinking of yourself and that's unfortunately how society is turning. [...] So that is all self-serving, isn't it? Because I'm just serving me and that's not what Christianity is, Christianity is about, you know, is not putting yourself first, it's putting our people first. [...] it's about loving people" (Selina, interview).

6. Discussion

The previous sections have provided reflections of participants over action that can be considered authentic. Self-transcendent authenticity captures the actor's awareness of truth. The consciousness of truth calls for conscious action in accordance with that truth. In this case study, participants' self-transcendence refers to the awareness of God and ensuing relationship. Accordingly, self-transcendent authenticity is having a relationship with God, which results in moral development. This is a continuous process requiring continuous effort. One's relationship with God happens through relationships with others and through serving. Serving is moral action. However, its most important aspect, from the participants' point of view, is the attitude of the doer. To be serving, the person needs to be conscious of the action being in service to God.

Actions might be moral, but do not necessarily constitute serving. The actor's frame of mind is essential. Being a Christian is a conscious choice, from which commitment ensues (cf. chapter eight). Christian authenticity rests on one's consciousness of being in a relationship with God and of practising a distinctive way of life. The authenticity of being a Christian lies not in performing moral actions, but in the intent to serve God. The distinctiveness of Christianity lies in the consciousness of the person to follow Jesus by expressing compassion in human relations. One's consciousness of following Jesus is essential for action to constitute serving; however, the boundaries of serving are sometimes disputed. Thus, the case of environmental protection was an example of the

absence of shared norms over what constitutes Christianity. Protection of the environment was considered ethical by all participants, but not sacralised by all. The sacralisation, or attribution of religious significance, to practices and beliefs, legitimises them as part of tradition. It thus defines the boundaries of authenticity.

Individual actors sacralise tradition. Therefore, the individual's personal and social background (cf. chapter six) informs the process of sacralisation. What the believer considers ethical or unethical reflects their personal background, the identity of the group and their interpretation of tradition. The interaction of one's social and personal identity (the 'social I' and the 'individual I') and tradition, as understood individually and communally, defines serving. The difference of opinions over the sacred status of environmental concerns reveals the social dynamics that construct religious tradition. Younger members tended to be more supportive of seeing environmental protection as serving. The topic was not discussed in depth, due to lack of time, and the discussion did not benefit from the relevant literature in eco-theology. So it is difficult to see at this point whether environmental concerns may be inscribed within tradition.

In contrast, Africa represents the experience of authenticity *par excellence*. It shows the transformative effect of serving. One's relationship with Jesus, and thus awareness of the truth, deepens through serving. Africa allows an intense form of self-transcendence, for it is physically the 'other side'. In Christian imaginary, Africa is truer because it lacks the artificiality of Western life. Once again, the anti-materialist motif resurfaces. Western materialism, selfish individualism, and comfort are lived as a challenge to being authentic.

This is understood by participants as a weakness on their part. Africa strips away the mundane from their consciousness, 'denudes' them to be in contact with their true self of being human. Thus, the feeling of compassion is not just 'suffering with', but seeing the "whole person", as Godwin described it, the same way as Jesus did.

Compassion, here, means seeing the other person as human. It entails going beyond the socio-economic status and situation in the life of a person. Just as self-transcendent authenticity is an experience, no matter how fleeting, of going beyond the social and personal construct of the self, the feeling of compassion allows the person to see the human beyond the social construct. This is the experience of informants. It should not be taken normatively, but descriptively. The ethic of compassion seems to have an effect not only on relationships with others, such as Celia's feelings for the parents of the parent and toddler group, but also on personal relationships, as Camden recounts his changed feelings for his wife, following his experience in Africa.

Compassion is expressed in relationships, which are central to Bethlehem's identity. Anecdotal evidence suggests that people who have visited Bethlehem centre consider it welcoming and friendly. The practice of relationships reflects the ethic of compassion embodied by Jesus. The ethical ideal of being like Jesus is articulated as "seeing others as God sees them". Compassion is not limited to feeling empathy for the suffering of others, but it inspires recognition of the dignity of others, regardless of their situation in life. The relational approach of compassion is in contrast with the abstract ethical principles of the ethic of purity. This is exemplified by Godwin's comments, which identify being like Jesus

as helping others by addressing their physical and emotional needs. Communicating the "message" of Christianity, evangelism, is thus secondary to being like Jesus to others.

As I have argued in chapter eight, Bethlehem exemplifies a move from doctrine to authenticity, which mirrors the move from an 'ethic of purity' towards an 'ethic of compassion'. This 'move' results from the construction of Christianity in opposition to wider society, seen as lonely and competitive, and in opposition to 'traditional religion', seen as norm-oriented. Participants also interpret the Old Testament as based on norms in contrast with the New Testament, which is seen as based on love. The authenticity of Christianity lies in being like Jesus: being compassionate in human relations, as Jesus would. This dichotomy between 'ethic of compassion' and 'ethic of purity' is a heuristic device to capture the antithesis expressed by participants in constructing the distinctiveness of Christianity. It also highlights the effect of a different ethical approach on boundary-making. Accordingly, Bethlehem's 'ethic of compassion' blurs the boundaries between Christian and non-Christian.

The compassion for humanity as a whole has the effect of stirring believers to be open to the stranger, to identify with one another's humanity. Authentic ethics is thus an attitude towards others. This should not be understood as a feeling; rather the emotion of compassion is fruit of a conscious decision. It results from the process of identity construction of authenticity that entails self-surrender, and the sacralisation of norms and practices, including the practice of relationships. The 'ethic of compassion' may contribute to destabilise ethical norms, such as those surrounding sexuality, and doctrine (cf. chapter

eight). It is worth reiterating that the 'ethic of compassion' should not be understood normatively as superior to that of purity or more progressive. The ethic of purity encapsulates judgement that is essential to concerns over justice. From an *etic* perspective, compassion and purity identify the different approaches in the boundary-making of a group as well as their ethical norms.

7. Conclusions

The chapter delved into the ethical aspects of authentic practices. The example of Jesus is moral, thus being a Christian implies living morally. However, serving is not merely moral, nor is it merely what is deemed normative within one's tradition. Serving is determined by an attitude and thus requires intentionality: action needs to be consciously performed to serve God. Intentionality stresses the choice made by the believer to follow Jesus. It is not mere choice, but consciousness of the religious/spiritual significance of one's actions. Being authentic calls for commitment to practise a distinctive way of life, which is in accordance with the image of Jesus informants hold. Serving, in turn, raises one's awareness of one's relationship with the divine.

Serving is moral, however not all moral concerns are serving. Thus, sacralisation defines the boundaries of what is considered Christian authenticity. The group discussion on environmental concerns showed that the protection of the environment is not framed by all

within a sacred perspective. It is deemed moral, but not sacred. This emphasises the necessity of avoiding conflation of ethics and religion. Authenticity is thus not confined to ethics or, even, truth. The recognition of truth in other traditions, as shown in chapter eight, reveals self-transcendent authenticity as tapping into something that is seen as universally human whilst, at the same time, stressing the privileged position of Christianity in relation to God. In other words, while people belonging to other religions or none might experience self-transcendence and "know something of truth", Christianity offers the "truer" way, a privileged way to truth.

This understanding of Christianity as a privileged path to truth, or God, is an example of the 'ethic of purity', which is based on norms. The emphasis on compassion in the local narrative and practice of relationships seems to suggest a tension between the privileged truth of Christianity and the recognition of others, implied by the 'ethic of compassion'. The understanding of compassion, presented in this chapter, rests on a shared humanity, shared ethics, and shared truth. Thus, the authenticity of compassion lies in the consciousness of one's behaviour being part of one's relationship with Jesus. Being authentic means being a true follower of Jesus, which entails a subjective process of conscious understanding of one's relation to Jesus and moral development. Authentic Christianity is thus distinctive in its intentionality, or consciousness, of practising the way of Jesus.

Chapter Eleven

Conclusions

1. A Contribution

The aim of this thesis has been to provide a better understanding of individual religious identity for contemporary times that went beyond the sociological construct of religion as a category identifying propositional belief, belonging to and attendance at religious establishments, and adherence to a set of cultural and ethical claims. It has been argued that such categorisation reifies the multifaceted phenomenon of religion and reflects an anachronistic Protestant model of religion, based on transcendence, which paradoxically fails to account for contemporary forms of Protestantism in a pluralistic society. Thus, chapter two has critiqued the notion of the 'disenchantment' of modernity for disregarding historical instances of rationalisation, differentiation, and individualisation and thus constructing a uniform pre-modern 'enchanted' world. The sociology of Georg Simmel provides a more nuanced understanding of the modern individualisation process and the

opportunity to explore how modern heightened subjectivity has shaped contemporary religiosity.

The object of the research was, therefore, to construct concepts that would capture how individual religiosity is expressed today, taking as a case study a group of committed Christian evangelicals. The focus on individual religiosity and the choice of ethnographic methods has enabled an appreciation of the individuals' aspirations and agency in constructing their identity. The dominant theories that have paid attention to the role of the individual – rational choice theory and spirituality studies – have assumed different conceptions of the individual. The former reduces the individual to self-interest, whilst the latter relies on a dichotomy between religion, as oriented towards the transcendent and experienced through official institutions, and spirituality, as oriented towards the immanent inner experience and self-expression. Both accounts neglect the role of commitment in religious life.

The conceptualisation of the individual in rational choice theory and spirituality studies emphasises individual autonomy, which is detached from cultural identity and group dynamics. The schema of the 'I and Thou', in chapter six, addresses this issue by proposing a relational account of the subject. The resulting relational self serves much better the research participants' own conception of the individual as interdependent. Thus, the thesis has made a theoretical contribution by providing a relational understanding of the subject; reconceptualising Simmel's religiosity (*Religiosität*) and belief (*belief in*) to provide a more

nuanced account of the narratives and practices of religious actors; and by formulating a concept of authenticity, which is not exhausted by ethical concerns.

The narratives and practices, emerging from the empirical research, revealed the construction of Christianity through the prism of authenticity, as a distinctive and 'truer' way of living. The theme of authenticity echoes transformations in the refashioning of church identity in the US, Australia and New Zealand. Viewed at a purely organisational level, Bethlehem church might seem an exercise in making Christianity relevant to contemporary lives by changing the physical environment of the church and providing services to the local community. However, this reading misunderstands the intentions of its members and leadership as well as their level of engagement with the wider culture. In the midst of pluralistic 'difference', this group of evangelicals has embraced the narrative of authenticity, which combines particularity and universality.

Pluralism offers Christian evangelicals the platform to articulate the particularistic tradition of Christianity, whilst retaining a claim to – what they understand to be – universal truth. Authenticity refers to a way of being a Christian, but also to Christian culture in a multicultural society. Christian evangelicals claim to follow a "truer" way of being, which is also distinctive by being in opposition to – what is understood as – mainstream culture. The *emic* notion of authenticity thus encapsulates a moral critique of society, but also religion as a practice sustained by a distinctive mind-set. The empirical chapters have shown that *emic* authenticity shares – what Simmel called – the 'unselfish surrender' of religiosity (Simmel 1912/1997, p. 161); yet it is not self-negation or unio

mystica, but affirmation of one's individuality. However, the individual's choice to lead the religious life engenders commitment. The person takes upon herself to live authentically by being true to what she considers truth.

The thesis thus developed the concept of authenticity to better reflect the actors' understanding of religious life. Authenticity, as proposed in this thesis, is a process of identity construction rather than a normative ideal. This enables a processual understanding of the individual and of identity-making. The authentic self is based on Simmel's underdeveloped notion of self-transcendence, which identifies the person's pull beyond her social and personal dimension, a sense of a wider horizon of 'truth'. Truth, here, is what is understood as such by actors within their particularistic cultural tradition. Truth may be felt as universal, however, it is always comprehended, experienced, and articulated through the framework of one's tradition.

This perspective stresses the agency of the actor in interpreting tradition to make sense of their search for self-transcendence. By identifying Simmel's notion of religiosity (*Religiosität*) with a sensitivity to self-transcendence, I linked the authentic self, and the concept of authenticity, with living the 'religious life'. Religiosity is thus not belief or belonging to a religious tradition or, even, religious manifestation, as argued by Simmel. Following Simmel, self-transcendence is not limited to religious life, but is present in other endeavours. The religious (self-transcendent) attitude thus includes spiritual practices, but it may also refer to political or artistic endeavours. It is, on one hand, a broad notion that applies across different fields, and yet, on the other hand, it captures a specific way of

being, which is present in different degrees depending on the person. Authenticity is thus a process of identity formation through the search for truth. Truth, within the theoretical framework, is social, reflecting one's social background and relationships; individual, striking a chord with the person; and self-transcendent, tapping into the universal.

It has been shown that, although expressed mostly in moral behaviour, *emic* authenticity does not identify an ethical ideal, but an attitude born out of one's relationship with Jesus. The boundaries of authenticity are thus not simply ethical, but are in relation to one's tradition. The individual religious actor participates in the formation of tradition by sacralising narratives and practices. The theoretical concept of sacralisation identifies the process of identification of the person with her religious tradition, interpretation and attribution of value to it. This process is essential in constituting the sacred, as the example of environmental concerns, in chapter ten, has demonstrated. Sacralisation also illustrates how the adherence to religious tradition stems from the person understanding herself as religious and leading a religious life. The intensity of the process of 'sacralisation' reflects one's religiosity, which, as conceived by Simmel, is present in people in varying degrees. Therefore, not all in one's life is seen within the religious framework and not all that is seen within the framework of faith is sacralised in the same way by people of the same group.

Through sacralisation one sees oneself within a religious framework and, as a consequence, invests with sacred value and legitimacy one's tradition and lived religion. The boundaries of authenticity are, therefore, constantly constructed by individuals and the group. Nevertheless, the empirical research has shown that the distinctive attitude of being an

authentic Christian is characterised by intentional commitment and compassion. The 'ethic of compassion' is the overarching ethic, but also the narrative that constructs the identity of Bethlehem church, as a "real" and "caring" church, and of individual Christians. The authentic identity of Bethlehem Christians is expressed primarily as the ethic and practice of compassion. In turn, the narrative and practice of compassion are articulated dialectically in contraposition to – what is seen as – the materialistic and lonely world outside, and the stern and judgemental "traditional" church. Christians at Bethlehem do not claim that informality and community-focussed church-doing represent 'true' Christianity, but a better way to communicate Christianity in the contemporary pluralistic world.

Compassion is interpreted as 'truer' and, thus, identifies a better way to walk on the path of authenticity. Authenticity is thus expressed through compassion. The practice of compassion has engendered an open attitude towards others in the local community thus blurring boundaries. The relationship with the 'outside' is, hence, two-fold: of opposition and criticism towards contemporary profit-making and selfish culture, but also of openness towards people looking for a community. This has been translated in activities that are of service to the local community without asking for any religious commitment or attendance, with the intent to provide a 'safe haven' from society. It has also posed a challenge in terms of the understanding of exclusive salvation, which for some has meant rethinking salvation altogether. The 'ethic of compassion' is pervasive of the local practices. It is from the narratives and practices of Christians, individually and communally, that the sacred emerges. The sacred is thus not the 'holy' (Otto 1917/1924), or what is considered ethical

in a given society at a given time (Lynch 2012); rather it constitutes the boundaries of authenticity.

The thesis thus makes a theoretical contribution by interpreting Simmelian sociology and philosophy and developing a theoretical framework for the understanding of individual religious identity in contemporary pluralistic society. Focussing on individual religiosity, it proposes its re-conceptualisation in the light of the 'search for authenticity', which emerged from the fieldwork. The thesis formulates the concept of authenticity, with its boundary-making process (sacralisation), for the study of contemporary evangelical Christianity. It thus integrates conceptual reflection and construction with ethnographic research bridging the gap between theory and empirical scholarship. Authenticity provides a frame for the understanding of how Christian religiosity is lived out in today's diverse society.

The concepts of compassion and purity begin to challenge the adversarial framing of ethical debates, which leads too often to an incommensurable divide between progressive and conservative positions. This fails to capture the internal logic and motivations of religious groups, and thus perpetuates antagonism rather than dialogue. A more nuanced portrayal of religious identity and practices, may help shift the focus onto possible paths of mutual understanding and cooperation. Drawing on Simmel's reflections on honour, the thesis has linked ethical practices and ideas to boundary-making and thus offered a broader perspective on the social dimension of ethics. Further research is needed to develop a robust framework for the understanding of ethical reasoning and practices within religious and non-religious communities. In particular, research should investigate what type of

ethical norms dominates in religious groups, how religious people draw ethical boundaries and identity boundaries, and how they negotiate different ethical ideals in practice. This issue is significant for policy-makers as well as academic scholars, given the current political polarisation in the public sphere. It has also the potential to illumine group dynamics in terms of cohesion and community engagement.

2. Becoming a Sociologist

When I began the study, during my M.Sc. in Social Science Research Methods, my focus was on learning how and why religious groups spend human and material resources to provide services for the community, to make a positive contribution to society, and, possibly, to change society for the better. I remember walking into Huw's office, at the end of my master's fieldwork, and saying "they are really caring". That's it. That was my 'finding': my research participants aimed to be caring and they were. I thought the Ph.D. would expand my knowledge of what made them care for each other and for people in the local community by investigating the role of leadership, the vision of the organisation, the internal structures, and narratives. 'Narratives' is what led me astray. They could not be labelled 'motivation', 'legitimation' of the internal order and philosophy of the group, or 'identity marker', alongside the other reifying categories of gender, class, and ethnicity much loved by sociologists. The sociological category of religion itself did not seem to fit and my informants did not want to identify with it.

I listened to participants, which I could have done better had I had a theory already in place and specific research questions; and without having ditched one theoretical concept after another. I thus began my descent into the Simmelian maelstrom without any theoretical anchor that might have kept me from drowning. The first draft of the 'I and Thou', right from the abyss of Simmel's Wechselwirkung, capsized Huw's boat as well, who swam quickly ashore, away from it. My only life-support were a few articles by Simmelian scholar Fréderick Vandenberghe. I once resolved to contact world-renown expert of Simmel, Professor David Frisby, for advice, but he died a week later. I put off contacting Dr Vandenberghe, just in case. So I lost sight of the scope of the research and how it should have been done. I carried on attending the home group and interviewed a few informants, following specific events in the field. As the 'I and Thou' and authenticity were taking shape, I interviewed Arthur and Camden. I wish I could have interviewed more people; I wish I could have interviewed them for longer; and I wish I could have asked them about compassion. Yet, the best data have come from informants opening up and revealing themselves and their lives as part of a conversation, rather than in response to specific questions.

They talked and with their words they made sense of life and gave meaning to it. The truth of which they spoke started to sound like the 'true self' of the prominent witch Starhawk²⁰ (1979). The traditional picture of Christian evangelicals as biblical literalists, staunch believers and utopian builders gave way to the complexity of people getting to grips with the sense of 'something bigger' than mundane chores or selfish pursuits. The community

²⁰ I am grateful to Dr Sara Delamont for introducing me to Starhawk's writings.

they sought to build reminded me of the ideal often talked about in Jewish circles. Therefore, I gave a workshop on community and faith at LimmudFest, a festival-conference of Jewish culture. I asked participants what they did not like of faith communities and what they should be like. I wrote the answers on a board and, then, I presented my findings on Bethlehem church by circling the words on the board. "They are Jews", someone exclaimed.

I might have turned Christian evangelicals into Jews because my eyes and ears saw and heard the human, rather than the numinous, but they sounded 'Jewish'. The belief in the inherent relationality of human beings articulated through community building rather than saving souls is, after all, overwhelmingly present in Jewish narratives. Indeed they are better at being Jewish. At the beginning of my Ph.D., while dining in the Rabbi's *sukkah*, I mentioned I studied a Christian evangelical church. A woman said to me: 'I know a good one: Bethlehem church'. I said nothing. She asked: 'why can't we be like them?'. Bethlehem Christians have made a conscious choice of building – what they call – a spiritual home. It is a home that requires commitment, but seeks to give acceptance and compassion. It is far from perfect and they are always questioning whether they are doing the right thing, or, as put by a church member during a discussion on vision: "people, do we like them?"

I began with an interest in the human and dispensed with God rather quickly. I could thus be faulted for not 'listening to the Spirit'; yet, for my research participants, rather than in a mystical quest, faith is lived in relationships, in everyday acts of kindness, in feelings of compassion, and in the struggle to overcome the temptation to be selfish, indifferent, and judgemental. I let them speak as much as possible through a perhaps intricate theoretical

framework. The theoretical framework helped me to discern the different aspects of the search for authenticity: the continuous work to be like Jesus. It separated the social from the personal, but also the normative. This distance helped me to see the process of sociation and how that impacts on the formation of ethics.

The concept of authenticity allowed me to capture the interplay between choice and commitment. Having choice implies a conscious decision, which requires commitment. Amidst diversity, choosing an identity means drawing boundaries. Bethlehem Christians have no evident rituals, look, or customs that would differentiate them from their neighbours. Thus, authenticity is an internal characteristic, an attitude, which is expressed in compassionate practices. My informants recognise that others engage in similar action, thus the framework of tradition makes Christian compassion distinctive and provides an identity to the individual. Yet, the theoretical framework distinguishing between the 'ethic of purity' and 'ethic of compassion' sought to capture the tension with which religious and non-religious people struggle in their daily lives.

The distinction derives from an attempt to inscribe the normative within sociological knowledge. Purity and compassion are, therefore, broad and nuanced concepts, because they need to contain boundary-making and boundary-breaking, relationality and individuality, ethical aspirations and concrete situations. Gaining a sociological perspective helped me see the video on pro-life campaigners in a different light, as the inherent tension between moral judgement and love of others, and how these emotions and ideals shape group and individual identities. In turn, a sociological perspective helped me appreciate

that ethics is a practice to which many factors contribute, such as habit, character, upbringing, social pressures, and beliefs. The sociological mind-set I have developed was useful in seeing the 'social' in life; yet, did that turn me into a heathen sociologist?

3. Pilgrimages

At night, sitting next to strangers, I waited to embark on a flight to Israel. It was 2012 and I had never been to Israel before. I kept on postponing, partly because all my holidays were spent at home in Italy seeing my family, after I left them 16 years ago. I hate flying and kept it to the absolutely necessary London – Bologna journey. Why on earth was I going to the Middle East? It was a pilgrimage, of sorts. I decided to see some friends and took the opportunity to present at a conference on spirituality. I slept in hotels and in the safety rooms in my friends' house. I stayed within the greenline and beyond it, with religious and with secular Israelis. I wandered in the streets of the old city of Jerusalem with a long skirt and scarf over my head. It was March and it was freezing. I noticed people reacting differently depending on what I was wearing. I disliked the 'inauthenticity' of the old city, with its fake show for tourists, and the tourists, with their photographs of walls, people and spices. I took copious sociological notes and wondered whether I had already become a heathen sociologist.

I had swapped my room with a perfect view of the golden dome of Al Aqsa for one with better wifi. I refused to queue to get a glimpse of the dome, I toured grudgingly the internal part of the Kotel, and uttered a perfunctory prayer outside. A friend told me that had I stayed at friends, away from the old city, had Shabbat together, it would have felt very different. As I passed the long queue of tourists waiting to enter Al Aqsa, I asked myself for what I would queue. "The Sistine chapel", came the answer straightaway. Notwithstanding the many documentaries and photographs, that ceiling was like an arrow going right through. I realised that my 'religiosity' needed an artistic expression, that only in Michelangelo's paintbrush could I glimpse at the transcendent. What captures my imagination is the artist's attempt at transforming everyday objects and images into symbols of the infinite, thus making me more sensitive to the unfolding relationship between transcendent and immanent.

Being Kantian, however, means aesthetic appreciation serves but to bring comfort to the daily struggle of fulfilling one's ethical duty in life. After all, this project sprang from the dissatisfaction at political and ethical debates, where conservative narratives seem to hold the monopoly of morality, but also at the alleged authenticity and legitimacy of 'orthodox' and conservative forms of religion. The focus on individuals of this research was, partly, an attempt at examining how they shape religion to include their understanding of authenticity, and how they legitimise change with themselves and others. What could not be part of the thesis is a discussion of power, which would have required a theory of power and, possibly, a wider investigation of the relationships between different religious institutions and actors and their production of legitimising discourses. I wanted to understand the role of individuals and their ability to 'own' the sacred. That led me into the discovery of how human beings create meaning and value in their lives. What I have

learned above everything else is that, besides our ethical or religious normative callings, there is a richness in social life that is worth exploring. Sociology is a window on the beauty of the human.

Appendix A – Statement of Faith

1. God

There is one God, who exists eternally in three Persons: the Father, the Son, and the Holy Spirit. God is perfect and unchanging in His character. He is the Almighty Creator, Saviour and Judge who controls everything for His own glory. God is love, and has demonstrated that love supremely in sending His Son into the world.

2. The Son of God, the Lord Jesus Christ

The Lord Jesus Christ is fully God and fully human. He was conceived by the Holy Spirit, born of a virgin, and lived a sinless life in obedience to God the Father. He taught with authority and all His words are true. He died on a cross in our place, bearing God's punishment for our sin, redeeming us by His blood. He rose from the dead and, in His resurrection body, went into heaven, where He now intercedes for us and is exalted as Lord of all.

3. The Holy Spirit

The Holy Spirit is fully God. He brings glory to the Lord Jesus Christ, convinces people of their wrongdoing, and by living in Christians He empowers them to become more like the Lord Jesus.

4. The Bible

God has revealed Himself in the Bible, which consists of the Old and New Testaments alone. Every word of the Bible, as originally given, has been spoken by God. It is, therefore, without error, fully reliable, and the supreme and final authority in faith and life. It shows us the way to God.

5. The Human Race

Everyone is made in the image of God, and therefore has equal dignity and worth. Our greatest purpose is to obey, worship and love God. However, as a result of disobedience by the parents of the human race, every aspect of human nature has been corrupted. Everyone is therefore without spiritual life, a sinner by nature and by choice, and faces the Judgement of God who is holy.

6. Salvation

Salvation cannot be earned or deserved: it is the free gift of God. He made it available through the death of the Lord Jesus Christ. We must turn from our sin and trust in Jesus Christ in order to receive the benefits of salvation. These include forgiveness, being credited with the righteousness of Christ, a restored relationship with God, the gift of the Holy Spirit, adoption into God's family, and eternal life.

7. Church

The church is the company of people who belong to the Lord Jesus Christ. It exists to reveal the glory of God. The church draws its life and direction from the Lord Jesus Christ. It is therefore an organism and not just an organization. Churches form locally as believers meet together to worship God, to encourage one another in their spiritual lives, and to serve their community. Obedience to the Word of God requires us to preach, evangelise, baptise, celebrate communion, and pray for all. All members of the church are equal and should use their varied gifts to work together in love and unity.

8. The Future

The Lord Jesus will come a second time. The dead will be raised. Believers in the Lord Jesus will enter into everlasting joy in fellowship with God. All unbelievers, whether living or dead, will be judged and suffer eternal punishment. God will make all things new and be glorified forever.

Appendix B – Vision of the Church

Mission Statement

Attracting the world to Jesus Christ through our everyday witness, discipling one another and those we invite to Him.

Vision Statement

Building an alternative community of Jesus' disciples, working together to make our witness accessible and visible to all.

Value 1 - Our God

Our God is Father, Son and Spirit loving one another, and we are His people who delight in Him. He is our Father, because we are united with His Son, Jesus, and empowered with His Holy Spirit. We trust Him to lead and guide us in everyday life, through His word by the Spirit.

Value 2 - Worship

Our God is awesome. There is none like Him. He has reached out to us when we deserved nothing and lavished upon us His very best. Our response is joyful worship and careful obedience to such a wonderful God. We gather together to worship him in small and large groups with a variety of styles, also recognising that our whole lives are worship to our glorious God.

Value 3 - Being Community

Our God is a community of diverse persons who has drawn us together into community by and for the love of Jesus. We are from a mix of backgrounds with diverse gifts, characters, past histories and families, yet we are one family, one body with Christ as our head, loving one another and letting the peace of Christ rule in our hearts.

Value 4 – Engaging Community

We are a community which does its best to present Jesus to the watching world, inviting others to taste and see not only the God who walks among us, but also to experience for themselves the outworking of Jesus' life and love in our lives.

Value 5 - Welcome

We are a people who aim to meet everyone where they are, being accepting of the situation in which they find themselves as they come to be part of the church. While God loves people just the way they are, He loves them too much to leave them that way, and our hope is for transformation for all into the likeness of and obedience to Christ.

Value 6 - Work in Progress

We are a people who acknowledge that we are not the finished article. We will remain humble, realising that in both success and defeat God is working through us for our good and always questioning how and why we are doing what we are doing.

Value 7 – Discipling

We are a people who are always building one another up in the knowledge and love of Jesus Christ. We know that trusting Jesus affects our obedience to Him in every area of life, and we will encourage and rebuke each other in love so that we might grow more into His likeness.

Value 8 - Prayer

We are a people who rejoice in fellowship with our God and love to talk to Him and listen to Him as an expression of our Sonship. We will cultivate this in big and small gatherings and will endeavour to make it a normal part of our Christian friendships and when we are on our own.

Value 9 - Creativity

We are a creative church expressing God's creativity among us and giving freedom to the creative gifts He has given us, stirring up interest through attractive ways of inviting people to Jesus. As creative people, our church structures will only ever serve our need but never govern us, refusing to let tradition set in.

Value 10 - The World

We are a community who recognise the needs of the world beyond ourselves. We will listen to, pray for and engage with God's mission in the whole world.

Appendix C – Membership

To become a member of Bethlehem Church, we ask that the following statement be confirmed:

In becoming a member of Bethlehem Church, I do so expressing my belief in the Almighty and Eternal God, Father, Son and Holy Spirit.

I affirm my faith in Jesus Christ as my Lord and my Saviour. I accept the Word of God as my rule in all matters of faith and conduct.

With God's help, I will seek to be a loving, faithful member of Bethlehem Church, making it a priority to read the Bible, pray, give of my talents and money, and be present at the Church meetings for teaching, fellowship, worship and prayer.

Bibliography

- ADAMCZYK, A. & PITT, C. 2009. Shaping attitudes about homosexuality: the role of religion and cultural context. *Social Science Research*, 38, pp. 338-351.
- AGAR, M. 1980. *The professional stranger: an informal introduction to ethnography.* New York, N.Y.: Academic Press.
- AHO, J. 1996. Popular Christianity and political extremism in the United States. *In:* SMITH, C. (ed.) *Disruptive religion. The force of faith in social-movement activism.* New York, N.Y.: Routledge, pp. 189-204.
- ALLES, G. D. 2009. Religious economies and rational choice: on Rodney Stark and Roger Finke, *Acts of Faith* (2000). *In:* STAUSBERG, M. (ed.) *Contemporary theories of religion. A critical companion.* London: Routledge, pp. 83-98.
- ALLPORT, G. W. & ROSS, J. M. 1967. Personal religious orientation and prejudice. *Journal of Personality and Social Psychology*, 5(4), pp. 423-443.
- AMMERMAN, N. T. 1997. Presidential Address 1996: Organized religion in a voluntaristic society. *Sociology of Religion*, 58(3), pp. 203-215.
- AMMERMAN, N. T. 2001. Still gathering after all these years: congregations in U.S. cities. *In:* WALSH, A. (ed.) *Can charitable choice work?: covering religion's impact on urban affairs and social services.* Hartford, CT: Trinity College and Leonard E. Greenberg Center for the Study of Religion in Public Life, pp. 6-22.
- AMMERMAN, N. T. 2002. Connecting mainline protestant churches *In:* WUTHNOW, R. & EVANS, J. H. (eds.) *The quiet hand of God. Faith-based activism and the public role of mainline Protestantism.* Berkeley, CA: University of California Press, pp. 129-158.
- AMMERMAN, N. T. 2005. *Pillars of faith. American congregations and their partners.*Berkeley, CA: University of California Press.
- AMMERMAN, N. T. (ed.) 2007. *Everyday religion: observing modern religious lives.* New York, N.Y.: Oxford University Press.

- AMMERMAN, N. T. 2009. Building religious communities, building the common good: a skeptical appreciation. *In:* LICHTERMAN, P. & POTTS, C. B. (eds.) *The civic life of American religion.* Stanford, CA: Stanford University Press, pp. 48-68.
- AMMERMAN, N. T. 2009. Congregations: local, social, and religious. *In:* CLARKE, P. B. (ed.) *The Oxford handbook of the sociology of religion.* Oxford: Oxford University Press, pp. 562-580.
- AMMERMAN, N. T. 2010. The challenges of pluralism: locating religion in a world of diversity. *Social Compass*, 57(2), pp. 154-167.
- AMMERMAN, N. T. 2013. Spiritual but not religious? Beyond binary choices in the study of religion. *Journal for the Scientific Study of Religion*, 52(2), pp. 258-278.
- ANDO, C. 2008. *The matter of the gods. Religion and the Roman Empire.* Berkeley, CA: University of California Press.
- ANDOLFI, F. 1996. Introduzione. *In:* SIMMEL, G. & ANDOLFI, F. (eds.) *I problemi fondamentali della filosofia.* Bari: Biblioteca Universale Laterza, pp. i-xxv.
- ANDOLFI, F. 2008. L'umanità degli individui eccellenti. Simmel interpreta Nietzsche. *In:* SIMMEL, G. & ANDOLFI, F. (ed.) Reggio Emilia: Diabasis, La Ginestra. pp. 7-28.
- ARDITI, J. 1996. Simmel's theory of alienation and the decline of the nonrational. *Sociological Theory*, 14(2), pp. 93-108.
- ARWECK, E. 2002. The insider/outsider problem in the study of new religious movements.

 In: ARWECK, E. & STRINGER, M. D. (eds.) Theorizing faith. The insider/outsider

 problem in the study of ritual. Birmingham: University of Birmingham Press, pp. 115132.
- ARWECK, E. & COLLINS, P. 2006. Reading religion in text and context: reflections of faith and practice in religious materials. Aldershot: Ashgate.
- ARWECK, E. & STRINGER, M. D. (eds.) 2002. *Theorizing faith. The insider/outsider problem in the study of ritual.* Birmingham: University of Birmingham Press.
- ASAD, T. 1993. *Genealogies of religion: discipline and reasons of power in Christianity and Islam.* Baltimore, MD: Johns Hopkins University Press.
- ATKINSON, P. 1992. *Understanding ethnographic texts.* Newbury Park, CA: Sage.
- ATKINSON, P. 2006. Rescuing autoethnography. *Journal of Contemporary Ethnography*, 35(4), pp. 400-404.

- ATKINSON, P., COFFEY, A. & DELAMONT, S. (eds.) 2001. *Handbook of ethnography.* London: Sage.
- ATKINSON, P., COFFEY, A. & DELAMONT, S. 2003. *Key themes in qualitative research:* continuities and changes. Walnut Creek, CA: AltaMira Press.
- ATKINSON, P. & SILVERMAN, D. 1997. Kundera's immortality. The interview society and the invention of the self. *Qualitative Inquiry*, 3(3), pp. 304-325.
- AUNE, K., SHARMA, S. & VINCETT, G. 2008. Women, religion and secularization: one size does not fit all. *In:* AUNE, K., SHARMA, S. & VINCETT, G. (eds.) *Women and religion in the west: challenging secularization.* Aldershot: Ashgate, pp. 1-22.
- AUNE, K., SHARMA, S. & VINCETT, G. (eds.) 2008. *Women and religion in the west:* challenging secularization. Aldershot: Ashgate.
- BACKHAUS, G. 2003. Husserlian affinities in Simmel's later philosophy of history. *Human Studies*, 26, pp. 223-258.
- BAERT, P. 2005. *Philosophy of the social sciences: towards pragmatism.* Cambridge: Polity Press.
- BAERT, P. & CARREIRA DA SILVA, F. 2010. *Social theory in the twentieth century and beyond.* Cambridge: Polity Press.
- BAERT, P. & TURNER, B. S. (eds.) 2007. *Pragmatism and European social theory.* Oxford: The Bardwell Press.
- BAKER, C. 2007. The hybrid church in the city. Aldershot: Ashgate.
- BAKER, C. 2008. Renewal in the centre of the city and the churches' contribution. *In:*BALLARD, P. H. (ed.) *The church at the centre of the city.* Peterborough: Epworth, pp. 29-44.
- BAKER, C. 2009. Blurred encounters? Religious literacy, spiritual capital and language. *In:* DINHAM, A., FURBEY, R. & LOWNDES, V. (eds.) *Faith in the public realm. Controversies, policies and practices.* Bristol: Policy Press, pp. 106-122.
- BAKER, C. & MILES-WATSON, J. 2010. Faith and traditional capitals: defining the public scope of spiritual and religious capital A literature Review *Implicit Religion*, 13(1), pp. 17-69.
- BAKER, C. & SKINNER, H. 2006. Faith in action: the dynamic connection between spiritual and religious capital. *Regenerating communities: a theological and strategic critique.*

- The William Temple Foundation's Research Project. [Online]. Available: http://www.williamtemplefoundation.org.uk/documents/faith-in-action.pdf
- BALMER, R. 1989. *Mine eyes have seen the glory: a journey into the evangelical subculture in America.* New York, N.Y.: Oxford University Press.
- BARBALET, J. 2007. Classical pragmatism, classical sociology: William James, religion and emotions. *In:* BAERT, P. & TURNER, B. S. (eds.) *Pragmatism and European social theory.* Oxford: Bardwell Press, pp. 17-43.
- BARKER, E. 2008. The church without and the God within: religiosity and/or spirituality? *In:* BARKER, E. (ed.) *The centrality of religion in social life: essays in honour of James A. Beckford.* Aldershot: Ashgate, pp. 187-202.
- BAR-TAL, D. 2000. *Shared beliefs in a society. Social psychological analysis.* Thousand Oaks, CA: Sage.
- BARTHEL-BOUCHIER, D. 2001. Authenticity and identity. Theme-parking the Amanas. *International Sociology*, 16(2), pp. 221-239.
- BARTKOWSKI, J. P. 1996. Beyond biblical literalism and inerrancy: conservative Protestants and the hermeneutic interpretation of scripture. *Sociology of Religion*, 57(3), pp. 259-272.
- BARTKOWSKI, J. P. & REGIS, H. A. 2003. Religious civility, civil society, and charitable choice: faith-based poverty relief in the post-welfare era. *In:* MCCONKEY, D. & LAWLER, P. A. (eds.) *Faith, morality, and civil society.* Lanham, MD: Lexington, pp. 127-144.
- BATSON, C. D., AHMAD, N. & TSANG, J.-A. 2002. Four motives for community involvement. *Journal of Social Issues*, 58, pp. 429-445.
- BAUMAN, Z. 2000. *Liquid modernity*. Cambridge: Polity Press.
- BEAUMONT, J. 2008. Introduction: faith-based organisations and urban social issues. *Urban Studies*, 45(10), pp. 2011-2017.
- BEAUMONT, J. & BAKER, C. (eds.) *Postsecular cities. Space, theory and practice.* London: Continuum
- BEBBINGTON, D. W. 1995. The decline and resurgence of evangelical social concern 1918 1980. *In:* WOLFFE, J. (ed.) *Evangelical faith and public zeal: evangelicals and society in Britain, 1780-1980.* London: SPCK, pp. 175-197.

- BECK, U. & BECK-GERNSHEIM, E. 2009. Individualization and 'precarious freedoms': perspectives and controversies of a subject-orientated sociology. *In:* ELLIOTT, A. & DU GAY,P. (eds.) Identity in question, London: Sage, pp. 13-36.
- BECKER, G. 1976. *The economic approach to human behavior.* Chicago, IL: University of Chicago Press.
- BECKER, H. S. 1967. Whose side are we on? *Social Problems*, 14(3), pp. 239-247.
- BECKER, H. S. 1996. The epistemology of qualitative research. *In:* JESSOR, R., COLBY, A. & SHWEDER, R. A. (eds.) *Ethnography and human development: context and meaning in social inquiry.* Chicago, IL: University of Chicago Press, pp. 53-71.
- BECKFORD, J. A. 1983. The restoration of "power" to the sociology of religion. *Sociological Analysis*, 44(1), pp. 11-32.
- BECKFORD, J. A. 1989. Religion and advanced industrial society. London: Unwin Hyman.
- BECKFORD, J. A. 1990. The sociology of religion and social problems. *Sociological Analysis*, 51(1), pp. 1-14.
- BECKFORD, J. A. 1996. Postmodernity, high modernity and new modernity: three concepts in search of religion. *In:* FLANAGAN, K. & JUPP, P. C. (eds.) *Postmodernity, sociology and religion.* Basingstoke: Macmillan, pp. 30-47.
- BECKFORD, J. A. 1997. The disenchantment of postmodernity. *New Blackfriars*, 78(913), pp. 121-128.
- BECKFORD, J. A. 2000. "Start together and finish together": shifts in the premises and paradigms underlying the scientific study of religion. *Journal for the Scientific Study of Religion*, 39(4), pp. 482-495.
- BECKFORD, J. A. 2003. Social theory and religion. Cambridge: Cambridge University Press.
- BECKFORD, J. A. 2006. 'A minimalist sociology of religion'? *In:* BECKFORD, J. A. & WALLISS, J. (eds.) *Theorising religion. Classical and contemporary debates.* Aldershot: Ashgate, pp. 182-196.
- BECKFORD, J. A. 2008. Secularism and coercive freedoms. *The British Journal of Sociology*, 59(1), pp. 41-45.
- BECKFORD, J. A. 2010a. Religious pluralism and diversity: response to Yang and Thériault. *Social Compass*, 57(2), pp. 217-223.

- BECKFORD, J. A. 2010b. The return of public religion? A critical assessment of a popular claim. *Nordic Journal of Religion and Society*, 23(2), pp. 121-136.
- BECKFORD, J. A. 2010c. Constructing religion in unexpected places: phishers of men and women. *Implicit Religion*, 13(1), pp. 71-83.
- BECKFORD, J. A. 2011. The continuum between "cults" and "normal" religion. *In:* CÔTÉ, P. (ed.) *Chercheurs de dieux dans l'espace public.* Ottawa: Les Presses de l'Université d'Ottawa, pp. 11-20.
- BECKFORD, J. A. 2012. Public religions and the post-secular: critical reflections. *Journal for the Scientific Study of Religion*, 51(1), pp. 1-19.
- BECKFORD, J. A. & LUCKMANN, T. 1989. The changing face of religion. London: Sage.
- BECKFORD, J. A. & WALLISS, J. (eds.) 2006. *Theorising religion. Classical and contemporary debates.* Aldershot: Ashgate.
- BELK, R. W. & COSTA, J. A. 1998. The mountain man myth: a contemporary consuming fantasy. *Journal of Consumer Research*, 25(3), pp. 218-240.
- BELK, R. W., WALLENDORF, M. & SHERRY, J. F. 1989. The sacred and the profane in consumer behavior: theodicy on the odyssey. *Journal of Consumer Research*, 16(1), pp. 1-38.
- BELL, D. M. 2008. Development of the religious self: a theoretical foundation for measuring religious identity. *In:* DAY, A. (ed.) *Religion and the individual: belief, practice, identity.* Aldershot: Ashgate, pp. 127-142.
- BELLAH, R. N. 1986. *Habits of the heart: individualism and commitment in American life.*New York, N.Y.: Harper & Row, Perennial Library.
- BENDER, C. 2003. *Heaven's kitchen. Living religion at God's love we deliver.* Chicago, IL: University of Chicago Press.
- BENDER, C. 2010. *The new metaphysicals. Spirituality and the American religious imagination.* Chicago, IL: University of Chicago Press.
- BENDER, C. & KLASSEN, P. E. 2010. After pluralism. Reimagining religious engagement. New York, N.Y.: Columbia University Press.
- BENDER, C. & KLASSEN, P. E. 2010. Introduction: habits of pluralism *In:* BENDER, C. & KLASSEN, P. E. (eds.) *After pluralism. Reimagining religious engagement.* New York, N.Y.: Columbia University Press, pp. 1-28.

- BERGER, P. L. 1967. *The sacred canopy: elements of a sociological theory of religion* New York, N.Y.: Anchor Books.
- BERGER, P. L. 1999. *The desecularization of the world: resurgent religion and world politics.*Washington, D.C.: Ethics and Public Policy Center; W.B. Eerdmans Publishing, pp. 1-18.
- BERGER, P. L. & HEFNER, R. W. 2003. Spiritual capital in comparative perspective.

 Templeton Foundation [Online]. Available:

 www.metanexus.net/spiritual_capital/pdf/Berger.pdf.
- BERMAN, M. 1970. *The politics of authenticity: radical individualism and the emergence of modern society.* New York, N.Y.: Atheneum.
- BERNSTEIN, R. J. 1992. The resurgence of pragmatism *Social Research*, 59(4), pp. 814-840.
- BESECKE, K. 2001. Speaking of meaning in modernity: reflexive spirituality as a cultural resource. *Sociology of Religion*, 62(3), pp. 365-381.
- BESECKE, K. 2005. Seeing invisible religion: religion as a societal conversation about transcendent meaning. *Sociological Theory*, 23(2), pp. 179-196.
- BEYER, P. 2003. Social forms of religion and religions in contemporary global society. *In:*DILLON, M. (ed.) *A handbook of the sociology of religion.* New York, N.Y.: Cambridge University Press, pp. 45-60.
- BIELO, J. S. 2004. Walking in the spirit of blood: moral identity among born-again Christians. *Ethnology*, 43(3), pp. 271-289.
- BIELO, J. S. 2008. On the failure of 'meaning': Bible reading in the anthropology of Christianity. *Culture and Religion*, 9(1), pp. 1-21.
- BIELO, J. S. 2009. The 'emerging church' in America: notes on the interaction of Christianities. *Religion*, 39(3), pp. 219-232.
- BIELO, J. S. 2011. *Emerging evangelicals: faith, modernity, and the desire for authenticity.*New York, N.Y.: New York University Press.
- BIRKHEAD, J. 2002. 'There's power in the blood'. Writing serpent handling as everyday life. *In:* SPICKARD, J. V., LANDRES, J. S. & MCGUIRE, M. B. (eds.) *Personal knowledge and beyond: reshaping the ethnography of religion.* New York, N.Y.: New York University Press, pp. 134-145.
- BLEICHER, J. 2006. Leben. *Theory Culture Society*, 23, pp. 343-345.

- BLEICHER, J. 2007. From Kant to Goethe: Georg Simmel on the way to Leben. *Theory Culture Society*, 24(7/8), pp. 139-158.
- BLUM, L. A. 1988. Gilligan and Kohlberg: implications for moral theory *Ethics*, 98(3), pp. 472-491.
- BLUMER, H. 1966. Sociological implications of the thought of George Herbert Mead. *American Journal of Sociology*, 71(5), pp. 535-544.
- BLUMER, H. 1969. *Symbolic interactionism: perspective and method.* Englewood Cliffs, N.J.: Prentice-Hall.
- BOBBIO, N. 1984/1991. Il futuro della democrazia. Torino: Einaudi.
- BOUDON, R. 1998. Limitations of Rational Choice Theory. *The American Journal of Sociology*, 104(3), pp. 817-828.
- BOUDON, R. 2003. Beyond rational choice theory. *Annual Review of Sociology*, 29, pp. 1-21.
- BOYARIN, D. 2004. *Border lines: the partition of Judaeo-Christianity.* Philadelphia, Penn.: University of Pennsylvania Press.
- BOYARIN, D. 2006. Apartheid comparative religion in the second century: some theory and a case study. *Journal of Medieval and Early Modern Studies*, 36(1), pp. 3-34.
- BOYARIN, D. 2009. Rethinking Jewish Christianity: an argument for dismantling a dubious category (to which is appended a correction of my Border Lines). *The Jewish Quarterly Review*, 99(1), pp. 7-36.
- BROWN, C. G. 2001. *The death of Christian Britain: understanding secularisation, 1800-2000.*London: Routledge.
- BROWN, C. G. 2010. What was the religious crisis of the 1960s? *Journal of Religious History*, 34(4), pp. 468–479.
- BROWN-SARACINO, J., THURK, J. & FINE, G. A. 2008. Beyond groups: seven pillars of peopled ethnnography in organizations and communities. *Qualitative Research*, 8(5), pp. 547-567.
- BRUCE, S. 1993. Religion and rational choice: a critique of economic explanations of religious behavior. *Sociology of Religion*, 54(2), pp. 193-205.
- BRUCE, S. 1999. *Choice and religion: a critique of rational choice theory.* Oxford: Oxford University Press.
- BRUCE, S. 2002. *God is dead: secularization in the West.* Oxford: Blackwell Publishers.

- BRUCE, S. & VOAS, D. 2010. Vicarious religion: an examination and critique. *Journal of Contemporary Religion*, 25(2), pp. 243-259.
- BUBER, M. & KAUFMANN, W. A. 1923/1970. I and thou. Edinburgh: T. & T. Clark.
- BURAWOY, M. 1991. The extended case method. *In:* BURAWOY, M. (ed) *Ethnography unbound: power and resistance in the modern metropolis.* Berkeley, CA: University of California Press, pp. 271-287.
- BURGESS, R. G. 1991. Sponsors, gatekeepers, members, and friends. Access in educational settings. *In:* SHAFFIR, W. & STEBBINS, R. A. (eds.) *Experiencing fieldwork: an inside view of qualitative research.* London: Sage, pp. 43-52.
- BURKE, P. 1997. Representation of the self from Petrarch to Descartes. *In:* PORTER, R. (ed) *Rewriting the self. Histories from the Renaissance to the present.* London: Routledge, pp. 17-28.
- BURKE, T. P. 1984. Must the description of a religion be acceptable to a believer? *Religious Studies*, 20(4), pp. 631-636.
- BUTTON, R. W. 2012. Fate, experience and tragedy in Simmel's dialogue with modernity. *Theory, Culture & Society,* 29(7-8), pp. 53-77.
- CADGE, W., SMILDE, D. & LEVITT, P. 2011. De-centering and re-centering: rethinking concepts and methods in the sociological study of religion. *Journal for the Scientific Study of Religion*, 50(3), pp. 437-449.
- CALABRÒ, G. 1913/2004. Postfazione. *In:* SIMMEL, G. and CALABRÒ, G. (ed) *L'etica e i problemi della cultura moderna.* Napoli: Guida, pp. 63-86.
- CALHOUN, C. J. 1991. Morality, identity, and historical explanation: Charles Taylor on the sources of the self. *Sociological Theory*, 9(2), pp. 232-263.
- CAMIA, F. 2009. Imperatori romani tra gli dei greci: riflessioni sull'associazione tra culto imperiale e culti tradizionali in Grecia a partire dalla documentazione epigrafica. *In:* CAMIA, F. & PRIVITERA, S. (eds.) *Obeloi. Contatti, scambi e valori nel Mediterraneo antico.* Paestum: Pandemos, pp. 205-222.
- CAMIA, F. 2012. Theoi Olympioi e Theoi Sebastoi: alcune considerazioni sull'associazione tra culto imperiale e culti tradizionali in Grecia. *In:* FRANCHI, E. & PROIETTI, G. (eds.) *Forme della memoria e dinamiche identitarie nell'antichità greco-romana* Trento: Università di Trento, pp. 93-106.

- CAMIA, F. & KANTIRÉA, M. 2010. The imperial cult in the Peloponnese. *In:* RIZAKIS, A. D. & LEPENIOTI, C. E. (eds.) *Roman Peloponnese III. Society, economy and culture under the Roman Empire: continuity and innovation.* Athens: The National Hellenic Research Foundation.
- CARRETTE, J. R. & KING, R. 2004. *Selling spirituality: the silent takeover of religion.* London: Routledge.
- CARSON, D. A. 1996. *The gagging of God: Christianity confronts pluralism.* Leicester: Apollos.
- CARTWRIGHT, D. E. 1984. Kant, Schopenhauer, and Nietzsche on the morality of pity. *Journal of the History of Ideas*, 45(1), pp. 83-98.
- CARTWRIGHT, D. E. 1988. Schopenhauer's compassion and Nietzsche's pity. *SchopenhauerJahrbuch*, 69(3), pp. 557-567.
- CARTWRIGHT, D. E. 2008. Compassion and solidarity with sufferers: the metaphysics of *Mitleid. European Journal of Philosophy*, 16(2), pp. 292-310.
- CASANOVA, J. 1994. *Public religions in the modern world.* Chicago, IL: University of Chicago Press.
- CASANOVA, J. 2006. Religion, European secular identities, and European integration. *In:*BYRNES, T. A. & KATZENSTEIN, P. J. (eds.) *Religion in an expanding Europe.*Cambridge: Cambridge University Press, pp. 65-92.
- CASANOVA, J. 2009. The religious situation in Europe. *In:* JOAS, H. & WIEGANDT, K. (eds.) *Secularization and the world religions.* Liverpool: Liverpool University Press, pp. 206-227.
- CASANOVA, J. 2011. From modernization to secularization to globalization. An autobiographical self-reflection *Religion and Society* 2(1), pp. 25-36.
- CELNICK, A. 2000. Ethics in the field. *In:* BURTON, D. (ed.) *Research training for social scientists.* London: Sage, pp. 97-108.
- CERULO, K. A. 1997. Identity construction: new issues, new directions. *Annual Review of Sociology*, 23, pp. 385-409.
- CHAMBERS, P. 2004. The effects of evangelical renewal on mainstream congregational identities: a Welsh case study. *In:* GUEST, M., TUSTING, K. & WOODHEAD, L. (eds.) *Congregational studies in the UK: Christianity in a post-Christian context.* Aldershot: Ashgate, pp. 57-70.

- CHAMBERS, P. 2006. Secularization: human rights and sacred texts. *In:* ARWECK, E. & COLLINS, P. (eds.) *Reading religion in text and context: reflections of faith and practice in religious materials.* Aldershot: Ashgate, pp. 50-64.
- CHANDLER, S. 2008. The social ethic of religiously unaffiliated spirituality. *Religion Compass* 2(2), pp. 240-256.
- CHAPMAN, R. 2009. Faith and the voluntary sector in urban governance: distinctive yet similar? *In:* DINHAM, A., FURBEY, R. & LOWNDES, V. (eds.) *Faith in the public realm. Controversies, policies and practices.* Bristol: Policy Press, pp. 203-222.
- CHAPMAN, R. & HAMALAINEN, L. 2011. Understanding faith-based engagement and volunteering in the postsecular society: motivations, rationales and translation. *In:*BEAUMONT, J. & BAKER, C. (eds.) *Postsecular cities. Space, theory and practice.*London: Continuum, pp. 184-200.
- CHAPMAN, R. & LOWNDES, V. 2008. Faith in governance? The potential and pitfalls of involving faith groups in urban governance. *Planning Practice and Research*, 23(1), pp. 57-75.
- CHARMÉ, S. 1992. The different voices in Sartre's ethics. *Bulletin de l'Association Americaine de Philosophie de Langue Française*, Fall, pp. 264-280.
- CHARMÉ, S. Z. 2000. Varieties of authenticity in contemporary Jewish identity. *Jewish Social Studies*, 6(2), pp. 133-155.
- CHAVES, M. 1994. Secularization as declining religious authority. *Social Forces*, 72(3), pp. 749-774.
- CHAVES, M. 1995. On the rational choice approach to religion. *Journal for the Scientific Study of Religion*, 34(1), pp. 98-104.
- CHAVES, M. 1999. Religious congregations and welfare reform: who will take advantage of "charitable choice"? *American Sociological Review*, 64(6), pp. 836-846.
- CHAVES, M. 2009. Congregations' significance to American civic life *In:* LICHTERMAN, P. & POTTS, C. B. (eds.) *The civic life of American religion.* Stanford, CA: Stanford University Press, pp. 69-81.
- CHAVES, M., GIESEL, H. M. & TSITSOS, W. 2002. Religious variations in public presence: evidence from the national congregation study *In:* WUTHNOW, R. & EVANS, J. H.

- (eds.) *The quiet hand of God. Faith-based activism and the public role of mainline Protestantism* Berkeley, CA: University of California Press, pp. 108-128.
- CHAVES, M. & GORSKI, P. S. 2001. Religious pluralism and religious participation. *Annual Review of Sociology*, 27, pp. 261-281.
- CHISERI-STRATER, E. 1996. Turning in upon ourselves: positionality, subjectivity, and reflexivity in case study and ethnographic research. *In:* MORTENSEN, P. & KIRSCH, G. (eds.) *Ethics and representation in qualitative studies.* Urbana: National Council of Teachers of English, pp. 115-132.
- CHRISTIANS, C. G. 2005. Ethics and politics in qualitative research. *In:* DENZIN, N. K. & LINCOLN, Y. S. (eds.) *The SAGE handbook of qualitative research.* Thousand Oaks, CA: Sage, pp. 139-164.
- CIPOLLA, C. M. 1969. *Literacy and development in the West.* Harmondsworth: Penguin.
- CIPOLLA, C. M. 1976/1993. *Before the industrial revolution*. London: Routledge.
- CIPRIANI, R. 1989. "Diffused religion" and new values in Italy. *In:* BECKFORD, J. A. & LUCKMANN, T. (eds.) *The changing face of religion.* London: Sage.
- CIPRIANI, R. 1998. Sociology and religion: Durkheim and Simmel. *In:* LAERMANS, R., WILSON, B. & BILLIET, J. (eds.) *Secularization and social integration.* Leuven: Leuven University Press, pp. 37-43.
- CIPRIANI, R. 2000. *Sociology of religion: an historical introduction.* New York, N.Y.: Aldine de Gruyter.
- CLOKE, P. 2011. Emerging postsecular rapprochement in the contemporary city *In:*BEAUMONT, J. & BAKER, C. (eds.) *Postsecular cities. Space, theory and practice.*London: Continuum, pp. 237-253.
- COFFEY, A. 1999. *The ethnographic self: fieldwork and the representation of identity.*Thousand Oaks, CA: Sage.
- COFFEY, A., HOLBROOK, B. & ATKINSON, P. 1996. Qualitative data analysis: technologies and representations. *Sociological Research Online*, 1(1),[Online]. Available: http://www.socresonline.org.uk/1/1/4.html.
- COHEN, A. B. 2001. Religion and the morality of mentality. *Journal of Personality and Social Psychology*, 81(4), pp. 697-710.

- COHEN, A. B., HALL, D. E., KOENIG, H. G. & MEADOR, K. G. 2005. Social versus individual motivation: implications for normative definitions of religious orientation. *Personality and Social Psychology Review,* 9(1), pp. 48-61.
- COLEMAN, S. 2002. 'But are they really Christian?' Contesting knowledge and identity in and out of the field. *In:* SPICKARD, J. V., LANDRES, J. S. & MCGUIRE, M. B. (eds.) *Personal knowledge and beyond: reshaping the ethnography of religion.* New York, N.Y.: New York University Press, pp. 75-87.
- COLLINS, P. 1998. Negotiating selves: reflections on 'unstructured' interviewing. *Sociological Research Online*, 3(3), [Online]. Available: http://www.socresonline.org.uk/3/3/2.html.
- COLLINS, P. 2008. Accommodating the individual and the social, the religious and the secular: modelling the parameters of discourse in 'religious' contexts *In:* DAY, A. (ed.) *Religion and the individual: belief, practice, identity.* Aldershot: Ashgate, pp. 143-156.
- COLLINS, R. 1994. Four sociological traditions. New York, N.Y.: Oxford University Press.
- CORTAZZI, M. 1993. *Narrative analysis*. London: Falmer Press.
- COSER, L. A. 1965. *Georg Simmel*. Englewood Cliffs: Prentice-Hall.
- CRESSWELL, J. W. & MILLER, D. L. 2000. Determining validity in qualitative inquiry. *Theory into Practice*, 39(3), pp. 124-131.
- CROMARTIE, M. 2003. *A public faith: evangelicals and civic engagement.* Lanham, MD: Rowman & Littlefield Publishers.
- DARMON, I. & FRADE, C. 2012. Beneath and beyond the fragments: the charms of Simmel's philosophical path for contemporary subjectivities. *Theory, Culture & Society,* 29(7-8), pp. 197-217.
- DAVIDMAN, L. 2002. Truth, subjectivity, and ethnographic research. *In:* SPICKARD, J. V., LANDRES, J. S. & MCGUIRE, M. B. (eds.) *Personal knowledge and beyond: reshaping the ethnography of religion.* New York, N.Y.: New York University Press, pp. 17-26.
- DAVIE, G. 1994. *Religion in Britain since 1945: believing without belonging.* Oxford: Blackwell.

- DAVIE, G. 1999. Europe: the exception that proves the rule? *In:* BERGER, P. L. (ed.) *The desecularization of the world. Resurgent religion and world politics.* Washington D.C.: Ethics and Public Policy Center; Wm. B. Eerdmans Publishing, pp. 65-84.
- DAVIE, G. 2002. *Europe the exceptional case: parameters of faith in the modern world.*London: Darton Longman & Todd.
- DAVIE, G. 2007. Vicarious religion: a methodological challenge. *In:* AMMERMAN, N. T. (ed.) *Everyday religion: observing modern religious lives.* New York, N.Y.: Oxford University Press, pp. 21-36.
- DAVIES, C. A. 1999. *Reflexive ethnography: a guide to researching selves and others.* London: Routledge.
- DAVIES, D. J. & GUEST, M. 2007. *Bishops, wives and children: spiritual capital across the generations.* Aldershot: Ashgate.
- DAVIES, T. 1997. Humanism. London: Routledge.
- DAVIS, F. 1973. The Martian and the convert: ontological polarities in social research. *Urban Life and Culture,* 2(3), pp. 333-343.
- DAVIS, M. S. 1997. Georg Simmel and Erving Goffman: legitimators of the sociological investigation of human experience. *Qualitative Sociology* 20(3), pp. 369-388.
- DAY, A. (ed.) 2008. *Religion and the individual: belief, practice, identity.* Aldershot: Ashgate.
- DAY, A. 2011. *Believing in belonging: belief and social identity in the modern world.* Oxford: Oxford University Press.
- DE LA FUENTE, E. 2008. The art of social forms and the social form of art: the sociology-aesthetics nexus in Georg Simmel's thought. *Sociological Theory*, 26(4), pp. 344-362.
- DE SIMONE, A. 2002. *Georg Simmel: i problemi dell'individualità moderna.* Urbino: Quattro Venti.
- DEFLEM, M. 2001. Ferdinand Tönnies (1855-1936). *In:* CRAIG, E. (ed.) *Routledge Encyclopedia of Philosophy.* London: Routledge.
- DEFLEM, M. 2003. The sociology of the sociology of money. Simmel and the contemporary battle of the classics. *Journal of Classical Sociology*, 3(1), pp. 67-96.
- DEHGHANI, M., ATRAN, S., ILIEV, R., SACHDEVA, S., MEDIN, D. & GINGES, J. 2010. Sacred values and conflict over Iran's nuclear program. *Judgment and Decision Making*, 5(7), pp. 540–546.

- DEHGHANI, M., ILIEV, R., SACHDEVA, S., ATRAN, S., GINGES, J. & MEDIN, D. 2009. Emerging sacred values: Iran's nuclear program. *Judgment and Decision Making*, 4(7), pp. 930–933.
- DELAMONT, S. 1981/2012. All too familiar? A decade of classroom research. *In:*DELAMONT, S. (ed) *Ethnographic methods in education.* Los Angeles, CA: Sage, pp. 279-294.
- DELAMONT, S. 1992. Fieldwork in educational settings methods, pitfalls and perspectives.

 London: Falmer Press.
- DELAMONT, S. 2009. The only honest thing: autoethnography, reflexivity and small crises in fieldwork. *Ethnography and Education* 4(1), pp. 51-63.
- DELAMONT, S. (ed) 2012. Ethnographic methods in education. Los Angeles, CA: Sage.
- DELAMONT, S., ATKINSON, P. & PUGSLEY, L. 2010. The concept smacks of magic: fighting familiarity today. *Teaching and Teacher Education*, 26(1), pp. 3-10.
- DEMERATH, N. J. 1995. Rational paradigms, A-rational religion, and the debate over secularization. *Journal for the Scientific Study of Religion*, 34(1), pp. 105-112.
- DENORA, T. 2000. Music in everyday life. Cambridge: Cambridge University Press.
- DENZIN, N. K. 2004. Symbolic interactionism. *In:* FLICK, U., VON KARDORFF, E. & STEINKE, I. (eds.) *A companion to qualitative research.* London: Sage, pp. 81-87.
- DENZIN, N. K. & LINCOLN, Y. S. 2005. *The SAGE handbook of qualitative research.* Thousand Oaks, CA: Sage.
- DEROCHE-GURCEL, L. 1988. Le conflit des méthodes en sciences sociales: la contribution de Simmel. *Persée. Revue de sociologie française,* 29(2), pp. 347-356.
- DINHAM, A. 2008. Commentary: from faith in the city to faithful cities: the 'third way', the Church of England and urban regeneration. *Urban Studies*, 45(10), pp. 2163.
- DINHAM, A. 2011. What is a 'faith community'? *Community Development Journal*, 46(4), pp. 526-541.
- DINHAM, A. 2012. *Faith and social capital after the debt crisis.* Houndmills, Basingstoke: Palgrave Macmillan.
- DINHAM, A., FURBEY, R. & LOWNDES, V. (eds.) 2009. *Faith in the public realm. Controversies, policies and practices.* Bristol: Policy Press.

- DINHAM, A. & LOWNDES, V. 2008. Religion, resources, and representation: Three narratives of faith engagement in British Urban governance. *Urban Affairs Review*, 43(6), pp. 817-845.
- DOEHRING, C., CLARKE, A., PARGAMENT, K. I., HAYES, A., HAMMER, D., NICKOLAS, M. & HUGHES, P. 2009. Perceiving sacredness in life: correlates and predictors *Archive for the Psychology of Religion*, 31, pp. 55-73.
- DÖRR-BACKES, F. 1995. The study of culture in Georg Simmel's writings on art. *In:* DÖRR-BACKES, F. & NIEDER, L. (eds.) *Georg Simmel between modernity and postmodernity. Georg Simmel zwischen moderne und postmoderne.* Würzburg, Germany:

 Königshausen & Neumann, pp. 119-128.
- DOUGLAS, M. 1966/1989. *Purity and danger: an analysis of concept of pollution and taboo.*London: Routledge.
- DROOGERS, A. 2007. Beyond secularisation versus sacralisation: lessons from a study of the Dutch case. *In:* FLANAGAN, K. & JUPP, P. C. (eds.) *A sociology of spirituality.*Aldershot: Ashgate, pp. 81-99.
- DURKHEIM, É. 1975. Textes. Religion, morale, anomie. Paris: Les Editions de Minuit.
- DURKHEIM, É., COSMAN, C. & CLADIS, M. S. 1912/2008. *The elementary forms of religious life.* Oxford: Oxford University Press.
- DUVENAGE, P. 2004. Alessandro Ferrara's reflective authenticity: rethinking the project of modernity. *Philosophy Social Criticism*, 30(1), pp. 127-134.
- EBAUGH, H. R., CHAFETZ, J. S. & PIPES, P. F. 2006. Where's the faith in faith-based organizations? *Social Forces*, 84, pp. 2259-2272.
- ECKHART, COLLEDGE, E. & MCGINN, B. 1981. *Meister Eckhart, the essential sermons, commentaries, treatises, and defense.* New York, N.Y.: Paulist Press.
- EDGELL BECKER, P. 1997. What is right? What is caring? Moral logics in local religious life.

 In: EDGELL, P. & EIESLAND, N. L. (eds.) Contemporary American religion: an

 ethnographic reader. Walnut Creek, CA: AltaMira Press, pp. 121-145.
- EDGELL BECKER, P. & EIESLAND, N. L. 1997. *Contemporary American religion: an ethnographic reader.* Walnut Creek, CA: AltaMira Press.
- EDGELL, P. 1999. *Congregations in conflict: cultural models of local religious life.* Cambridge: Cambridge University Press.

- ELIOT, G. 1872/2008. *Middlemarch: a study of provincial life.* Oxford: Oxford University Press.
- ELISHA, O. 2008a. Moral ambitions of grace: the paradox of compassion and accountability in evangelical faith-based activism. *Cultural Anthropology*, 23(1), pp. 154-189.
- ELISHA, O. 2008b. Faith beyond belief. Evangelical Protestant conceptions of faith and the resonance of anti-humanism. *Social Analysis*, 52(1), pp. 56-78.
- ELISHA, O. 2011. *Moral ambition. Mobilization and social outreach in evangelical megachurches.* Berkeley, CA: University of California Press.
- ELLISON, C. G. 1995. Rational choice explanations of individual religious behavior: notes on the problem of social embeddedness. *Journal for the Scientific Study of Religion*, 34(1), pp. 89-97.
- ELSTER, J. 2009. Interpretation and rational choice. *Rationality and Society*, 21(1), pp. 5-33.
- EMERSON, R. M., FRETZ, R. I. & SHAW, L. L. 2001. Participant observation and fieldnotes. *In:* ATKINSON, P., COFFEY, A. & DELAMONT, S. (eds.) *Handbook of ethnography.* London: Sage, pp. 352-368.
- EMMONS, R. A. 2000. Is spirituality an intelligence? Motivation, cognition, and the psychology of ultimate concern. *International Journal for the Psychology of Religion*, 10(1), pp. 3-26.
- EMMONS, R. A. 2003. *The psychology of ultimate concerns: motivation and spirituality in personality.* New York, N.Y.: Guilford Press.
- EMMONS, R. A. 2005. Emotion and religion. *In:* PALOUTZIAN, R. F. & PARK, C. L. (eds.) *Handbook of the psychology of religion and spirituality.* New York, N.Y.: Guilford Press, pp. 235-252.
- EMMONS, R. A. 2005. Striving for the sacred: personal goals, life meaning, and religion. *Journal of Social Issues*, 61(4), pp. 731-745.
- EMMONS, R. A. & CRUMPLER, C. A. 1999. Religion *and* spirituality? The roles of sanctification and the concept of God *The International Journal for the Psychology of Religion*, 9(1), pp. 17-24.
- ERICKSON, V. L. 2001. Georg Simmel: American sociology chooses the stone the builders refused. *In:* FENN, R. K. (ed) *The Blackwell companion to sociology of religion.* Oxford: Blackwell, pp. 105-119.

- FARNELL, R. 2001. Faith communities, regeneration and social exclusion: developing a research agenda. *Community Development Journal*, 36(4), pp. 263-272.
- FARNELL, R. 2009. Faiths, government and regeneration: a contested discourse. *In:* DINHAM, A., FURBEY, R. & LOWNDES, V. (eds.) *Faith in the public realm. Controversies, policies and practices.* Bristol: Policy Press, pp. 183-202.
- FARNELL, R., FURBEY, R., MACEY, M., SMITH, G. & AL-HAQQ HILLS, S. S. (eds.) 2003. 'Faith' in urban regeneration? Bristol: Policy Press.
- FARR, J. 2004. Social capital: a conceptual history *Political Theory*, 32(1), pp. 6-33.
- FAUGHT, J. 1985. Neglected affinities: Max Weber and Georg Simmel. *The British Journal of Sociology*, 36(2), pp. 155-174.
- FERRARA, A. 1998. *Reflective authenticity: rethinking the project of modernity.* London: Routledge.
- FERRARA, A. 2004. the relation of authenticity to normativity: a response to Larmore and Honneth. *Philosophy Social Criticism*, 30(1), pp. 17-24.
- FERRARA, A. 2009. The separation of religion and politics in a post-secular society. *Philosophical Social Criticism*, 35(1-2), pp. 77-91.
- FETTERMAN, D. M. 1998. Ethnography: step by step. London: Sage.
- FEUERBACH, L. & ELIOT, G. 1841/1989. *The essence of Christianity*. Buffalo, N.Y: Prometheus Books.
- FINE, G. A. 1993. The sad demise, mysterious disappearance, and glorious triumph of symbolic interactionism. *Annual Review of Sociology*, 19, pp. 61-87.
- FINE, G. A. 2003. Towards a peopled ethnography: developing theory from group life. *Ethnography*, 4(1), pp. 41-60.
- FINE, G. A. 2010. The sociology of the local: action and its publics. *Sociological Theory*, 28(4), pp. 355-376.
- FINKE, R. 2003. Spiritual capital: definitions, applications, and new frontiers. *Templeton Foundation*, [Online]. Available: www.metanexus.net/spiritual_capital/pdf/finke.pdf.
- FINKE, R. & STARK, R. 2003. The dynamics of religious economies *In:* DILLON, M. (ed.) *A handbook of the sociology of religion.* Cambridge: Cambridge University Press, pp. 96-109.

- FISKE, A. P. & TETLOCK, P. E. 1997. Taboo trade-offs: reactions to transactions that transgress the spheres of justice. *Political Psychology*, 18(2), pp. 255-297.
- FITZI, G. 2012. A 'transformative' view of society building: Simmel's sociological epistemology and philosophical anthropology of complex societies. *Theory, Culture & Society*, 29(7-8), pp. 177-196.
- FLANAGAN, K. 1996. *The enchantment of sociology: a study of theology and culture.*Basingstoke: Macmillan.
- FLANAGAN, K. 2007. Visual spirituality: an eye for religion. *In:* FLANAGAN, K. & JUPP, P. C. (eds.) *A sociology of spirituality.* Aldershot Ashgate, pp. 219-250.
- FLANAGAN, K. 2008. Sociology into theology: the unacceptable leap. *Theory Culture Society*, 25(7-8), pp. 236-261.
- FLANAGAN, K. & JUPP, P. C. 1996. *Postmodernity, sociology and religion.* Basingstoke: Macmillan.
- FLANAGAN, K. & JUPP, P. C. (eds.) 2007. A sociology of spirituality. Aldershot: Ashgate.
- FLANAGAN, O. & JACKSON, K. 1987. Justice, care, and gender: the Kohlberg-Gilligan debate revisited. *Ethics*, 97(3), pp. 622-637.
- FLETCHER, S. K. 2004. Religion and life meaning: differentiating between religious beliefs and religious community in constructing life meaning. *Journal of Aging Studies*, 18(2), pp. 171-185.
- FLICK, U. 2004. Constructivism. *In:* FLICK, U., VON KARDORFF, E. & STEINKE, I. (eds.) *A companion to qualitative research.* London: Sage, pp. 88-94.
- FLICK, U. 2006. An introduction to qualitative research. London: Sage.
- FLORY, R. & MILLER, D. E. 2007. The embodied spirituality of the post-boomer generations. *In:* FLANAGAN, K. & JUPP, P. C. (eds.) *A sociology of spirituality.* Aldershot: Ashgate, pp. 201-218.
- FLORY, R. W. & MILLER, D. E. 2008. *Finding faith: the spiritual quest of the post-boomer generation.* New Brunswick, N.J.: Rutgers University Press.
- FOLEY, D. E. 2002. Critical ethnography: the reflexive turn. *Qualitative Studies in Education*, 15(5), pp. 469-490.

- FONTANA, A. & FREY, J. H. 2005. The interview: from neutral stance to political involvement. *In:* DENZIN, N. K. & LINCOLN, Y. S. (eds.) *Handbook of qualitative research.* London: Sage, pp. 695-728.
- FORSEY, M. G. 2010. Ethnography as participant listening. *Ethnography*, 11(4), pp. 558-572.
- FRISBY, D. 1981. *Sociological impressionism: a reassessment of Georg Simmel's social theory.*London: Heinemann.
- FRISBY, D. 1985. Georg Simmel, first sociologist of modernity. *Theory, Culture & Society,* 2(3), pp. 49-67.
- FRISBY, D. 1992. *Simmel and since: essays on Georg Simmel's social theory.* London: Routledge.
- FRISBY, D. (ed.) 1994. Georg Simmel: critical assessments. London: Routledge.
- FRISBY, D. 2002. Georg Simmel. Chichester: Horwood.
- FROHNEN, B. 1996. *The new communitarians and the crisis of modern liberalism.* Lawrence: University Press of Kansas.
- FULKERSON, G. M. & THOMPSON, G. H. 2008. The evolution of a contested concept: a metaanalysis of social capital definitions and trends (1988-2006). *Sociological Inquiry* 78(4), pp. 536-557.
- FURBEY, R. 1999. Urban 'regeneration': reflections on a metaphor. *Critical Social Policy*, 19(4), pp. 419-445.
- FURBEY, R. 2009. Controversies of 'public faith'. *In:* DINHAM, A., FURBEY, R. & LOWNDES, V. (eds.) *Faith in the public realm. Controversies, policies and practices.* Bristol: Policy Press, pp. 21-40.
- FURBEY, R. & MACEY, M. 2005. Religion and urban regeneration: a place for faith? *Policy and Politics*, 33(1), pp. 95-116.
- FURBEY, R. A., DINHAM, A., FARNELL, R., FINNERON, D. & WILKINSON, G. 2006. *Faith as social capital: connecting or dividing?* Bristol: Policy Press.
- GALLAGHER, S. K. 2009. Defining spiritual growth: congregations, community, and connectedness. *Sociology of Religion*, 70(3), pp. 232-261.
- GANIEL, G. & MITCHELL, C. 2006. Turning the categories inside-out: complex identifications and multiple interactions in religious ethnography. *Sociology of Religion*, 67(1), pp. 3-21.

- GEERTZ, A. 2002. As the Other sees us: on reciprocity and mutual reflection in the study of Native American religions *In:* SPICKARD, J. V., LANDRES, J. S. & MCGUIRE, M. B. (eds.) *Personal knowledge and beyond: reshaping the ethnography of religion.* New York, N.Y.: New York University Press, pp. 225-236.
- GEERTZ, C. 1973. The interpretation of cultures: selected essays. New York, N.Y.: Basic Books.
- GEERTZ, C. 1999. 'From the native's point of view': on the nature of anthropological understanding. *In:* MCCUTCHEON, R. T. (ed.) *The insider/outsider problem in the study of religion: a reader.* London: Cassell, pp. 50-63.
- GEERTZ, C. 2005. Deep play: notes on the Balinese cockfight. *Daedalus*, 134(4), pp. 56-86.
- GENTILE, E. 1990. Fascism as political religion. *Journal of Contemporary History*, 25(2/3), pp. 229-251.
- GENTILE, E. 2000. The sacralisation of politics: definitions, interpretations and reflections on the question of secular religion and totalitarianism. *Totalitarian Movements and Political Religions*, 1(1), pp. 18-55.
- GENTILE, E. 2001. Politics as religion. Princeton, N.J.: Princeton University Press.
- GENTILE, E. 2004. *God's democracy. American religion after September 11.* Westport, CT.: Praeger.
- GENTILE, E. 2005. Political religion: a concept and its critics A critical survey. *Totalitarian Movements and Political Religions*, 6(1), pp. 19-32.
- GERHARDS, J. 1986. Georg Simmel's contribution to a theory of emotions. *Social Science Information*, 25(4), pp. 901-924.
- GHERARDI, S. & TURNER, B. 1987/1999. Real men don't collect soft data. *In:* BRYMAN, A. A. B., ROBERT G. (ed.) *Qualitative research.* London: Sage, pp. 103-119.
- GILBERT, G. N. & MULKAY, M. 1984. Experiments are the key: participants' histories and historians' histories of science. *Isis*, 75(1), pp. 105-125.
- GILLIGAN, C. 1982. *In a different voice: psychological theory and women's development.*Cambridge, MA: Harvard University Press.
- GILLINGAN, C. 1977. In a different voice: women's conceptions of self and of morality. *Harvard Educational Review*, 47(4), pp. 481-516.
- GIORDAN, G. 2004. Dalla religione alla spiritualità: una nuova legittimazione del sacro? *Quaderni di Sociologia*, 35, pp. 105-117.

- GIORDAN, G. 2012. La spiritualità come nuovo paradigma in sociologia della religione. *Caminhos*, 10(1), pp. 61-72.
- GIORDANO-ZECHARYA, M. 2005. As Socrates shows, the Athenians did not believe in gods. *Numen*, 52(3), pp. 326-354.
- GLASER, B. G. & STRAUSS, A. L. 1967. *The discovery of grounded theory: strategies for qualitative research.* New York, N.Y.: Aldine Publishing Co.
- GOBO, G. 2008. Doing ethnography. London: Sage.
- GOLDBERG, D. T. 1994. *Multiculturalism: a critical reader.* Cambridge, MA: Blackwell Publishers.
- GOLDBERG, J. J. 2013. Agriprocessors raid made us, briefly, rethink kashrut. *The Jewish Daily Forward,* [Online]. Available:

 http://forward.com/articles/176455/agriprocessors-raid-made-us-briefly-rethink-kashru/?p=all.
- GOLDSTEIN, W. S. 2011. The dialectics of religious conflict: church-sect, denomination and the culture wars. *Culture and Religion*, 12(1), pp. 77-99.
- GORDON, D. F. 1987. Getting close by staying distant: fieldwork with proselytizing groups. *Qualitative Sociology*, 10(3), pp. 267-287.
- GORSKI, P. S. 2000. Historicizing the secularization debate: church, state, and society in late medieval and early modern Europe, ca. 1300 to 1700. *American Sociological Review*, 65(1), pp. 138-167.
- GORSKI, P. S. & ALTINORDU, A. 2008. After secularization? *Annual Review of Sociology*, 34, pp. 55-85.
- GRAHAM, E. 2004. Introduction: 'religious capital in declining communities'. *In:* GRAHAM, E. & BAKER, C., eds. Religious capital in regenerating communities, Manchester Centre for Public Theology. [Online]. Available: http://www.williamtemplefoundation.org.uk/documents/religious-capital.pdf
- GRAHAM, E. 2008. What makes a good city? Reflections on urban life and faith *International Journal of Public Theology*, 2(1), pp. 7-26.
- GUEST, M. 2002. 'Alternative' worship: challenging the boundaries of the Christian faith. *In:* ARWECK, E. & STRINGER, M. D. (eds.) *Theorizing faith. The insider/outsider problem in the study of ritual.* Birmingham: University of Birmingham Press, pp. 35-56.

- GUEST, M. 2004. 'Friendship, fellowship and acceptance': the public discourse of a thriving evangelical congregation. *In:* GUEST, M., TUSTING, K. & WOODHEAD, L. (eds.) *Congregational studies in the UK: Christianity in a post-Christian context.* Aldershot: Ashgate, pp. 71-84.
- GUEST, M. 2005. Sociological strand: worship and action. *In:* CAMERON, H., RICHTER, P., DAVIES, D. & WARD, F. (eds.) *Studying local churches: a handbook.* Norwich: SCM Press, pp. 98-109.
- GUEST, M. 2007a. *Evangelical identity and contemporary culture: a congregational study in innovation.* Milton Keynes: Paternoster.
- GUEST, M. 2007b. Reconceiving the congregation as a source of authenticity. *In:* GARNETT, J., GRIMLEY, M. & HARRIS, A. (eds.) *Redefining Christian Britain. Post 1945*Perspectives. Norwich: SCM Press, pp. 63-72.
- GUEST, M. J. 2007c. In search of spiritual capital: the spiritual as a cultural resource. *In:* FLANAGAN, K. & JUPP, P. C. (eds.) *The sociology of spirituality.* Ashgate: Aldershot, pp. 181-200.
- GUEST, M. 2008. The reproduction and transmission of religion. *In:* CLARKE, P. B. (ed.) *The Oxford handbook of the sociology of religion.* Oxford: Oxford University Press, pp. 651-670.
- GUEST, M. & TAYLOR, S. 2006. The post-evangelical emerging church: innovations in New Zealand and the UK. *International Journal for the Study of the Christian Church,* 6(1), pp. 49-64.
- GUEST, M., TUSTING, K. & WOODHEAD, L. 2004. *Congregational studies in the UK: Christianity in a post-Christian context.* Aldershot: Ashgate.
- GUIGNON, C. B. 2004. *On being authentic.* London: Routledge.
- GUYER, P. 2012. Schopenhauer, Kant and compassion. *Kantian Review*, 17(3), pp. 403-429.
- HABERMAS, J. 2006. Religion in the public sphere. *European Journal of Philosophy*, 14(1), pp. 1-25.
- HADDOCK, B. A. & SUTCH, P. (eds.) 2003. *Multiculturalism, identity, and rights.* London: Routledge.
- HALL, D. (ed.) 1997. *Lived religion in America: toward a history of practice.* Princeton, N.J.: Princeton University Press.

- HALL, P. A. 1999. Social capital in Britain. *British Journal of Political Science*, 29, pp. 417–461.
- HAMILTON, M. 2000. Eating ethically: 'spiritual' and 'quasi-religious' aspects of vegetarianism. *Journal of Contemporary Religion*, 15(1), pp. 65-83.
- HAMMERSLEY, M. 1992/2002. Ethnography and realism. *In:* HUBERMAN, A. M. & MILES, M. B. (eds.) *The qualitative researcher's companion.* Thousand Oaks, CA: Sage, pp. 65-80.
- HAMMERSLEY, M. & ATKINSON, P. 2007. *Ethnography: principles in practice.* London: Routledge.
- HAMPSHER-MONK, I. & HINDMOOR, A. 2010. Rational choice and interpretive evidence: caught between a rock and a hard place? *Political Studies*, 58(1), pp. 47-65.
- HARAWAY, D. 1988. Situated knowledges: the science question in feminism and the privilege of partial perspective. *Feminist Studies*, 14(3), pp. 575-599.
- HARDEN, B. 2005. The greening of evangelicals. *The Washington Post*, February 6.
- HARDING, S. 1993. Rethinking standpoint epistemology: what is 'strong objectivity'? *In:* ALCOFF, L. & POTTER, E. (eds.) *Feminist epistemologies.* New York, N.Y.: Routledge, pp. 49-82.
- HARDING, S. G. 1986. *The science question in feminism.* Ithaca, NY: Cornell University Press.
- HARRINGTON, A. 2001. Dilthey, empathy and Verstehen. *European Journal of Social Theory*, 4(3), pp. 311-329.
- HARRINGTON, A. 2005. Introduction to Georg Simmel's essay 'Europe and America in world history'. *European Journal of Social Theory*, 8(1), pp. 63-72.
- HARRINGTON, A. & KEMPLE, T. M. 2012. Introduction: Georg Simmel's 'sociological metaphysics': money, sociality, and precarious life. *Theory, Culture & Society,* 29(7-8), pp. 7-25.
- HARVEY, D. 1989. The condition of postmodernity. Oxford: Basil Blackwell.
- HATCHER, A. C. 2010. The British invasion and the social transformation of American evangelicals. *Midwest Political Science Association Conference*. Chicago, IL, April 22-25.
- HECHTER, M. & KANAZAWA, S. 1997. Sociological rational choice theory. *Annual Review of Sociology*, 23, pp. 191-214.

- HEDSTRÖM, P. & STERN, C. 2008. Rational choice and sociology. *In:* DURLAUF, S. N. & BLUME, L. E. (eds.) *The new Palgrave dictionary of economics*.[Online]. Available: http://www.dictionaryofeconomics.com/article?id=pde2008_R000249 Palgrave Macmillan.
- HEDSTRÖM, P. & SWEDBERG, R. 1996. Rational choice, empirical research and the sociological tradition. *European Sociological Review*, 12(2), pp. 127-146.
- HEELAS, P. 1996. *The New Age movement*. Oxford: Blackwell.
- HEELAS, P. 1998. New age authenticity and social roles: a response to Steve Bruce. *Journal of Contemporary Religion*, 13(2), pp. 257-264.
- HEELAS, P. 2006a. Challenging secularization theory: the growth of New Age' spiritualities of life. *Hedgehog Review*, Spring & Summer, pp. 46-58.
- HEELAS, P. 2006b. The infirmity debate: on the viability of New Age spiritualities of life. *Journal of Contemporary Religion*, 21(2), pp. 223-240.
- HEELAS, P. 2007. The holistic milieu and spirituality: reflections on Voas and Bruce. *In:* FLANAGAN, K. & JUPP, P. C. (eds.) *A sociology of spirituality.* Aldershot: Ashgate, pp. 63-80.
- HEELAS, P. 2008. *Spiritualities of life: new age Romanticism and consumptive capitalism.*Malden, MA: Blackwell.
- HEELAS, P., MARTIN, D. & MORRIS, P. 1998. *Religion, modernity, and postmodernity.*Malden, MA: Blackwell.
- HEELAS, P., MORRIS, P. & LASH, S. 1996. *De-traditionalization: critical reflections on authority and identity.* Cambridge, MA: Blackwell.
- HEELAS, P., WOODHEAD, L., SEEL, B., SZERSZYNSKI, B. & TUSTING, K. 2005. *The spiritual revolution: why religion is giving way to spirituality.* Malden, MA: Blackwell
- HEILMAN, S. 2009. Ethnography behind *Defenders of the Faith. In:* PUDDEPHATT, A. J., SHAFFIR, W. & KLEINKNECHT, S. W. (eds.) *Ethnographies revisited. Constructing theory in the field.* New York, N.Y.: Routledge, pp. 197-211.
- HELLE, H. J. 1997. Introduction. *In:* SIMMEL, G. & HELLE, H. J. (ed.) *Essays on religion.* New Haven, CT: Yale University Press, pp. xi-xx.

- HENNIS, W., BRISSON, E. & BRISSON, R. 1994. The meaning of 'Wertfreiheit' on the background and motives of Max Weber's 'Postulate'. *Sociological theory,* 12(2), pp. 113-125.
- HEPWORTH, C. & STITT, S. 2007. Social capital and faith-based organisations *The Heythrop Journal*, 48(6), pp. 895-910.
- HERTZ, R. (ed.) 1997. Reflexivity & voice. Thousand Oaks, CA: Sage.
- HERVIEU-LÉGER, D. 2002. Space and religion: new approaches to religious spatiality in modernity. *International Journal of Urban and Regional Research* 26(1), pp. 99-105.
- HERVIEU-LÉGER, D. 2003. Individualism, the validation of faith, and the social nature of religion in modernity. *In:* FENN, R. K. (ed.) *The Blackwell companion to sociology of religion*. Oxford: Blackwell, pp. 161-174.
- HERVIEU-LÉGER, D. 2006. In search of certainties: the paradoxes of religiosity in societies of high modernity. *Hedgehog Review,* Spring & Summer, pp. 59-68.
- HERZFELD, M. 2009. The cultural politics of gesture: reflections on the embodiment of ethnographic practice. *Ethnography*, 10(2), pp. 131-152.
- HOBSBAWM, E. J. 1987. The age of empire, 1875-1914. London: Weidenfeld and Nicolson.
- HOBSBAWM, E. 1994. *Age of extremes: the short twentieth century 1914-1991.* London: Abacus.
- HOBSBAWM, E. J. 1995. *The age of capital, 1848-1875.* London: Weidenfeld & Nicolson.
- HOLLANDER, P. 2010. Why the celebrity cult? *Society*, 47(5), pp. 388-391
- HORNSEY, M. J. & JOLANDA, J. 2004. The individual within the group: balancing the need to belong with the need to be different. *Personality and Social Psychology Review*, 8(3), pp. 248-264.
- HOUTMAN, D. & AUPERS, S. 2007. The spiritual turn and the decline of tradition: the spread of post-Christian spirituality in 14 Western countries, 1981-2000. *Journal for the Scientific Study of Religion*, 46(3), pp. 305-320.
- HOUTMAN, D., AUPERS, S. & HEELAS, P. 2009. Christian religiosity and new age spirituality: a cross-cultural comparison. *Journal for the Scientific Study of Religion*, 48(1), pp. 169-178.
- HULSETHER, M. 1995. Interpreting the 'popular' in popular religion. *American Studies,* 36(2), pp. 127-137.

- HUNSBERGER, B. & JACKSON, L. M. 2005. Religion, meaning, and prejudice. *Journal of Social Issues*, 61(4), pp. 807-826.
- HUNT, S. 2002. The 'health and wealth' gospel in the UK: variations on a theme. *Culture and Religion*, 3(1), pp. 89-104.
- HUNT, S. 2002. Religion in Western society. New York, N.Y.: Palgrave.
- HUNTER, J. D. 1985. Conservative Protestantism on the American scene. *Social Compass*, 32(2-3), pp. 233-243
- HUNTER, J. D. 1987. *Evangelicalism: the coming generation.* Chicago, IL: University of Chicago Press.
- IANNACCONE, L. R. 1990. Religious practice: a human capital approach. *Journal for the Scientific Study of Religion*, 29(3), pp. 297-314.
- IANNACCONE, L. R. 1991. The consequences of religious market structure. Adam Smith and the economics of religion. *Rationality and Society*, 3(2), pp. 156-177.
- IANNACCONE, L. R. 1994. Why strict churches are strong. *American Journal of Sociology*, 99(5), pp. 1180-1211.
- IANNACCONE, L. R. 1995a. Voodoo economics? Reviewing the rational choice approach to religion. *Journal for the Scientific Study of Religion*, 34(1), pp. 76-89.
- IANNACCONE, L. R. 1995b. Second thoughts: a response to Chaves, Demerath, and Ellison. *Journal for the Scientific Study of Religion*, 34(1), pp. 113-120.
- IANNACCONE, L. R. 1997a. Rational choice: framework for the scientific study of religion. *In:* Young, L. A. (ed) *Rational choice theory and religion: summary and assessment.*New York, N.Y.: Routledge, pp. 25-44.
- IANNACCONE, L. R. 1997b. Skewness explained: a rational choice model of religious giving. *Journal for the Scientific Study of Religion*, 36(2), pp. 141-157.
- IANNACCONE, L. R. 1998. Introduction to the economics of religion. *Journal of Economic Literature*, 36, pp. 1465–1496.
- IANNACCONE, L. R. & KLICK, J. 2003. Spiritual capital: an introduction and literature review. *Templeton Foundation,* [Online]. Available: www.spiritualcapitalresearchprogram.com/pdf/review.pdf

- IANNACCONE, L. R., STARK, R. & FINKE, R. 1997. Deregulating religion: the economics of church and state. *Economic Inquiry* 35(2), pp. 350-364.
- IMBACH, R. 1996. *Dante, la philosophie et les laïcs. I : initiations à la philosophie médiévale.*Paris: Cerf.
- INGERSOLL, J. 2002. Against univocality. Re-reading ethnographies of conservative Protestant women. *In:* SPICKARD, J. V., LANDRES, J. S. & MCGUIRE, M. B. (eds.) *Personal knowledge and beyond: reshaping the ethnography of religion.* New York, N.Y.: New York University Press, pp. 162-174.
- INGLEHART, R. 1981. Post-materialism in an environment of insecurity. *The American Political Science Review* 75(4), pp. 880-900.
- INGLEHART, R. 2008. Changing values among Western publics from 1970 to 2006. *West European Politics*, 31(1-2), pp. 130 146.
- INGLEHART, R. & BAKER, W. E. 2000. Modernization, cultural change, and the persistence of traditional values. *American Sociological Review*, 65(1), pp. 19-51.
- INSOLE, C. J. 2004. Against radical orthodoxy: the dangers of overcoming political liberalism. *Modern Theology*, 20(2), pp. 213–241.
- ISRAEL, M. & HERSH, D. 2006. Ethics. *In:* GILBERT, G. N. (ed.) *From postgraduate to social scientist: a guide to key skills.* London: Sage, pp. 43-58.
- JACOBS, J. L. 2002. Transitional identities: self, other, and the ethnographic process. *In:* SPICKARD, J. V., LANDRES, J. S. & MCGUIRE, M. B. (eds.) *Personal knowledge and beyond: reshaping the ethnography of religion.* New York, N.Y.: New York University Press, pp. 88-99.
- JAMES, W. 1902/1960. The varieties of religious experience: a study in human nature, Collins.
- JENKINS, R. 2000. Disenchantment, enchantment and re-enchantment: Max Weber at the Millennium. *Max Weber studies,* 1(1), pp. 11-32.
- JENSEN, J. S. 2011. Revisiting the insider-outsider debate: dismantling a pseudo-problem in the study of religion. *Method and Theory in the Study of Religion*, 23(1), pp. 29-47.
- JEROLMACK, C. & PORPORA, D. 2004. Religion, rationality, and experience: a response to the new rational choice theory of religion. *Sociological Theory*, 22(1), pp. 140-160.

- JINDRA, M. 1994. Star Trek fandom as a religious phenomenon. *Sociology of Religion*, 55(1), pp. 27-51.
- JINDRA, M. 2005. It's about faith in our future. Star Trek fandom as cultural religion. *In:*FORBES, B. D. & MAHAN, J. H. (eds.) *Religion and popular culture in America.* 2nd ed.
 Berkeley, CA: University of California Press, pp. 165-179.
- JOAS, H. 2000. The genesis of values. Cambridge: Polity Press.
- JOAS, H. 2000. Social theory and the sacred: a response to John Milbank. *Ethical Perspectives*, 7(4), pp. 233-243.
- JOAS, H. 2001. Values versus norms: a pragmatist account of moral objectivity. *The Hedgehog Review*, 3(3), pp. 42-56.
- JOAS, H. 2004. The changing role of the social sciences. *International Sociology*, 19(3), pp. 301-313.
- JOAS, H. 2004. Morality in an age of contingency. *Acta Sociologica*, 47(4), pp. 392-399.
- JOAS, H. 2008. *Do we need religion? On the experience of self-transcendence.* London: Paradigm.
- JOAS, H. 2009. The emergence of universalism: an affirmative genealogy. *In:* HEDSTRÖM, P. & WITTROCK, B. (eds.) *Frontiers of sociology.* The Annals of the International Institute of Sociology, pp. 15-24.
- JORGENSEN, G. 2006. Kohlberg and Gilligan: duet or duel? *Journal of Moral Education*, 35(2), pp. 179-196.
- KEANE, J. 1984. Public life and late capitalism. Cambridge: Cambridge University Press.
- KAERN, M. 1990. The world as human construction. *In:* KAERN, M., PHILLIPS, B. S. & COHEN, R. S. (eds.) *Georg Simmel and contemporary sociology.* Dordrecht: Kluwer Academic Publishers, pp. 75-98.
- KAERN, M., PHILLIPS, B. S. & COHEN, R. S. 1990. *Georg Simmel and contemporary sociology.*Dordrecht: Kluwer Academic Publishers.
- KEMPLE, T. 2007. Introduction Allosociality: bridges and doors to Simmel's social theory of the limit. *Theory, Culture & Society*, 24(7/8), pp. 1-19.
- KENNEDY, D. 2004. The disenchantment of logically formal legal rationality, or Max Weber's sociology in the genealogy of the contemporary mode of Western legal thought. *Hastings Law Journal*, 55, pp. 1031-1076.

- KENT, G. 2000a. Ethical principles. *In:* BURTON, D. (ed.) *Research training for social scientists.* London: Sage, pp. 61-67.
- KENT, G. 2000b. Informed consent. *In:* BURTON, D. (ed.) *Research training for social scientists.* London: Sage, pp. 81-96.
- KING, U. 2001. Introduction: spirituality, society and the new millennium wasteland, wilderness or new vision? *In:* KING, U. (ed.) *Spirituality and society in the new millennium.* Brighton: Sussex Academic Press, pp. 1-13.
- KISER, E. & HECHTER, M. 1998. The debate on historical sociology: rational choice theory and its critics. *The American Journal of Sociology*, 104(3), pp. 785-816.
- KITCHEN, G. 1999. Charles Taylor: the malaises of modernity and the moral sources of the self. *Philosophical Social Criticism*, 25(3), pp. 29-55.
- KNIPPENBERG, H. 2010. Secular Europe versus religious America: explaining the gap. *Prešov,* 15, pp. 42-54.
- KNOTT, K. 2005. Insider/outsider perspectives. *In:* HINNELLS, J. R. (ed.) *The Routledge companion to the study of religion.* New York, N.Y.: Routledge, pp. 243-258.
- KNUDSEN, B. T. & WAADE, A. M. (eds.) 2010. *Re-investing authenticity: tourism, place and emotions.* Bristol: Channel View.
- KOHLBERG, L. 1973. The claim to moral adequacy of a highest stage of moral judgment. *The Journal of Philosophy*, 70(18), pp. 630–646.
- KÖHNKE, K. C. 1990. Four concepts of social science at Berlin University: Dilthey, Lazarus, Schmoller and Simmel. *In:* KAERN, M., PHILLIPS, B. S. & COHEN, R. S. (eds.) *Georg Simmel and contemporary sociology.* Dordrecht: Kluwer Academic Publishers, pp. 99-107.
- KUSNETZ, D. 2012. Expanding the definition of kosher. *Colgate Academic Review*, 5(3), pp. 4-11.
- KVALE, S. 1996. *Interviews: an introduction to qualitative research interviewing.* Thousand Oaks, CA: Sage.
- LABANOW, C. E. 2009. Evangelicalism and the emerging church. Farnham: Ashgate.
- LACTANTIUS. 302-311. *Divinarum Institutionum*. [Online]. Available at: http://www.documentacatholicaomnia.eu/02m/0240-0320, Lactantius, Divinarum_Institutionum_Liber_IV,_MLT.pdf

- LAERMANS, R. 2006. The ambivalence of religiosity and religion: a reading of Georg Simmel. *Social Compass*, 53(4), pp. 479-489.
- LAMINE, A.-S. 2008. Croyances et transcendances: variations en modes mineurs. *Social Compass*, 55(2), pp. 154-167.
- LAMINE, A.-S. 2010. Les croyances religieuses: entre raison, symbolisation et expérience. L'Ann'ee sociologique, 60(1), pp. 93-114.
- LANDRES, J. S. 2002. Being (in) the field: defining ethnography in southern California and central Slovakia. *In:* SPICKARD, J. V., LANDRES, J. S. & MCGUIRE, M. B. (eds.) *Personal knowledge and beyond: reshaping the ethnography of religion.* New York, N.Y.: New York University Press, pp. 100-112.
- LANE, L. & POWER, A. 2009. Soup runs in central London: 'the right help in the right place at the right time?' London: LSE Housing Report, Centre for Analysis of Social Exclusion.
- LARMORE, C. 2004. Alessandro Ferrara's theory of authenticity. *Philosophy Social Criticism*, 30(5), pp. 5-9.
- LARSEN, T. & TREIER, D. J. 2007. *The Cambridge companion to evangelical theology.*Cambridge: Cambridge University Press.
- LASCH, C. 1979. The culture of narcissism: American life in an age of diminishing expectations. New York, N.Y.: Norton.
- LASH, S. 1996. Tradition and the limits of difference. *In:* HEELAS, P., LASH, S. & MORRIS, P. (eds.) *Detraditionalization. Critical reflections on authority and identity.* Oxford: Blackwell Publishers, pp. 250-274.
- LASH, S. 2006. Life (Vitalism). *Theory Culture Society*, 23, pp. 323-329.
- LAWLESS, E. J. 1992. "I was afraid someone like you ... an outsider ... would misunderstand"

 Negotiating interpretive differences between ethnographers and subjects. *Journal of American Folklore*, 105, pp. 302-313.
- LAWRENCE, P. A. 1976. *Georg Simmel: sociologist and European.* Sunbury-on-Thames: Nelson.
- LAWSON, M. P. 1997. Struggles for mutual reverence. Social strategies and religious stories. *In:* EDGELL, P. & EIESLAND, N. L. (eds.) *Contemporary American religion: an ethnographic reader.* Walnut Creek, CA: AltaMira Press, pp. 51-77.

- LECHNER, F. J. 1990. Social differentiation and modernity: on Simmel's macrosociology. *In:* KAERN, M., PHILLIPS, B. S. & COHEN, R. S. (eds.) *Georg Simmel and contemporary sociology.* Dordrecht: Kluwer Academic Publishers, pp. 155-179.
- LECHNER, F. J. 2006. Rational choice and religious economies. *In:* BECKFORD, J. A. & DEMERATH, N. J. (eds.) *SAGE handbook of the sociology of religion.* pp. 81-98.
- LEE, M. & SILVER, D. 2012. Simmel's law of the individual and the ethics of the relational self. *Theory, Culture & Society*, 29(7/8), pp. 124-145.
- LEHMANN, D. 2010. An idea, a tribe, and their critics: rational choice and the sociology of religion. *In:* TURNER, B. (ed.) *The new Blackwell companion to the sociology of religion.* Oxford: Blackwell Publishing, pp. 181-200.
- LEHTONEN, T.-K. & PYYHTINEN, O. 2008. On Simmel's conception of philosophy. *Continental Philosophical Review*, 41, pp. 301-322.
- LEVINE, D. N. (ed.) 1971. *On individuality and social forms: selected writings [of] Georg Simmel.* Chicago, IL: University of Chicago Press.
- LEVINE, D. N. 1991. Simmel as educator: on individuality and modern culture. *Theory, Culture & Society,* 8(3), pp. 99-117.
- LEVINE, D. N. 2010. Introduction. *In:* SIMMEL, G. and LEVINE, D. N. (ed.) *The view of life.*Four metaphysical essays with journal aphorisms. Chicago, IL: University of Chicago, pp. ix-xxxii.
- LEVINE, D. N. 2012. *Soziologie* and *Lebensanschauung*: two approaches to synthesizing 'Kant' and 'Goethe' in Simmel's work. *Theory, Culture & Society,* 29(7/8), pp. 26-52.
- LICHTBLAU, K. 1991. Causality or interaction? Simmel, Weber and interpretive sociology. *Theory, Culture & Society*, 8(3), pp. 33-62.
- LICHTERMAN, P. 2006. Social capital or group style? Rescuing Tocqueville's insights on civic engagement. *Theory and society.* 35(5-6), pp. 529-563.
- LICHTERMAN, P. 2009. How religion circulates in American's local public square *In:*LICHTERMAN, P. & POTTS, C. B. (eds.) *The civic life of American religion.* Stanford,
 CA: Stanford University Press, pp. 100-122.
- LICHTERMAN, P. & POTTS, C. B. 2009. The civic life of American religion: an introductory statement. *In:* LICHTERMAN, P. & POTTS, C. B. (eds.) *The civic life of American religion.* Stanford, CA: Stanford University Press, pp. 1-22.

- LINKOGLE, S. & LEE-TREWEEK, G. 2000. *Danger in the field: risk and ethics in social research.* London: Routledge.
- LINNEKIN, J. 1991. Cultural invention and the dilemma of authenticity. *American Anthropologist*, 93(2), pp. 446-449.
- LIVEZEY, L. W. 2000. The new context of urban religion. *In:* LIVEZEY, L. W. (ed.) *Public religion and urban transformation. Faith in the city.* New York, N.Y.: New York University Press, pp. 3-25.
- LIVEZEY, L. W. 2002. Epilogue: the ethnographer and the quest for meanings. *In:* ARWECK, E. & STRINGER, M. D. (eds.) *Theorizing faith. The insider/outsider problem in the study of ritual.* Birmingham: University of Birmingham Press, pp. 155-165.
- LOW, J. 2008. Structure, agency, and social reality in Blumerian symbolic interactionism: the influence of Georg Simmel. *Symbolic Interaction*, 31(3), pp. 325-343.
- LUCKMANN, T. 1967. The invisible religion, New York, N.Y.: Macmillan.
- LUCKMANN, T. 1990. Shrinking transcendence, expanding religion? *Sociology of Religion.* 51(2), pp. 127-138.
- LUCKMANN, T. 2003. Transformations of religion and morality in modern Europe. *Social Compass*, 50(3), pp. 275-285
- LUND, S., FARNELL, R., FURBEY, R., LAWLESS, P., WISHART, B. & ELSE, P. 1995.

 Empowering communities: The experience of the church urban fund. *Local Economy*10(3), pp. 218 228
- LYNCH, G. 2007. *The new spirituality: an introduction to progressive belief in the twenty-first century.* London: I. B. Tauris.
- LYNCH, G. 2012. *On the Sacred.* London: Acumen.
- LYNCH, M. 2000. Against reflexivity as an academic virtue and source of privileged knowledge. *Theory, Culture & Society,* 17(3), pp. 26-54.
- LYON, D. 2000. *Jesus in Disneyland: religion in postmodern times.* Cambridge: Polity Press in association with Blackwell Publishers.
- MACHACEK, D. W. 2003. The problem of pluralism. *Sociology of Religion*, 64(2), pp. 145-161.
- MACHAMER, P. 2009. Philosophy of science and the study of religion. *Religion*, 39, pp. 356-360.

- MACINTYRE, A. C. 1981. *After virtue: a study in moral theory.* London: Duckworth.
- MACINTYRE, A. C. 1999. Is understanding religion compatible with believing? *In:*MCCUTCHEON, R. T. (ed.) *The insider/outsider problem in the study of religion: a reader.* London: Cassell, pp. 37-49.
- MACINTYRE, A. C. 2007. Prologue. *After Virtue* after a quarter of a century. *In:* MACINTYRE, A. C. *After virtue: a study in moral theory.* 3rd ed. London: Duckworth, pp. ix-xvi.
- MACKINNON, M. H. 2001. Max Weber's disenchantment. *Journal of Classical Sociology*, 1(3), pp. 329-351.
- MACLEAN, M. A., WALKER, L. J. & MATSUBA, M. K. 2004. Transcendence and the moral self: identity integration, religion, and moral life. *Journal for the Scientific Study of Religion*, 43(3), pp. 429-437.
- MARTIN, D. 2005. *On secularization: towards a revised general theory.* Aldershot: Ashgate.
- MARTINELLI, M. 2012. Idea y experiencia de la libertad. Algunas consideraciones sobre el pensamiento de Georg Simmel. *Sociológica*, 27(76), pp. 89-114.
- MARWICK, A. 2005. The cultural revolution of the long sixties: voices of reaction, protest, and permeation. *The International History Review*, 27(4), pp. 709-944.
- MARX, K., ENGELS, F. & STEDMAN JONES, G. 2002. *The communist manifesto.* London: Penguin.
- MARX, K. 1883/1976. *Capital: a critique of political economy / Vol.1.* Harmondsworth: Penguin.
- MARX, K. & RAINES, J. C. 2002. *Marx on religion*. Philadelphia, Penn.: Temple University Press.
- MASON, J. 1996. *Planning and designing qualitative research*. London: Sage.
- MAUTHNER, N. S. & DOUCET, A. 2003. Reflexive accounts and accounts of reflexivity in qualitative data analysis. *Sociology*, 37(3), pp. 413-431.
- MAXWELL, J. A. 1992/2002. Understanding and validity in qualitative research. *In:*HUBERMAN, A. M. & MILES, M. B. (eds.) *The qualitative researcher's companion.*Thousand Oaks, CA: Sage, pp. 37-64.
- MAXWELL, J. A. 1996. *Qualitative research design: an interactive approach.* Thousand Oaks: CA, Sage.

- MAXWELL, J. A. 2004. Causal explanation, qualitative research, and scientific inquiry in education. *Educational Researcher*, 33(2), pp. 3-11.
- MAY, T. 1999. Reflexivity and sociological practice. *Sociological Research Online*, 4(3), [Online]. Available: http://www.socresonline.org.uk/4/3/may.html.
- MCCARTHY BROWN, K. 2002. Writing about "the Other," revisited. *In:* SPICKARD, J. V., LANDRES, J. S. & MCGUIRE, M. B. (eds.) *Personal knowledge and beyond: reshaping the ethnography of religion.* New York, N.Y.: New York University Press, pp. 127-133.
- MCCOLE, J. 2005. Georg Simmel and the philosophy of religion. *New German Critique*, 94, pp. 8-35.
- MCCONKEY, D. 2001. Whither Hunter's culture war? Shifts in evangelical morality, 1988-1998. *Sociology of Religion*, 62(2), pp. 149-174.
- MCCUTCHEON, R. T. (ed.) 1999. *The insider/outsider problem in the study of religion: a reader.* London: Cassell.
- MCCUTCHEON, R. T. 2001. The jargon of authenticity and the study of religion. *Religion & Theology*, 8(3-4), pp. 229-252.
- MCGUIRE, M. B. 2002. New-old directions in the social scientific study of religion.

 Ethnography, phenomenology, and the human body. *In:* SPICKARD, J. V., LANDRES, J. S. & MCGUIRE, M. B. (eds.) *Personal knowledge and beyond: reshaping the ethnography of religion.* New York, N.Y.: New York University Press, pp. 195-211.
- MCGUIRE, M. B. 2008. *Lived religion: faith and practice in everyday life.* Oxford: Oxford University Press.
- MCINTOSH, D. N. 1997. Religion-as-schema, with implications for the relation between religion and coping *In:* SPILKA, B. & MCINTOSH, D. N. (eds.) *The psychology of religion: theoretical approaches.* Boulder, Co.: Westview Press, pp. 171-183.
- MCKINNON, A. 2010. Elective affinities of the Protestant ethic: Weber and the chemistry of capitalism. *Sociological Theory*, 28(1), pp. 108-126.
- MCLEOD, H. 2007. *The religious crisis of the 1960s.* Oxford: Oxford University Press.
- MELIA, K. M. 1997. Producing 'plausible stories': interviewing student nurses. *In:* MILLER, G. & DINGWALL, R. (eds.) *Context and method in qualitative research.* London: Sage, pp. 26-36.

- MERTON, R. K. 1972. Insiders and outsiders: a chapter in the sociology of knowledge. *American Journal of Sociology*, 78(1), pp. 9-47.
- MESTROVIC, S. 1990. Simmel's sociology in relation to Schopenhauer's philosophy. *In:* KAERN, M., PHILLIPS, B. S. & COHEN, R. S. (eds.) *Georg Simmel and contemporary sociology.* Dordrecht: Kluwer Academic Publishers, pp. 182-197.
- MILÀ, N. C. 2005. *A sociological theory of value: Georg Simmel's sociological relationism.*Bielefeld: transcript Verlag.
- MILLER, D. E. 1997. *Reinventing American Protestantism. Christianity in the new millennium.*Berkeley, CA: University of California Press.
- MILLER, G. & DINGWALL, R. 1997. *Context and method in qualitative research.* London: Sage.
- MILLER, V. J. 2004. *Consuming religion: Christian faith and practice in a consumer culture.*New York, N.Y.: Continuum.
- MITCHELL, J. P. 1997. A moment with Christ: the importance of feelings in the analysis of belief. *The Journal of the Royal Anthropological Institute,* 3(1), pp. 79-94.
- MOL, H. 1976. *Identity and the sacred : a sketch for a new social-scientific theory of religion.*Oxford: Blackwell.
- MONROE, K. R. 1994. A fat lady in a corset: altruism and social theory. *American Journal of Political Science* 38(4), pp. 861-93.
- MONTEMAGGI, F. E. S. 2010. Misunderstanding faith: when 'capital' does not fit the 'spiritual'. *The International Journal of Interdisciplinary Social Sciences,* 5(5), pp. 179-192.
- MONTEMAGGI, F. E. S. 2011. The enchanting dream of "spiritual capital". *Implicit Religion*, 14(1), pp. 67-86.
- MONTEMAGGI, F. E. S. 2013. Shopping for a church? Choice and commitment in religious behavior. *In:* GAUTHIER, F. & MARTIKAINEN, T. (eds.) *Religion in Consumer Society.* London: Ashgate, pp. 109-125.
- MOODY, M. 2002. Caring for creation: environmental advocacy by mainline protestant organizations. *In:* WUTHNOW, R. & EVANS, J. H. (eds.) *The quiet hand of God. Faithbased activism and the public role of mainline Protestantism.* Berkeley, CA: University of California Press, pp. 237-264.

- MOON, D. 2009. A tacit sense of evil? The meaning of politics in United Methodist debates over homosexuality. *In:* LICHTERMAN, P. & POTTS, C. B. (eds.) *The civic life of American religion.* Stanford, CA: Stanford University Press, pp. 122-139.
- MOORE, L. 1994. *Selling God. American religion in the marketplace of culture.* Oxford: Oxford University Press.
- MORRIS-REICH, A. 2004. From autonomous subject to free individual in Simmel and Lacan. *History of European Ideas*, 31, pp. 103-127.
- MORSE, J. M. & RICHARDS, L. 2002. *Readme first for a reader's guide to qualitative methods.*Thousand Oaks, CA: Sage.
- MOUFFE, C. 1993. The return of the political. London: Verso.
- MOUFFE, C. 1994. Political liberalism. Neutrality and the political. *Ratio Juris*, 7(3), pp. 314-324.
- MOUFFE, C. 1999. Deliberative democracy or agonistic pluralism? *Social Research*, 66(3), pp. 745-758.
- MOUFFE, C. 2005. The limits of John Rawls's pluralism. *Politics, Philosophy & Economics,* 4(2), pp. 221-231.
- MUESSE, M. W. 1999. Religious studies and 'Heaven's Gate': making the strange familiar and the familiar strange. *In:* MCCUTCHEON, R. T. (ed.) *The insider/outsider problem in the study of religion: a reader.* London: Cassell, pp. 390-394.
- MULLIN, A. 1996. Purity and pollution: resisting the rehabilitation of a virtue. *Journal of the History of Ideas*, 57(3), pp. 509-524.
- MURPHY, E. & DINGWALL, R. 2001. The ethics of ethnography. *In:* ATKINSON, P., COFFEY, A. & DELAMONT, S. (eds.) *Handbook of ethnography.* London: Sage, pp. 339-351.
- MYERS, I. B., KIRBY, L. K. & MYERS, K. D. 2000. *Introduction to type: a guide to understanding your results on the Myers-Briggs Type Indicator*. Oxford: Oxford Psychologists Press.
- MYERS, I. B. & MYERS, P. B. 1995. *Gifts differing: understanding personality type.* Palo Alto, CA: Davies-Black Pub.
- NASH, M. & STEWART, B. 2002. *Spirituality and social care: contributing to personal and community well-being.* London: Jessica Kingsley Publishers.

- NEDELMANN, B. 1990. On the concept of 'elrleben' in Georg Simmel's sociology. *In:* KAERN, M., PHILLIPS, B. S. & COHEN, R. S. (eds.) *Georg Simmel and contemporary sociology.*Dordrecht: Kluwer Academic Publishers, pp. 182-197.
- NEEDHAM, R. 1972. Belief, language, and experience. Oxford: Blackwell.
- NEITZ, M. J. 2002. Walking between the worlds: permeable boundaries, ambiguous identities. *In:* SPICKARD, J. V., LANDRES, J. S. & MCGUIRE, M. B. (eds.) *Personal knowledge and beyond: reshaping the ethnography of religion.* New York, N.Y.: New York University Press, pp. 47-60.
- NEITZ, M. J. & SPICKARD, J. V. 1990. Steps toward a sociology of religious experience: the theories of Mihaly Csikszentmihalyi and Alfred Schutz. *Sociological Analysis*, 51(1), pp. 15-33.
- NOLL, M. A. 1995. *The Scandal of the evangelical mind.* Grand Rapids, MI: W.B. Eerdmans.
- OAKES, G. 1980. Introduction. *In:* OAKES, G. (ed.) *Essays on interpretation in social science.*Manchester: Manchester University Press, pp. 3-94.
- OAKES, G. (ed.) 1984. *Georg Simmel: on women, sexuality, and love.* New Haven, CT: Yale University Press.
- O'GUINN, T. C. 1991. Touching greatness: the central midwest Barry Manilow fan club. *In:* BELK, R. W. (ed.) *Highways and buyways: naturalistic research from the consumer behavior odyssey.* Provo, UT: Association for Consumer Research, pp. 102-111.
- O'GUINN, T. C. & BELK, R. W. 1989. Heaven on earth: consumption at Heritage Village, USA. *Journal of Consumer Research*, 16(2), pp. 227-238.
- OH, S. & SARKISIAN, N. 2012. Spiritual individualism or engaged spirituality? Social implications of holistic spirituality among mind-body-spirit practitioners. *Sociology of Religion*, 73(3), pp. 299-322.
- OLSON, D. V. A. & PERL, P. 2001. Variations in strictness and religious commitment within and among five denominations. *Journal for the Scientific Study of Religion,* 40(4), pp. 757-764.
- ORSI, R. A. 1997. Everyday miracles: the study of lived religion. *In:* HALL, D. (ed.) *Lived religion in America: toward a history of practice.* Princeton, N.J.: Princeton University Press, pp. 3-21.

- ORSI, R. A. 2003. Is the study of lived religion irrelevant to the world we live in? *Journal for the Scientific Study of Religion*, 42(2), pp. 169–174.
- ORSI, R. A. 2005. *Between heaven and earth. The religious worlds people make and the scholars who study them.* Princeton, N.J.: Princeton University Press.
- OTTO, R. 1917/1924. The idea of the holy: an inquiry into the non-rational factor in the idea of the divine and its relation to the rational. London: Humphrey Milford, Oxford University Press.
- OWSLEY, R. & BACKHAUS, G. 2003. Simmel's four components of historical science. *Human Studies*, 26, pp. 209-222.
- PAGNUCCO, R. 1996. A comparison of the political behavior of faith-based and secular peace groups. *In:* SMITH, C. (ed.) *Disruptive religion. The force of faith in social-movement activism.* New York, N.Y.: Routledge, pp. 205-222.
- PALLY, M. 2011. *The new evangelicals: expanding the vision of the common good.* Grand Rapids, MI: W.B. Eerdmans Publishing Company.
- PALMISANO, S. 2009. Reconstructors: reinventing the spiritual path within Italian Catholicism. *Fieldwork in Religion*, 4(1), pp. 29-45.
- PALMISANO, S. 2013. The paradoxes of new monasticism in consumer society. *In:*GAUTHIER, F. & MARTIKAINEN, T. (eds.) *Religion in Consumer Society.* London:
 Ashgate, pp. 75-90.
- PALOUTZIAN, R. F. 1996. *Invitation to the psychology of religion*. Boston: Allyn and Bacon.
- PALOUTZIAN, R. F. 2005. Religious conversion and spiritual transformation: a meaning-system analysis. *In:* PALOUTZIAN, R. F. & PARK, C. L. (eds.) *Handbook of the psychology of religion and spirituality.* New York, N.Y.: Guilford Press, pp. 331-347.
- PALOUTZIAN, R. F. & PARK, C. L. 2005. *Handbook of the psychology of religion and spirituality.* New York, N.Y.: Guilford Press.
- PALOUTZIAN, R. F., RICHARDSON, J. T. & RAMBO, L. R. 1999. Religious conversion and personality change. *Journal of Personality*, 67(6), pp. 1047-1079.
- PALS, D. L. 1999. Reductionism and belief: an appraisal of recent attacks on the doctrine of irreducible religion. *In:* MCCUTCHEON, R. T. (ed.) *The insider/outsider problem in the study of religion: a reader.* London: Cassell, pp. 179-196.

- PAPILLOUD, C. 2000. Georg Simmel. La dimension sociologique de la Wechselwirkung. *Revue européenne des sciences sociales*, 38(119), pp. 103-129.
- PAPILLOUD, C. & ROL, C. 2004. Compromise, social justice and resistance: an introduction to the political sociology of Georg Simmel. *Social Science Information*, 43(2), pp. 205-230.
- PARGAMENT, K. I. 1997. *The psychology of religion and coping: theory, research, practice.*New York, N.Y.: Guilford Press.
- PARGAMENT, K. I. 2007. *Spiritually integrated psychotherapy. Understanding and addressing the sacred.* New York, N.Y.: Guildford Press.
- PARGAMENT, K. I. 2008. The sacred character of community life. *American Journal of Community Psychology*, 41, pp. 22-34.
- PARGAMENT, K. I., MAGYAR, G. M., BENORE, E. & MAHONEY, A. 2005b. Sacrilege: a study of sacred loss and desecration and their implications for health and well-being in a community sample. *Journal for the Scientific Study of Religion*, 44(1), pp. 59–78.
- PARGAMENT, K. I., MAGYAR-RUSSELL, G. M. & MURRAY-SWANK, N. A. 2005a. The sacred and the search for significance: religion as a unique process. *Journal of Social Issues*, 61(4), pp. 665-687.
- PARGAMENT, K. I. & MAHONEY, A. 2002. Spirituality: discovering and conserving the sacred. *In:* SNYDER, C. R. & LOPEZ, S. J. (eds.) *Handbook of positive psychology.* New York, N.Y.: Oxford University Press, pp. 646-659.
- PARGAMENT, K. I., SULLIVAN, M. S., BALZER, W. K., VAN HAITSMA, K. S. & RAYMARK, P. H. 1995. The many meanings of religiousness: a policy-capturing approach. *Journal of Personality*, 63(4), pp. 953-983.
- PARISH, S. M. 2009. Are we condemned to authenticity? *Ethos*, 37(1), pp. 139-153.
- PARK, C. L. 2005. Religion and meaning. *In:* PALOUTZIAN, R. F. & PARK, C. L. (eds.) *Handbook of the psychology of religion and spirituality.* New York, N.Y.: Guilford Press, pp. 295-314.
- PARK, C. L. 2005. Religion as a meaning-making framework in coping with life stress *Journal of Social Issues*, 61(4), pp. 707-729.

- PARK, J. Z. & SMITH, C. 2000. "To whom much has been given...": religious capital and community voluntarism among churchgoing protestants. *Journal for the Scientific Study of Religion*, 39(3), pp. 272-286.
- PARSONS, T. 1965. Introduction. *In:* WEBER, M. & PARSONS, T. (ed.) *The sociology of religion.* London: Methuen, pp. xix-lxvii.
- PEARMAIN, C. 2007. *Authenticity: an ethic of capacity realisation.* Ph.D. Thesis. University of Warwick.
- PEARSON, J. 2002. 'Going native in reverse': the insider as researcher in British Wicca. *In:*ARWECK, E. & STRINGER, M. D. (eds.) *Theorizing faith. The insider/outsider problem in the study of ritual.* Birmingham: University of Birmingham Press, pp. 57-113.
- PESHKIN, A. 1984. Odd man out: the participant observer in an absolutist setting. *Sociology of Education*, 57(4), pp. 254-264.
- PESHKIN, A. 1986. *God's choice: the total world of a fundamentalist Christian school.* Chicago, IL: University of Chicago Press.
- PESSI, A. B. 2013. Privatized religiosity revisited: building an authenticity model of individual-church relations. *Social Compass*, 60(1), pp. 3-21.
- PESSI, A. B. & JELDTOFT, N. 2012. Authenticity matters: reflections on Lutheran and Muslim perspectives of individual religiosity in post-secularity. *In:* NYNÄS, P., LASSANDER, M. & UTRIAINEN, T. (eds.) *Post-secular society.* New Brunswick: Transactions Publishers, pp. 157-186.
- PETERSON, R. A. 2005. In search of authenticity. *Journal of Management Studies*, 42(5), pp. 1083-1098.
- PETERSON, R. A. & N., A. 2004. The production of culture perspective. *Annual Review of Sociology* 303, pp. 11-34.
- PHILLIPS, B. S. 1990. Simmel, individuality, and fundamental change. *In:* KAERN, M., PHILLIPS, B. S. & COHEN, R. S. (ed.) *Georg Simmel and contemporary sociology.*Dordrecht: Kluwer Academic Publishers, pp. 259-281.
- PIKE, K. L. 1999. Etic and emic standpoints for the description of behavior. *In:*MCCUTCHEON, R. T. (ed.) *The insider/outsider problem in the study of religion: a reader.* London: Cassell, pp. 28-36.

- PODOKSIK, E. 2010. Georg Simmel: three forms of individualism and historical understanding. *New German Critique*, 37(1), pp. 119-145.
- PODOKSIK, E. 2012. In search of unity: Georg Simmel on Italian cities as works of art. *Theory, Culture & Society,* 29(7-8), pp. 101-123.
- POLOMA, M. M. 1982. Toward a Christian sociological perspective: religious values, theory and methodology. *Sociological Analysis*, 43(2), pp. 95-108.
- PORPORA, D. 2006. Methodological atheism, methodological agnosticism and religious experience. *Journal for the Theory of Social Behaviour*, 36(1), pp. 57-75.
- PORTER, R. (ed.) 1997. *Rewriting the self. Histories from the Renaissance to the present.*London: Routledge.
- PUDDEPHATT, A. J., SHAFFIR, W. & KLEINKNECHT, S. W. (eds.) 2009. *Ethnographies revisited. Constructing theory in the field.* New York, N.Y.: Routledge.
- PUTNAM, R. D. 1995. Tuning in, tuning out: the strange disappearance of social capital in America *PS: Political Science and Politics.* 28(4), pp. 664-683
- PUTNAM, R. D. 2000. *Bowling alone: the collapse and revival of American community.* New York, N.Y.: Simon & Schuster.
- PUTNAM, R. D. 2002. *Democracies in flux: the evolution of social capital in contemporary society.* Oxford: Oxford University Press.
- PUTNAM, R. D., CAMPBELL, D. E. & GARRETT, S. R. 2010. *American grace: how religion divides and unites us.* New York, N.Y.: Simon & Schuster.
- PUTNAM, R. D., LEONARDI, R. & NANETTI, R. Y. 1993. *Making democracy work: civic traditions in modern Italy.* Princeton, N.J. Princeton University Press.
- PYYHTINEN, O. 2008a. Ambiguous individuality: Georg Simmel on the 'who' and the 'what' of the individual. *Human Studies*, 31, pp. 279-298.
- PYYHTINEN, O. 2008b. *Bringing the social alive. Essays on Georg Simmel's social theory.*Ph.D. Thesis. University of Turku, Finland.
- PYYHTINEN, O. 2009. Being-with: Georg Simmel's sociology of association. *Theory Culture Society*, 26(5), pp. 108-128.
- PYYHTINEN, O. 2009. From metaphysics as dogma to metaphysics as life. *Process Studies*, 38(2), pp. 253-278.
- PYYHTINEN, O. 2010. Simmel and 'the social' New York, N.Y.: Palgrave Macmillan.

- PYYHTINEN, O. 2012. Life, death and individuation: Simmel on the problem of life itself. *Theory, Culture & Society,* 29(7-8), pp. 78-100.
- RADFORD, S. K. & BLOCH, P. H. 2012. Grief, commiseration, and consumption following the death of a celebrity. *Journal of Consumer Culture* 12(2), pp. 137-155.
- RAWLS, J. 1972. A theory of justice. Oxford: Clarendon Press.
- RAWLS, J. 1985. Justice as fairness: political not metaphysical. *Philosophy and Public Affairs*, 14(3), pp. 223-251.
- RAWLS, J. 1987. The idea of an overlapping consensus. *Oxford Journal of Legal Studies*, 7(1), pp. 1-25.
- RAWLS, J. 1993. Political liberalism. New York, N.Y.: Columbia University Press.
- REICHERTZ, J. 2004. Abduction, deduction and induction in qualitative research. *In:* FLICK, U., VON KARDORFF, E. & STEINKE, I. (eds.) *A companion to qualitative research.*London: Sage, pp. 159-164.
- REINHARZ, S. 1997. Who am I? The need for a variety of selves in the field. *In:* HERTZ, R. (ed.) *Reflexivity & voice.* Thousand Oaks, CA: Sage, pp. 3-20.
- REISINGER, Y. & STEINER, C. J. 2006. Reconceptualizing object authenticity. *Annals of Tourism Research*, 33(1), pp. 65–86.
- REPSTAD, P. 2003. The powerlessness of religious power in a pluralist society. *Social Compass*, 50(2), pp. 161-173.
- REPSTAD, P. 2008. From sin to a gift from God: constructions of change in conservative Christian organizations. *Journal of Contemporary Religion*, 23(1), pp. 17-31.
- REPSTAD, P. 2009. A softer God and a more positive anthropology: changes in a religiously strict region in Norway. *Religion*, 39(1), pp. 126-131.
- RICHARDSON, J. T. 1991. Experiencing research on new religions and cults. Practical and ethical considerations. *In:* SHAFFIR, W. & STEBBINS, R. A. (eds.) *Experiencing fieldwork: an inside view of qualitative research.* London: Sage, pp. 62-71.
- RIESSMAN, C. K. 1993. Narrative analysis. Newbury Park, CA: Sage.
- RITZER, G. 1993. *The McDonaldization of society: an investigation into the changing character of contemporary social life.* Thousand Oaks, CA: Pine Forge Press.
- ROCCAS, S. 2005. Religion and value systems *Journal of Social Issues*, 61(4), pp. 747-759.

- ROELAND, J., AUPERS, S., HOUTMAN, D., DE KONING, M. & NOOMEN, I. 2010. The quest for religious purity in New Age, Evangelicalism and Islam. Religious renditions of Dutch youth and the Luckmann legacy. *Annual Review of the Sociology of Religion*, pp. 289-306.
- ROOF, W. C. 1993. *A generation of seekers. The spiritual journeys of the baby-boom generation.* San Francisco, CA: Harper.
- ROOF, W. C. 1999. *Spiritual marketplace: baby boomers and the remaking of American religion.* Princeton, N.J.: Princeton University Press.
- ROOF, W. C. 2003. Religion and spirituality: toward an integrated analysis. *In:* DILLON, M. (ed.) *A handbook of the sociology of religion.* New York, N.Y.: Cambridge University Press, pp. 137-150.
- ROOF, W. C. 2007. Pluralism as a culture: religion and civility in southern California. *The Annals of the American Academy of Political and Social Science,* 612, pp. 82-99.
- ROTHENBUHLER, E. W. 2005. The church of the cult of the individual. *In:* ROTHENBUHLER, E. W. & COMAN, M. (eds.) *Media anthropology.* Thousand Oaks, CA: Sage, pp. 91-100.
- ROTHENBUHLER, E. W. 2006. The self as a sacred object in media. *In:* SUMIALA-SEPPÄNEN, J., LUNDBY, K. & SALOKANGAS, R. (eds.) *Implications of the sacred in (post)modern media.* Göteborg: Nordicom, pp. 31-41.
- RUBIN, H. J. & RUBIN, I. 1995. *Qualitative interviewing: the art of hearing data.* Thousand Oaks, CA: Sage.
- RÜPKE, J. & SPICKERMANN, W. (eds.) 2012. *Reflections on religious individuality: Greco-Roman and Judaeo-Christian texts and practices.* Berlin: Walter de Gruyter.
- SAIEDI, N. 1987. Simmel's epistemic road to multidimensionality. *The Social Science journal*, 24(2), pp. 181-193.
- SALER, M. 2006. Modernity and enchantment: a historiographical review. *The American Historical Review* 111(3), pp. 692-716.
- SAMBUR, B. 2002. From the dichotomy of spiritualism/ritualism to the dichotomy of insider/outsider. *In:* ARWECK, E. & STRINGER, M. D. (eds.) *Theorizing faith. The insider/outsider problem in the study of ritual.* Birmingham: University of Birmingham Press, pp. 21-33.
- SANTAELLA, L. 2005. Abduction: the logic of guessing. *Semiotica*, 153(1-4), pp. 175–198.

- SASSATELLI, M. 2008. An interview with Alessandro Ferrara: Europe, authenticity and unavailable identities. *European Journal of Social Theory*, 11(3), pp. 421-437.
- SATZ, D. & FEREJOHN, J. 1994. Rational choice and social theory. *The Journal of Philosophy*, 91(2), pp. 71-87.
- SCAFF, L. A. 1990. Georg Simmel's theory of culture. *In:* KAERN, M., PHILLIPS, B. S. & COHEN, R. S. (eds.) *Georg Simmel and contemporary sociology.* Dordrecht: Kluwer Academic Publishers, pp. 283-296.
- SCAFF, L. A. 2005. The mind of the modernist: Simmel on time. *Time Society*, 14(1), pp. 5-23.
- SCHEID, J. 2003. *An introduction to Roman religion.* Bloomington, IN: Indiana University Press.
- SCHEITLE, C. P. & ADAMCZYK, A. 2009. It takes two: the interplay of individual and group theology on social embeddedness. *Journal for the Scientific Study of Religion*, 48(1), pp. 16-29.
- SCHLEHOFER, M. M., OMOTO, A. M. & ADELMAN, J. R. 2008. How do "religion" and "spirituality" differ? Lay definitions among older adults. *Journal for the Scientific Study of Religion*. 47(3), pp. 411–425.
- SCHOPENHAUER, A. 1839/1995. On the basis of morality. Indianapolis, In.: Hackett.
- SCHUDSON, M. 2009. The varieties of civic experience. *In:* LICHTERMAN, P. & POTTS, C. B. (eds.) *The civic life of American religion.* Stanford, CA: Stanford University Press, pp. 23-47.
- SEDIKIDES, C. & BREWER, M. B. 2001. *Individual self, relational self, collective self.* Hove: Psychology Press.
- SEGAL, R. A. 1999. In defense of reductionism. *In:* MCCUTCHEON, R. T. (ed.) *The insider/outsider problem in the study of religion: a reader.* London: Cassell, pp. 139-163.
- SEGAL, R. A. 2005. Theories of religion. *In:* HINNELLS, J. R. (ed.) *The Routledge companion to the study of religion.* New York, N.Y.: Routledge, pp. 49-60.
- SEGAL, R. A. 2009. Kühn and the science of religion. *Religion*, 39(4), pp. 352-355.
- SEGRE, S. 1990. Principi metodologici nella sociologia di Simmel. Milano: EGEA.

- SÉGUY, J. 1964. Aux enfances de la sociologie des religions: Georg Simmel. *Archives des Sciences Sociales des Religions*, 17(1), pp. 5-11.
- SELLERBERG, A.-M. 1994. *A blend of contradictions. Georg Simmel in theory and practice.*New Brunswick: Transactions Publishers.
- SHAFFIR, W. 1991. Managing a convincing self-presentation. Some personal reflections on entering the field. *In:* SHAFFIR, W. & STEBBINS, R. A. (eds.) *Experiencing fieldwork:* an inside view of qualitative research. London: Sage, pp. 72-81.
- SHAFFIR, W. 2009. On piecing the puzzle. researching Hassidic Jews. *In:* PUDDEPHATT, A. J., SHAFFIR, W. & KLEINKNECHT, S. W. (eds.) *Ethnographies revisited. Constructing theory in the field.* New York, N.Y.: Routledge, pp. 212-224.
- SHALIN, D. N. 1986. Pragmatism and social interactionism. *American Sociological Review,* pp. 519-29.
- SHAROT, S. 2002. Beyond Christianity: a critique of the rational choice theory of religion from a Weberian and comparative religions perspective. *Sociology of Religion*, 63(4), pp. 427-454.
- SHIBLEY, M. A. 1996. *Resurgent evangelicalism in the United States: mapping cultural change since 1970.* Columbia, S.C.: University of South Carolina.
- SILBERMAN, I. 2005. Religion as a meaning system: implications for the new millennium. *Journal of Social Issues,* 61(4), pp. 641-663.
- SILVER, D., LEE, M. & MOORE, R. 2007. The View of Life: a Simmelian reading of Simmel's 'testament'. *Simmel Studies*, 17(2), pp. 262-290.
- SILVERMAN, D. 1997. The logics of qualitative research. *In:* MILLER, G. & DINGWALL, R. (eds) *Context and method in qualitative research.* London: Sage, pp. 12-25.
- SILVERMAN, D. 2000. *Doing qualitative research : a practical handbook.* Thousand Oaks, CA: Sage.
- SIMMEL, G. 1882/1968. Psychological and ethnological studies on music. *In:* ETZKORN, K. P. (ed.) *Georg Simmel. The conflict in modern culture and other essays.* . New York, N.Y.: Teachers College Press, pp. 98-130.
- SIMMEL, G. 1890/1982. *La differenziazione sociale.* Rome: Laterza.
- SIMMEL, G. 1893. Moral deficiencies as determining intellectual functions. *International Journal of Ethics*, 3(4), pp. 490-507.

- SIMMEL, G. 1896/2008. Friedrich Nietzsche. Un profile filosofico-morale. *In:* SIMMEL, G. & ANDOLFI, F. (ed.) *Friedrich Nietzsche. Filosofo morale.* Reggio Emilia: Diabasis, La Ginestra, pp. 29-45.
- SIMMEL, G. 1897/2008. Il culto di Nietzsche di Ferdinand Tönnies *In:* SIMMEL, G. & ANDOLFI, F. (ed) *Friedrich Nietzsche. Filosofo morale.* Reggio Emilia: Diabasis, La Ginestra, pp. 46-55.
- SIMMEL, G. 1898/1955. A contribution to the sociology of religion. *The American Journal of Sociology*, 60(6), pp. 1-18
- SIMMEL, G. 1898a. The persistence of social groups, I. *The American Journal of Sociology,* 3(5), pp. 662-698.
- SIMMEL, G. 1898b. The persistence of social groups, II. *The American Journal of Sociology*, 3(6), pp. 829-836.
- SIMMEL, G. 1898c. The persistence of social groups, III. *The American Journal of Sociology,* 4(1), pp. 35-50.
- SIMMEL, G. 1900. A chapter in the philosophy of value. *The American Journal of Sociology*, 5(5), pp. 577-603.
- SIMMEL, G. 1902/1997. Contributions to the epistemology of religion *In:* SIMMEL, G. & HELLE, H. J. (ed.) *Essays on religion* New Haven, CT: Yale University Press, pp. 121-133.
- SIMMEL, G. 1902/2008. Per comprendere Nietzsche. *In:* SIMMEL, G. & ANDOLFI, F. (ed.) *Friedrich Nietzsche. Filosofo morale.* Reggio Emilia: Diabasis, La Ginestra, pp. 56-63.
- SIMMEL, G. 1902/2012. Selections from Simmel's writings for the journal *Jugend. Theory, Culture & Society,* 29(7-8), pp. 263-278.
- SIMMEL, G. 1903/1997. On the salvation of the soul *In:* SIMMEL, G. & HELLE, H. J. (ed.) *Essays on religion* New Haven, CT: Yale University Press, pp. 29-35.
- SIMMEL, G. 1904/1957 Fashion. The American Journal of Sociology, 62(6), pp. 541-558.
- SIMMEL, G. 1904/1997. Religion and the contradictions of life *In:* SIMMEL, G. & HELLE, H. J. (ed.) *Essays on religion* New Haven, CT: Yale University Press, pp. 36-44.
- SIMMEL, G. 1906/1968. On the third dimension in art. *In:* ETZKORN, K. P. (ed.) *Georg Simmel. The conflict in modern culture and other essays.* New York, N.Y.: Teachers College Press, pp. 86-90.

- SIMMEL, G. 1906/2008. Nietzsche e Kant. *In:* SIMMEL, G. & ANDOLFI, F. (ed.) *Friedrich Nietzsche. Filosofo morale.* Reggio Emilia: Diabasis, La Ginestra, pp. 64-74.
- SIMMEL, G. 1907/1991. *Schopenhauer and Nietzsche.* Urbana, IL: University of Illinois Press.
- SIMMEL, G. 1907/2008. I valori dell'umanità e la decadenza. *In:* SIMMEL, G. & ANDOLFI, F. (ed.) *Friedrich Nietzsche. Filosofo morale.* Reggio Emilia: Diabasis, La Ginestra, pp. 87-124.
- SIMMEL, G. 1908/1955. Conflict. New York, N.Y.: Free Press.
- SIMMEL, G. 1908/2007. The social boundary. *Theory, Culture & Society*, 24(7-8), pp. 53-56.
- SIMMEL, G. 1909/1976. The future of our culture. *In:* LAWRENCE, P. A. (ed.) *Georg Simmel: sociologist and European.* Sunbury-on-Thames, Middlesex: Nelson, pp. 250-252.
- SIMMEL, G. 1909/1997. Fundamental religious ideas and modern science: an inquiry. *In:* SIMMEL, G. & HELLE, H. J. (ed.) *Essays on religion* New Haven, CT: Yale University Press, pp. 3-6.
- SIMMEL, G. 1910/2007 The metaphysics of death. *Theory Culture Society*, 24(7/8), pp. 72-77
- SIMMEL, G. 1910. How is society possible? *The American Journal of Sociology,* 16(3), pp. 372-391.
- SIMMEL, G. 1910/1949. The sociology of sociability *The American Journal of Sociology*, 55(3), pp. 254-261.
- SIMMEL, G. 1911a/1997. The problem of religion today. *In:* SIMMEL, G. & HELLE, H. J. (ed.) *Essays on religion* New Haven, CT: Yale University Press, pp. 7-19.
- SIMMEL, G. 1911b/1997. The personality of God. *In:* SIMMEL, G. & HELLE, H. J. (ed.) *Essays on religion* New Haven, CT: Yale University Press, pp. 45-62.
- SIMMEL, G. 1912/1997. Religion. *In:* SIMMEL, G. & HELLE, H. J. (ed.) *Essays on religion.* New Haven, CT: Yale University Press, pp. 137-214.
- SIMMEL, G. 1913/2004. L'etica e i problemi della cultura moderna. Napoli: Guida.
- SIMMEL, G. 1913/2007. The problem of fate. *Theory, Culture & Society*, 24(7-8), pp. 78-84.
- SIMMEL, G. 1914/2007. Goethe and youth. *Theory, Culture & Society*, 24(7-8), pp. 85-90.
- SIMMEL, G. 1915/2007. 'Become what you are'. *Theory, Culture & Society,* 24(7-8), pp. 57-60.

- SIMMEL, G. 1917/1976. The crisis of culture. *In:* LAWRENCE, P. A. (ed.) *Georg Simmel: sociologist and European.* Sunbury-on-Thames, Middlesex: Nelson, pp. 253-266.
- SIMMEL, G. 1917/2007 Individualism. *Theory Culture Society*, 24(7/8), pp. 66-71.
- SIMMEL, G. 1918/1968. The conflict in modern culture *In:* ETZKORN, K. P. (ed.) *Georg Simmel. The conflict in modern culture and other essays.* New York, N.Y.: Teachers College Press, pp. 11-26.
- SIMMEL, G. 1918/1971. The transcendent character of life. *In:* LEVINE, D. N. (ed.) *Georg Simmel on individuality and social forms.* Chicago, IL: University of Chicago Press, pp. 353-374.
- SIMMEL, G. 1918/1997. The conflict of modern culture. *In:* SIMMEL, G. & HELLE, H. J. (ed.) *Essays on religion* New Haven, CT: Yale University Press, pp. 20-25.
- SIMMEL, G. 1919/1984. Flirtation. *In:* OAKES, G. (ed.) *Georg Simmel: on women, sexuality, and love* New Haven, CT: Yale University Press, pp. 133-152.
- SIMMEL, G. 1919/1984. The problem of the sexes. *In:* OAKES, G. (ed.) *Georg Simmel: on women, sexuality, and love* New Haven, CT: Yale University Press, pp. 102-132.
- SIMMEL, G. 1950. The stranger. *In:* SIMMEL, G. & WOLFF, K. H. (ed.) *The sociology of Georg Simmel.* Glencoe, IL: Free Press, pp. 402-408.
- SIMMEL, G. 1911/1968. The dramatic actor and reality. *In:* ETZKORN, K. P. (ed.) *Georg Simmel. The conflict in modern culture and other essays.* . New York, N.Y.: Teachers College Press, pp. 91-97.
- SIMMEL, G. 1971. Freedom and the individual. *In:* LEVINE, D. N. (ed.) *On individuality and social forms: selected writings [of] Georg Simmel.* Chicago, IL: University of Chicago Press, pp. 217-226.
- SIMMEL, G. 1984. On love. *In:* OAKES, G. (ed.) *Georg Simmel: on women, sexuality, and love*New Haven, CT: Yale University Press, pp. 153-192.
- SIMMEL, G. 1994. Bridge and door. *Theory Culture Society*, 11(1), pp. 5-10.
- SIMMEL, G. & WOLFF, K. H. 1908/1950. *The sociology of Georg Simmel.* Glencoe, IL: Free Press.
- SIMMEL, G. & OAKES, G. (ed.) 1918/1980. *Essays on interpretation in social science.*Manchester: Manchester University Press.

- SIMMEL, G. & ANDOLFI, F. 1910/1996. *I problemi fondamentali della filosofia.* Bari: Biblioteca Universale Laterza.
- SIMMEL, G., FRISBY, D. & BOTTOMORE, T. B. 1900/2004. *The philosophy of money.* London: Routledge.
- SIMMEL, G., FRISBY, D. & FEATHERSTONE, M. 1997. *Simmel on culture: selected writings.*London: Sage.
- SIMMEL, G., HELLE, H. J. & NIEDER, L. (eds.) 1997. *Essays on religion.* New Haven, CT: Yale University Press.
- SIMMEL, G. & LUKÁCS, G. (ed.) 1910/2003. Michelangelo, Milano, Abscondita.
- SIMMEL, G. & ANDOLFI, F. (ed) 2008. *Friedrich Nietzsche. Filosofo morale.* Reggio Emilia: Diabasis, *La Ginestra.*
- SIMMEL, G. & LEVINE, D. N. 1918/2010. *The view of life. Four metaphysical essays with journal aphorisms.* Chicago, IL: University of Chicago Press.
- SIMON, D. 2001. Persona, formazione, agire sociale in Durkheim, Simmel e Weber. Elementi di lettura. *Sociologia*, 2, pp. 23-20.
- SIMON, D. 2006. Religione e religiosità. Significato e rilevanza attuale del contributo di Émil Durkheim e Georg Simmel. *Sociologia*, 1, pp. 145-52.
- SMILDE, D. 2007. *Reason to believe. Cultural agency in Latin American evangelicalism.*Berkeley, CA: University of California Press.
- SMITH, B. G. & JOHNSON, B. 2010. The liberalization of young evangelicals: a research note. *Journal for the Scientific Study of Religion*, 49(2), pp. 351–360.
- SMITH, C. (ed.) 1996. *Disruptive religion. The force of faith in social-movement activism.* New York, N.Y.: Routledge.
- SMITH, C. 1998. *American evangelicalism: embattled and thriving* Chicago, IL: University of Chicago Press.
- SMITH, C. 2000. *Christian America? What evangelicals really want.* Berkeley, CA: University of California Press.
- SMITH, R. 1997. Self-reflection and the self. *In:* PORTER, R. (ed.) *Rewriting the self. Histories from the Renaissance to the present.* London: Routledge, pp. 49-57.
- SNOW, D. A., MORRILL, C. & ANDERSON, L. 2003. Elaborating analytic ethnography: linking fieldwork and theory. *Ethnography*, 4(2), pp. 181-200.

- SOMERS, M. R. 1994. The narrative constitution of identity: a relational and network approach. *Theory and Society*, 23, pp. 605-649.
- SOMERS, M. R. 1998. 'We're no angels': realism, rational choice, and relationality in social science. *The American Journal of Sociology*, 104(3), pp. 722-784.
- SPICKARD, J. V. 1991. Experiencing religious rituals: a Schutzian analysis of Navajo ceremonies. *Sociological Analysis*, 52(2), pp. 191-204.
- SPICKARD, J. V. 1998. Rethinking religious social action: what is "rational" about Rational-Choice Theory? *Sociology of Religion*, 59(2), pp. 99-115.
- SPICKARD, J. V. 2002. On the epistemology of post-colonial ethnography. *In:* SPICKARD, J. V., LANDRES, J. S. & MCGUIRE, M. B. (eds.) *Personal knowledge and beyond: reshaping the ethnography of religion.* New York, N.Y.: New York University Press, pp. 237-252.
- SPICKARD, J. V. 2006. What is Happening to Religion? Six Sociological Narratives. *Nordic Journal of Religion and Society*, 19(1), pp. 13-29.
- SPICKARD, J. V., LANDRES, J. S. & MCGUIRE, M. B. 2002. *Personal knowledge and beyond:* reshaping the ethnography of religion. New York, N.Y.: New York University Press.
- SPILKA, B. & MCINTOSH, D. N. 1997. *The psychology of religion: theoretical approaches.*Boulder, Co.: Westview Press.
- STANCZAK, G. C. 2006. *Engaged spirituality: social change and American religion* New Brunswick: Rutgers University Press.
- STARHAWK 1979. *The spiral dance: a rebirth of the ancient religion of the great Goddess.*San Francisco: Harper & Row.
- STARK, R. 1999. Secularization, R.I.P. Sociology of Religion, 60(3), pp. 249-273.
- STARK, R. 2001. Gods, rituals and the moral order. *Journal for the Scientific Study of Religion*, 40(4), pp. 619-636.
- STARKS, B. & ROBINSON, R. V. 2009. Two approaches to religion and politics: moral cosmology and subcultural identity. *Journal for the Scientific Study of Religion*, 48(4), pp. 650-669.
- STARR, B. E. 1996. The tragedy of the kingdom. Simmel and Troeltsch on prophetic religion. *Journal of Religious Ethics*, 24(1), pp. 141-167.
- STEIN, P. 1999. Roman law in European history. Cambridge: Cambridge University Press.

- STOCKS, J. 1997. To stay or to leave? Organizational legitimacy in the struggle for change among evangelical feminists. *In:* EDGELL, P. & EIESLAND, N. L. (eds.) *Contemporary American religion: an ethnographic reader.* Walnut Creek, CA: AltaMira Press, pp. 99-119.
- STONE, J. R. 1996. Defining Protestant orthodoxy: religious authentication and the 'new' evangelicalism, 1940-1960. *Religion and the social order. The Issue of Authenticity in the Study of Religions*, 667-92.
- STREIB, H. & HOOD, R. W. 2011. "Spirituality" as privatized experience-oriented religion: empirical and conceptual Perspectives. *Implicit Religion*, 14(4), pp. 433-353.
- STROUMSA, G. G. 2012. Reading practices in early Christianity and the individualisation process. *In:* RÜPKE, J. & SPICKERMANN, W. (eds.) *Reflections on religious individuality: Greco-Roman and Judaeo-Christian texts and practices.* Berlin: Walter de Gruyter, pp. 175-192.
- STYLES, J. 1979. Outsider/Insider: researching gay baths. *Urban Life*, 8(2), pp. 135-152.
- SVEJENOVA, S. 2005. 'The path with the heart': creating the authentic career. *Journal of Management Studies*, 42(5), pp. 947-973.
- SWATOS, W. H. & CHRISTIANO, K. J. 1999. Secularization theory: the course of a concept *Sociology of Religion*, 60(3), pp. 209-228.
- SWEDBERG, R. & REICH, W. 2010. Georg Simmel's aphorisms. *Theory, Culture & Society,* 27(1), pp. 24-51.
- SYMONDS, M. & PUDSEY, J. 2006. The forms of brotherly love in max Weber's sociology of religion. *Sociological theory*, 24(2), pp. 133-149.
- SYMONDS, M. & PUDSEY, J. 2008. The concept of 'paradox' in the work of Max Weber. *Sociology*, 42(2), pp. 223–241.
- SZTOMPKA, P. 2009. The return to values in recent sociological theory. *In:* HEDSTRÖM, P. & WITTROCK, B. (eds.) *Frontiers of sociology.* The Annals of the International Institute of Sociology, pp. 38-57.
- TACEY, D. 2004. *The spirituality revolution: the emergence of contemporary spirituality.*Hove: Brunner-Routledge.
- TAVORY, I. & GOODMAN, Y. C. 2009. "A collective of individuals": between self and solidarity in a rainbow gathering. *Sociology of Religion*, 70(3), pp. 262-284.

- TAYLOR, C. 1989. *Sources of the self: the making of the modern identity.* Cambridge: Cambridge University Press.
- TAYLOR, C. 1991. The ethics of authenticity. Cambridge, MA: Harvard University Press.
- TAYLOR, C. 1995. Two theories of modernity. *The Hastings Center Report*, 25(2), pp. 24-33.
- TAYLOR, C. 2002. *Varieties of religion today: William James revisited.* Cambridge, MA: Harvard University Press.
- TAYLOR, C. 2007. A secular age. Cambridge, MA: Belknap Press of Harvard University Press.
- TAYLOR, C. 2010. The meaning of secularism. The Hedgehog Review, Fall.
- TAYLOR, J. 2001. Authenticity and sincerity in tourism. *Annals of Tourism Research*, 28(1), pp. 7–26.
- TENBRUK, F. H. 1994. Formal sociology. *In:* FRISBY, D. (ed.) *Georg Simmel: critical assessments.* London: Routledge, pp. 347-373.
- THOMAS, J. N. & OLSON, D. V. A. 2012. Evangelicals' elites' changing responses to homosexuality 1960-2009. *Sociology of Religion*, 73(3), pp. 239-272.
- THOMPSON, J. B. 1996. Tradition and self in a mediated world. *In:* HEELAS, P., LASH, S. & MORRIS, P. (eds.) *Detraditionalization. Critical reflections on authority and identity.*Oxford: Blackwell Publishers, pp. 89-108.
- TIDBALL, D. J. 1994. *Who are the evangelicals? tracing the roots of the modern movements.*London: Marshall Pickering.
- TOMLINSON, D. 1995. The post-evangelical. London: Triangle.
- TÖNNIES, F. 1887/1963. Community & society. New York, N.Y.: Harper & Row.
- TRILLING, L. 1972. *Sincerity and authenticity.* London: Oxford University Press.
- TSCHANNEN, O. 1991. The secularization paradigm: a systematization. *Journal for the Scientific Study of Religion*. 30(4), pp. 395-415.
- TURNER, B. S. 2006. Classical sociology and cosmopolitanism: a critical defence of the social. *British Journal of Sociology*, 57(1), pp. 133-151.
- TURNER, B. S. 2007. Religious authority and the new media. *Theory, Culture & Society,* 24(2), pp. 117-134.
- TURNER, B. S. 2010. Religion in a post-secular society. *In:* TURNER, B. S. (ed.) *The new Blackwell companion to sociology of religion.* Oxford: Blackwell, pp. 649-667.

- TYLER, T. R. 2006. Psychological perspectives on legitimacy and legitimation. *Annual Review of Psychology*, 57, pp. 375-400.
- UNRUH, H. R. & SIDER, R. J. 2005. *Saving souls, serving society: understanding the faith factor in church-based social ministry.* Oxford: Oxford University Press.
- VALDES, F. 1996. Unpacking hetero-patriarchy: tracing the conflation of sex, gender & (and) sexual orientation to its origins. *Yale Journal of Law & the Humanities*, 8(1), pp. 161-212.
- VALLIN, J. 2006. Europe's demographic transition, 1740-1940. *In:* CASELLI, G., VALLIN, J. & WUNSCH, G. (eds.) *Demography: analysis and synthesis. A treatise in population.* San Diego: Academic Press, pp. 41-66.
- VAN DYCK, L. 2007. The church in evangelical theology and practice. . *In:* LARSEN, T. & TREIER, D. J. (eds.) *The Cambridge companion to evangelical theology.* Cambridge: Cambridge University Press, pp. 125-142.
- VAN MAANEN, J. 1991. Playing back the tape: early days in the field. *In:* SHAFFIR, W. & STEBBINS, R. A. (eds.) *Experiencing fieldwork: an inside view of qualitative research.* London: Sage, pp. 31-42.
- VANDENBERGHE, F. 1999a. La double dualité comme principe d'unité de la pensée simmelienne. *Simmel Newsletter*. Bielefeld: Georg Simmel-Gesellschaft.
- VANDENBERGHE, F. 1999b. Simmel and Weber as ideal-typical founders of sociology *Philosophy and Social Criticism,* 25(4), pp. 57-80.
- VANDENBERGHE, F. 2000. La loi individuelle: une éthique nietzschéo-bergsonienne. Simmel Studies, 10(2), pp. 169-178.
- VANDENBERGHE, F. 2001. La sociologie de Georg Simmel. Paris: La Découverte.
- VANDENBERGHE, F. 2002. Relativisme, relationnisme, structuralisme. *Simmel Studies*, 12(1), pp. 41-84.
- VANDENBERGHE, F. 2010. Immanent transcendence in Georg Simmel's sociology of religion. *Journal of Classical Sociology*, 10(1), pp. 5-32.
- VANNINI, P. & WILLIAMS, J. P. Authenticity in culture, self, and society. Authenticity in culture, self, and society. *In:* VANNINI, P. & WILLIAMS, J. P. (eds.). Farnham: Ashgate, pp. 1-18.

- VARGA, I. 2007. Georg Simmel: religion and spirituality. *In:* FLANAGAN, K. & JUPP, P. C. (eds.) *A sociology of spirituality.* Aldershot: Ashgate, pp. 145-160.
- VARGA, S. 2011. The paradox of authenticity. *Telos*, 156(Fall), pp. 113-130
- VARGA, S. 2012. Authenticity as an ethical ideal. New York, N.Y.: Routledge.
- VERSTEEG, P. G. A. 2001. *Draw me close: an ethnography of experience in a Dutch charismatic church.* Amsterdam: P. Gerrit.
- VERSTEEG, P. G. A. 2007. Spirituality on the margin of the church: Christian spiritual centres in the Netherlands. *In:* FLANAGAN, K. & JUPP, P. C. (eds.) *A sociology of spirituality.* Aldershot: Ashgate, pp. 101-114.
- VERTER, B. 2003. Spiritual capital: theorizing religion with Bourdieu against Bourdieu. *Sociological Theory*, 21(2), pp. 150-174.
- VILLEGAS, F. G. 1997. El fundamento filosófico de la teoría de la modernidad en Simmel. *Estudios Sociológicos,* 15(45), pp. 3-46.
- VOAS, D. & BRUCE, S. 2007. The spiritual revolution: another false dawn for the sacred. *In:* FLANAGAN, K. & JUPP, P. C. (eds.) *A sociology of spirituality.* Aldershot: Ashgate, pp. 43-62.
- VOAS, D. & CROCKETT, A. 2005. Religion in Britain: neither believing nor belonging. *Sociology*, 39(1), pp. 11-28.
- WAGNER, P. 2005. Your spiritual gifts can help your church grow Ventura, CA: Regal Books.
- WARNER, R. S. 1993. Work in progress toward a new paradigm for the sociological study of religion in the United States. . *American Journal of Sociology*, 98, pp. 1044-1093.
- WARNER, R. S. 1997. Religion, boundaries, and bridges. *Sociology of Religion*, 58(3), pp. 217-238.
- WARNER, R. S. 2000. Building religious communities at the turn of the century. *In:* LIVEZEY, L. W. (ed.) *Public religion and urban transformation. Faith in the city.* New York, N.Y.: New York University Press, pp. 295-307.
- WARNER, R. S. 2008. Autonomous conformism the paradox of entrepreneurical Protestantism. Spring harvest: a case study. *In:* DAY, A. (ed.) *Religion and the individual: belief, practice, identity.* Aldershot: Ashgate, pp. 95-109.

- WARREN, C. 2002. Qualitative interviewing. *In:* GUBRIUM, J. F. & HOLSTEIN, J. A. (eds.) *Handbook of interview research, context and method.* Thousand Oaks, CA: Sage, pp. 83-102.
- WARREN, R. 1995. *The purpose driven church: growth without compromising your message and mission* Grand Rapids, MI: Zondervan.
- WARREN, R. 2002. *The purpose driven life. What on earth am I here for?* Grand Rapids, MI: Zondervan.
- WASKOW, A. 1992. What is eco-kosher? Jewish Quarterly 39(4), pp. 5-10.
- WATIER, P. 1996. Georg Simmel: religion, sociologie et sociologie de la religion. *Archives des sciences sociales des religions*, 93(janvier/mars), pp. 23-50.
- WEBER, M. 1904/1949. *The methodology of the social sciences*. Glencoe, IL: Free Press.
- WEBER, M. 1919/1970. Science as a vocation. *In:* GERTH, H. H. & MILLS, C. W. (eds.) *From Max Weber: essays in sociology.* London: Routledge and Kegan Paul.
- WEBER, M. 1919/1970. Politics as a vocation. *In:* GERTH, H. H. & MILLS, C. W. (eds.) *From Max Weber: essays in sociology.* London: Routledge and Kegan Paul.
- WEBER, M. 1922/1965. *The sociology of religion.* London: Methuen.
- WEBER, M. 1922/1978. *Economy and society: an outline of interpretive sociology.* Berkeley, CA: University of California Press.
- WEBER, M., GERTH, H. H. & MILLS, C. W. (eds.)1970. From Max Weber: essays in sociology.

 London: Routledge and Kegan Paul.
- WEBER, M. & PARSONS, T. 1905/2001. *The Protestant ethic and the spirit of capitalism.* London: Routledge.
- WEDAM, E. 1997. Splitting interests or common causes. Styles of moral reasoning in opposing abortion. *In:* EDGELL, P. & EIESLAND, N. L. (eds.) *Contemporary American religion: an ethnographic reader.* Walnut Creek, CA: AltaMira Press, pp. 147-168.
- WEDAM, E. 2000. "God doesn't ask what language I pray in". Community and culture on Chicago's Southwest Side. *In:* LIVEZEY, L. W. (ed.) *Public religion and urban transformation. Faith in the city.* New York, N.Y.: New York University Press, pp. 107-132.

- WEINSTEIN, D. & WEINSTEIN, M. A. 1990. Dimensions of conflict: Georg Simmel on modern life. *In:* KAERN, M., PHILLIPS, B. S. & COHEN, R. S. (eds.) *Georg Simmel and contemporary sociology.* Dordrecht: Kluwer Academic Publishers.
- WEINSTEIN, D. & WEINSTEIN, M. 1993. *Postmodern(ized) Simmel.* New York, N.Y.: Routledge.
- WELLS, D. 1994. On being evangelical: some theological differences and similarities. *In:*NOLL, M. A., RAWLYK, G. A. & BEBBINGTON, D. W. (eds.) *Evangelicalism:*comparative studies of popular Protestantism in North America, the British Isles, and beyond 1700-1900. New York, N.Y.: Oxford University Press, pp. 389-410.
- WENMAN, M. 2003. 'Agonistic pluralism' and three archetypal forms of politics. *Contemporary Political Theory,* 2, pp. 165–186.
- WESSELY, A. 1990. Simmel's metaphysics. *In:* KAERN, M., PHILLIPS, B. S. & COHEN, R. S. (eds.) *Georg Simmel and contemporary sociology.* Dordrecht: Kluwer Academic Publishers, pp. 357-373.
- WIEBE, D. 1999. Does understanding religion require a religious understanding? *In:*MCCUTCHEON, R. T. (ed.) *The insider/outsider problem in the study of religion: a reader.* London: Cassell, pp. 260-273.
- WILCOX, M. M. 2002. Dancing on the fence. Researching lesbian, gay, bisexual, and transgender Christians. *In:* SPICKARD, J. V., LANDRES, J. S. & MCGUIRE, M. B. (eds.) *Personal knowledge and beyond: reshaping the ethnography of religion.* New York, N.Y.: New York University Press, pp. 47-60.
- WILLAIME, J.-P. 1993. Du Protestantisme comme objet sociologique. *Archives des Sciences Sociales des Religions*, 83(1), pp. 159-178.
- WILLAIME, J.-P. 1993. Du Protestantisme comme objet sociologique. *Archives des Sciences Sociales des Religions*, 83(1), pp. 159-178.
- WILLAIME, J.-P. 1993. Le croire, l'acteur et le chercheur. Introduction au dossier "Croire et modernité". *Archives des Sciences Sociales des Religions*, 81(1), pp. 7-16.
- WILLAIME, J.-P. 1993. Le croire, l'acteur et le chercheur. Introduction au dossier "Croire et modernité". *Archives des Sciences Sociales des Religions*, 81(1), pp. 7-16.

- WILLIAMS, R. H. & BLACKBURN, J. 1996. Many are called but few obey: ideological commitment and activism in operation rescue. *In:* SMITH, C. (ed.) *Disruptive religion. The force of faith in social-movement activism.* New York, N.Y.: Routledge, pp. 167-185.
- WILSON, B. 2002. Salvation, secularization and de-moralization. *In:* FENN, R. K. (ed.) *The Blackwell companion to sociology of religion.* Oxford: Blackwell, pp. 39-51.
- WILSON, B. R. 1966. *Religion in secular society: a sociological comment.* London: Watts.
- WITTE, J. 1987. Blest be the ties that bind: covenant and community in Puritan thought. *Emory Law Journal,* 36, pp. 579-601.
- WOLCOTT, H. F. 1981/2012. Confessions of a 'trained' observer *In:* DELAMONT, S. (ed.) *Ethnographic methods in education.* Los Angeles, CA: Sage, pp. 263-277.
- WOLFF, K. H. 1958. The challenge of Durkheim and Simmel. *The American Journal of Sociology*, 63(6), pp. 590-596.
- WOOD, M. 2007. *Possession, power and the New Age: ambiguities of authority in neoliberal societies.* Aldershot: Ashgate.
- WOOD, M. 2009. The nonformative elements of religious life: questioning the 'sociology of spirituality' paradigm. *Social Compass*, 56(2), pp. 237-248.
- WOOD, M. 2010. The sociology of spirituality. Reflections on a problematic endeavor. *In:*TURNER, B. S. (ed.) *New Blackwell companion to the sociology of religion.* Chichester:

 Blackwell, pp. 267-285.
- WOODHEAD, L. 2011. Five concepts of religion. *International Review of Sociology,* 21(1), pp. 121-143.
- WOODHEAD, L. 2012. Spirituality and Christianity: the unfolding of a tangled relationship. *In:* GIORDAN, G. & SWATOS, W. H. (eds.) *Religion, Spirituality and Everyday Practice.*Dordrecht: Springer, pp. 3-21.
- WOODHEAD, L. & HEELAS, P. 2000. *Religion in modern times: an interpretive anthology.*Malden, MA: Blackwell.
- WUTHNOW, R. 1997. The cultural turn: stories, logics, and the quest for identity in American religion. *In:* EDGELL, P. & EIESLAND, N. L. (eds.) *Contemporary American religion: an ethnographic reader.* Walnut Creek, CA: AltaMira Press, pp. 245-265.

- WUTHNOW, R. 1998. *After heaven: spirituality in America since the 1950s.* Berkeley, CA: University of California Press.
- WUTHNOW, R. 2001. Spirituality and spiritual practice. *In:* FENN, R. K. (ed.) *The Blackwell companion to sociology of religion.* Oxford: Blackwell, pp. 306-320.
- WUTHNOW, R. 2002. Beyond quiet influence? Possibilities for the Protestant mainline. *In:* WUTHNOW, R. & EVANS, J. H. (eds.) *The quiet hand of God. Faith-based activism and the public role of mainline Protestantism.* Berkeley, CA: University of California Press, pp. 381-403.
- WUTHNOW, R. 2004. The challenge of diversity. *Journal for the Scientific Study of Religion*, 43(2), pp. 159-170.
- WUTHNOW, R. 2004. "The religious factor" revisited. *Sociological Theory*, 22(2), pp. 205-218.
- WUTHNOW, R. 2007. Cognition and religion. Sociology of Religion, 68(4), pp. 341-360.
- WUTHNOW, R., HACKETT, C. & BECKY, Y. H. 2004. The effectiveness and trustworthiness of faith-based and other service organizations: a study of recipients' perceptions *Journal for the Scientific Study of Religion*, 43(1), pp. 1-17.
- YAMANE, D. 1997. Secularization on trial: in defense of a neosecularization paradigm. *Journal for the Scientific Study of Religion*, 36(1), pp. 109-122.
- YAMANE, D. 2000. Narrative and religious experience. *Sociology of Religion*, 61(2), pp. I71-189.
- YORK, M. 2001. New Age commodification and appropriation of spirituality. *Journal of Contemporary Religion*, 16(3), pp. 361-372.
- YUVAL-DAVIS, N. 2004. Borders, boundaries, and the politics of belonging. *In:* MAY, S., MODOOD, T. & SQUIRES, J. (eds.) *Ethnicity, nationalism and minority rights.*Cambridge: Cambridge University Press pp. 214-230.
- ZARET, D. 1985. *The heavenly contract: ideology and organization in pre-revolutionary Puritanism.* Chicago, IL: University of Chicago Press.
- ZARET, D. 1989. Religion and the rise of liberal-democratic ideology in 17th-Century England. *American Sociological Review,* 163-179.
- ZARET, D. 1992. Calvin, covenant theology, and the Weber thesis. *The British Journal of Sociology*, 43(3), pp. 369-391.

- ZARET, D. 2000. *Origins of democratic culture: printing, petitions, and the public sphere in early-modern England.* Princeton, N.J.: Princeton University Press.
- ZINNBAUER, B. J. & PARGAMENT, K. I. 2005. Religiousness and spirituality. *In:*PALOUTZIAN, R. F. & PARK, C. L. (eds.) *Handbook of the psychology of religion and spirituality.* New York, N.Y.: Guilford Press, pp. 21-42.
- ZINNBAUER, B. J., PARGAMENT, K. I. & SCOTT, A. B. 1999. The emerging meanings of religiousness. *Journal of Personality*, 67(6), pp. 890-919.